UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, DC 20549

FORM 8-K

CURRENT REPORT Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of report (Date of earliest event reported): November 7, 2023

APOLLO MEDICAL HOLDINGS, INC.

(Exact Name of Registrant as Specified in its Charter)

Delaware (State or Other Jurisdiction of Incorporation) 001-37392 (Commission File Number) 95-4472349 (I.R.S. Employer Identification No.)

1668 S. Garfield Avenue, 2nd Floor, Alhambra, California 91801 (Address of Principal Executive Offices) (Zip Code)

(626) 282-0288

Registrant's Telephone Number, Including Area Code

(Former Name or Former Address, if Changed Since Last Report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

□ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common Stock	AMEH	The Nasdaq Stock Market LLC

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company \Box

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

Item 1.01 Entry into a Material Definitive Agreement.

On November 7, 2023, Apollo Medical Holdings, Inc. (the "Company"), Metropolitan IPA, a California professional corporation ("CFC Buyer"), ApolloCare Enablement of CA, LLC, a wholly-owned subsidiary of the Company ("GP Buyer"), and Network Medical Management, Inc., a wholly-owned subsidiary of the Company ("MMM"), entered into an Asset and Equity Purchase Agreement, dated November 7, 2023, with Community Family Care Medical Group IPA, Inc. ("CFC"), Advanced Health Management Systems, L.P. ("AHMS"), Accie M. Mitchell and Gloria C. Mitchell, as Co-Trustees of the Mitchell Family Trust dated July 2, 2003, CFC Management, LLC, the limited partners of AHMS and Marc Mitchell, as the Equityholder Representative (the "CFC/AHMS Purchase Agreement"). Under the terms of the CFC/AHMS Purchase Agreement, subject to satisfaction of customary conditions: (i) GP Buyer and NMM will purchase all of the outstanding general and limited partnership interests of AHMS for an aggregate purchase price of \$52 million, subject to customary adjustments, and (ii) CFC Buyer will purchase substantially all the assets of CFC for an aggregate purchase price of \$93.8 million in cash and 631,712 shares of common stock of the Company), subject to customary adjustments in an aggregate amount of up to \$15 million. The purchase pursuant to the CFC/AHMS Purchase Agreement of the assets of, and the assumption of the liabilities of, CFC by CFC Buyer and the purchase of all of the outstanding general and limited partnership interests of AHMS by the GP Buyer and NMM are referred to collectively as the "Transaction."

CFC is an independent medical practice association that has entered into agreements with organizations such as insurance companies, health plans, self-insured employers, government payors, health maintenance organizations, medical groups, independent practice associations and other third party payors for the arrangement of the provision of healthcare services to subscribers or enrollers of such plans. AHMS is engaged in the business of providing management, consulting, administrative and other support services to entities that provide or arrange for the provision of professional healthcare services.

The GP Buyer and NMM are direct or indirect wholly-owned subsidiaries of the Company. The CFC Buyer is a wholly-owned professional corporation subsidiary of AP-AMH 2 Medical Corporation ("AP-AMH 2"), which is a professional corporation of which Dr. Thomas Lam, a licensed physician and the co-Chief Executive Officer and President, and a director, of the Company, is the sole shareholder. AP-AMH 2 is a variable interest entity of the Company, the financial results of which are consolidated in the Company's financial statements. As a wholly-owned subsidiary of AP-AMH 2, the financial results of the CFC Buyer will be consolidated into the financial results of AP-AMH 2.

The Company intends to finance the Transaction with cash on hand and borrowings under its amended and restated credit agreement with Truist Bank, as administrative agent, and the financial institutions that are lenders thereunder. The Company anticipates that it will lend the funds required for the purchase of the assets of CFC to AP-AMH 2 on a secured basis, and that AP-AMH 2 will use those funds to purchase preferred stock of the CFC Buyer.

The CFC/AHMS Purchase Agreement includes customary representations, warranties, covenants, conditions and other agreements. The obligations of the parties to complete the Transaction is subject to the satisfaction, or waiver, of customary closing conditions, including receipt of applicable regulatory approvals. It is currently anticipated that the purchase of the assets of CFC and the purchase of the outstanding partnership interests of AMHS will occur in two separate closings, both of which are currently expected to occur during the first calendar quarter of 2024.

In connection with the Transaction, NMM has entered into a Stock Purchase Agreement, dated November 7, 2023 (the "I Health Purchase Agreement"), with Ronald Brandt and Allison Brandt, each in their individual capacities and in their capacities as Co-Trustees of the Ronald Lee Brandt and Allison Leigh Brandt Family Trust dated December 16, 2003, and I Health, Inc. ("I Health"). Under the terms of the I Health Purchase Agreement, subject to satisfaction of customary conditions, NMM will purchase 25% of the outstanding shares of common stock of I Health, subject to customary adjustments, and will have a call option to purchase the remaining outstanding shares of common stock of I Health Purchase Agreement closing will occur during the first calendar quarter of 2024.

I Health is engaged in the business of providing management, consulting, administrative and other support services to entities that provide or arrange for the provision of professional healthcare services.

The foregoing description of the CFC/AHMS Purchase Agreement and the Transaction and the I Health Purchase Agreement does not purport to be complete and is subject to, and qualified in its entirety by, the full text of the CFC/AHMS Purchase Agreement and the I Health Purchase Agreement, copies of which are filed as Exhibit 10.1 and Exhibit 10.2, respectively, to this Current Report on Form 8-K (this "Report") and are incorporated herein by reference.

The CFC/AHMS Purchase Agreement and I Health Purchase Agreement have been included with this Report to provide investors and security holders with information regarding the terms of the Transaction. It is not intended to provide any other factual information about the Company, CFC, AHMS or I Health or any of the other parties to the Transaction or the I Health Purchase Agreement or the assets or equity to be purchased in the Transaction or such agreement. The representations, warranties, covenants and agreements contained in the CFC/AHMS Purchase Agreement and the I Health Purchase Agreement, which are made only for purposes of the respective agreement and as of specific dates, are solely for the benefit of the parties to the respective agreement, may be subject to limitations agreed upon by the parties (including being qualified by confidential disclosures made for the purposes of allocating contractual risk between the parties to the applicable to investors and security holders. Company investors and security holders should not rely on the representations, warranties, covenants and agreements should not rely on the representations, warranties to the Transaction or any descriptions thereof as characterizations of the actual state of facts or condition of the Company, CFC, AHMS or I Health or any of the other parties to the Transaction or the I Health Purchase Agreement. Moreover, information concerning the subject matter of the representations and warranties may change after the date of the applicable agreement, which subsequent information may or may not be fully reflected in the Company's public disclosures.

Item 8.01 Other Events

On November 7, 2023, the Company issued a press release announcing that it had entered into the CFC/AHMS Purchase Agreement and the I Health Purchase Agreement. A copy of the press release is filed as Exhibit 99.1 to this Report and incorporated herein by reference.

Forward-Looking Statements

This Report contains forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended. These statements include words such as "forecast," "guidance," "projects," "estimates," "anticipates," "believes," "expects," "intends," "may," "plans," "seeks," "should," or "will," or the negative of these words or similar words. Forward-looking statements involve certain risks and uncertainties, and actual results may differ materially from those discussed in each such statement. A number of important factors could cause actual results to differ materially from those included within or contemplated by the forward-looking statements, including, but not limited to, risks arising from the diversion of management's attention from the Company's ongoing business operations, an increase in the amount of costs, fees and expenses and other charges related to the Transaction described in this Report, outcome of any litigation that the Company or the sellers may become subject to relating to such Transaction, the extent of, and the time necessary to obtain, any regulatory approvals required for completion of the Transaction, the other torus of the Transaction, the occurrence of any event, change or other circumstance that could give rise to the closing of the Transaction to be satisfied or waived by the applicable party, the occurrence of any event, change or other circumstance that could give rise to the termination of any of the agreements to the Transaction, a decline in the market price for the Company's completed, risks that the

Transaction disrupt current plans and operations of the Company or sellers and potential difficulties in sellers' employee retention as a result of the Transaction, and the ability to implement business plans, forecasts and other expectations after the completion of the Transaction, realize the intended benefits of the Transaction, and identify and realize additional opportunities following the Transaction, as well as the other risks and uncertainties identified in filings by the Company with the Securities and Exchange Commission, including the Company's Annual Report on Form 10-K/A for the year ended December 31, 2022 and subsequent quarterly reports on Form 10-Q. The Company does not undertake any responsibility to update any of these factors or to announce publicly any revisions to any of the forward-looking statements contained in this or any other document, whether as a result of new information, future events, or otherwise, except as may be required by any applicable securities laws.

Item 9.01 Financial Statements and Exhibits.

(d) Exhibits.

Exhibit No.	Description
<u>10.1*</u>	Asset and Equity Purchase Agreement, dated as of November 7, 2023, by and among Metropolitan IPA, a California professional corporation, ApolloCare
	Enablement of CA, LLC, Network Medical Management, Inc., Apollo Medical Holdings, Inc., Community Family Care Medical Group IPA, Inc.,
	Advanced Health Management Systems, L.P., Accie M. Mitchell and Gloria C. Mitchell, as Co-Trustees of the Mitchell Family Trust dated July 2, 2003,
	CFC Management, LLC, the other parties thereto and Marc Mitchell, as the Equityholder Representative.
10.2*	Stock Purchase Agreement, dated as of November 7, 2023, by and among Network Medical Management, Inc., I Health, Inc., Ronald Brandt and Allison
	Brandt.
<u>99.1</u>	Press Release issued by the Company on November 7, 2023.
104	Cover Page Interactive Data File (the cover page XBRL tags are embedded within the inline XBRL document).

* Certain of the exhibits and schedules to this exhibit have been omitted in accordance with Item 601(a)(5) of Regulation S-K. The Company agrees to furnish a copy of all omitted exhibits and schedules to the Securities and Exchange Commission upon its request.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

APOLLO MEDICAL HOLDINGS, INC.

Date: November 7, 2023

By: /s/ Thomas S. Lam

Name: Thomas S. Lam, M.D., M.P.H.

Title: Co-Chief Executive Officer and President

<u>Page</u>

ASSET AND EQUITY PURCHASE AGREEMENT

by and among

METROPOLITAN IPA, A CALIFORNIA PROFESSIONAL CORPORATION,

APOLLOCARE ENABLEMENT OF CA, LLC,

NETWORK MEDICAL MANAGEMENT, INC.,

APOLLO MEDICAL HOLDINGS, INC.,

COMMUNITY FAMILY CARE MEDICAL GROUP IPA, INC.,

ADVANCED HEALTH MANAGEMENT SYSTEMS, L.P.,

ACCIE M. MITCHELL AND GLORIA C. MITCHELL, AS CO-TRUSTEES OF THE MITCHELL FAMILY TRUST DATED JULY 2, 2003,

CFC MANAGEMENT, LLC,

THE OTHER PARTIES HERETO, AND

MARC MITCHELL, AS THE EQUITYHOLDER REPRESENTATIVE

DATED AS OF November 7, 2023

TABLE OF CONTENTS

ARTICLE 1 I	Purchase and sale; Closing	3
1.1	Purchase and Sale of Assets	3
1.2	Purchase and Sale of Purchased Interests	5
1.3	CFC IPA Deferred Payment.	5
1.4	Purchase Price and Closing Payments	7
1.5	Purchase Price Allocation	8
1.6	Closing	ç
1.7	Withholding	ç
1.8	Non-Assignable Contracts	10
ARTICLE 2 I	Purchase Price Adjustments	10
2.1	Closing Consideration Adjustment	10
2.2	Post-Closing Determinations.	11
2.3	Post-Closing Adjustment Amount.	14
ARTICLE 3 I	Representations and Warranties of the Equityholders	16
3.1	Capacity	17
3.2	Authority and Enforceability	17
3.3	Title to Equity Interests	17
3.4	No Conflict	18
3.5	Legal Proceedings	18
3.6	No Brokerage	18
ARTICLE 4 I	Representations and Warranties of the Group Companies	18
4.1	Organization and Good Standing: Subsidiaries	19
4.2	Authorization and Effect	19
4.3	Capitalization	19
4.4	Non-Contravention; Consents	21
4.5	Financial Statements	21
4.6	Absence of Certain Changes, Events and Conditions	23
4.7	<u>Material Contracts</u>	23
4.8	Tangible Assets	25
4.9	Real Property	25
4.10	Legal Proceedings; Judgments	26
4.11	Compliance with Applicable Laws	27
4.12	Permits	27
4.13	Healthcare Regulatory Matters	28
4.14	Intellectual Property	29

4.15	Privacy and Security Compliance	31
4.16	Insurance	31
4.17	Taxes	31
4.18	Employment Matters	33
4.19	Employee Benefit Matters	35
4.20	Environmental Matters	37
4.21	Affiliate Transactions	37
4.22	Illegal Payments	37
4.23	Brokers	38
4.24	Investment Company Status	38
4.25	Investment Representations and Warranties	38
ARTICLE 5 Re	presentations and Warranties of Buyer Parties	40
5.1	Organization and Authority of Buyer	40
5.2	Non-Contravention; Consents	40
5.3	Investment Intention	41
5.4	Sufficiency of Funds; Solvency	41
5.5	Legal Proceedings	41
5.6	Title	42
5.7	Brokers	42
5.8	R&W Insurance Policy	42
5.9	Issuance of the Buyer Parent Shares	43
5.10	Exclusive Representations	43
ARTICLE 6 Co	venants	43
6.1	Confidentiality	43
6.2	Publicity	43
6.3	Further Assurances	43
6.4	Conduct of Business Prior to the Applicable Closing	44
6.5	Access to Information	47
6.6	Approvals and Consents	47
6.7	No Negotiation	48
6.8	Regulatory Matters	48
6.9	Employee Matters	49
6.10	Tail Policies.	49
6.11	Efforts: Regulatory Filings and Consents	50
6.12	Excluded Real Property	52
6.13	R&W Insurance Policy	52
6.14	Advisory Committee	52
6.15	Termination of 401(k) Plan	52
6.16	Non-Competition; Non-Solicitation; Non-Disparagement	53
6.17	Financing Cooperation	54
6.18	Permit Transfers	55
6.19	<u>Further Transfers</u>	55

- 6.15 6.16 Termination of 401(k) Plan Non-Competition; Non-Solicitation; Non-Disparagement
- 6.17 Financing Cooperation
- 6.18 6.19 Permit Transfers Further Transfers

6.20	Use of Names	55
6.21	Wrong Pockets: Payments	55
6.22	Additional Restrictions on Transfers	56
6.23	<u>Rule 144</u>	56
6.24	Insurance	57
6.25	Bulk Sales	57
6.26	Letter of Credit Requirements	57
6.27	Additional Covenants	57
6.28	Disclosure Schedule Supplement	58
ARTICLE 7 T	Fax matters	58
7.1	Tax Covenants	58
7.2	Straddle Period	59
7.3	Tax Indemnification	60
7.4	Refunds	61
7.5	Contests	61
7.6	Cooperation and Exchange of Information	61
7.7	Post-Closing Actions	62
7.8	Intended Tax Treatment; Purchase Price Allocation	62
7.9	Transfer Taxes	64
7.10	Overlap	64
ARTICLE 8 C	Conditions to Closing	64
8.1	Conditions to the Obligations of All Parties with Respect to the Initial Closing	64
8.2	Initial Closing Conditions to Obligations of Buyer Parties	64

8.3	<u>Initial Closing Conditions to Obligations of CFC IPA and the IPA Equityholder</u>	67
8.4	<u>Second Closing Conditions to Obligations of All Parties</u>	68
8.5	<u>Second Closing Conditions to Obligations of Buyer Parties</u>	69
8.6	<u>Second Closing Conditions to Obligations of the Plan Companies and the AHMS Equityholders</u>	71
ARTICLE 9 Te	rmination	72
9.1	Termination of the Agreement	72
9.2	Effect of Termination	73
ARTICLE 10 I	ndemnification	73
10.1	Survival	73
10.2	Indemnification by the Equityholders	74
10.3	Indemnification by the Buyer Parties	77
10.4	Certain Limitations	78
10.5	Indemnification Procedures	79
10.6	Tax Treatment of Indemnification Payments	81

-iii-

10.7	Recourse	81
10.8	Exclusive Remedies	83
10.9	R&W Insurance Policy	83
10.10	No Contribution	83
ARTICLE 11 I	Definitions	84
11.1	Certain Definitions	84
11.2	Interpretation	106
ARTICLE 12 N	Miscellaneous	106
Incriede 121		100
12.1	Equityholder Representative	106
12.2	Notices	108
12.3	Disclosure Schedule	109
12.4	Assignment	110
12.5	No Third Party Beneficiaries	110
12.6	<u>Counterparts</u>	110
12.7	Amendment and Modification: Waiver	111
12.8	Expenses	111
12.9	Governing Law; Submission to Jurisdiction	111
12.10	Specific Performance	112
12.11	Remedies Cumulative	112
12.12	Construction	112
12.13	Severability	112
12.14	Entire Agreement	113
12.15	Time is of the Essence	113
12.16	No Rescission	113
12.17	Certain Matters Regarding Representation of the Equityholders and the Companies	113
12.18	Release	114

-iv-

EXHIBITS, ATTACHMENTS AND SCHEDULES

Exhibit B-1Calculations of Cash-to-Claims Ratio for CFC IPA, Minimum IPA Cash-to-Claim Ratio Amount, CFC IPA TNE, and Tangible Net EquityExhibit B-2Calculations of CFCHP TNE, and Tangible Net EquityExhibit CDeposit Escrow Amount Joint InstructionExhibit DBill of SaleExhibit FForm of Assignment AgreementExhibit GForm of Logal Opinion of Counsel to the Buyer Parent Share TransfereesSchedule 1.3(a)Member Month Amount Example CalculationsSchedule 1.3(c)CFC IPA Deferred Payment CovenantsSchedule 6.9Continuing EmployeesSchedule 6.9Conduct of Business Prior to the Applicable ClosingSchedule 8.2(e)Initial Closing Required ConsentsSchedule 8.2(f)Initial Closing Required ConsentsSchedule 8.5(f)Second Closing Required ConsentsSchedule 8.5(g)Second Closing Related Party Contract TerminationsSchedule 8.5(g)Second Closing Related Party Contract TerminationsSchedule 1.1(b)Excluded AssetsSchedule 1.1(b)Permited Liens	Exhibit A	Escrow Agreement
Exhibit B-2Calculations of CFCHP TNE, and Tangible Net EquityExhibit CDeposit Escrow Amount Joint InstructionExhibit CDeposit Escrow Amount Joint InstructionExhibit DBill of SaleExhibit EForm of Assignment AgreementExhibit FForm of Joinder AgreementExhibit GForm of Legal Opinion of Counsel to the Buyer Parent Share TransfereesSchedule 1.3(a)Member Month Amount Example CalculationsSchedule 2.1(b)Tangible Net Equity MethodologySchedule 2.1(b)Tangible Net Equity MethodologySchedule 6.4Conduct of Business Prior to the Applicable ClosingSchedule 7.8Draft AHMS Purchase Price AllocationSchedule 8.2(e)Initial Closing Required ConsentsSchedule 8.2(f)Initial Closing Required ConsentsSchedule 8.5(g)Second Closing Related Party Contract TerminationsSchedule 8.5(g)Second Closing Related Party Contract TerminationsSchedule 1.1(a)Assumed ContractsSchedule 1.1(b)Excluded Assets		6
Exhibit DBill of SaleExhibit EForm of Assignment AgreementExhibit FForm of Joinder AgreementExhibit GForm of Legal Opinion of Counsel to the Buyer Parent Share TransfereesSchedule 1.3(a)Member Month Amount Example CalculationsSchedule 1.3(c)CFC IPA Deferred Payment CovenantsSchedule 2.1(b)Tangible Net Equity MethodologySchedule 6.4Conduct of Business Prior to the Applicable ClosingSchedule 7.8Draft AHMS Purchase Price AllocationSchedule 8.2(e)Initial Closing Required ConsentsSchedule 8.2(f)Initial Closing Related Party Contract TerminationsSchedule 8.5(g)Second Closing Related Party Contract TerminationsSchedule 8.5(g)Second Closing Related Party Contract TerminationsSchedule 1.1(a)Assumed ContractsSchedule 1.1(b)Excluded Assets	Exhibit B-2	
Exhibit DBill of SaleExhibit EForm of Assignment AgreementExhibit FForm of Joinder AgreementExhibit GForm of Legal Opinion of Counsel to the Buyer Parent Share TransfereesSchedule 1.3(a)Member Month Amount Example CalculationsSchedule 1.3(c)CFC IPA Deferred Payment CovenantsSchedule 2.1(b)Tangible Net Equity MethodologySchedule 6.4Conduct of Business Prior to the Applicable ClosingSchedule 7.8Draft AHMS Purchase Price AllocationSchedule 8.2(e)Initial Closing Required ConsentsSchedule 8.2(f)Initial Closing Related Party Contract TerminationsSchedule 8.5(g)Second Closing Related Party Contract TerminationsSchedule 8.5(g)Second Closing Related Party Contract TerminationsSchedule 1.1(a)Assumed ContractsSchedule 1.1(b)Excluded Assets	Exhibit C	Deposit Escrow Amount Joint Instruction
Exhibit FForm of Joinder AgreementExhibit GForm of Legal Opinion of Counsel to the Buyer Parent Share TransfereesSchedule 1.3(a)Member Month Amount Example CalculationsSchedule 1.3(c)CFC IPA Deferred Payment CovenantsSchedule 2.1(b)Tangible Net Equity MethodologySchedule 6.4Conduct of Business Prior to the Applicable ClosingSchedule 6.9Continuing EmployeesSchedule 7.8Draft AHMS Purchase Price AllocationSchedule 8.2(e)Initial Closing Required ConsentsSchedule 8.5(f)Second Closing Required ConsentsSchedule 8.5(g)Second Closing Required ConsentsSchedule 8.5(g)Second Closing Required Contract TerminationsSchedule 1.1(a)Assumed ContractsSchedule 1.1(b)Excluded Assets	Exhibit D	1
Exhibit FForm of Joinder AgreementExhibit GForm of Legal Opinion of Counsel to the Buyer Parent Share TransfereesSchedule 1.3(a)Member Month Amount Example CalculationsSchedule 1.3(c)CFC IPA Deferred Payment CovenantsSchedule 2.1(b)Tangible Net Equity MethodologySchedule 6.4Conduct of Business Prior to the Applicable ClosingSchedule 6.9Continuing EmployeesSchedule 7.8Draft AHMS Purchase Price AllocationSchedule 8.2(e)Initial Closing Required ConsentsSchedule 8.5(f)Second Closing Required ConsentsSchedule 8.5(g)Second Closing Required ConsentsSchedule 8.5(g)Second Closing Required Contract TerminationsSchedule 1.1(a)Assumed ContractsSchedule 1.1(b)Excluded Assets	Exhibit E	Form of Assignment Agreement
Schedule 1.3(a)Member Month Amount Example CalculationsSchedule 1.3(c)CFC IPA Deferred Payment CovenantsSchedule 2.1(b)Tangible Net Equity MethodologySchedule 6.4Conduct of Business Prior to the Applicable ClosingSchedule 6.9Continuing EmployeesSchedule 7.8Draft AHMS Purchase Price AllocationSchedule 8.2(e)Initial Closing Required ConsentsSchedule 8.2(f)Initial Closing Required ConsentsSchedule 8.5(f)Second Closing Required ConsentsSchedule 8.5(g)Second Closing Related Party Contract TerminationsSchedule 11.1(a)Assumed ContractsSchedule 11.1(b)Excluded Assets	Exhibit F	
Schedule 1.3(c)CFC IPA Deferred Payment CovenantsSchedule 2.1(b)Tangible Net Equity MethodologySchedule 6.4Conduct of Business Prior to the Applicable ClosingSchedule 6.9Continuing EmployeesSchedule 7.8Draft AHMS Purchase Price AllocationSchedule 8.2(e)Initial Closing Required ConsentsSchedule 8.2(f)Initial Closing Related Party Contract TerminationsSchedule 8.5(f)Second Closing Required ConsentsSchedule 8.5(g)Second Closing Related Party Contract TerminationsSchedule 11.1(a)Assumed ContractsSchedule 11.1(b)Excluded Assets	Exhibit G	Form of Legal Opinion of Counsel to the Buyer Parent Share Transferees
Schedule 2.1(b)Tangible Net Equity MethodologySchedule 6.4Conduct of Business Prior to the Applicable ClosingSchedule 6.9Continuing EmployeesSchedule 7.8Draft AHMS Purchase Price AllocationSchedule 8.2(e)Initial Closing Required ConsentsSchedule 8.2(f)Initial Closing Related Party Contract TerminationsSchedule 8.5(f)Second Closing Required ConsentsSchedule 8.5(g)Second Closing Related Party Contract TerminationsSchedule 11.1(a)Assumed ContractsSchedule 11.1(b)Excluded Assets	Schedule 1.3(a)	
Schedule 6.4Conduct of Business Prior to the Applicable ClosingSchedule 6.9Continuing EmployeesSchedule 7.8Draft AHMS Purchase Price AllocationSchedule 8.2(e)Initial Closing Required ConsentsSchedule 8.2(f)Initial Closing Related Party Contract TerminationsSchedule 8.5(f)Second Closing Required ConsentsSchedule 8.5(g)Second Closing Related Party Contract TerminationsSchedule 11.1(a)Assumed ContractsSchedule 11.1(b)Excluded Assets	Schedule 1.3(c)	CFC IPA Deferred Payment Covenants
Schedule 6.9Continuing EmployeesSchedule 7.8Draft AHMS Purchase Price AllocationSchedule 8.2(e)Initial Closing Required ConsentsSchedule 8.2(f)Initial Closing Related Party Contract TerminationsSchedule 8.5(f)Second Closing Required ConsentsSchedule 8.5(g)Second Closing Related Party Contract TerminationsSchedule 11.1(a)Assumed ContractsSchedule 11.1(b)Excluded Assets	Schedule 2.1(b)	Tangible Net Equity Methodology
Schedule 7.8Draft AHMS Purchase Price AllocationSchedule 8.2(e)Initial Closing Required ConsentsSchedule 8.2(f)Initial Closing Related Party Contract TerminationsSchedule 8.5(f)Second Closing Required ConsentsSchedule 8.5(g)Second Closing Related Party Contract TerminationsSchedule 11.1(a)Assumed ContractsSchedule 11.1(b)Excluded Assets	Schedule 6.4	Conduct of Business Prior to the Applicable Closing
Schedule 8.2(e)Initial Closing Required ConsentsSchedule 8.2(f)Initial Closing Related Party Contract TerminationsSchedule 8.5(f)Second Closing Required ConsentsSchedule 8.5(g)Second Closing Related Party Contract TerminationsSchedule 11.1(a)Assumed ContractsSchedule 11.1(b)Excluded Assets	Schedule 6.9	Continuing Employees
Schedule 8.2(f)Initial Closing Related Party Contract TerminationsSchedule 8.5(f)Second Closing Required ConsentsSchedule 8.5(g)Second Closing Related Party Contract TerminationsSchedule 11.1(a)Assumed ContractsSchedule 11.1(b)Excluded Assets	Schedule 7.8	Draft AHMS Purchase Price Allocation
Schedule 8.5(f)Second Closing Required ConsentsSchedule 8.5(g)Second Closing Related Party Contract TerminationsSchedule 11.1(a)Assumed ContractsSchedule 11.1(b)Excluded Assets	Schedule 8.2(e)	Initial Closing Required Consents
Schedule 8.5(g)Second Closing Related Party Contract TerminationsSchedule 11.1(a)Assumed ContractsSchedule 11.1(b)Excluded Assets	Schedule 8.2(f)	Initial Closing Related Party Contract Terminations
Schedule 11.1(a)Assumed ContractsSchedule 11.1(b)Excluded Assets	Schedule 8.5(f)	Second Closing Required Consents
Schedule 11.1(b) Excluded Assets	Schedule 8.5(g)	Second Closing Related Party Contract Terminations
	Schedule 11.1(a)	Assumed Contracts
Schedule 11.1(c) Permitted Liens	Schedule 11.1(b)	Excluded Assets
	Schedule 11.1(c)	Permitted Liens

ASSET AND EQUITY PURCHASE AGREEMENT

This ASSET AND EQUITY PURCHASE AGREEMENT (this "Agreement"), dated as of November7, 2023, is entered into by and among METROPOLITAN IPA, a California professional corporation ("<u>PC Buyer</u>"); APOLLOCARE ENABLEMENT OF CA, LLC, a California limited liability company ("<u>MSO GP Buyer</u>"); NETWORK MEDICAL MANAGEMENT,INC., a California corporation ("<u>MSO LP Buyer</u>" and, together with MSO GP Buyer, the "<u>MSO Buyers</u>" and together with PC Buyer, the "<u>Buyers</u>" and each a, "<u>Buyer</u>"); APOLLO MEDICAL HOLDINGS,INC., a Delaware corporation the stock of which is publicly traded on the Nasdaq ("<u>Buyer Parent</u>" and together with Buyers, "<u>Buyer Parties</u>" and each, a "<u>Buyer Party</u>"); COMMUNITY FAMILY CARE MEDICAL GROUP IPA,INC., a California corporation (" <u>CFC IPA</u>"); ADVANCED HEALTH MANAGEMENT SYSTEMS, L.P., a California limited partnership ("<u>AHMS</u>" and together with the CFC IPA, the "<u>Companies</u>" and each, a "<u>Companies</u>" and each a "<u>Companies</u>" and the general partner of AHMS General Partner of AHMS Eimited Partners" (AHMS General Partners") the other limited partners of AHMS set forth in the signatu

RECITALS

WHEREAS, Buyer Parent is an Affiliate of MSO LP Buyer and MSO GP Buyer, and Buyer Parent owns one hundred percent (100%) of the issued and outstanding Equity Interests of MSO LP Buyer, and MSO LP Buyer owns one hundred percent (100%) of the issued and outstanding Equity Interests of MSO GP Buyer;

WHEREAS, CFC IPA operates as an independent practice association, which may enter into agreements with organizations such as insurance companies, health plans, self-insured employers, government payors, health maintenance organizations, medical groups, independent practice associations, and other third party payors (collectively referred to as "<u>Plans</u>") for the arrangement of the provision of healthcare services to subscribers or enrollees of the Plans (together with any other business activity conducted by CFC IPA during the twelve (12) month period immediately preceding the Initial Closing or currently contemplated to be conducted, the "<u>IPA Business</u>");

WHEREAS, Community Family Care Health Plan, Inc., a California corporation ("<u>CFCHP</u>") operates a California managed care services business pursuant to its license as a health care service plan by the California Department of Managed Health Care ("<u>DMHC</u>") under the Knox-Keene Act (the "<u>Plan Business</u>");

-1-

WHEREAS, AHMS is in the business of providing management, consulting, administrative, and other support services to entities that provide or arrange for the provision of professional healthcare services;

WHEREAS, pursuant to that certain Management Services Agreement, dated as of September15, 2017, by and between the Companies, AHMS agreed to provide to CFC IPA certain management, consulting, administrative, and support services ("CFC Management Services");

WHEREAS, IPA Beneficial Owner beneficially owns one hundred percent (100%) of the issued and outstanding shares of CFC IPA (the "CFC IPA Shares") and Marc Mitchell and Alex Mitchell as Trustees of the CFC Management Irrevocable Trust, own one hundred percent (100%) of the issued and outstanding general partnership interest of AHMS General Partner;

WHEREAS, AHMS General Partner owns all of the issued and outstanding general partnership interest of AHMS (collectively, the 'AHMS GP Interests');

WHEREAS, the AHMS Limited Partners collectively own all of the issued and outstanding limited partnership interest of AHMS (collectively, the "<u>AHMS LP</u> <u>Interests</u>" and together with the AHMS GP Interests, the "<u>AHMS Interests</u>"), all as set forth in <u>Section 4.3(a)</u> of the Disclosure Schedule;

WHEREAS, (i)AHMS owns all of the issued and outstanding Equity Interests of Community Family Care Holdings, LLC, a California limited liability company ("<u>CFCH</u>"), and (ii)CFCH owns all of the issued and outstanding Equity Interests of CFCHP (<u>CFCHP</u> together with AHMS and CFCH, the "<u>Plan Companies</u>" and, together with CFC IPA, the "<u>Group Companies</u>");

WHEREAS, pursuant to the terms and conditions set forth herein, effective as of the Initial Closing, CFC IPA wishes to sell, transfer and assign to PC Buyer, and PC Buyer wishes to purchase, acquire and assume from CFC IPA the Acquired Assets and Assumed Liabilities, all as more specifically provided herein;

WHEREAS, pursuant to the terms and conditions set forth herein, (i)AHMS General Partner wishes to sell to MSO GP Buyer, and MSO GP Buyer wishes to purchase from AHMS General Partner, the AHMS GP Interests; and (ii)each AHMS Limited Partner wishes to sell to MSO LP Buyer, and MSO LP Buyer wishes to purchase from AHMS Limited Partner, the AHMS LP Interests, in each case for the cash purchase price specified herein;

WHEREAS, Ron Brandt is also an AHMS Limited Partner; and

WHEREAS, in furtherance of the transactions contemplated hereunder, on June7, 2022, Buyer Parent deposited One Million Six Hundred Fifty Thousand Dollars (\$1,650,000.00) into an escrow account ("Deposit Escrow Account") with City National Bank, a National Banking Association (the "Deposit Escrow Agent"), pursuant to the terms of that certain Escrow Agreement, dated as of June7, 2022 (the "Deposit Escrow Agreement"), by and among Buyer Parent, Accie Mitchell, as representative of all the shareholders and partners, as applicable, of the Group Companies, and the Escrow Agent, and on June15, 2023, a portion of the escrowed funds were released to the applicable Group Company from the Deposit Escrow Account, leaving Eight Hundred Twenty-Five Thousand Dollars (\$825,000) (the "Deposit Escrow Amount") remaining in the Deposit Escrow Account.

AGREEMENT

ARTICLE 1 Purchase and sale; Closing

1.1 Purchase and Sale of Assets.

(a) <u>Acquired Assets</u>. On the terms and subject to the conditions of this Agreement, at the Initial Closing, PC Buyer shall purchase, acquire and accept from CFC IPA, and CFC IPA shall sell, convey, assign, transfer and deliver to PC Buyer, all assets, properties, rights, titles and interests of CFC IPA, whether tangible or intangible, real or personal and wherever located, including the IPA Business and the Enumerated Assets, but excluding the Excluded Assets (collectively, the "<u>Acquired Assets</u>"), in each case, free and clear of all Liens, other than Permitted Liens). Notwithstanding the foregoing, CFC IPA may retain copies of any Contracts, documents, correspondence or records which are required to be retained pursuant to any Applicable Law or are subject to the attorney-client privilege, for financial reporting purposes, for Tax purposes, for legal defense or prosecution purposes or otherwise in connection with the Excluded Liabilities or the Excluded Assets. The Parties acknowledge and agree that CFC IPA shall have the right, in its sole discretion, to assign its right to receive fifty percent (50%) of the Buyer Parent Shares to AMFIT, LLC, a Delaware limited liability company ("<u>AMFIT</u>"), and the remaining fifty percent (50%) to MMFIT, LLC, a Delaware limited liability company (<u>MMFIT</u>," and together with AMFIT, the "<u>Buyer Parent Shares Transferees</u>") pursuant to assignment agreements each substantially in the form attached as<u>Exhibit E</u> hereto and that, subject to the limited Closing, in the event that CFC IPA assigns its rights pursuant to the preceding sentence, CFC IPA shall cause to be delivered to the Buyer Parties (i)joinders to this Agreement each substantially in the form attached as <u>Exhibit F</u> hereto and executed by each of the Buyer Parent Share Transferees and (ii)a legal opinion of counsel substantially in the form attached as <u>Exhibit F</u> hereto and executed by each of the Buyer Parent Share Transferees via assignments referenced in the preceding sentence are exempt from registration under

(b) Assumed Liabilities. On and subject to the terms set forth in this Agreement, at the Initial Closing, as additional consideration for the Acquired Assets, PC Buyer hereby assumes only the following Liabilities of CFC IPA (and thereafter shall pay, perform, discharge or otherwise satisfy in accordance with their respective terms) (the "Assumed Liabilities"): (i)Liabilities of the IPA Business and the Acquired Assets to the extent incurred (A)in the Ordinary Course of Business (excluding any Liability arising out of any breach of contract, breach of warranty, violation of Applicable Law, tort, misappropriation, infringement, misappropriation, or Action outstanding or to the extent arising from facts or circumstances that occurred prior to the Initial Closing), or (B)on or after the Initial Closing Date; (ii)Liabilities of CFC IPA under any Assumed Contracts but only to the extent (A)such Assumed Contracts are assigned to PC Buyer or PC Buyer otherwise receives the rights and benefits of such Assumed Contracts pursuant to <u>Section 1.8</u>, and (B)such obligations did not arise from a breach by CFC IPA thereunder that occurred prior to the Initial Closing Date; (iii) any Liability for the Acquired Assets, or operation, or management of the IPA Business or Acquired Assets from and after the Initial Closing.

-3-

Excluded Liabilities. Other than the Assumed Liabilities, PC Buyer is not assuming, and shall not be liable for, any Liabilities of CFC IPA or its (c) owners (including without limitation IPA Beneficial Owner) with respect to with the IPA Business or Acquired Assets (" Excluded Liabilities"), including without limitation (1) any liability or obligation of CFC IPA for Taxes to the extent related to the Acquired Assets or the IPA Business (other than those described in the prior clause (iv)) (2) any employment or payroll Taxes attributable to the applicable Pre-Closing Tax Period the payment of which CFC IPA has elected to defer until after the Initial Closing pursuant to the CARES Act or any similar legislation enacted by any Governmental Authority or pursuant to the Presidential Memorandum on Deferring Payroll Tax Obligations in Light of the Ongoing COVID-19 Disaster, as issued on August 8, 2020 (such clauses (1) and (2), "<u>Seller Taxes</u>"), (3) all Liabilities arising under or in connection with (i) any Employee Benefit Plan sponsored or maintained by CFC IPA, or (ii) any Employee Benefit Plan that was sponsored or maintained by CFC IPA at the time of its termination, in each case as indicated in Section 4.19(a) of the Disclosure Schedule, (4) any Liabilities arising out of the employment or engagement of any Person with CFC IPA or termination of employment, engagement or service of any Person from CFC IPA, (5) any Indebtedness of CFC IPA, (6) any Company Transaction Expenses attributable to CFC IPA; (7) any Liability arising from CFC IPA's violation of Environmental Requirements, (8) Liabilities with respect to, or arising out of, the Excluded Assets or Excluded Real Property, (9) any Liability arising under Healthcare Laws, (10) any allegation, claim or threatened claim or charge arising out of equity ownership of CFC IPA, (11) any transfer or assignment of rights to receive any consideration for the Acquired Assets, (12) any allegation, claim, charge or threatened claim or charge by any current or former employee or independent contractor of CFC IPA involving or relating to facts or circumstances that occurred prior to the Initial Closing Date, (13) obligations arising from checks issued but not presented for payment; (14) any Liabilities of CFC IPA arising out of the Actions filed against CFC IPA set forth on in Section 4.10 of the Disclosure Schedule; and (15) outstanding balances owed to Memorial Hospital of Gardena for quarterly settlements received in 2022 in excess of its portion of the final fiscal year 2022 risk pool. CFC IPA hereby acknowledges that it is retaining all Excluded Liabilities and shall be fully responsible and liable for all Excluded Liabilities.

(d) <u>CFC IPA Purchase Price</u>. In consideration for the sale, transfer, conveyance, assignment and delivery of the Acquired Assets pursuant to <u>Section 1.1</u>, (i)at the Initial Closing, the PC Buyer shall pay to CFC IPA an aggregate amount equal to One Hundred Thirteen Million One Hundred Sixty-Eight Thousand Dollars (\$113,800,000), consisting of Ninety-Three Million Eight Hundred Thousand Dollars (\$93,800,000) of cash (the "<u>CFC IPA Cash Purchase Price</u>"), and Buyer Parent Shares (which shares have an aggregate value equal to the Buyer Parent Share Consideration, which shall be payable and subject to adjustment pursuant to <u>Article 2</u>, and (ii)thereafter, as described in and contingent upon satisfaction of the conditions described in <u>Section 1.3</u>, the PC Buyer shall pay to CFC IPA, or an applicable assignee, earnout payments of up to Fifteen Million Dollars (\$15,000,000) (the "<u>CFC IPA Deferred Payments</u>"), the payment of which is subject to the terms and conditions set forth in<u>Section 1.3</u> (the "<u>Initial Closing Purchase Price</u>").

-4-

1.2 Purchase and Sale of Purchased Interests.

(a) <u>Purchase and Sale of Purchased Interests</u>. On the terms and subject to the conditions of this Agreement:

(i) At the Second Closing, AHMS General Partner shall sell, transfer, convey, assign and deliver to MSO GP Buyer, and MSO GP Buyer shall purchase, acquire and accept from AHMS General Partner, all of the right, title and interests of AHMS General Partner in, to and under the AHMS GP Interests, free and clear of all Liens (other than those imposed by securities Applicable Laws); and

(ii) At the Second Closing, each AHMS Limited Partner shall sell, transfer, convey, assign and deliver to MSO LP Buyer, and MSO LP Buyer shall purchase, acquire and accept from AHMS Limited Partner, all of the right, title and interests of AHMS Limited Partner in, to and under the AHMS LP Interests, free and clear of all Liens (other than those imposed by securities Applicable Laws) (the AHMS GP Interests and the AHMS LP Interests are referred to as the "<u>Purchased Interests</u>").

(b) <u>AHMS Purchase Price</u>. In consideration for the sale, transfer, conveyance, assignment and delivery of the AHMS GP Interests and the AHMS LP Interests pursuant to <u>Section 1.2(a)</u>, at the Second Closing, the applicable MSO Buyer shall pay to the AHMS General Partner and the AHMS Limited Partners cash in immediately available funds in an aggregate amount equal to Fifty-Two Million Dollars (\$52,000,000) (the "<u>Second Closing Purchase Price</u>"), which shall be payable and

subject to adjustment pursuant to Article 2 and allocated among the AHMS General Partner and the AHMS Limited Partners in accordance with the Closing Payments Schedule.

1.3 CFC IPA Deferred Payment.

(a) <u>CFC IPA Deferred Payment</u>. If earned, and subject to the terms and conditions set forth in this Agreement and <u>Schedule 1.3(c)</u>, the Buyer Parties shall pay CFC IPA, or an applicable assignee, Five Million Dollars (\$5,000,000) within thirty (30) days following the date on which it is finally determined pursuant to this Agreement that the Actual Member Month Amount during any Measurement Period, measured as of the last day of any Measurement Period, equals or exceeds the Target Member Month Amount (such condition, the "<u>CFC IPA Deferred Payment Condition</u>"); provided, that if the CFC IPA Deferred Payment Condition is not achieved with respect to any Measurement Period, CFC IPA, or an applicable assignee, shall be eligible for a Catch-Up Payment, which shall be paid in accordance with the terms set forth in this Agreement.

(b) Catch-Up Payments. In the event that the CFC IPA Deferred Payment Condition is not achieved with respect to either or both of the First Measurement Period and/or the Second Measurement Period, CFC IPA, or an applicable assignee, shall be entitled to receive a Catch-Up Payment that shall be paid concurrent with the payment(s) of the CFC IPA Deferred Payment(s) for the Second Measurement Period and/or Third Measurement Period, as applicable, if (x) the sum of the Actual Member Month Amount in the First Measurement Period and Second Measurement Period, divided by two (2), equals or exceeds the Target Member Month Amount, (y) the sum of the Actual Member Month Amount in the Second Measurement Period and Third Measurement Period, divided by two (2), equals or exceeds the Target Member Month Amount, (s) the sum of the Actual Member Month Amount in the First Measurement Period and Third Measurement Period, divided by two (2), equals or exceeds the Target Member Month Amount; or (z) the sum of the Actual Member Month Amount in the First Measurement Period, Second Measurement Period and Third Measurement Period, divided by two (2), equals or exceeds the Target Member Month Amount; provided, that, for the avoidance of doubt, no additional Catch-Up Payments shall be earned after the expiration of the Third Measurement Period regardless of whether the Target Member Month Amount is achieved after such time; provided, however, that to the extent any Catch-Up Payment is earned pursuant to clause (x), (y)and/or (z)of this Section 1.3(b), CFC IPA, or an applicable assignee, shall be entitled to receive such Catch-Up Payment even if such payment is not made until after the expiration of the Third Measurement Period. An example illustration is included in Schedule 1.3(a).

(c) <u>CFC IPA Deferred Payment Covenants</u>. Until the end of the Third Measurement Period, neither the Buyer Parties nor any of their Affiliates shall (i)directly or indirectly, take any action the primary purpose or effect of which is to decrease the maximum CFC IPA Deferred Payment payable, or (ii)take any of the actions set forth on <u>Schedule 1.3(c)</u> without the prior written consent of CFC IPA (not to be unreasonably withheld). It is understood and agreed that this <u>Section 1.3(c)</u> and the terms set forth on <u>Schedule 1.3(c)</u> are material covenants on which IPA Equityholder has relied in respect of the CFC IPA Deferred Payments and entering into this Agreement. If a Buyer Party breaches any of the covenants set forth on <u>Schedule 1.3(c)</u> at any time prior to the end of the Third Measurement Period, the Buyer Parties shall within five (5)Business Days after becoming aware of such breach deliver written notification to IPA Equityholder of such breach. The Buyer Parties have thirty (30) days to cure any such breach. If the Buyer Parties do not cure such breach within such thirty (30) day period as determined in good faith by the Parties, IPA Equityholder shall be entitled to acceleration and payment of the CFC IPA Deferred Payments"). In the event of a dispute between Buyer Parties and IPA Equityholder regarding whether such breach has occurred or whether a breach has been cured, such dispute shall be resolved in accordance with <u>Section 12.9</u>. Within thirty (30) days of final determination that such a breach has occurred (whether pursuant to <u>Section 12.9</u> if such breach has been disputed or by written notice by outside counsel to the Buyer Parties articulating that a breach has occurred), the Buyer Parties shall pay to the IPA Equityholder the Unpaid CFC IPA Deferred Payments. The Parties acknowledge and agree that in no event will the Unpaid CFC IPA Deferred Payments be payable more than once. The Unpaid CFC IPA Deferred Payments shall be the sole and exclusive remedy of IPA Equityholder against the Buyer Par

-6-

(d) Determination. On or before forty-five (45) days following each Measurement Period, the Buyer Parties shall provide to CFC IPA a statement, which sets forth the Actual Member Month Amount with respect to such Measurement Period and all supporting information related thereto (the "<u>Initial CFC IPA Deferred Payment Statement</u>"). CFC IPA shall have thirty (30) calendar days following receipt of the Initial CFC IPA Deferred Payment Statement to review the Initial CFC IPA Deferred Payment Statement ("<u>CFC IPA Deferred Payment Review Period</u>"). During each month of each Measurement Period, the Buyer Parties shall deliver to the CFC IPA a monthly report setting forth the Actual Member Month Amount for such month. On or prior to the last day of the CFC IPA Deferred Payment Review Period, CFC IPA may object to the Initial CFC IPA Deferred Payment Statement, and, after the expiration of the CFC IPA Deferred Payment Review Period, all undisputed calculations set forth in the Initial CFC IPA Deferred Payment Statement, and, after the expiration of the CFC IPA Deferred Payment Review Period, all undisputed calculations in the Initial CFC IPA Deferred Payment Statement shall be final, conclusive and binding on the Parties. If CFC IPA timely disputes any calculations in the Initial CFC IPA Deferred Payment Review Period either Buyer Parties or CFC IPA may elect to have the Accounting Expert resolve any such amounts that remain in dispute, in which event the Buyer Parties and CFC IPA will instruct the Accounting Expert to resolve such dispute in accordance with <u>Section 2.2</u>, applied *mutatis mutandis*.

(e) All payments made pursuant to this <u>Section1.3</u> shall be treated by all parties hereto for income Tax purposes as adjustments to the Initial Closing Purchase Price.

1.4 <u>Purchase Price and Closing Payments</u>. The aggregate consideration for the Acquired Assets and the Purchased Interests shall be the Buyer Parent Shares, the CFC IPA Cash Purchase Price, the Second Closing Purchase Price and to the extent earned, the CFC IPA Deferred Payments including all Catch-Up Payments which are paid (collectively, the "<u>Purchase Price</u>").

- (a) <u>At the Initial Closing</u>:
- Section 1.1(a);

(i) as consideration for the Acquired Assets, PC Buyer shall deliver or cause to be delivered the Buyer Parent Shares in accordance with

(ii) as consideration for the Acquired Assets, PC Buyer shall, on behalf of CFC IPA, deliver the Initial Closing Payment by wire transfer of immediately available funds to the account(s) specified in the Closing Payments Schedule;

(iii) the applicable Parties shall sign and deliver the joint instructions attached hereto as Exhibit C instructing the Deposit Escrow Agent to (A)release (1)Five Hundred and Four Thousand Dollars (\$504,000) (the "<u>CFC Indemnity Escrow Amount</u>") of the Deposit Escrow Amount into an account (the "<u>Escrow Account</u>") with the Escrow Agent, and (2)Three Hundred Thousand Dollars (\$300,000) (the "<u>CFC Adjustment Escrow Amount</u>") of the Deposit Escrow Amount into the Escrow Account, pursuant to the terms of the Escrow Agreement in the form attached hereto as <u>Exhibit A</u> (the "<u>Escrow Agreement</u>"), and (B) release the remaining Twenty-One Thousand Dollars (\$21,000) of the Deposit Escrow Amount to CFC IPA;

(iv) on behalf of CFC IPA, PC Buyer shall deliver, by wire transfer of immediately available funds, Fifty Thousand Dollars (\$50,000) (the <u>'IPA</u> <u>Representative Fund</u>'') to the account(s) specified in the Closing Payments Schedule; and

(v) on behalf of CFC IPA, PC Buyer shall pay or cause to be paid the amount for all the Initial Closing Estimated Transaction Expenses set forth on a schedule to be delivered by CFC IPA no later than five (5)Business Days prior to the Initial Closing Date by wire transfer of immediately available funds to accounts specified in the Closing Payments Schedule.

-7-

(b) <u>At the Second Closing</u>:

(i) as consideration for the AHMS GP Interests and AHMS LP Interests, the applicable MSO Buyer shall deliver, by wire transfer of immediately available funds, the Second Closing Payment to the account(s) specified in the Closing Payments Schedule;

(ii) on behalf of the AHMS Equityholders, the applicable MSO Buyer shall deliver, by wire transfer of immediately available funds, Fifty Thousand Dollars (\$50,000) (the "<u>AHMS Representative Fund</u>" and together with the IPA Representative Fund, the "<u>Equityholder Representative Fund</u>") to the account(s) specified in the Closing Payments Schedule;

(iii) on behalf of the AHMS Equityholders, the applicable MSO Buyer shall deposit or cause to be deposited into the Escrow Account the AHMS Indemnity Escrow Amount with the Escrow Agent, in accordance with the Escrow Agreement;

(iv) on behalf of the AHMS Equityholders, the applicable MSO Buyer shall deposit or cause to be deposited into the Escrow Account the AHMS Adjustment Escrow Amount, in accordance with the Escrow Agreement;

(v) on behalf of the AHMS Equityholders, the applicable MSO Buyer shall deposit or cause to be deposited into the Escrow Account the AV Escrow Amount, in accordance with the Escrow Agreement (the "<u>AV Escrow</u>"); and

(vi) on behalf of the applicable Plan Companies, the applicable MSO Buyer shall pay or cause to be paid the amount for all the Second Closing Estimated Transaction Expenses set forth on a schedule to be delivered by the Equityholder Representative no later than five (5)Business Days prior to the Second Closing by wire transfer of immediately available funds to accounts specified in the Closing Payments Schedule; <u>provided however</u>, that in the case of Second Closing Estimated Transaction Expenses owed to employees of a Plan Company, the applicable Buyer Party shall contribute such amounts to such Plan Company and cause such Plan Company to pay such amounts (less applicable withholding and any Taxes required to be paid by such Plan Company with respect thereto) to the applicable employees as soon as practicable following the Second Closing.

1.5 <u>Purchase Price Allocation</u>. At least five (5)Business Days prior to the Initial Closing, the Equityholder Representative shall deliver to the Buyer Parties a schedule (the "<u>Closing Payments Schedule</u>"), allocating the Initial Closing Payment and Second Closing Purchase Price amongst CFC IPA, the AHMS General Partner and the AHMS Limited Partners, as applicable, and other payees described above and including wire instructions for any applicable Persons receiving payments from the Buyer Parties in connection with each Closing. The Buyer Parties shall not be responsible for any amounts paid in accordance with the Closing Payments Schedule.

-8-

1.6 <u>Closing</u>.

(a) The consummation of the sale of the Acquired Assets and the assumption of the Assumed Liabilities (the "<u>Initial Closing</u>") shall take place electronically via mutual exchange of facsimile or portable document format (.PDF) signatures commencing at 10:00 a.m. Pacific Standard Time on the third (3rd) Business Day after the last of the conditions set forth in <u>Section 8.1</u>, <u>Section 8.2</u> and <u>Section 8.3</u> have been satisfied or waived (other than conditions which, by their nature, are to be satisfied on the Initial Closing Date), or at such other time as shall be agreed upon in writing by the Buyer Parties and CFC IPA. The date on which the Initial Closing is actually held is referred to herein as the "<u>Initial Closing Date</u>." The Initial Closing shall be deemed to be effective at 12:01 A.M.Pacific Standard Time on the Initial Closing Date (the <u>Initial Effective Time</u>") for all purposes, except as may otherwise be expressly provided herein.

(b) The consummation of the sale of the AHMS GP Interests and the AHMS LP Interests (the "Second Closing") shall take place electronically via mutual exchange of facsimile or portable document format (.PDF) signatures commencing at 10:00 a.m.Pacific Standard Time on the third (3rd) Business Day after the last of the conditions set forth in Section 8.4, Section 8.5 and Section 8.6 have been satisfied, or at such other time as shall be agreed upon in writing by the Buyer Parties and the Equityholder Representative. The date on which the Second Closing is actually held is referred to herein as the "Second Closing Date." The Second Closing shall be deemed to be effective at 12:01 A.M.Pacific Standard Time on the Second Closing Date (the "Second Effective Time") for all purposes, except as may otherwise be expressly provided herein.

1.7 Withholding. Buyers and Buyer Parent shall be entitled to deduct and withhold from the consideration or other amounts otherwise payable pursuant to this Agreement to any Person such amounts as they are required to deduct and withhold with respect to such payment under the Code, or any provision of state, local or foreign law. Except for any withholding required as a result of a failure to deliver an IRS FormW-9 and any withholding on payments under any compensatory payments made in connection with the Transactions, before making any such deduction or withholding, Buyers and Buyer Parent shall deliver to the Equityholder Representative notice of the intention to make such deduction or withholding, and such notice shall be given at least three (3)Business Days before such deduction or withholding is required, in order for the Equityholder Representative to obtain reduction of or relief from such deduction or withholding. To the extent that amounts are so withheld by Buyers or Buyer Parent, such withheld amounts shall be (a)paid to the appropriate Tax Authority and (b)treated for all purposes of this Agreement as having been paid to the appropriate recipient in respect of which such deduction and withholding was made by Buyers.

1.8 Non-Assignable Contracts. Notwithstanding anything to the contrary in this Agreement, to the extent that the assignment hereunder by CFC IPA to PC Buyer of any Assumed Contract or Permit is not permitted or is not permitted without the consent of any other party to such Assumed Contract or any Governmental Authority, neither this Agreement nor the Bill of Sale shall be deemed to constitute an assignment of any such Assumed Contract or Permit if such consent is not given or if such assignment otherwise would constitute a breach of, violation of or cause a loss of benefits under, any such Assumed Contract or Permit, and except as otherwise provided in this Agreement, PC Buyer shall not assume any obligations or liabilities under any such Assumed Contract or Permit. CFC IPA shall use Commercially Reasonable Efforts to obtain all consents and waivers necessary for the sale, transfer, assignment, conveyance and delivery of the Assumed Contracts, Permits and the Acquired Assets to PC Buyer hereunder and, if any such consent is not obtained or if such assignment is not permitide irrespective of such consent, CFC IPA shall, for a period of twelve (12) months following the Initial Closing Date, cooperate with PC Buyer following the Initial Closing Date in any reasonable and lawful arrangement designed to provide PC Buyer or its

Affiliates with the rights and benefits under any such Assumed Contract or Permit, including enforcement for the benefit of PC Buyer, at the request of and cost and expense of PC Buyer, of any and all rights of CFC IPA thereunder (including rights against any other party thereto arising out of any breach or cancellation of any such Assumed Contract by such other party) and, if requested by PC Buyer, at PC Buyer's sole cost and expense, to the extent legally permitted, acting as an agent on behalf of PC Buyer or as PC Buyer shall otherwise reasonably require; provided, that, if PC Buyer is provided the benefits of any Assumed Contracts or Permits, then PC Buyer shall assume the Liabilities of CFC IPA under such Assumed Contract or Permits, as applicable, but only to the extent the Liabilities thereunder arise after the Initial Closing Date.

ARTICLE 2

Purchase Price Adjustments

2.1 Closing Consideration Adjustment.

(a) At least five (5)business days prior to the Initial Closing Date, CFC IPA shall calculate and deliver to the Buyer Parties a statement (the "Estimated Initial Closing Statement") of (it being clarified, that the following shall be calculated without duplication so as to avoid "double-counting" of assets or amounts otherwise taken into account in determining the Purchase Price and/or any adjustments thereto): (i)the estimated amount of the Closing IPA Cash-to-Claims Amount (the "Estimated Closing IPA Cash-to-Claims Amount"), (ii) a statement setting forth the estimated amount of the Tangible Net Equity applicable to CFC IPA only (and no other Group Companies) (the "Estimated Closing IPA TNE"), and the Estimated Closing IPA TNE Surplus or the Estimated Closing IPA TNE Deficiency, as the case may be, which shall be calculated onsistent with the historical methodology employed by CFC IPA when making regulatory filings to the DMHC, (iii)the estimated amount of the Initial Closing Funded Indebtedness as of the Initial Closing thereto (the "Initial Closing Estimated Transaction Expenses"), and (v)the calculation of the Initial Closing Payment based on the calculation of the foregoing amounts. Notwithstanding anything contained herein to the contrary, the amounts set forth on the Estimated Initial Closing Statement shall be prepared in accordance with GAAP.

-10-

(b) At least five (5)business days prior to the Second Closing Date, the Plan Companies shall calculate and deliver to the Buyer Parties a statement (the "Estimated Second Closing Statement") of (it being clarified, that the following shall be calculated without duplication so as to avoid "double-counting" of assets or amounts otherwise taken into account in determining the Purchase Price and/or any adjustments thereto): (i)the estimated amount of the Closing Cash for the Group Companies other than CFC IPA (the "Estimated Closing Cash"), (ii) a statement setting forth the estimated amount of the Tangible Net Equity applicable to CFCHP and the Plan Companies) (the "Estimated Closing CFCHP TNE"), and the Estimated Closing CFCHP TNE Surplus or the Estimated Closing CFCHP TNE Deficiency, as the case may be, which shall be calculated consistent with the methodology set forth in <u>Schedule 2.1(b</u>)attached hereto, (iii)the estimated amount of the Second Closing Funded Indebtedness"), (iv) the estimated amount of the Group Companies' Transaction Expenses as of the Second Closing (but excluding the Initial Closing Estimated Transaction Expenses"), and (v) the calculation of the Second Closing Payment based on the calculation of the foregoing amounts, in each case calculated without giving effect to the consummation of the transactions contemplated by this Agreement. Notwithstanding anything contained herein to the contrary, the amounts set forth on the Estimated Second Closing Statement shall not reflect any purchase accounting adjustments as a result of the Transactions and shall be prepared in accordance with GAAP.

2.2 Post-Closing Determinations.

(a) Within one hundred eighty (180) days following the Initial Closing Date, the Buyer Parties shall cause to be prepared and delivered to the Equityholder Representative a statement (the "<u>Initial Closing Statement</u>") of the Buyer Parties' calculations of the following amounts (it being clarified, that the following shall be calculated without duplication so as to avoid "double-counting" of assets or amounts otherwise taken into account in determining the Purchase Price and/or any adjustments thereto): (i) as of the Initial Closing Date, (A)the Closing IPA Cash-to-Claims Amount, (B)the Target IPA TNE (the "<u>Closing IPA TNE</u>"), (C)the Initial Closing Funded Indebtedness, and (D)the Transaction Expenses of the Group Companies as of the Initial Closing ("<u>Initial Closing Company Transaction Expenses</u>"), and (ii) a recalculation of the Initial Closing Payment based on the calculation of the foregoing amounts.

(b) Within twelve (12) months following the Second Closing Date, the Buyer Parties shall cause to be prepared and delivered to the Equityholder Representative a statement (the "Second Closing Statement") of the Buyer Parties' calculations of the following amounts (it being clarified, that the following shall be calculated without duplication so as to avoid "double-counting" of assets or amounts otherwise taken into account in determining the Purchase Price and/or any adjustments thereto): (i)as of the Second Closing Date, (A)the Target CFCHP TNE (the "Closing CFCHP TNE"), (B)the Second Closing Funded Indebtedness, and (C)the Transaction Expenses of the Group Companies as of the Second Closing ("Second Closing Company Transaction Expenses"), and (ii)a recalculation of the Second Closing Payment based on the calculation of the foregoing amounts.

-11-

(c) Notwithstanding anything contained herein to the contrary, the amounts set forth on the Initial Closing Statement or Second Closing Statement, as applicable, shall not reflect any purchase accounting adjustments as a result of the Transactions and shall be prepared in accordance with GAAP. If the Buyer Parties fail to deliver the Initial Closing Statement or Second Closing Statement, as applicable, by its due date, then the Estimated Initial Closing Statement and Estimated Second Closing Statement (and all components thereof) shall be deemed (x)accepted by, and final and binding upon, the Buyer Parties and conclusive for all purposes hereunder and (y)to be the Initial Closing Statement or Second Closing Statement, as applicable. The "First Calculation of the Initial Closing Paymenf" shall mean the recalculation of the Initial Closing Payment using the Initial Closing Statement proposed by the Buyer Parties under Section 2.2(a); provided, that if any of the amounts on such Initial Closing Payment using the Second Closing Payment. The "Final Calculation of the Second Closing Paymenf" shall mean the recalculation of the Second Closing Payment using the Second Closing Payment to Section 2.2(b); provided, that if any of the amounts on such Second Closing Payment using the Equityholder Representative pursuant to Section 2.2(b); provided, that if any of the amounts on such Second Closing Payment using the Equityholder Representative pursuant to Section 2.2(b); provided, that if any of the amounts on such Second Closing Statement are objected to by the Equityholder Representative pursuant to Section 2.2(c), the final determination of such amounts on such Second Closing Statement are objected to by the Equityholder Representative pursuant to Section 2.2(b); provided, that if any of the amounts on such Second Closing Statement are objected to by the Equityholder Representative pursuant to Section 2.2(c), the final determination of such amounts in accordance with thisSection 2.2 shall be used in the Final Calculation

(d) Within twelve (12) months following the Initial Closing Date, the Buyer Parties shall cause to be prepared and delivered to the Equityholder Representative a statement (the "<u>12-Month Closing Statement</u>") of the Buyer Parties' calculations of (i)the same amounts and line items (it being clarified, that the following shall be calculated without duplication so as to avoid "double-counting" of assets or amounts otherwise taken into account in determining the Purchase Price and/or any adjustments thereto) set forth in the Estimated Initial Closing Statement or the Initial Closing Statement, as applicable, or the dates set forth in <u>Section 2.2(a)</u>, as the case may be, and (ii)a recalculation of the Initial Closing Payment, based on the calculation of the foregoing amounts. Notwithstanding anything contained herein to the contrary, the amounts set forth on the 12-Month Closing Statement shall not reflect any purchase accounting adjustments as a result of the Transactions and shall be prepared in accordance with GAAP. If the applicable, shall be deemed (x)accepted by, and final and binding upon, the Buyer Parties and conclusive for all purposes hereunder, and (y)to be the 12-Month Closing Statement, delivered as of the twelve (12) month anniversary of the Initial Closing Date. The <u>Final Calculation of the Initial Closing Payment</u>' shall mean the recalculation of the Initial Closing Statement roposed by the Buyer Parties under this <u>Section 2.2(d)</u>; provided, that if any of the amounts on such 12-Month Closing Statement are objected to by the Equityholder Representative pursuant to <u>Section 2.2(e)</u>, the final determination of such amounts in

accordance with this Section 2.2 shall be used in the Final Calculation of the Initial Closing Payment.

(c) The Equityholder Representative shall notify the Buyer Parties in writing, within the applicable Review Period, whether the Equityholder Representative accepts the Initial Closing Statement, the Second Closing Statement or the 12-Month Closing Statement. If the Equityholder Representative notifies the Buyer Parties that the Equityholder Representative does not accept the Initial Closing Statement, the Second Closing Statement, the Second Closing Statement or the 12-Month Closing Statement or the 12-Month Closing Statement (or any component thereof), the Equityholder Representative shall include with such notice its reasons for such non-acceptance and the adjustments the Equityholder Representative believes should be made to the Initial Closing Statement, the Second Closing Statement or the 12-Month Closing Statement, as applicable. In the event that the Equityholder Representative does not accepted the Initial Closing Statement (and all components thereof), as applicable, which shall be final, binding and conclusive for all purposes hereunder. If the Equityholder Representative has submitted a notice of non-acceptance, then thereafter, the Buyer Parties, on the one hand, and the Equityholder Representative, on the other hand, shall negotiate in good faith with each other to reach an agreement in respect of the disputed portions of the Initial Closing Statement, as applicable delivered by the Buyer Parties to the Equityholder Representative for a period of thirty (30) days following the Buyer Parties' receipt of such notice of non-acceptance.

-12-

(f) During the applicable Review Period, the Buyer Parties shall provide the Equityholder Representative and any of its Representatives with reasonable access to all books and records (including work papers in accordance with the access policies of each Group Companies' accountants) in its possession or control and personnel who prepared such information as may be reasonably required to enable the Equityholder Representative to evaluate the accuracy of the Initial Closing Statement, the Second Closing Statement and the 12-Month Closing Statement, and the Buyer Parties shall respond promptly and in good faith to inquiries from the Equityholder Representative, as applicable, or their respective Representatives related to such review.

(g) If the Buyer Parties and the Equityholder Representative are unable to resolve all disagreements identified by the Equityholder Representative pursuant to <u>Section 2.2(e)</u>, within such applicable thirty (30) days after delivery to the Buyer Parties of written notice of such disagreements, then such disagreements shall be submitted for final and binding resolution to Deloitte or, if Deloitte is unable to serve, Buyer and the Equityholder Representative, as applicable, shall appoint by mutual agreement a Neutral Accounting Firm (the "Accounting Expert") who, acting as experts and not arbitrators, shall resolve the disputed items only.

(h) Each Party shall be permitted to present a supporting brief to the Accounting Expert (which supporting brief shall also be concurrently provided to the other Party) within ten (10)days following the appointment of the Accounting Expert. Within five (5)days following receipt of a supporting brief, the receiving Party may present a responsive brief to the Accounting Expert (which responsive brief shall also be concurrently provided to the other Party). No discovery will be permitted and no arbitration hearing will be held. The Accounting Expert shall only consider the briefs of the Parties, and shall not conduct any independent review, in determining those items and amounts disputed by the Parties, and the Accounting Expert shall consider only those items and amounts which are identified by the Parties as being in dispute. The Accounting Expert shall resolve the dispute by selecting either the position of the Buyer Parties or the Equityholder Representative (and may not propose or make any alternative position) and must resolve the matter in accordance with the terms and provisions of this Agreement. The Accounting Expert shall deliver to the Buyer Parties and the accounting Expert shall on any event within sixty (60) days after its appointment, a written report setting forth the resolution of any such disagreement determined in accordance with the terms of this Agreement. The determination of the Accounting Expert shall be final, binding and non-appealable.

(i) All fees of the Accounting Expert shall be allocated between the Buyer Parties, on the one hand, and the Equityholder Representative, on the other hand, such that the amount paid by the Equityholder Representative bears the same proportion that the aggregate dollar amount unsuccessfully disputed by the Equityholder Representative bears to the total dollar amount of the disputed items that were submitted for resolution to the Accounting Expert, and the Buyer Parties shall pay the balance. For purposes of illustration only, if the Final Calculation of the Initial Closing Payment or the First Calculation of the Initial Closing Payment is disputed to be \$1,000 by the Equityholder Representative and \$900 by the Buyer Parties and is determined by the Accounting Expert to be \$960, then the Equityholder Representative would bear forty percent (40%) of the fees of the Accounting Expert and the Buyer Parties would bear sixty percent (60%), because the amount disputed was \$100 and the amount unsuccessfully disputed by the Equityholder Representative was \$40.

-13-

2.3 <u>Post-Closing Adjustment Amount</u>.

(a) Following the final determination of the First Calculation of the Initial Closing Payment or the Final Calculation of the Second Closing Payment, as applicable, in accordance with Section 2.2, a payment pursuant to this Section 2.3 shall be made as follows:

(i) If the Closing Actual Adjustment with respect to CFC IPA is a positive amount, then Buyer Parent and the Equityholder Representative shall execute joint written instructions to the Escrow Agent instructing the Escrow Agent to release from the Escrow Account to CFC IPA an amount equal to fifty percent (50%) of the CFC Adjustment Escrow Amount, and then Buyer Parent shall pay to CFC IPA an aggregate amount equal to the absolute value of such positive Closing Actual Adjustment.

(ii) If the Closing Actual Adjustment with respect to AHMS is a positive amount, then Buyer Parent and the Equityholder Representative shall execute joint written instructions to the Escrow Agent instructing the Escrow Agent to release from the Escrow Account to the AHMS Equityholders an amount equal to the AHMS Adjustment Escrow Amount, and then Buyer Parent shall pay to the AHMS Equityholders an aggregate amount equal to the absolute value of such positive Closing Actual Adjustment.

(iii) If the Closing Actual Adjustment with respect to CFC IPA is a negative amount, then Buyer Parent and the Equityholder Representative shall execute joint written instructions to the Escrow Agent instructing the Escrow Agent to release from the Escrow Account to the applicable Buyer an amount equal to the lesser of (A)the absolute value of such Closing Actual Adjustment, and (B)the CFC Adjustment Escrow Amount, and in the event that the absolute value of such negative Closing Actual Adjustment Escrow Amount, CFC IPA shall pay to the applicable Buyer an aggregate amount equal to the absolute value of such negative Closing Actual Adjustment less the CFC Adjustment Escrow Amount;

(iv) If the Closing Actual Adjustment with respect to AHMS is a negative amount, then Buyer Parent and the Equityholder Representative shall execute joint written instructions to the Escrow Agent instructing the Escrow Agent to release from the Escrow Account to the applicable Buyer an amount equal to the lesser of (A)the absolute value of such Closing Actual Adjustment, and (B)the AHMS Adjustment Escrow Amount; <u>provided</u>, that in the event that the absolute value of such negative Closing Actual Adjustment Escrow Amount, then such joint written instructions to the Escrow Agent to release to the remaining amount of the AHMS Adjustment Escrow Amount, then such joint written instructions to the Escrow Agent shall instruct the Escrow Agent to release to the remaining amount of the AHMS Adjustment Escrow Amount to the AHMS Equityholders in accordance with their respective Pro Rata Portion; <u>provided</u>, however, that in the event that the absolute value of such negative Closing Actual Adjustment is more than the AHMS Adjustment Escrow Amount, the AHMS Equityholders shall pay in accordance with their respective Pro Rata Portion to the applicable Buyer an aggregate amount equal to the absolute value of such negative Closing Actual Adjustment less the AHMS Adjustment Escrow Amount.

(v) Any Closing Actual Adjustment payable by either Party pursuant to this <u>Section 2.3(a)</u>shall be paid promptly by the party required to pay such Closing Actual Adjustment, but in no event later than five (5)Business Days following the date on which the First Calculation of the Initial Closing Payment or the Final Calculation of the Second Closing Payment, as applicable, becomes binding pursuant to <u>Section 2.2</u>. Payment by any Party of the Closing Actual Adjustment, or any portion thereof, shall be made in immediately available funds via wire transfer to an account(s)designated in writing by the Party entitled to receive such payment. Any joint written instructions to be delivered to the Escrow Agent pursuant to this <u>Section 2.3(a)</u> shall be delivered promptly, but in no event later than five (5) calendar days following the date on which the First Calculation of the Initial Closing Payment or Final Calculation of the Second Closing Payment, as applicable, becomes binding pursuant to <u>Section 2.3(a)</u> shall be delivered promptly, but in no event later than five (5) calendar days following the date on which the First Calculation of the Initial Closing Payment or Final Calculation of the Second Closing Payment, as applicable, becomes binding pursuant to <u>Section 2.2</u>.

(b) Following the final determination of the Final Calculation of the Initial Closing Payment, in accordance with Section 2.2, a payment pursuant to this Section 2.3 shall be made as follows:

(i) If the Final Initial Closing Adjustment is a positive amount, then Buyer Parent and the Equityholder Representative shall execute joint written instructions to the Escrow Agent instructing the Escrow Agent to release from the Escrow Account to CFC IPA the remaining amount of the CFC Adjustment Escrow Amount, and Buyer Parent shall pay to the Equityholder Representative an aggregate amount equal to the Final Initial Closing Adjustment.

(ii) If the Final Initial Closing Adjustment is a negative amount, then Buyer Parent and the Equityholder Representative shall execute joint written instructions to the Escrow Agent instructing the Escrow Agent to release from the Escrow Account to the applicable Buyer an amount equal to the absolute value of such Final Initial Closing Adjustment up to a maximum amount equal to the remaining balance of the CFC Adjustment Escrow Amount, and in the event that the absolute value of such Final Initial Closing Adjustment is more than the remaining balance of the CFC Adjustment Escrow Amount, CFC IPA shall pay to the applicable Buyer Party an aggregate amount equal to the absolute value of such Final Initial Closing Adjustment is such Final Initial Closing Adjustment Escrow Amount.

(iii) If after the payment of any such Final Initial Closing Adjustment, there remains any portion of the CFC Adjustment Escrow Amount in the Escrow Account, then Buyer Parent and the Equityholder Representative shall execute joint written instructions to the Escrow Agent instructing the Escrow Agent to release the remaining portion of the CFC Adjustment Escrow Amount to CFC IPA.

-15-

(iv) Any Final Initial Closing Adjustment payable by either Party pursuant to this <u>Section 2.3(a)</u>shall be paid promptly by the party required to pay such Final Initial Closing Adjustment, but in no event later than five (5)Business Days following the date on which the Final Calculation of the Initial Closing Payment, becomes binding pursuant to <u>Section 2.2</u>. Payment by any Party of the Final Initial Closing Adjustment, or any portion thereof, shall be made in immediately available funds via wire transfer to an account(s)designated in writing by the Party entitled to receive such payment. Any joint written instructions to be delivered to the Escrow Agent pursuant to this <u>Section 2.2(a)(a)</u>shall be delivered promptly, but in no event later than five (5)calendar days following the date on which the Final Calculation of the Initial Closing Payment becomes binding pursuant to <u>Section 2.2</u>.

(c) All payments made pursuant to this <u>Article 2</u> shall be treated by all parties hereto for Tax purposes as adjustments to the Purchase Price.

(d) <u>Calculation Methodology</u>. The Parties hereto agree that no amount shall be (or is intended to be) included, in whole or in part (either as an increase or a reduction), more than once in the calculation of the First Calculation of the Initial Closing Payment, the Final Calculation of the Initial Closing Payment, as applicable, or any other calculated amount pursuant to this Agreement. All of the amounts set forth on the Estimated Initial Closing Statement, Estimated Second Closing Statement, the Initial Closing Statement, the Second Closing Statement (and the individual elements thereof, as applicable) shall be (i)determined in accordance with the methodologies set forth on <u>Exhibit B-1</u> and <u>Exhibit B-2</u> and in accordance with GAAP and the definitions included in this Agreement (with the terms in this Agreement and <u>Exhibit B-1</u> and <u>Exhibit B-2</u> taking precedence if they are not in accordance with GAAP) and (ii)perpeared with the same rigor as if such accounts were being prepared as of a fiscal year-end, and (iii)not include any "purchase accounting" or other adjustment arising out of the consummation of the Transactions.

ARTICLE 3 Representations and Warranties of the Equityholders

Each Equityholder hereby, severally and not jointly, represents and warrants to the applicable Buyer Parties that the statements contained in this <u>Article 3</u> are true and correct as to such Equityholder only (and no other Equityholder) on the date hereof and each Closing Date (except as set forth in the corresponding Section of the Disclosure Schedule or in any other Sections of the Disclosure Schedule if the application of the disclosure is reasonably apparent on its face):

3.1 Capacity.

(a) Such Equityholder who is an individual has all legal capacity to execute and deliver this Agreement and each of the Transaction Documents to which such Equityholder is a party, to perform his or her obligations hereunder and thereunder and to consummate the transactions contemplated hereby or thereby, or will have such capacity in the case of the Transaction Documents that are to be executed after the date hereof.

(b) AHMS General Partner is a limited liability company duly organized, validly existing and in good standing under the Laws of California. AHMS General Partner has all requisite limited liability company power and authority to own and operate its properties and assets and to carry on its business as currently conducted. AHMS General Partner is duly qualified to do business and is in good standing in each jurisdiction where the ownership or operation of its properties and assets or the conduct of its business requires such qualification, except for failures to be so qualified or in good standing that would not, individually or in the aggregate, reasonably be expected to materially adversely affect AHMS General Partner's ability to execute, deliver or perform this Agreement or any Transaction Document, or to timely consummate the Transactions.

(c) Such Equityholder that is a trust is duly organized under the Applicable Law of its jurisdiction of formation. Each such Equityholder is a validly created trust and is in existence as at the date of this Agreement. The validly appointed trustees of such Equityholder are set forth on <u>Section 3.1(c)</u> of the Disclosure Schedule and no action has been taken or is proposed to remove such Person as trustee of such trust.

3.2 <u>Authority and Enforceability</u>. Such Equityholder has the requisite power and authority, and has taken all action necessary, to execute and deliver this Agreement and each Transaction Document and to perform its obligations hereunder and thereunder. The execution, delivery and performance of this Agreement has been, and each Transaction Document will be prior to the applicable Closing, duly authorized, executed and delivered by such Equityholder, and this Agreement constitutes, and each

Transaction Document to which he or it is a party when so executed and delivered will constitute, the legal, valid and binding obligations of such Equityholder, enforceable against such Equityholder in accordance with their terms; subject to bankruptcy, insolvency, reorganization, moratorium and similar Applicable Laws of general application relating to or affecting creditors' rights and to general equity principles. No other proceedings on the part of such Equityholder are necessary to approve and authorize the execution, delivery and performance of this Agreement and the Transaction Documents to which it is a party and the consummation of the transactions contemplated hereby and thereby.

3.3 <u>Title to Equity Interests</u>. Such Equityholder is the record and beneficial owner of, and has good and valid title to, the CFC IPA Shares and/or the AHMS Interests set forth opposite such Equityholder's name on <u>Section 3.3</u> of the Disclosure Schedule, free and clear of all Liens, other than those imposed by securities or healthcare Applicable Laws or provided in the Governing Documents of CFC IPA, or that result from any actions taken by or on behalf of any Buyer Party. Such Equityholder is not a party to any option, warrant, right, contract, call, put or other agreement or commitment providing for the disposition or acquisition of any of the Purchased Interests (other than this Agreement). Other than as set forth on <u>Section 3.3</u> of the Disclosure Schedule, such Equityholder does not have any other equity ownership interest in any of the Companies.

3.4 <u>No Conflict</u>. The execution and delivery by such Equityholder of this Agreement and each Transaction Document to which such Equityholder is a party, and the performance by him/her/it of any actions contemplated hereunder or thereunder, does not and will not, directly or indirectly (with or without notice or lapse of time):

(a) conflict with or violate any provision of the Governing Documents of such Equityholder;

(b) require notice, consent or approval under, conflict with, violate, result in a breach of, result in the acceleration of obligations, loss of a benefit or increase in Liabilities or fees under, create in any Person the right to terminate, cancel or modify, or cause a default under or give rise to any rights or penalties under (i)any provision of Applicable Law relating to such Equityholder, (ii)any provision of any Governmental Order to which such Equityholder or any of his/her/its properties are subject, or (iii) any provision of any Material Contract to which such Equityholder or its properties are bound; or

(c) cause such Equityholder to become subject to a Lien (other than Permitted Liens); or

(d) except for filings as may be required under the HSR Act, require a registration, filing, application, notice, consent, approval, order, qualification or waiver with, to or from any Governmental Authority.

3.5 Legal Proceedings. There are no Actions pending or, to the actual knowledge of such Equityholder after reasonable inquiry, threatened against or by such Equityholder or any of his/her/its applicable Affiliates that, individually or in the aggregate, would have a Material Adverse Effect on such Equityholder's ability to execute, deliver or perform this Agreement or any Transaction Document to which it is a party, or to timely consummate the transactions contemplated hereby. Such Equityholder is not subject to any order, writ, judgment, award, injunction or decree of any Governmental Authority of competent jurisdiction or any arbitrators.

3.6 <u>No Brokerage</u>. No broker, finder, financial advisor, investment banker or other Person is entitled to any brokerage commissions, finders' fees or similar compensation in connection with this Agreement or the Transactions or any other Transaction Document based on any arrangement or agreement made by or on behalf of such Equityholder.

ARTICLE 4

Representations and Warranties of the Group Companies

Each Company hereby, severally and not jointly, represents and warrants to the applicable Buyer Parties that the statements contained in this <u>Article 4</u> with respect to such Company and any of its Affiliated Group Companies, as applicable, are true and correct on the date hereof and each Closing Date (except as set forth in the corresponding Section of the Disclosure Schedule or in any other Section of the Disclosure Schedule if the application of the disclosure to the first Section is reasonably apparent):

-18-

4.1 Organization and Good Standing; Subsidiaries.

(a) Each Group Company is duly organized and validly existing, under the Applicable Law of its jurisdiction of organization, has all requisite corporate, limited liability company or limited partnership, as applicable, power and authority to own or lease and operate its properties and assets and to carry on its business as presently conducted, and is duly qualified or licensed to do business and is in good standing in each jurisdiction where the conduct of its business requires such qualification or license, except for such failure to be so qualified, licensed or in good standing, or to have such power or authority, that would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect.

(b) Except as set forth in <u>Section 4.1(b)</u>of the Disclosure Schedule, the Group Companies do not, and have never, (i)own(ed) of record or beneficially, directly or indirectly, (A)any shares of capital stock or securities convertible into capital stock of any Entity, (B)any participating interest in any Entity, or (C)any assets, outside the Ordinary Course of Business, comprising the business or obligations of any other Entity or (ii) control(led), directly or indirectly, any other Entity.

4.2 Authorization and Effect. Each Company has full corporate or limited partnership power and authority to enter into and perform its obligations under this Agreement and the Transaction Documents to which it is a party and to consummate the transactions contemplated hereunder and thereunder. The execution, delivery and performance by each applicable Company of this Agreement and the Transaction Documents to which such Company is a party and the consummation by such Company of the transactions contemplated hereunder and thereunder have been duly authorized by all requisite corporate or limited partnership action on the part of such Company and no other corporate or limited partnership proceedings on the part of such Company is necessary to authorize the execution, delivery and performance of this Agreement or to consummate the Transactions. This Agreement has been duly executed and delivered by each Company, and (assuming due authorization, execution and delivery by each other Party) this Agreement constitutes a legal, valid and binding obligation of such Company enforceable against such Company in accordance with its terms; subject to bankruptcy, insolvency, reorganization, moratorium and similar Applicable Laws of general application relating to or affecting creditors' rights and to general equity principles. When each terms; subject to bankruptcy, insolvency, reorganization, moratorium and similar Applicable Laws of general application relating to or affecting creditors' rights and to general equity rinciples. Without limiting the generality of the foregoing, CFC IPA has the unrestricted power and authority to sell, assign, transfer, convey and deliver the Acquired Assets to PC Buyer.

4.3 <u>Capitalization</u>.

authorized, issued and outstanding Equity Interests of each Group Company, including the identity of the Persons that are the record and beneficial owners thereof. All of the Equity Interests of each Group Company have been duly authorized, were validly issued, fully paid (as applicable), are free and clear of any and all Liens (other than those imposed by securities or healthcare Applicable Laws or provided in the Governing Documents of CFC IPA, or that result from any actions taken by or on behalf of any Buyer Party), and have been offered, issued, sold and delivered in compliance with applicable federal and state securities laws and the organizational documents of such Group Company. Except for the Equity Interests set forth on Section 4.3(a) of the Disclosure Schedule or as otherwise set forth on Section 4.3(a) of the Disclosure Schedule, no voting or non-voting units, other Equity Interests or other voting securities of any Group Company are issued, reserved for issuance or outstanding.

(b) (i)Except as set forth on <u>Section 4.3(b)(i)</u> of the Disclosure Schedule, no subscription, warrant, option, convertible or exchangeable security, or other right (contingent or otherwise) to purchase or otherwise acquire Equity Interests of any Group Company is authorized or outstanding, and (ii)there is no commitment by a Group Company to issue shares, subscriptions, warrants, options, convertible or exchangeable securities, or other such rights or to distribute to holders of any of its Equity Interests any evidence of indebtedness or asset, to repurchase or redeem any securities of any Group Company or to grant, extend, accelerate the vesting of, change the price of, or otherwise amend any warrant, option, convertible or exchangeable security or other such right.

(c) Except as set forth on <u>Section 4.3(c)</u> of the Disclosure Schedule, none of the Equity Interests of any Group Company is subject to any purchase option, call option, right of first refusal, first offer, co-sale or participation, preemptive right, subscription right or any similar right. Except for the Governing Documents for each of the Group Companies, there are no documents, instruments or agreements relating to the voting of any Equity Interests of any Group Company.

(d) Except as set forth on <u>Section 4.3(d)</u> of the Disclosure Schedule, there are no existing, authorized, issued or outstanding securities, options, warrants, calls, rights (including conversion rights, preemptive rights, co-sale rights, rights of first refusal, equity appreciation rights, "phantom" equity or equity-based rights, and similar rights), convertible or exchangeable securities or Contracts or obligations of any kind (contingent or otherwise) to which any Group Company or, to the Knowledge of the applicable Company, any equityholder of any Group Company, is a party or by which it is bound obligating such Group Company, directly or indirectly, to issue, deliver or sell, or cause to be issued, delivered or sold, additional Equity Interests of such Group Company. There are no outstanding obligations of any Group Company (contingent or otherwise) to (i)repurchase, redeem or otherwise acquire, directly or indirectly, any Equity Interests of any Group Company or (ii)make any investment (in the form of a loan, capital contribution or otherwise) in, or to provide any guarantee with respect to the obligations of, any Person.

(e) There are no bonds, debentures, notes or other Indebtedness of any Group Company having the right to vote or consent (or, convertible into, or exercisable or exchangeable for, securities having the right to vote or consent) on any matters on which the equityholders of any Group Company may vote. There are no voting trusts, irrevocable proxies or other contracts or understandings to which any Group Company, or, to the Company's Knowledge, any equityholder of any Group Company is a party or is bound with respect to the voting or consent of any Equity Interests.

(f) All distributions, dividends, repurchases and redemptions of the capital stock (or other equity interests) of the Group Companies were undertaken in compliance with the applicable Governing Documents then in effect, any agreement to which the applicable Group Company then was a party and in compliance with Applicable Law.

(g) The Group Companies have made available to the applicable Buyer Parties true, correct and complete copies of the Governing Documents for each of the Group Companies, which Governing Documents are in full force and effect.

-20-

4.4 <u>Non-Contravention: Consents</u>. The execution, delivery and performance by the applicable Group Company of this Agreement and each Transaction Document to which it is a party, and the performance by it of any actions contemplated hereunder or thereunder, does not and will not, directly or indirectly (with or without notice or lapse of time):

(a) conflict with, require any consent under, accelerate any obligation under or violate any provision of the Governing Documents of such Company;

(b) contravene, require notice, consent or approval under, conflict with, violate, result in a breach or violation of, conflict with or constitute or result in a breach or default (whether after the giving of notice, lapse of time or both), accelerate any obligation under, or give rise to a right of acceleration, cancellation or termination of, loss of a benefit or increase in Liabilities or fees under, create in any Person the right to terminate, cancel or modify, or cause a default under or give rise to any rights or penalties under (i) any provision of Applicable Law relating to such Group Company, (ii) any provision of any Governmental Order to which such Company or any of its properties are subject, (iii) any Permit, license or authorization to which any Group Company is a party or by which its assets are bound or (iv) any provision of any Material Contract, except as would not, individually or in the aggregate, reasonably be expected to constitute or result in a Material Adverse Effect;

(c) cause any of the assets of any Group Company to be subject to a Lien (other than Permitted Liens); or

(d) except for filings as may be required under the HSR Act, require a registration, filing, application, notice, consent, approval, order, qualification or waiver with, to or from any Governmental Authority or third party, except as would not, individually or in the aggregate, reasonably be expected to constitute or result in a Material Adverse Effect.

4.5 <u>Financial Statements</u>.

(a) The Group Companies have made available to the Buyer Parties (i)the unaudited financial statements consisting of the balance sheet of the Plan Companies, and (ii)the audited financial statements consisting of the balance sheet of CFC IPA, in each case, dated as of 2021 and 2022 and the related statements of income and retained earnings, and cash flow for the years then ended (collectively, the "<u>Year-End Financial Statements</u>"), and unaudited financial statements consisting of the balance sheet of the Plan Companies and CFC IPA, each dated as of September30, 2023, and the related statements of income and retained earnings for the 9-month period then ended (collectively, the "<u>Interim Financial Statements</u>" and together with the Year-End Financial Statements, the "<u>Financial Statements</u>"). The Financial Statements (i)were prepared in accordance with GAAP applied on a consistent basis throughout the periods covered, (ii)fairly present in all material respects the financial condition of the Plan Companies or CFC IPA, as the case may be, each as of the unaudited financial statements, to normal year-end audit adjustments that are neither individually nor in the aggregate material in amount and a lack of footnotes, and (iii)have been prepared from, and are in accordance with, the books and records of the applicable Group Company in all material respects. The balance sheet of the Plan Companies and the balance sheet of CFC IPA, in each case, as of December31, 2022 is referred to herein collectively, as the <u>"Year-End Balance Sheet</u>" and the date thereof as the "<u>Year-End Balance Sheet</u>" and the date thereof as the "<u>Interim Balance Sheet</u>" and the date thereof as the "<u>Interim Balance Sheet</u>" and the date thereof as the "<u>Interim Balance Sheet</u>" and the date thereof as the "<u>Interim Balance Sheet</u>" and the date thereof as the "<u>Interim Balance Sheet</u>" and the date thereof as the "<u>Interim Balance Sheet</u>" and the date thereof as the "<u>Interim Balance Sheet</u>".

(b) The books of account and financial records of the Group Companies are true and correct in all material respects and have been prepared and are maintained on a consistent basis throughout the period involved. The Group Companies have not made any changes in their accounting practice since the Year-End Balance Sheet Date. The Group Companies maintain a standard system of accounting established and administered on a consistent basis.

(c) There are no Liabilities of the Group Companies of any nature, whether absolute, accrued, contingent or otherwise or whether due or to become due, other than (i)Liabilities reflected on the Year-End Balance Sheet and (ii)individual Liabilities that have arisen after the Year-End Balance Sheet in the Ordinary Course of Business.

(d) All of the accounts receivable of each Group Company are valid and enforceable claims, and are subject to no set-off or counterclaim, except for normal course recoupment and related actions undertaken by payors or other set-offs or counterclaims that occur in the Ordinary Course of Business. Since the Year-End Balance Sheet Date, each Group Company has collected its accounts receivable in the Ordinary Course of Business and has not accelerated any such collections. No Group Company has any accounts receivable or loans receivable from any Person who is a Related Party.

(e) All accounts payable and notes payable of each Group Company arose in bona fide arm's length transactions in the Ordinary Course of Business and no such account payable or note payable is materially delinquent in its payment. Since the Year-End Balance Sheet Date, such Group Company has paid its accounts payable in the Ordinary Course of Business and in a manner which is consistent with its past practices. No Group Company has any accounts payable or note payable to any Person who is a Related Party.

(f) No Group Company has entered into any transactions involving the use of special purpose entities for any off-balance sheet activity other than as specifically described in the Financial Statements.

-22-

4.6 <u>Absence of Certain Changes, Events and Conditions</u> Since the Interim Balance Sheet Date, there has occurred no fact, event or circumstance which, individually or in the aggregate, has had, or could reasonably be expected to have a Material Adverse Effect and no Group Company has suffered any theft, damage, destruction or casualty loss, whether or not covered by insurance. Except as set forth in <u>Section 4.6</u> of the Disclosure Schedule, since the Interim Balance Sheet Date, the Group Companies have conducted the Business in the Ordinary Course of Business, and, without limiting the generality of the foregoing, there has not been with respect to any Group Company any action that would have been prohibited or otherwise restricted under <u>Section 6.4</u>, had such action been taken during the period between the Interim Balance Sheet Date and the Second Closing Date.

4.7 <u>Material Contracts</u>.

(a) <u>Section 4.7(a)</u> of the Disclosure Schedulesets forth an accurate and complete list of the following Contracts to which any Group Company is a party or the properties, rights or assets of the Group Companies bound (in each case, with specific reference to the applicable clause(s) below that relates to such Contract):

(i) any Contract or Judgment to which any of the Companies is a party or by which it is bound that involve obligations (contingent or otherwise) of, or payments to, the Group Companies in excess of One Hundred Thousand Dollars (\$100,000) per fiscal year, or that may not be terminated without penalty on sixty (60) days' notice or less;

on the other hand.

(ii) any Contract between a Group Company, on the one hand, and any of its Affiliates, Related Parties, or any Affiliate of any Related Party,

(iii) any Contract for employment or the provision of independent contractor or consulting services (other than Contracts with medical services providers) to which any Group Company is a party;

(iv) any Contract to which any Group Company other than CFC IPA is a party involving payment of a bonus, commissions, severance benefits, nonqualified deferred compensation, profit sharing, employee equity purchase, or other similar agreement to or for any employee, independent contractor or consultant (other than Contracts with medical services providers) of any Group Company;

-23-

(v) any bonds, debentures, mortgages, notes or other similar Indebtedness or Liabilities whatsoever, or any Contract to create or issue any bonds, debentures, mortgages, notes or other similar Indebtedness;

(vi) any Contract pursuant to which a Group Company guarantees the performance, Liabilities or obligations of any other Person;

(vii) any Contract limiting the freedom of the Companies to (A)engage in any line of business, (B)compete with any other Person or (C)solicit any employees, customers or suppliers of any other Person;

(viii) any (i)lease or sublease or similar Contract with any Person under which any Group Company is a lessee or sublease of any real property (collectively, the "Leased Real Property Documents"), or (ii) any lease, sublease or similar Contract with any Person under which any Group Company is a lessee or sublease of any tangible personal property;

(ix) any Contract that involves (A)the acquisition, merger or purchase of all or substantially all of the assets or business of a third party, or (B)the purchase or sale of assets, or a series of purchases and sales of assets, involving aggregate consideration of One Hundred Thousand Dollars (\$100,000) or more outside of the Ordinary Course of Business;

(x) any Contract pursuant to which any Group Company has (i)agreed to provide "most favored nation" status, "best pricing" or any other arrangement whereby any Group Company or Equityholder has agreed with any Person that such Person will receive at least as favorable terms and conditions that are provided by any Group Company to any other Person, or (ii) granted "exclusivity" or agreed to deal exclusively with, or granted exclusive rights or rights of first refusal to, any customer, vendor, supplier, distributor, contractor or other Person, or requiring the purchase of all or substantially all of such Group Company's requirements of a particular product from a vendor, supplier or other Person;

except as entered into in the Ordinary Course of Business;

(xii) any Contract relating to the settlement of any actions, suits, proceedings, orders, judgments, decrees, claims or investigations, including administrative charge or investigation by a Governmental Authority or other dispute;

- (xiii) any Contract with any Governmental Authority;
- (xiv) any Contract to which any Group Company is a party with any Plan;
- (xv) any Contract providing for liquidated damages or penalties on event of transfer, assignment or default;

-24-

(xvi) any Contract not executed in the Ordinary Course of Business that is not otherwise set forth on Section 4.7(a) of the Disclosure Schedule; or

(xvii) any Contract not otherwise referred to in this Section 4.7(a) that if terminated or expired without being renewed would result in a Material

Adverse Effect.

(b) The Contracts required to be disclosed on Section 4.7(a) of the Disclosure Schedule are referred to herein as the <u>Material Contracts</u>. Each Material Contract is valid and binding on the applicable Group Company in accordance with its terms and is in full force and effect. Except as set forth on Section 4.7(b) of the Disclosure Schedule, none of the Group Companies and to the Knowledge of the applicable Company, no other party thereto is in breach of or default under (or is alleged to be in breach of or default under) in any material respect, or has provided or received any written or, to the Knowledge of the applicable Company, oral notice of any intention to terminate, any Material Contract prior to the expiration of the term (including renewal terms) thereof, or to amend the material terms of any Material Contract outside of the Ordinary Course of Business. No party to a Material Contract has exercised any termination rights with respect thereto or has given written notice of any material dispute with respect thereto. Complete and correct copies of each Material Contract (including all modifications, amendments and supplements thereto and waivers thereunder) have been made available to Buyer.

4.8 <u>Tangible Assets</u>. Each Group Company has legal title to all of the material personal, tangible or intangible properties and assets it purports to own, free and clear of all Liens (except Permitted Liens). Each Group Company owns, leases, licenses or otherwise has the right to use all of the assets, properties and rights necessary to conduct its Business as presently being conducted. All material tangible assets of the Group Companies are in operational condition, normal wear and tear excepted, have been regularly and properly serviced and maintained in a manner that, to the Knowledge of the applicable Company, would not void or limit the coverage of any warranty thereon, other than items currently under, or scheduled for, repair or construction, and are adequate and fit to be used for the purposes for which they are currently used in the manner they are currently used, except as would not reasonably be expected to have a Material Adverse Effect. The Acquired Assets and the Excluded Assets constitute all of the assets used by CFC IPA in operating the IPA Business.

4.9 <u>Real Property</u>.

(a) AHMS owns real property located at 1102 Elden Avenue Los Angeles, CA 90006 (the "Excluded Real Property"). Prior to the Second Closing, the Excluded Real Property will be transferred to certain AHMS Limited Partners. No other Group Company owns any interest in real property.

(b) <u>Section 4.9(b)</u> of the Disclosure Schedule sets forth a correct and complete list of all real property that is leased and occupied by any Group Company (the "<u>Leased Real Property</u>"). The Group Companies have made available to Buyers correct and complete copies of the Leased Real Property Documents. To the Knowledge of the applicable Company, the occupancy, use and operation of the Leased Real Property complies in all material respects with all Applicable Law and Governmental Orders. There are no material pending or, to the Knowledge of the applicable Company, threatened, appropriation, condemnation, eminent domain or like proceedings, or any other Actions relating to the Leased Real Property.

-25-

(c) None of the Group Companies have leased or sublet, as lessor, sublessor, licensor or the like, any of the Leased Real Property to any Person and each of the Group Companies has the right to transfer its leasehold interest in the Leased Real Property as contemplated by this Agreement pursuant to the terms and conditions of the applicable Leased Real Property Document.

(d) The Leased Real Property has access, in all material respects, sufficient for the conduct of the Business in the Ordinary Course of Business, including to public roads.

(c) The Leased Real Property is in good condition and repair, including, without limitation, the roof, foundation, and other structural components of the Leased Real Property, as well as any improvements to the Leased Real Property, and furniture, fixtures, and equipment within the premises of the Leased Real Property, sufficient for the conduct of the Business as is currently conducted, with good and working utility connections, including, without limitation, electricity, internet, sanitary and storm sewer, potable water, natural gas and other utilities used in the operation of the Business at that location.

(f) The lessors and/or sublessors that are party to the Leased Real Property Documents are not in default of the applicable Leased Real Property

Document(s).

4.10 Legal Proceedings; Judgments.

(a) Except as set forth on Section 4.10(a) of the Disclosure Schedule, there is no Action pending or to, the Knowledge of the applicable Company, threatened, (i) against any Group Company or, to the Knowledge of the applicable Company, any officer, manager, or director thereof, in his/her capacity as an officer, manager or director or any of the assets or property owned or used by any Group Company, or any Person whose liability any Group Company has retained or assumed, either contractually or by operation of law, or (ii) that challenge, or that may have the effect of preventing, delaying, making illegal or otherwise interfering with, any of the other transactions contemplated by this Agreement or the Transaction Documents to which any Group Company or Equityholder is or will be a party, in each case at law or in equity, or before or by any Governmental Authority, and to the Knowledge of the applicable Company, there is no basis for any of the foregoing. Neither any Group Company is currently, or has been during the past four (4) years, subject to any outstanding investigation, order, writ, injunction, judgment or decree issued by any Governmental Authority.

(b) <u>Section 4.10(b)</u> of the Disclosure Schedule includes a description of all litigation, claims, proceedings or, to the Company's Knowledge, investigations involving any Group Company or any of its directors, members, managers, officers or employees, in their capacity as such, occurring, arising or existing during the past four (4)years. To the Company's Knowledge, no director, member, manager, officer or employee of any Group Company is, or during the past four (4)years has been,

subject to any order, writ, injunction, judgment or decree that prohibits such director, member, manager, officer or employee from engaging in or continuing any conduct, activity or practice relating to the Group Companies' business.

-26-

(c) No current Equityholder, officer, director or manager of any Group Company has been during the past four (4)years: (i)subject to voluntary or involuntary petition under the federal bankruptcy laws or any state insolvency law or the appointment of a receiver, fiscal agent or similar officer by a court for his or her business or property or that of any partnership of which he or she was a general partner or any corporation or business association of which he or she was an executive officer; (ii)convicted in a criminal proceeding or named as a subject of a pending criminal proceeding (excluding traffic violations and other minor offenses) or been otherwise accused of any act of moral turpitude; (iii)the subject of any order, judgment, or decree (not subsequently reversed, suspended or vacated) of any court of competent jurisdiction permanently or temporarily enjoining him or her from, or otherwise imposing limits or conditions on his or her ability to engage in any securities, investment advisory, banking, insurance or other type of business or acting as an officer or director of a public company; or (iv)found by a court of competent jurisdiction to have violated any federal or state commodities, securities or unfair trade practices law, which judgment or finding has not been subsequently reversed, suspended, or vacated.

4.11 Compliance with Applicable Laws. Except as set forth in Section 4.11 of the Disclosure Schedule, each Group Company is, and for the past four (4)years has been, in compliance with all Applicable Laws in all material respects. To the Knowledge of the applicable Company, no Group Company is, and has not been in the past four (4)years, in violation in any material respect of any Applicable Law or Order of any Governmental Authority, payment network, governmental program or self-regulatory organization, by which such Group Company is bound, whether by operation of Applicable Law or by Contract, or to which such Group Company is subject. No Group Company has, nor has it in the past four (4)years, received any written notice, or to the Company's Knowledge, oral notice, from any Governmental Authority of an investigation with respect to any violation with respect to any such Applicable Law or Order. There is no, and there has not been for the past five (5)years, pending or, to the Company's Knowledge, threatened regulatory action, investigation or inquiry (other than non-material routine or periodic inspections or reviews) against or affecting any Group Company. In the past four (4)years, no Group Company has received any written notices or other communications regarding any actual or alleged violation of, or any material failure to comply with, any Applicable Laws, or orders or instructions of any Governmental Authority.

4.12 <u>Permits</u>.

(a) The Group Companies owns or possesses, and has owned and possessed during the past four (4) years, all right, title and interest in and to all Permits necessary to (A)conduct the Business in the manner currently conducted and (B) permit the Group Companies to own and use the assets in the manner in which they are currently owned and used (the "<u>Material Permits</u>"). The Group Companies have delivered to Buyers accurate and complete copies of all Material Permits including all renewals and all amendments. The Material Permits are valid and in full force and effect.

-27-

Each Group Company is in compliance in all material respects with the terms and requirements of the Material Permits and all such Permits are valid, (h)in full force and effect and sufficient for the services provided by the Group Companies. In the past four (4) years, no Group Company has received any written notice from any Governmental Authority regarding (i) any actual or possible material violation of, or failure to comply with, any term or requirement of any Permits, or (ii) any actual or possible revocation, withdrawal, suspension, cancellation, termination or modification of any Permits. To the Knowledge of the Company, no event has occurred or condition or state of facts exists which constitutes or, after notice or lapse of time or both, would constitute a material breach or default under any such Permit or which, after notice or lapse of time or both, would permit revocation or termination of any such Permit. No Group Company has received written notice, or to the Company's Knowledge, oral communication, of the pending or threatened revocation, suspension, lapse, modification, or limitation of any such Permit; (iii)there are no provisions in, or agreements relating to, any such Permits that preclude or limit any Group Company from operating and carrying on its business as currently conducted; and (iv)no notice of cancellation, breach, termination, revocation or default or of any dispute concerning any Permit has been received in writing by any Group Company, in each case, other than routine expiration in accordance with the terms of any such Permit. Each Group Company has made available to the Buyer Parties correct and complete copies of any Governmental Authority or third-party audit or survey reports covering the last four (4)years (the "Audit Reports") and all plans of correction, if any, that any Group Company was required to submit in response to the Audit Reports, and all plans of correction have been accepted by the applicable Governmental Authority and have been or are in the process of being implemented. There are no material outstanding deficiencies or liabilities which any Governmental Authority has asserted in writing to any Group Company with respect to any Audit Report, and, for the past four (4)years, no material fine or penalty has been imposed on any Group Company by any Governmental Authority. None of the Equityholders nor any Group Company: (i)is party to any corporate integrity agreement with any Governmental Authority or (ii)has reporting obligations pursuant to any settlement agreement entered into with any Governmental Authority.

(c) All applications required to have been filed for the renewal of the Material Permits have been duly filed on a timely basis with the appropriate Governmental Authorities, and each other notice or filing required to have been given or made with respect to such Material Permits has been duly given or made on a timely basis with the appropriate Governmental Authorities.

(d) Notwithstanding anything to the contrary contained herein, no representations or warranties are made in this Agreement as to whether any Buyer Party or any of their Affiliates will be approved or otherwise granted the right to continue operating the Plan Business under any Material Permits.

4.13 Healthcare Regulatory Matters. Except as set forth on Section 4.13 of the Disclosure Schedule, during the four (4) -year period ending on the date hereof:

(a) The Group Companies are and have been in compliance in all material respects with all applicable Healthcare Laws and have timely filed all material reports, data and other information required to be filed with Governmental Authorities, including DMHC.

(b) The Group Companies, collectively and individually, have not been charged with, given written notice of, placed under any obligation to take remedial action under, is subject to any material Action with respect to, or, to the Knowledge of the applicable Company, placed under investigation with respect to, any applicable Healthcare Laws.

(c) Without limiting the generality of the foregoing, neither the Group Companies nor any Equityholder or Representative thereof, has directly or indirectly offered, paid, solicited or received, or made arrangements to offer, pay, solicit or receive, any remuneration, in cash or in kind, to or from any past, present, or potential customers, past, present, or potential suppliers, patients, physicians, healthcare professionals, medical staff members, contractors, referral sources, or third party payors either in material violation of any Healthcare Law or in order to obtain business or payments from such Persons that would reasonably be expected to subject the Group Companies or the Business to any material damage or penalty in any civil, criminal or governmental Action.

(d) Neither the Group Companies nor any Equityholder or Representative thereof, has made or is in the process of making a voluntary self-disclosure under the Self-Referral Disclosure Protocol established by the Secretary of the United States Department of HHS pursuant to Section6409 of the Patient Protection and Affordable Care Act, or under the self-disclosure protocol established and maintained by the HHS Office of the Inspector General, or any United States Attorney, the U.S. Department of Justice, a state Medicaid program or fraud unit, a state Attorney General, or other Governmental Authority.

(e) The Group Companies are not and have not been (i)a party to a corporate integrity agreement with the HHS Office of Inspector General; (ii)subject to any reporting obligations pursuant to any settlement agreement entered into with any Governmental Authority; (iii)a defendant in any qui tam/False Claims Act litigation; or (iv)the recipient of or served with any search warrant, subpoena, civil investigation demand, contact letter or any other material inquiry related to compliance with Healthcare Laws from any Governmental Authority.

(f) The Group Companies maintain and adhere to a corporate compliance program for oversight of the Business designed to ensure compliance with Healthcare Laws. Certain policies and procedures of CFCHP have been submitted to and approved by DMHC.

4.14 Intellectual Property.

(a) Except as set forth on Section 4.14(a) of the Disclosure Schedule, none of the Group Companies have any (i)Company Registered Intellectual Property, or (ii)Company Licensed Intellectual Property. The Group Companies are the sole owners of all right, title and interest in and to the Company Intellectual Property and have the right to use all other Intellectual Property necessary to conduct their business as currently conducted.

-29-

(b) The Group Companies have taken commercially reasonable steps to obtain, maintain and protect their material Company Intellectual Property. Unless otherwise set forth on Section 4.14(b) of the Disclosure Schedule: (i)to the Knowledge of the applicable Company, no Group Company is nor has in the past six years been infringing upon, nor has any Group Company received any notice with respect to any alleged infringement or unlawful or improper use of, any Intellectual Property owned or alleged to be owned by others, (ii)to the Knowledge of the applicable Company, no reasonable basis exists upon which a claim may be asserted against such Group Company for infringement, misappropriation or unlawful or improper use of any Intellectual Property, (iii)no Group Company has in the past six years made any written claim or sent any written notice that a Person is infringing on the Company Intellectual Property, (iv)to the Knowledge of the applicable Company, no Person is infringing on the Company Intellectual Property, (iv)to the Knowledge of the applicable Company, no Person is infringing on the Company Intellectual Property, divito the Knowledge of the applicable Company, no Person is infringing on the Company Intellectual Property, (iv)to the Knowledge of the applicable Company, no Person is infringing on the Company Intellectual Property and (v)no order, holding, decision or judgment has been rendered by any Governmental Authority or court, and no agreement, consent or stipulation exists which would limit any Group Company's use or enjoyment of, any right in the Company Intellectual Property necessary to conduct its business as currently conducted.

(c) Each Group Company has obtained and possesses valid licenses to use all of the software programs that are used in the conduct of its business.

(d) All former and current employees, consultants and contractors of each Group Company who have been involved in the creation or development of any material Company Intellectual Property have assigned to such Group Company, via written agreement or by operation of law, all rights, title and interest in and to any such Company Intellectual Property. To the Knowledge of the applicable Group Company, no party to any such agreement is in material breach thereof, and no employee or independent contractor has, or has asserted, any right, title or interest in or to any of the Company Intellectual Property.

(e) To the Knowledge of the applicable Company, the IT Systems (i)are sufficient and in good working condition to effectively perform for the operation of the business, (ii)have not had any material malfunction or failure that has not been remedied in all material respects, and (iii)do not contain any virus, worm, Trojan horse or similar disabling code or program. To the Knowledge of the Group Companies, there have been no unauthorized intrusions or breaches of security with respect to the IT Systems. The Group Companies have implemented and maintain commercially reasonable measures designed to protect the integrity and security of the IT Systems under their control, as well as commercially reasonable data backup and continuity, system redundancy, and disaster avoidance and recovery procedures.

(f) Following the Initial Closing, PC Buyer will be permitted to cause to be exercised all of the rights of CFC IPA under the Company Intellectual Property to substantially the same extent the CFC IPA would have been able had the Transactions not occurred, and following the Second Closing, the Buyer Parties will be permitted to cause to be exercised all of the rights of the Group Companies under the Company Intellectual Property to the same extent the Group Companies would have been able had the Transactions not occurred. Following the Initial Closing, all Company Intellectual Property will be fully transferable, alienable or licensable by PC Buyer without restriction and without payment of any kind to any Person, and following the Second Closing, all Company Intellectual Property will be fully transferable, alienable or licensable or licensable by the Buyer Parties without restriction and without payment of any kind to any Person.

-30-

4.15 <u>Privacy and Security Compliance</u>. Except as set forth on <u>Section 4.15</u> of the Disclosure Schedule, the Group Companies are in compliance in all material respects with (i)the applicable state and federal requirements of the regulations governing the privacy of individually identifiable health information and the regulations governing the security of such information maintained in electronic form, including those promulgated pursuant to the Health Insurance Portability and Accountability Act of 1996 ("<u>HIPAA</u>") the implementing regulations adopted pursuant thereto at 45 C.F.R. Parts 160, 162 and 164 and any such other any other applicable federal privacy law or rule, including the American Recovery and Reinvestment Act of 2009, Public Law 111-5 ("<u>HITECH</u>") and any applicable state privacy laws, and (ii)all other applicable Privacy Laws, to the extent not preempted by HIPAA. The Group Companies are in compliance in all material respects with all regulations and agreements regarding business associates, as that term is defined by HIPAA. Within the past four (4)years, no Group Company or any of its Representatives has improperly disclosed an individual's protected health information. Within the past four (4)years, the Group Companies have not had any reportable data breaches, as that term is defined by HIPAA, involving any patient protected health information. No information security or privacy incident has occurred in the past five (5)years that would require notification to any Governmental Authority.

4.16 <u>Insurance</u>. The Companies have made available to the Buyer Parties all of the insurance policies and binders maintained by or on behalf of any Group Company with respect to its directors and officers, properties, assets, equipment, operations, employees, consultants and business (collectively, "<u>Company Insurance Agreements</u>"), together with a claims history for the past five (5)years. All Company Insurance Agreements are valid, binding and in full force and effect in accordance with their terms and all premiums with respect thereto covering all periods up to and including the applicable Closing Date have been paid or will be paid when due, and no Group Company has ever, in the past five (5)years, been: (a)in default with respect to its Liabilities under any such insurance policies or (b)denied insurance coverage. No material default exists with respect to the obligations of any Group Companies under any such Company Insurance Agreements. No Group Company has received any written notice of cancellation or termination with respect to any such policy which has not been replaced on substantially similar terms. This <u>Section 4.16</u> shall not apply to any Employee Benefit Plans or other employee benefit arrangements.

4.17 Taxes. Except as set forth in Section 4.17 of the Disclosure Schedule, and with respect to CFC IPA, solely in respect of the Acquired Assets and the IPA

Business:

(a) AHMS is treated as a "partnership" for federal income Tax purposes (and any corresponding provision of state or local law) and has not made an election to be classified as an association taxable as a corporation pursuant to Treasury Regulation section 301.7701-3 or as an Scorporation under Internal Revenue Service guidance with respect thereto, and will not make such an election up to and including the Second Closing Date.

(b) CFCH is treated as a disregarded entity with respect to AHMS for federal income Tax purposes (and any corresponding applicable provision of state or local income Tax law) and has not made an election to be classified as an association taxable as a corporation pursuant to Treasury Regulation section 301.7701.3 or as an S corporation under Internal Revenue Service guidance with respect thereto, and will not make such an election up to and including the Second Closing Date.

(c) Each Group Company has duly and timely filed all income and other material Tax Returns required to be filed by it under Applicable Laws with the appropriate Tax Authority in all jurisdictions in which such Tax Returns are required to be filed (after giving effect to any valid extensions of time in which to make such filings). All such Tax Returns are true, accurate and complete in all material respects and have been prepared in compliance with all Applicable Laws. All Taxes (whether or not shown on any Tax Return) imposed or otherwise due and owing by or with respect to the business and activities of a Group Company have been paid in full.

(d) CFC IPA and AHMS have made available to Buyers copies of all income Tax Returns of the Group Companies for all taxable years ending on or after December 31, 2019.

(e) No outstanding deficiency or adjustment in respect of Taxes has been proposed or asserted in writing or assessed by any Tax Authority against or with respect to a Group Company. There are no outstanding refund claims with respect to any Tax or Tax Return of the Group Companies, or the business or activities of the Group Companies.

(f) Each Plan Company has withheld and paid all Taxes required to have been withheld and paid in connection with any amounts paid or owing to any employee, independent contractor, creditor, stockholder, or other third party.

(g) There are no liens for Taxes (other than current Taxes not yet due and payable or not yet delinquent) on any of the assets of the Group Companies.

(h) No Group Company has ever consented to extend, nor is a beneficiary of an agreement to extend, the time in which any Tax may be assessed or collected by any Tax Authority other than consents with respect to periods for which the applicable statute of limitations has since expired.

(i) No Group Company has ever requested or been granted, nor is the beneficiary of, an extension of the time for filing any Tax Return that has not yet been filed (other than any automatic extensions of time that are allowable under Applicable Law).

(j) No Action related to Taxes or Tax audit has been initiated in writing or is currently pending against a Group Company (or, to the Knowledge of an applicable Group Company, threatened) in any jurisdiction.

(k) There are no outstanding rulings of, or requests for rulings by, any Tax Authority addressed to a Group Company that are, or if issued would be, binding on a Group Company after the applicable Closing Date.

(1) There are no tax sharing, allocation, indemnification or similar agreements (other than Contracts that do not primarily relate to Taxes) in effect as between any Group Company or any predecessor or Affiliate thereof and any other party under which Buyers or any Group Company could be liable for any Taxes or other claims of any party after the applicable Closing Date.

-32-

(m) Neither the Group Companies nor Buyers(by reason of owning any interest in the Group Companies) will be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion thereof) ending after the applicable Closing Date as a result of any (i)installment sale or open transaction disposition made prior to the applicable Closing Date; (ii)prepaid amount received prior to the applicable Closing Date, or (iii)closing agreement under Code Section 7121 (or other comparable agreement) entered into prior to the applicable Closing Date.

(n) No Group Company is nor has ever been a party to a "listed" transaction within the meaning of Section6707A of the Code or Treasury Regulations Section 1.6011-4.

(o) CFCHP (i)has not been, in the past three (3)years, a party to a transaction reported or intended to qualify as a reorganization under Section368 of the Code and (ii)has not constituted a "distributing corporation" or a "controlled corporation" (within the meaning of Section355(a)(1)(A) of the Code) in a distribution of shares that was reported or otherwise constitutes a distribution of shares under Section355 of the Code in the two (2)years prior to the date of this Agreement or that could otherwise constitute part of a "plan" or "series of related transactions" (within the meaning of Section355(e)of the Code) that includes the transactions contemplated by this Agreement.

4.18 Employment Matters.

(a) Section 4.18(a) of the Disclosure Schedule contains a complete and accurate list of all current employees of each Group Company and sets forth for each such individual the following information as of the date of the applicable Closing: (i)date of hire; (ii)title/position; (iii)full-time or part-time status; (iv)hourly rate or salary amount; (v)incentive-based compensation to which they are eligible to receive, if any; (vi)accrued paid time off, vacation, and/or paid sick leave; (vii)classification as exempt or non-exempt under theFair Labor Standards Act (" <u>FLSA</u>") and other Applicable Laws; (viii)employment status (including whether any such employee is on approved leave of absence (whether family leave, workers' maternity or parental leave, workers' compensation, short-term and long-term disability, medical leave or otherwise)); and (ix)employer of record. Each employee identified as "exempt" from overtime and/or minimum wage requirements of the FLSA or other Applicable Laws is properly and correctly so classified. Section 4.18(a) of the Disclosure Schedule also contains a separate accurate and complete list of the independent contractors currently engaged to provide services to each Group Company and the individuals listed in <u>Section 4.18(a)</u> of the Disclosure Schedule as independent contractors or employees have been properly so classified, as the case may be.

(b) Except as set forth in <u>Section 4.18(b)</u>of the Disclosure Schedule, the engagement of each employee and independent contractor of the Group Companies is terminable without any penalty, liability or severance obligation.

(c) The Group Companies have paid or will timely pay all compensation, including wages, commissions, bonuses, fees, or other compensation payable to their respective employees or independent contractors for services performed on or before the Closing Date. Other than as set forth in <u>Section 4.18(c)</u> of the Disclosure Schedule, there are no Contracts between the Group Companies and any employees or independent contractors regarding any compensation, wages, commissions, bonuses, or

(d) No Group Company is a party to any collective bargaining agreement, nor has it made any proposals regarding the terms of any collective bargaining agreement. No Group Company is subject to any: (i) pending or, to the Knowledge of the applicable Company, threatened unfair labor practice complaint before a Governmental Authority; (ii)pending or, to the Knowledge of the applicable Company, threatened labor strike, slowdown, work stoppage, lockout, or other organized labor disturbance, or (iii) to the Knowledge of the applicable Group Company, union organization efforts or attempts by any union to represent employees as a collective bargaining agent.

(e) There are no claims, disputes, grievances, or controversies pending or, to the Knowledge of the applicable Company, threatened involving any of their respective prospective or actual employees or group of employees, or independent contractors. To the Knowledge of the applicable Company, there are no written threats, charges, investigations, administrative proceedings or formal complaints of discrimination (including discrimination based upon sex, age, marital status, race, national origin, sexual orientation, disability or veteran status) pending before the Equal Employment Opportunity Commission, the National Labor Relations Board, the U.S. Department of Labor, the U.S. Occupational Safety and Health Administration, the Workers Compensation Appeals Board, or any other Governmental Authority against any Group Company pertaining to any of their respective prospective or actual employees or group of employees, interns, volunteers or independent contractors.

(f) In the last five (5)years:(i)to the Knowledge of the applicable Company, no allegations of sexual harassment or sexual misconduct have been made involving their respective current or former directors, officers, managers or independent contractors; and(ii)no Group Company has entered into any settlement agreements related to allegations of sexual harassment or sexual misconduct by any current or former director, officer, manager or independent contractor.

(g) The Group Companies are in compliance in all material respects with Applicable Laws regarding employment and employment practices, including all Applicable Laws relating to wages, hours, paid sick leave, overtime, collective bargaining, employment discrimination, civil rights, safety and health, workers' compensation, pay equity, immigration and employment eligibility verification laws (including the Immigration and Nationality Act and the Immigration Reform and Control Act of 1986, and all related regulations promulgated thereunder), classifications of employees and independent contractors for wage-and-hour and tax purposes, and the collection and payment of withholding and/or social security Taxes.

(h) During the past five (5)years, no Group Company has effectuated: (i)a "plant closing" (as defined in the Worker Adjustment and Retraining Notification Act (the "<u>WARN Act</u>") or any similar state or local Applicable Law) affecting the Leased Real Property or (ii)a "mass layoff" (as defined in the WARN Act, or any similar state or local Applicable Law) affecting the Leased Real Property.

-34-

4.19 Employee Benefit Matters.

(a) Section 4.19(a) of the Disclosure Schedule sets forth an accurate and complete list of all material benefit plans, policies, or arrangements under which any current or former employee, director, or consultant of a Group Company, or any beneficiary of such individuals, has any present or future right to benefits, or to which a Group Company has or may reasonably be expected to have any Liability or obligation to contribute (including on behalf of an ERISA Affiliate), including: (i)any material "employee benefit plan" (within the meaning of Section3(3) of ERISA, whether or not subject to ERISA); (ii)any material profit-sharing, pension, Code Section125 "cafeteria" or flexible spending account, deferred compensation, bonus, stock or equity-based incentive, educational assistance, severance, retention, change in control, vacation or paid time off, or health or welfare plan, agreement or arrangement; or (iii) any other material plan, agreement or arrangement, but without regard to materiality, an "Employee Benefit Plan"). Section 4.19(a) of the Disclosure Schedule also specifies the Group Company that sponsors or maintains each Employee Benefit Plan.

(b) With respect to each material Employee Benefit Plan, the Group Companies have made available to Buyers a copy of the following documents as applicable to each such Employee Benefit Plan: (i)the plan document or, if an oral arrangement, a written summary of the material terms of the Employee Benefit Plan; (ii)the most recent annual report (Form5500); (iii)each trust agreement, the most recent summary plan description and any summaries of material modification; (iv)all nondiscrimination testing reports for the three (3)most recently completed plan years; (v)all related insurance contracts and administrative service agreements, (vi)the most recent Internal Revenue Service determination or opinion letter, and (vii)all material and non-routine correspondence with any Governmental Authority within the past three (3) years.

(c) Except as set forth in Section 4.19(c) of the Disclosure Schedule, none of the Group Companies nor any ERISA Affiliate maintains, sponsors, contributes to, has any obligation to contribute to, or has any current or contingent obligation or liability under or with respect to any (i)"defined benefit plan" (as defined in Section3(35) of ERISA) or any other plan subject to the funding requirements of Section412 or 430 of the Code or Section302 or Title IV of ERISA, (ii)"multiemployer plan" (as defined in Section3(37) of ERISA), (iii)plan, program, agreement or arrangement that provides for post-retirement or post-termination medical, life insurance or other similar benefits (other than health continuation coverage required by COBRA for which the covered individual pays the full cost of coverage), (iv)multiple employer plan (as described in section 413(c)of Code or Section210 of ERISA), or (v)"multiple employer welfare arrangement" (as defined in Section3(40) of ERISA). None of the Group Companies nor any ERISA Affiliate has any current or contingent liability or obligation under Title IV of ERISA.

(d) Each Employee Benefit Plan has been operated and administered in all material respects in accordance with its terms, and is and has been in compliance in all material respects with ERISA, the Code and all other Applicable Laws. All material respects, forms and other documents required to be filed with any Governmental Authority or furnished to employees with respect to any Employee Benefit Plan (including summary plan descriptions, Forms 5500 and summary annual reports) have been timely filed or furnished in all material respects. Without limiting the generality of the foregoing, each of the Group Companies is, and during all relevant times has been, in compliance in all material respects with the applicable requirements of (i)COBRA, (ii)the applicable requirements of the Health Insurance Portability and Accountability Act of 1996, as amended, and the Applicable Laws (including the proposed regulations) thereunder, and (iii)the Patient Protection and Affordable Care Act of 2010, and all rulesand official guidance promulgated thereunder, and no circumstance exists or event has occurred, which reasonably could be expected to result in a material violation of, or material penalty or liability under, any of the foregoing. The obligations of all Employee Benefit Plans that provide health, welfare or similar insurance are fully insured by bona fide third-party insurers. No Employee Benefit Plan is maintained through a human resources and benefits outsourcing entity, professional employer organization, or other similar vendor or provider, except to provide third-party administrative or recordkeeping services to an Employee Benefit Plan, all payments, premiums, contributions, distributions, reimbursements and accruals for all periods ending prior to or as of the Closing Date shall have been made or accrued on the Interim Balance Sheet in accordance with GAAP and there is no unfunded liability which is not reflected on the Interim Balance Sheet.

(e) Each Employee Benefit Plan which is intended to be qualified under Section401(a) of the Code is so qualified and is the subject of a favorable determination letter or, in the case of a prototype or volume submitter plan, an opinion letter from the Internal Revenue Service that the form of the plan document meets the applicable requirements of Section401(a) of the Code, and nothing has occurred since the date of such determination or opinion letter that could reasonably be expect to adversely affect the qualified status of such Employee Benefit Plan.

(f) With respect to each Employee Benefit Plan, (i) no "prohibited transaction" by any Group Company or, to the Knowledge of the applicable Company, by any other party, within the meaning of Section4975 of the Code or Sections406 and 407 of ERISA, and not otherwise exempt under Section408 of ERISA, has occurred; (ii) there are no actions, suits or claims pending, or, to the Knowledge of the applicable Company, threatened or anticipated (other than routine claims for benefits) against any such Employee Benefit Plan or fiduciary thereto or against the assets of any such Employee Benefit Plan; (iii) no breach of fiduciary duty (as determined under ERISA) by any Group Company or, to the Knowledge of the applicable Company, by any other party, or any other failure to act or comply in connection with the administration or investment of the assets of such Employee Benefit Plan has occurred; and (iv) there are no audits, inquiries or proceedings pending or, to the Knowledge of the applicable Company, threatened by any Governmental Authority with respect to any Employee Benefit Plan.

(g) Except as set forth in Section 4.19(c) of the Disclosure Schedule, the execution and delivery of this Agreement and the consummation of the Transactions will not, either alone or in connection with any other event or circumstance, (i) result in any termination or severance payment becoming due to any employee of a Group Company; (ii) accelerate the time of payment or vesting, or otherwise increase the amount of compensation or benefits due any officer, director, employee, or consultant of a Group Company; or (iii) result in any amount of compensation failing to be deductible by reasons of Section 280G of the Code.

(h) Each Employee Benefit Plan that is a "nonqualified deferred compensation plan" (as defined under Section409A(d)(1) of the Code) is and has been, in all material respects, in documentary and operational compliance with Section409A of the Code. No Group Company has any "gross-up" or indemnity obligations for Taxes imposed under Section 4999 or 409A of the Code.

-36-

(i) Each of the Plan Companies has, for purposes of each Employee Benefit Plan sponsored or maintained by a Plan Company, correctly classified those individuals performing services for a Plan Company as common law employees or independent contractors of the Plan Company.

(j) None of the Employee Benefit Plans are maintained, contributed to, or required to be contributed to outside of the United States or for the benefit of any employee or other service provider that provides services or resides outside of the United States.

4.20 <u>Environmental Matters</u>. Each Group Company is, and has been for the past five (5)years, in compliance in all material respects with all Environmental Requirements and has no Liabilities, including corrective, investigatory or remedial obligations arising under Environmental Requirements, and no Group Company has received any written, or to the Knowledge of the applicable Company oral, notice, report or information regarding any Liabilities, including corrective, investigatory or remedial obligations arising under Environmental Requirements which relate to the Group Companies or any of their properties or facilities. Without limiting the generality of the foregoing, each Group Company has in the past five (5)years obtained and complied with, and is currently in compliance with, all Permits and other authorizations that may be required pursuant to any Environmental Requirements for the occupancy of its properties or facilities or the operation of its business as currently conducted.

4.21 Affiliate Transactions. Except as set forth on Section 4.21 of the Disclosure Schedule, no Related Party, is, or has been within the past five (5)years, directly or indirectly, a party to any Contract or transaction with any Group Company or which is pertaining to the business of the Group Companies or has any interest in any property, real or personal or mixed, tangible or intangible, used in or pertaining to the business of the Group Companies. Section 4.21 of the Disclosure Schedule hereto describes: (a)all affiliated services provided to or on behalf of any Group Company by the Equityholders or their respective Affiliates and to or on behalf of the Equityholders and their respective Affiliates by any Group Company and all affiliate transactions or Contracts among any Group Company), (b)all affiliate transactions or Contracts among the Group Company), (b)all affiliate transactions or Contracts among all parties to any arrangements or transactions disclosed in accordance with clauses (a)and (b)of this Section 4.21.

4.22 <u>Illegal Payments</u>. None of the Group Companies, the Equityholders nor, to the Knowledge of the applicable Company, any of their respective directors, officers, employees or agents have offered, authorized, made or received on behalf of any Group Company any illegal payment or contribution of any kind, including any payment in violation of any Applicable Laws, directly or indirectly, including, without limitation, any kickbacks, bribes, payments, gifts or gratuities, to any Person, or U.S. or foreign national, state or local government officials, employees or agents or candidates therefor or other Persons. The Group Companies have maintained systems of internal controls (including, but not limited to, accounting systems, purchasing systems and billing systems) and written policies to ensure compliance with Applicable Laws (in all material respects) regarding illegal payments and to ensure that all books and records accurately and fairly reflect, in reasonable detail, all transactions and dispositions of funds and assets. None of the Group Companies nor any Equityholder, nor, to the Knowledge of the applicable Company, any of their respective directors, officers, employees or agents, are the subject of any material allegation, voluntary disclosure, investigation, prosecution or other enforcement action related to illegal payment laws.

-37-

4.23 <u>Brokers.</u> Except as set forth in <u>Section 4.23</u> of the Disclosure Schedule, no broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with the Transactions or any other transactions, whether past, present or future, based upon arrangements made by or on behalf of any Group Company.

4.24 <u>Investment Company Status</u>. None of the Group Companies is an "investment company" as defined in, or subject to regulation under, the Investment Company Act of 1940, as amended.

4.25 Investment Representations and Warranties.

(a) Evaluation of Risk. CFC IPA has such knowledge, skill and experience in business, financial and investment matters that CFC IPA is capable of evaluating the merits and risks of an investment in the Buyer Parent Shares. With the assistance of CFC IPA's own professional advisors, to the extent that CFC IPA has deemed appropriate, CFC IPA has made its own legal, tax, accounting, and financial evaluation of the merits and risks of an investment in the Buyer Parent Shares and the consequences of this Agreement. CFC IPA has considered the suitability of the Buyer Parent Shares as an investment in light of its own circumstances and financial condition, and it is able to bear the risks associated with an investment in the Buyer Parent Shares.

(b) <u>Access to Information</u>. CFC IPA has: (i)become familiar with the business, financial condition and operations of Buyer Parent, has been given access to and an opportunity to examine such documents, materials and information concerning Buyer Parent as CFC IPA deems necessary or advisable in order to reach an informed decision regarding an investment in Buyer Parent, and has carefully reviewed and understands such documents, materials and information CFC IPA's full satisfaction any and all questions regarding such documents, materials and information; (ii)made such independent investigation of the Buyer Parent Shares and

Buyer Parent and such other matters as CFC IPA deems to be necessary or advisable in connection with the acquisition of the Buyer Parent Shares; and (iii)not been offered the Buyer Parent Shares by any means of general solicitation or general advertising. CFC IPA has consulted with, and relied on the advice of, only CFC IPA's own tax and other advisors and the representations and warranties set forth in <u>Article 5</u> in evaluating the tax consequences of an investment in the Buyer Parent Shares to CFC IPA. CFC IPA acknowledges that no federal or state agency has passed upon the Buyer Parent Shares nor has any agency made any finding or determination of the fairness of an investment in the Buyer Parent Shares.

(c) Accredited Investor. CFC IPA is an "accredited investor" as defined in Rule501(a)under the Securities Act. CFC IPA is a professional corporation, with total assets in excess of \$5,000,000, not formed for the specific purpose of acquiring the Buyer Parent Shares, whose acquisition of the Buyer Parent Shares has been directed by a Person who has such knowledge and experience in financial and business matters that such Person is capable of understanding the merits and risks of an investment in the Buyer Parent Shares. CFC IPA shall furnish any additional information requested by Buyer Parent or any of its affiliates to assure compliance with applicable U.S. federal and state securities laws in connection with CFC IPA's purchase of the Buyer Parent Shares meets the qualifications set forth in the preceding sentence). Any information that has been furnished or that will be furnished by CFC IPA to evidence its status as an accredited investor is accurate and complete, and does not contain any misrepresentation or omission. CFC IPA's principal place of business and residence is in the state of California.

(d) Investment Intent. CFC IPA is acquiring the Buyer Parent Shares (subject to its assignment thereof as provided in <u>Section 1.1(a)</u>) solely for CFC IPA's own beneficial account, for investment purposes, and not with a view to, or for resale in connection with, any distribution of the Buyer Parent Shares, as such terms are defined or otherwise used under the Securities Act. CFC IPA understands that the Buyer Parent Shares issued pursuant to this Agreement have not been registered under the Securities Act or any state securities laws by reason of exemptions under the provisions of the Securities Act and such state securities laws which depend in part upon the investment intent of CFC IPA and the other representations made by CFC IPA in this Agreement, and CFC IPA understands that Buyer Parent is relying upon the representations and agreements set forth in this Agreement (and any additional information provided by CFC IPA at the requires of Buyer Parent) for the purpose of determining whether the transactions contemplated by this Agreement meet the requirements for such exemptions.

(e) <u>Restrictions on Transfer</u>. CFC IPA acknowledges that the Buyer Parent Shares are restricted securities under the Securities Act, that additional restrictions on transfer of the Buyer Parent Shares are imposed by <u>Section 6.22</u>; and that, subject to <u>Section 6.23</u>, any stock certificates representing the Buyer Parent Shares will bear restrictive legends referring to such restrictions.

(f) Non-reliance. CFC IPA (i)is not relying on (and will not at any time rely on) any communication (written or oral) of Buyer Parent, as investment advice or as a recommendation to acquire the Buyer Parent Shares; (ii)acknowledges that neither Buyer Parent nor any Person acting on behalf of Buyer Parent has: (A)given any guarantee or representation as to the potential success, return, effect or benefit (either legal, regulatory, tax, financial, accounting or otherwise) of an investment in the Buyer Parent Shares or (B)made any representation to CFC IPA regarding the legality of an investment in Buyer Parent Shares under applicable legal investment or similar law or regulations, in each case of clauses (A)and (B), other than any representations or warranties set forth in <u>Article 5</u>. In deciding to invest in the Buyer Parent Shares, CFC IPA is not relying on the advice or recommendations of Buyer Parent or any Person acting on behalf of Buyer Parent and CFC IPA has made his own independent decision that the investment in the Buyer Parent Shares is suitable and appropriate for the CFC IPA. Other than the representations and warranties of Buyer Parent expressly set forth in <u>Article 5</u>, neither CFC IPA nor any other Person has made any representation or warranty, expressed or implied, as to the accuracy or completeness of the information provided or to be provided to CFC IPA by or on behalf of Buyer Parent.

-39-

ARTICLE 5 Representations and Warranties of Buyer Parties

Each Buyer Party hereby jointly and severally represents and warrants to the Companies and the Equityholders that the statements contained in this<u>Article5</u> are true and correct on the date hereof and shall be true and correct on the applicable Closing Date as if made thereon.

Organization and Authority of Buyer; MSO Relationship. MSO LP Buyer is a corporation duly organized, validly existing and in good standing under the 5.1 laws of the state of California. MSO GP Buyer is a limited liability company duly organized, validly existing and in good standing under the laws of the state of California. PC Buyer is a medical professional corporation duly organized, validly existing and in good standing under the laws of the state of California. Buyer Parent is a corporation duly organized, validly existing and in good standing under the laws of the state of Delaware. Each Buyer Party has full power and authority to enter into this Agreement and the other Transaction Documents to which such Buyer Party is a party, to carry out its obligations hereunder and thereunder and to consummate the transactions hereunder and thereunder. The execution and delivery by each Buyer Party of this Agreement and any other Transaction Document to which such Buyer Party is a party, the performance by each Buyer Party of its obligations hereunder and the consummation by such Buyer Party of the transactions contemplated hereunder and thereunder have been duly authorized by all requisite corporate action on the part of such Buyer Party. This Agreement has been duly executed and delivered by each Buyer Party, and (assuming due authorization, execution and delivery by the Equityholders and the Companies) this Agreement constitutes a legal, valid and binding obligation of such Buyer Party enforceable against it in accordance with its terms; subject to bankruptcy, insolvency, reorganization, moratorium and similar Applicable Laws of general application relating to or affecting creditors' rights and to general equity principles. When each other Transaction Document to which each Buyer Party is or will be party has been duly executed and delivered by such Buyer Party (assuming due authorization, execution and delivery by each other Party thereto), such Transaction Document will constitute a legal and binding obligation of such Buyer Party enforceable against such Buyer Party in accordance with its terms; subject to bankruptcy, insolvency, reorganization, moratorium and similar Applicable Laws of general application relating to or affecting creditors' rights and to general equity principles. MSO LP Buyer was formed in 1994 as a management service organization for the purposes of providing management services to medical companies and independent practice associations.

5.2 <u>Non-Contravention; Consents</u>. The execution and delivery by the applicable Buyer Party of this Agreement and each Transaction Document to which each Buyer Party is a party, and, assuming the accuracy of the Equityholders and Group Companies' representations and warranties set forth in <u>Article 3</u> and <u>Article 4</u>, the performance by such Buyer Party of any actions contemplated hereunder or thereunder, does not and will not, directly or indirectly (with or without notice or lapse of time):

(a) conflict with or violate any provision of the Governing Documents of each such Buyer Party;

in material Liabilities or fees under, create in any Person the right to terminate, cancel or materially modify, or cause a material default under or give rise to any material rights or penalties under (i) any provision of Applicable Law relating to such Buyer Party, (ii) any provision of any Governmental Order to which such Buyer Party or any of its properties are subject, or (iii) any provision of any material Contracts to which such Buyer Party is a party, except as, individually or in the aggregate, would not reasonably be expected to constitute or result in a Material Adverse Effect; or

(c) except for filings as may be required under the Securities Act, the Exchange Act, the HSR Act or other applicable Antitrust Laws, applicable state securities and "blue sky" laws or the rulesand regulations of the Nasdaq, require a registration, filing, application, notice, consent, approval, order, qualification or waiver with, to or from any Governmental Authority or third party, except as, individually or in the aggregate, would not reasonably be expected to constitute or result in a Material Adverse Effect.

5.3 <u>Investment Intention</u>. The Buyer Parties have such knowledge and experience in financial and business matters that it is capable of evaluating the merits and risks of acquiring the Group Companies. The Buyer Parties confirm that the Group Companies have made available to the Buyer Parties the opportunity to ask questions of the officers and management of the Group Companies to acquire additional information about the business, assets and financial condition of the Group Companies. The Buyer Parties will acquire their respective interests in the Group Companies for investment only, and not with a view toward or for sale in connection with any distribution thereof or with any present intention of distributing or selling any interest therein. The Buyer Parties understand that the Transactions have not been, and will not be registered or qualified under the Securities Act of 1933, as amended, nor any state or any other applicable securities law, by reason of a specific exemption from the registration or qualification provisions of those laws, based in part upon the Buyer Parties' representations in this Agreement. The Buyer Parties understand that no part of the interest in the Group Companies which the Buyer Parties acquire may be resold unless such resale is registered under the Securities Act of 1933, as amended, and registered or qualified under applicable state securities laws or an exemption from such registration and qualification is available.

5.4 <u>Sufficiency of Funds; Solvency</u>. Each Buyer Party has, and will have, on the applicable Closing Date sufficient cash from such Buyer Party's immediately available internal organization funds or available under a currently established committed credit facility or unutilized lines of credit with financial institutions, to consummate the Transactions and perform its obligations hereunder and under the Transaction Documents.

5.5 Legal Proceedings. There are no Actions pending or, to each Buyer Parties' knowledge, threatened against or by each Buyer Party or any respective Affiliate of such Buyer Party that, individually or in the aggregate, would have a Material Adverse Effect on any Buyer Parties' ability to execute, deliver or perform this Agreement or any Transaction Document to which it is a party, or to timely consummate the Transactions.

-41-

5.6 <u>Title</u>. Assuming delivery of the Acquired Assets at the Initial Closing in accordance with <u>Section 1.1</u>, at the Initial Closing, (a)the Buyer Parent Shares issued in exchange thereof shall be duly authorized, validly issued and outstanding, fully paid and nonassessable, and not in violation of any preemptive or similar rights, or similar rights, and (b)CFC IPA shall acquire good and valid title thereto, free and clear of any Liens, other than restrictions on transfer imposed by securities Applicable Laws or provided in the Governing Documents of Buyer Parent or in this Agreement, or that result from any actions taken by or on behalf of CFC IPA, IPA Equityholder, its owners or Buyer Parent Shares Transferees. The issued and outstanding shares of Buyer Parent's common stock are registered pursuant to Section 12(b) of the Exchange Act, and are listed for trading on The Nasdaq Capital Market (the "<u>Nasdaq</u>") under the symbol "AMEH," and Buyer Parent shares. Assuming the accuracy of IPA Equityholder's representations and warranties set forth in <u>Article 4</u> of this Agreement and assuming the accuracy of the Buyer Parent Shares made by the Buyer Parent Shares in the joinders to this Agreement referred to in <u>Section 1.1(c)</u>(as if such representations and warranties were made on the date hereof), no registration under the Securities Act is required for the offer and sale of the Buyer Parent Shares by Buyer Parent.

5.7 <u>Brokers.</u> Except for such amounts as the Buyer Parties shall pay, no broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with the Transactions or any other Transaction Document based upon arrangements made by or on behalf of the Buyer Parties.

5.8 <u>R&W Insurance Policy</u>. The Buyer Parties have obtained, at the Buyer Parties' sole expense, third-party insurance (i.e., bound and incepted) from AXA XL Insurance as of the date of this Agreement a representation and warranty insurance policy covering the representations and warranties given by the Equityholders and Group Companies under this Agreement for the benefit of the Buyer Parties or its designee as the named insured (the "<u>R&W Insurance Policy</u>"), which R&W Insurance Policy is subject to a conditional binder in the form mutually agreed by the Parties. The R&W Insurance Policy includes a provision whereby the insurer expressly waives, and irrevocably agrees not to pursue, directly or indirectly, any subrogation rights or other rights of recovery against any Equityholders or any claim made by any insured thereunder, which waiver any Equityholder may enforce directly against the insurer of the R&W Insurance Policy or Article 4. The foregoing waiver in the R&W Insurance Policy may not be amended by any party in any manner materially adverse to the Equityholders or any of their Representatives without the prior written consent of the Equityholder Representative.

-42-

5.9 <u>Issuance of the Buyer Parent Shares.</u> The Buyer Parent Shares are duly authorized and, when issued at the Initial Closing in accordance with the terms of this Agreement, the Buyer Parent Shares (1)will be validly issued to the Buyer Parent Shares Transferees, fully paid and non-assessable, and free and clear of any Liens, other than any restrictions on transfer created by any state of federal securities laws and (2)will not have been issued in violation of or subject to any preemptive or similar rights created under Buyer Parent's certificate of incorporation or bylaws (each as amended through the Closing), by contract or under the Delaware General Corporation Law. The Buyer Parent Shares are registered pursuant to Section 12(b)of the Securities Exchange Act of 1934, and are listed for trading on the Nasdaq under the symbol "AMEH," and Buyer Parent has either timely complied with (or the issuance of the Buyer Parent Shares is exempted from) Rule5250(e)of the Nasdaq Listing Rulesin respect of the issuance of the Buyer Parent Shares.

5.10 <u>Exclusive Representations</u>. Each Buyer Party acknowledges that none of the Equityholders, nor anyone acting on behalf of any such Persons, has made any representation or warranty, express or implied, including as to the accuracy or completeness of any memoranda, charts, summaries, presentations, schedules or other information heretofore made available by the Equityholders to the Buyer Parties, except as expressly set forth in this Agreement.

ARTICLE 6 Covenants

6.1 <u>Confidentiality</u>. The Parties to this Agreement acknowledge that the information being exchanged in connection with the Transactions is subject to the terms and conditions of that certain Mutual Confidentiality and Mutual Non-Disclosure Agreement, dated as of August8, 2021, by and between Buyer Parent and CFC IPA (as the same may be amended, restated, modified or supplemented from time to time, the "<u>Confidentiality Agreement</u>"), the terms of which are incorporated herein by reference. If this Agreement is, for any reason, terminated prior to the Initial Closing, the Confidentiality Agreement shall continue in full force and effect in respect of such confidential information in accordance with its terms.

6.2 <u>Publicity</u>. Notwithstanding anything to the contrary contained in this Agreement, except as may be required to comply with the requirements of any Applicable Law or by the applicable rulesof any securities exchange upon the reasonable advice of counsel (it being understood that each such Party shall use reasonable commercial efforts to provide the other Parties to this Agreement a draft of such press release, public statement or public communication in advance of such disclosure), from and after the date hereof, no Party shall make, and each Party shall cause their respective Affiliates not to make, any press release or similar public announcement or public communication relating to this Agreement or the Transactions unless specifically approved in advance by the other Parties, which approval shall not be unreasonably withheld, conditioned or delayed.

6.3 <u>Further Assurances</u>. From time to time, as and when reasonably requested by any Party in writing, each Party shall execute and deliver, or cause to be executed and delivered, all such documents and instruments and shall take, or cause to be taken, all such further or other actions as such other Party may reasonably deem necessary or desirable to consummate the Transactions, including executing and delivering to the other Parties hereto such assignments, deeds, bills of sale, consents and other instruments as the other Parties hereto or their counsel may reasonably request as necessary or desirable for such purpose. For a period of twelve (12) months after the applicable Closing, the applicable Equityholders shall provide information not subject to attorney-client privilege that is reasonably requested by Buyers in writing regarding the Group Companies and the Business in order to assist Buyers in its filings for regulatory approvals.

-43-

6.4 Conduct of Business Prior to the Applicable Closing

(a) Except for matters set forth in <u>Schedule 6.4</u> or as permitted or required by the terms of this Agreement or Transaction Documents or otherwise required by the Buyer Parties in writing in connection with the Transactions, or except as required by Applicable Law, from the date of this Agreement through the applicable Closing, the applicable Group Companies shall conduct the Business in the Ordinary Course of Business and preserve intact the current business organization, use their Commercially Reasonable Efforts to keep available the services of current officers, employees and agents of each Group Company and maintain in good state relations and good will with customers, suppliers, licensors, licensees, distributors, landlords, creditors, employees, agents and others having business relationships with a Group Company. The Group Companies shall refrain from taking any action that would result in any of the conditions set forth in <u>Section 8.1</u> or <u>Section 8.2</u> not being satisfied. In addition (and without limiting the generality of the foregoing), except as set forth in <u>Schedule 6.4</u> or otherwise contemplated by the terms of this Agreement or the Transaction Documents, or except as required by Applicable Law, from the date of this Agreement through the applicable Closing, the applicable Group Companies shall not do any of the following without the prior written consent of MSO Buyer or PC Buyer, as applicable:

(i) amend their Governing Documents;

(ii) issue any capital stock or other equity interests or any option, warrant or right relating thereto or any securities convertible into or exchangeable for any capital stock or other equity interests;

(iii) hire any employee or retain any independent contractor whose annual base compensation exceeds \$150,000, other than in the Ordinary Course of Business or in connection with the performance of this Agreement;

(iv) borrow any amount, receive any committed loan facilities or, incur or guarantee any Indebtedness (including contingently as a guarantor or otherwise) or other Liabilities, except for Transaction Expenses and current liabilities incurred in the Ordinary Course of Business or grant, incur or suffer to exist any Lien, or received any notice of any Lien on any Equity Interests issued by it;

(v) permit or allow the Purchased Interests, Acquired Assets or any of the assets or properties of the Group Companies to become subjected to any Lien (other than Permitted Liens, Liens imposed by securities or healthcare Applicable Laws or that result from any actions taken by or on behalf of any Buyer Party);

(vi) pay, loan or advance any amount to, or sell, transfer or lease any of its assets to, or enter into any agreement or arrangement with any of its respective Affiliates or Related Parties;

-44-

(vii) make any material change in any method of accounting or accounting practice or policy;

(viii) acquire by merging or consolidating with, or by purchasing a substantial portion of the assets or capital stock or other equity interest of, or by any other manner, any business or any corporation, partnership, association or other business organization or division thereof or otherwise acquire any assets (other than inventory) other than in the Ordinary Course of Business;

(ix) make or incur capital expenditures (including entering into any capital lease) that are not currently budgeted and that, in the aggregate, are in excess of Ten Thousand Dollars (\$10,000);

(x) other than in the Ordinary Course of Business, sell, lease, license or otherwise dispose of any of their assets that are material, individually or in the aggregate, to the Business;

(xi) enter into, amend or terminate any lease of real property or material tangible personal property or waive any material term or condition thereof or grant any consents thereunder; create any Lien (other than Permitted Liens) on any material property or assets; grant or otherwise create or consent to the creation of any easement, covenant, restriction, assessment or charge affecting any Leased Real Property or any part thereof; commit any waste or nuisance on any Leased Real Property; or make any material changes in the construction or condition of any Leased Real Property;

(xii) terminate or allow to be terminated any insurance policy in effect as of the date hereof, or fail to maintain, with financially responsible insurance companies, insurance on tangible assets of the Group Companies and on the Business in such amounts and against such risks and losses as are consistent with past practice;

(xiii) form any Subsidiary or acquire the equity (or right to acquire equity) of any Person;

(xiv) commence or settle any material Action, other than the settlement of an Action for amounts less than Two Hundred and Fifty Thousand Dollars (\$250,000) which only involve the payment of monetary damages and does not impose any equitable remedies upon a Group Company and there is no admission of fault or any violation of Applicable Law;

(xv) enter into, amend or make any modification to any Material Contract or Permit, in each case, outside the Ordinary Course of Business;

(xvi) without limiting the generality of the foregoing, amend or make any modification to any Contract between CFC IPA, on the one hand, and any Plan Company, on the other hand;

(xvii) make any material change to any of the cash management practices of the Business, including but not limited to, discontinue the payment of its accounts payable that are payable in the Ordinary Course of Business or deviate from or alter any of its practices, policies or procedures in paying accounts payable or collecting accounts receivable;

-45-

(xviii) commence, institute, cancel, compromise, assign, waive, release, settle or agree to settle any right or claim (or series of related rights and claims) in excess of Twenty-Five Thousand Dollars (\$25,000) or outside the Ordinary Course of Business;

(xix) make, change or revoke any Tax election (other than one that is immaterial), adopt or change any method of Tax accounting, settle or compromise any material Action with respect to Taxes (other than one that is immaterial), amend any Tax Return (other than one that is immaterial), consent to any extension or waiver of the statute of limitations for the assessment or collection of any Tax, or take any reporting position with respect to Taxes that is inconsistent with past practice or that is otherwise not in the Ordinary Course of Business, in each case if such action would have the effect of materially increasing the Tax Liability or materially reducing any Tax asset of Buyers or the Group Companies in respect of any Post-Closing Tax Period (for purposes of this <u>Section 6.4(xix)</u>, a matter is "immaterial" if it affects the Tax treatment and liability solely of one or more Equityholders); provided that, for the avoidance of doubt, this parenthetical shall only apply to the word "immaterial" or "materially"; or

(xx) other than as required under any Employee Benefit Plan, as required by Applicable Law, or in the Ordinary Course of Business, (i)terminate any employee whose current annual base compensation exceeds \$150,000; (ii)enter into any employment Contract, written or oral, or modification of the terms of any such existing Contract, except with respect to the hiring of any employee pursuant to the terms of this Agreement; (iii)enter into any collective bargaining agreement or relationship, (iv)grant any increase in the base compensation of any of its current or former directors, officers, managers, members, partners, employees or consultants, other than in connection with an annual review of the base compensation of such individuals, provided such increase is no larger than ten percent (10%) of any individual's base compensation, or (v)pay or agree to become obligated to pay any bonus, severance or change of control payments or other consideration of any nature whatsoever (other than salary, commissions or consulting fees) to any of its current or former directors, officers, managers, members, partners, employees or consultants;

(xxi) other than as required under <u>Section 6.15</u> or by such Employee Benefit Plan, as required by Applicable Law, or in the Ordinary Course of Business, adopt, amend, modify, or terminate any material Employee Benefit Plan;

(xxii) declare, set aside, make or pay any non-cash dividend or other non-cash distribution on or with respect to any Equity Interest or ownership interest in any Group Company;

(xxiii) accelerated the collection of or discount any accounts receivable, delay the payment of accounts payable or defer expenses, reduce inventories or otherwise increase cash on hand, except in the Ordinary Course of Business;

-46-

(xxiv) agree, commit or offer (in writing or otherwise) to take any of the actions described in clauses (i) through (xxiii) of this Section 6.4.

(b) Notwithstanding the foregoing, Buyers specifically agree that prior to the applicable Closing, the Group Companies may distribute any or all excess cash to the applicable Equityholders so long as any such distributions do not result in a failure by such Group Company to satisfy its financial requirements imposed by Applicable Laws.

6.5 Access to Information. The applicable Company shall afford to the applicable Buyer Party(ies) and its Representatives, reasonable access, upon notice during normal business hours during the period prior to the applicable Closing, to all the personnel, independent accountants, properties, books, Contracts, commitments, Tax Returns and records of Plan Companies or CFC IPA, as appropriate, and, during such period shall make available to Buyers any information concerning the Group Companies as Buyers may reasonably request; provided, that no Group Company shall be required to violate any attorney-client privilege. In particular, but without limitation, from and after the date of this Agreement, the applicable Buyer Party and its Representatives shall have the right and privilege, subject to compliance with Applicable Law, upon reasonable written request, of entering upon all properties leased or occupied by a Group Company and of reviewing the books and records of any such Group Company regarding such properties from time to time as needed to make any inspections, evaluations, surveys or tests which Buyers may deem necessary or appropriate. Any visit or investigation by Buyers shall be conducted in such a manner subject to compliance with Applicable Law and as not to interfere unreasonably with the conduct of the Business.

6.6 <u>Approvals and Consents</u>. On the terms and subject to the conditions of this Agreement, until the earlier of the applicable Closing Date or such time as this Agreement shall be terminated pursuant to <u>Article 9</u>:

(a) Each Party shall use its reasonable best efforts to cause each Closing to occur, including taking all reasonable actions necessary to comply promptly with all Applicable Laws that may be imposed on it or any of its Affiliates with respect to each Closing.

(b) The applicable Group Company shall use their respective best efforts to obtain and deliver to the Buyer Parties evidence reasonably satisfactory to Buyer Parties that the Initial Closing Required Consents and the Second Closing Required Consents, as the case may be, have been obtained. The applicable Group Companies shall request the Second Closing Required Consents from the applicable third-parties prior to the occurrence of the Initial Closing at a date mutually agreed by the Parties. If the Group Companies receive a notice of termination or breach from any counterparty to an Initial Closing Required Consent or a Second Closing Required Consent, then Buyer Parties shall be entitled to terminate this Agreement pursuant to Section 9.1(b).

(c) The Equityholders shall, subject to applicable legal limitations, provide information within their possession that is reasonably requested by Buyers in writing regarding the Group Companies and the Business in order to assist Buyers in its filings for regulatory approvals.

(d) Prior to the Second Closing and in accordance with <u>Section 6.8</u>, the DMHC shall have issued an order approving the Notice of Material Modification and such order shall be in full force.

6.7 No Negotiation. Until the earlier of the applicable Closing Date or such time as this Agreement shall be terminated pursuant to Article 9, neither the Group Companies nor any of their respective Affiliates or Representatives shall directly or indirectly solicit, initiate, encourage, entertain or respond to any inquiries or proposals from, discuss or negotiate with, provide any information to or consider the merits of any inquiries or proposals from any Person (other than Buyer) or otherwise participate in a transaction or potential transaction concerning any sale of all or substantially all of the assets of any Group Company, any direct or indirect sale of equity interests or other securities any Group Company, or any merger, consolidation or similar transaction involving the Equityholders or any Group Company. The Equityholder Representative shall notify Buyers of any such inquiry or proposal, and the material terms thereof within forty-eight (48) hours of receipt or awareness of the same by any Group Company or the Equityholders. As of the date of this Agreement, the Group Companies and each of their respective Affiliates and Representatives shall immediately cease and cause to be terminated any activities, discussions or negotiations being conducted, if any, prior to the date of this Agreement with any Persons other than the Buyer Parties with respect to any of the foregoing. The Group Companies will be responsible for any breach of this Section by their Affiliates or Representatives.

6.8 <u>Regulatory Matters</u>.

(a) In furtherance and not in limitation of the terms of <u>Section 6.6</u>, until the earlier of the Second Closing Date or such time as this Agreement shall be terminated pursuant to <u>Article 9</u>, the applicable Group Companies and Buyers shall promptly execute and file, or join in the execution and filing of, any application, notification, or other document that may be necessary in order to obtain the authorization, approval, or consent of any Governmental Authority, whether federal, state, local, or foreign, that may be reasonably required, or that Buyer Parties may reasonably request, in connection with the consummation of the Transactions.

(b) AHMS shall cause CFCHP to prepare, complete and properly submit the Notice of Material Modification and all supporting documentation to DMHC and PC Buyer and the Companies shall jointly meet with appropriate DMHC personnel to apprise the DMHC of the Transactions. From and after the date of execution of this Agreement and through the Second Closing Date, CFCHP and PC Buyer shall take any and all action as shall be reasonably requested by the other Party or required by the DMHC to maintain, protect and preserve the Business and the Parties' ability to consummate the Transactions. The Parties acknowledge that DMHC must have issued an order approving the Notice of Material Modification prior to the Second Closing Date. PC Buyer shall pay one hundred percent (100%) of all filing fees associated with the filings required by this <u>Section 6.6</u>.

-48-

6.9 Employee Matters.

(a) Prior to the Second Closing Date, the Plan Companies shall terminate all employees and independent contractors of such Plan Companies, and I Health or one of its Affiliates shall offer employment, effective as of the Second Closing Date, to those individuals listed on <u>Schedule 6.9</u>, if any (the "<u>Continuing Employees</u>"). For a period of one (1)year following the Second Closing Date (the "<u>Continuation Period</u>"),I Health shall, or shall cause its Affiliates to, provide to each of the Continuing Employees base salary, incentive compensation opportunities, and 401(k)benefits, substantially similar to those provided to the Continuing Employees immediately prior to the Second Closing Date. Nothing herein shall prevent I Health, the Buyer Parties, their Affiliates or the Group Companies from terminating the employment of any employee during the Continuation Period in compliance with Applicable Law.

(b) In connection with their termination of employment, the Plan Companies shall pay to employees all obligations due to them in connection with such termination under applicable law. To the extent that service is relevant for purposes of eligibility, vesting or, with respect to paid time off only, benefit accrual, under any employee plans and benefit arrangements of I Health which are established or maintained by I Health or one of its Affiliates for the benefit of Continuing Employees ("<u>I Health</u> <u>Plans</u>"),I Health shall cause such I Health Plans to credit such Continuing Employees for service with the Plan Companies prior to the Second Closing Date, except (x) as would cause a duplication of benefits or coverage for the same period of service or (y) to the extent such service was not recognized under the corresponding Employee Benefit Plan.

(c) At the Second Closing, the Plan Companies shall pay any accrued and unused paid time off due to the Continuing Employees.

(d) Notwithstanding any other provision of this Agreement, nothing contained in this <u>Section 6.9</u> shall (i)be deemed to be the adoption of, or an amendment to, any employee benefit plan, program, arrangement, contract or practice, or otherwise limit the right of the Group Companies, Buyer Parties or their respective Affiliates, to amend, modify or terminate any employee benefit plan, program, arrangement, contract or practice or (ii)give any third party any right to enforce the provisions of this <u>Section 6.9</u>.

6.10 <u>Tail Policies</u>.

(a) At or prior to the applicable Closing, CFC IPA or AHMS, as applicable, shall purchase a "tail" prepaid directors' and officers' liability, employment practices liability insurance policy, and any other insurance policies maintained by any Group Company as of immediately prior to the Closing, effective as of the applicable Closing Date, in each case, providing coverage for a period of six (6)years after the applicable Closing Date with respect to claims arising from facts or events that occurred on or before the Closing, and which tail policies shall contain substantially the same coverage and amounts as, and contain terms and conditions no less advantageous than, in the aggregate, the coverage urrently provided by such current policies (each, a "<u>Tail Policy</u>"). The Buyer Parties covenant and agree not to cancel, redeem or take any action that would adversely affect the terms and conditions of the Tail Policy. The cost and expense of purchasing the Tail Policy shall be shared equally between the applicable Group Company and the applicable Buyer Party.

-49-

(b) The Buyer Parties agree that all rights to indemnification and exculpation for acts or omissions occurring prior to the applicable Closing now existing in favor of the current or former managers, directors or officers (or persons holding similar positions) of the Group Companies who have the right to indemnification or exculpation by the Group Companies (collectively, the "<u>Covered Persons</u>") as provided in their respective Governing Documents and as provided under Applicable Law as in effect as of the date hereof shall survive the Transactions and shall continue in full force and effect in accordance with their terms for a period of six (6)years from the applicable Closing. Without limiting the foregoing, for a period of not less than six (6)years from the Closing, the Buyer Parties shall not, and shall not permit any Group Company to, amend, modify or terminate any Governing Document regarding or related to such indemnification matters in any manner that would adversely affect the rights thereunder of the Covered Persons, unless such modification is required by Applicable Law.

(c) The provisions of this <u>Section 6.10</u> are: (i)intended to be for the benefit of, and shall be enforceable by, each Covered Person and each other Person entitled to indemnification or coverage under a policy referenced in this <u>Section 6.10</u>, and each such Person's heirs, legatees, representatives and successors, it being expressly agreed that such Persons shall be third party beneficiaries of this <u>Section 6.10</u>, and (ii)in addition to, and not in substitution for, any other rights to indemnification or contribution that any such Person may have by Contract or otherwise.

(d) If the applicable Buyer or its successors or assigns: (i)shall consolidate with or merge into any other Person and shall not be the continuing or surviving corporation or entity of such consolidation or merger or (ii)shall transfer all or substantially all of its properties and assets to any Person, then, and in each such case,

proper provisions shall be made so that the successors and assigns of the applicable Buyer shall assume all of the obligations of such Buyer set forth in this Section 6.10.

6.11 Efforts; Regulatory Filings and Consents.

(a) Subject to the terms hereof, the Buyer Parties and the Group Companies shall each, as applicable:

(i) use its reasonable best efforts to take, or cause to be taken, all actions, and do, or cause to be done, and to assist and cooperate with the other Party in doing, all things necessary, proper or advisable to consummate and make effective the Transactions as promptly as practicable;

(ii) within five (5)Business Days after the date of this Agreement, make all necessary filings under the HSR Act, and thereafter make any other required submissions under the HSR Act and satisfy any related request of a Governmental Authority thereunder in each case as promptly as practicable (it being agreed that Buyer Parties shall not extend any waiting period or agree to refile under the HSR Act without the express written consent of the Companies and Equityholders);

(iii) use its reasonable best efforts to make, as promptly as practicable, all necessary filings, and thereafter make any other required submissions, with respect to this Agreement and the Transactions required under any other applicable law;

-50-

(iv) use its reasonable best efforts to obtain, as promptly as practicable, from any Governmental Authority or any other third party any consents, licenses, permits, waivers, approvals, authorizations or orders required to be obtained or made by it or any of its subsidiaries in connection with the authorization, execution and delivery of this Agreement and the consummation of the Transactions; and

(v) execute or deliver any additional instruments or documents necessary to consummate the transactions contemplated by, and to fully carry out the purposes of, this Agreement.

The Buyer Parties and the Group Companies, shall cooperate with each other in connection with the making of all such filings, as applicable, including providing copies of all such documents to the non-filing party and its advisors prior to filing and considering in good faith their comments in connection therewith. The Buyer Parties and the Group Companies shall furnish to each other, as applicable, all information required for any application or other filing to be made pursuant to the rulesand regulations of any applicable law in connection with the Transactions. For the avoidance of doubt, nothing contained in this <u>Section 6.11(a)</u>shall modify or affect the rights and responsibilities under <u>Section 6.11(b)</u>.

(b) Subject to the terms hereof, the Buyers Parties and the Companies, as applicable, shall, and shall cause each of their respective subsidiaries to, cooperate and use their respective reasonable best efforts to obtain any government clearances or approvals required for the Initial Closing under applicable Antitrust Laws and respond promptly to any government requests for additional information and documentation under any Antitrust Laws. Except to the extent prohibited by applicable law, the Parties will consult and cooperate with one another, and consider in good faith the views of one another, in connection with, and provide to the other in advance, any analyses, appearances, presentations, memoranda, briefs, arguments, opinions and proposals made or submitted by or on behalf of any Party in connection with proceedings under or relating to any Antitrust Law.

(c) The filing fees under the HSR Act and other applicable Antitrust Laws shall be paid by the Buyer Parties.

(d) To the extent any of the documents or information provided pursuant to this <u>Section 6.11</u> are commercially or competitively sensitive, the Equityholders and the Companies or the Buyers, as the case may be, may satisfy their respective obligations by providing such documents or information to the other Party's outside counsel, and materials may be redacted as necessary to comply with contractual arrangements, and as necessary to address attorney-client or other privilege concerns.

(e) The Group Companies and Buyer Parties acknowledge that if a Party fails to satisfy any conditions to Closing set forth in <u>Article 8</u>, including failure to obtain the regulatory approvals or consents that are required pursuant to <u>Article 8</u>, the other Party shall have no obligation to consummate the applicable Closing.

-51-

6.12 <u>Excluded Real Property</u>. Prior to the Second Closing, AHMS shall transfer the Excluded Real Property to certain AHMS Limited Partners such that AHMS shall not own any interest in the Excluded Real Property at the Initial Closing.

6.13 <u>R&W Insurance Policy</u>. The Buyer Parties shall cause the R&W Insurance Policy to be in full force and effect as of the Initial Closing Date. Any costs and expenses related to the R&W Insurance Policy, including the total premium, underwriting costs, brokerage commissions for the Buyer Party's broker, Taxes related to such policy and other fees and expenses of such policy shall be borne by the Buyer Parties. The R&W Insurance Policy provides that (a)the underwriters shall only be entitled to subrogate against the Companies, the Equityholders, the Equityholders' Representative or their Representatives if and to the extent that the payment of Damages thereunder arose out of Fraud, (b)the Equityholders may rely upon and enforce the preceding <u>Section 6.13(a)</u>as an express third-party beneficiary, and (c)the foregoing provisions of <u>Section 6.13(a)</u>and (b)in the R&W Insurance Policy may only be amended with the written consent of the Equityholder Representative. From and after the date hereof, the Buyer Parties (i)shall provide the Equityholder Representative with prompt written notice of any claim made against the R&W Insurance Policy after the applicable Closing Date and (ii)shall not amend, modify, supplement or otherwise change, terminate or waive any provision of the R&W Insurance Policy in a manner materially adverse to the Companies, the Equityholders Representative without the prior written consent of the Equityholder Representative (which consent shall be in the sole and absolute discretion of such Person). CFC IPA and/or the Equityholder Representative, as applicable, shall use Commercially Reasonable Efforts to cooperate with the Buyer Parties in connection with any claim made by any Buyer Party under the R&W Insurance Policy, to the extent that such reasonable cooperation would assist the Buyer Parties in pursuing and obtaining the maximum recovery available in respect of such claim.

6.14 <u>Advisory Committee</u>. Within thirty (30) days after the Initial Closing Date, the applicable Buyer Party shall form an advisory committee, consisting of three (3)members, one of which will include Accie Mitchell, M.D. The committee will meet semiannually and will discuss matters pertaining to the Business. Accie Mitchell, M.D. shall provide no more than four (4) hours of services per month and shall receive compensation in the amount of \$2,000 per month.

6.15 <u>Termination of 401(k)Plan</u>. The Group Companies shall take all actions necessary and appropriate (including fully vesting participants in their account balances) to terminate all Company 401(k)Plans, effective no later than the day immediately preceding the Second Closing Date, provided, however, that termination of such 401(k)Plans may be made contingent upon the occurrence of the Second Closing. The Buyer Parties shall be afforded a reasonable opportunity to review and comment on such resolutions. The provisions of this <u>Section 6.15</u> are solely for the benefit of the parties to this Agreement, and no current or former employee of the Group Companies shall be regarded for any purpose as a third-party beneficiary of this Agreement. In no event shall the terms of this <u>Section 6.15</u> be deemed to (a)establish, amend or modify any Employee Benefit Plan, (b)alter or limit the ability of the Group Companies or Buyer Parties to amend, modify or terminate any Employee Benefit Plan or any other benefit or employment plan, program or Contract, or (c)confer upon any current or former employee, officer, director or consultant any right to employment or continued employment or continued service with the Group Companies, Buyer Parties, or any of their respective Affiliates.

6.16 <u>Non-Competition; Non-Solicitation; Non-Disparagement</u>.

(a) In further consideration for the Buyer Parties' payment of the amounts to be paid to the Equityholders under this Agreement and the Transaction Documents, and in order to protect the value of the Group Companies and the business, including its goodwill, acquired by the Buyer Parties under this Agreement, each Equityholder hereby agrees, for itself and on behalf of its owners, as applicable, that, during the period commencing on the applicable Closing Date and ending on the fifth (5th) anniversary of the Initial Closing Date (the "<u>Restricted Period</u>"), each Equityholder shall not, and shall cause its Affiliates not to, directly or indirectly, acquire, engage or participate in (including by lending the Equityholder's name, providing financing, or owning or assisting another in), hold any economic, financial or other interest (including equity or debt securities) in, act as a stockholder, partner, consultant, agent, employee, co-venturer, member or representative of, render any services to, otherwise operate or hold an interest (whether an equity interest or otherwise) in, enter the employ of, or render any services to, any Person engaged or seeking to engage in any business the same or similar to the Protected Business as of the Closing Date within the State of California (the "<u>Restricted Territory</u>"); provided, that nothing contained herein shall be construed as prohibiting or prohibit the Equityholders from engaging in such restricted activities on behalf of, or for the benefit of, any Buyer Party or purchasing up to an aggregate of two percent (2%) of any class of the outstanding voting securities of any other Entity whose securities are listed on a national securities exchange, so long as the Equityholders and their respective Affiliates do not participate in any way in the management or control of such Person.

(b) In further consideration for the Buyer Parties' payment of the amounts to be paid to the Equityholders under this Agreement and the Transaction Documents, and in order to protect the value of the Group Companies and the Protected Business, the Equityholders (on his, her or their behalf) hereby agree that during the Restricted Period, the Equityholders shall not, directly or indirectly, solicit, interfere with, induce or attempt to solicit, interfere with or induce any (i) referral source or patient of the Protected Business, (ii)employee or contractor of the Protected Business, (iii)business relationship of the Protected Business to divert, reduce, adversely modify or terminate its business relationship with any Buyer Party or any Group Company or (iv)planned or contemplated potential acquisition or investment candidates of any Buyer Parties or any Group Companies.

(c) The Equityholders hereby agree that, during the Restricted Period, the Equityholders shall not make any negative comments or otherwise disparage any Group Company, Buyer Party or any of their respective Affiliates as of the date of the applicable Closing. Nothing herein or otherwise shall preclude any Equityholder from making truthful statements that are necessary to comply with Applicable Law or Governmental Authority, or to defend or enforce such Person's or any Group Company's rights under this Agreement, any Transaction Document or any other agreement between or among the Parties.

-53-

(d) The Equityholders understand that the restrictions set forth in this <u>Section 6.16</u> are intended to protect the interests of the Group Companies, Buyer Parties and their respective Affiliates in the Group Companies' proprietary information, goodwill and established employee, customer, supplier, consultant and vendor relationships, and agree that such restrictions are reasonable and appropriate for this purpose. The Equityholders further acknowledge and agree that the time, scope, geographic area and other provisions of this <u>Section 6.16</u> have been specifically negotiated by sophisticated parties and absent the Equityholders' agreement to and compliance with the restrictions set forth in this <u>Section 6.16</u>, the Buyer Parties would not have entered into the transactions contemplated by this Agreement or the Transaction Documents. The Restricted Period shall be extended with respect to any particular Equityholder by each day that such Equityholder, as applicable, is in breach of this <u>Section 6.16</u> are independent of any other obligation of any Group Company, Buyer Party or their respective Affiliates has to the Equityholders, including any such obligation under this <u>Agreement</u>.

(c) The Parties expressly agree that the subject matter, length of time, geographical scope, and range of activities, as applicable, contained in this <u>Section 6.16</u> are reasonable in light of the circumstances as they exist on the date upon which this Agreement has been executed. However, should a determination nonetheless be made by a court of competent jurisdiction at a later date that the character, duration or geographic scope of the restrictive covenants set forth in this <u>Section 6.16</u> is unreasonable in light of the circumstances as they then exist, then it is the intention and the agreement of the Parties hereto that this Agreement shall be construed by the court in such a manner as to impose only those restrictions on the conduct of the Equityholders that may be enforceable under Applicable Law, to the fullest extent of such enforceability to assure the Buyer Parties of the intended benefit of this Agreement. If, in any judicial proceeding, a court shall refuse to enforce all of the separate covenants that, if eliminated, would permit the remaining separate covenants to be enforced in such proceeding shall, for the purpose of such proceeding, be deemed eliminated from the provisions hereof.

(f) For the avoidance of doubt, nothing in this <u>Section 6.16</u> shall restrict the ability of the Equityholders to own or manage the Plan Companies and operate the Plan Business during the period between the date of this Agreement and the Second Closing. Additionally, in the event that the Second Closing does not occur, this <u>Section 6.16</u> shall in no way restrict the Equityholders' ownership and management of the Plan Companies or operation of the Plan Business.

6.17 <u>Financing Cooperation</u>. From and after the Initial Closing Date the Group Companies and each Equityholder shall use each of its commercially reasonable efforts to cooperate with Buyer Parties (at Buyer's cost and expense with respect to any reasonable documented out of pocket costs) in connection with Buyer Parties' obligation to arrange the financing by the Buyer Parties or its Affiliates for the Second Closing Payment (the "<u>Financing</u>"); provided, however, that nothing herein shall require (i)such cooperation to the extent that it would interfere materially and unreasonably with the Business and operations of the Plan Companies; or (ii)the taking of any action that would conflict with or violate the Governing Documents of any Plan Company or any Applicable Law. Without limiting the foregoing, nothing in this Section6.17 shall require any Equityholder or Plan Company to (a)incur any liability (or cause their respective directors, officers or employees to incur any liability) in connection with the Financing; or (b) enter into any definitive agreement or commitment related to Buyer Parties' financing obligations.

6.18 <u>Permit Transfers</u>. At the written request of PC Buyer, CFC IPA and IPA Equityholder shall prepare and, subject to PC Buyer's reasonable review, submit such filings, notices, submissions or requests for approval as are required by Applicable Law or by any Governmental Authority in order to transfer or assign any legally or contractually transferrable Permits not transferred or assigned to PC Buyer at the Initial Closing from CFC IPA to PC Buyer, or otherwise reasonably cooperate with PC Buyer in obtaining a reissuance or replacement of any such Permits, for PC Buyer's ownership or operation of the IPA Business or Acquired Assets, to the applicable Governmental Authority within a reasonable time as requested by PC Buyer in writing.

6.19 <u>Further Transfers</u>. CFC IPA and IPA Equityholder shall execute and deliver such further instruments of conveyance and transfer and take such additional action as PC Buyer may reasonably request to effect, consummate, confirm or evidence the transfer to PC Buyer of the Acquired Assets, the assumption by PC Buyer of the

Assumed Liabilities and the conduct by PC Buyer of the IPA Business (including with respect to transferring all transferrable Permits necessary or desirable in connection therewith) and CFC IPA and IPA Equityholder shall execute such instruments and other documents as necessary to reasonably assist PC Buyer in preserving or perfecting its rights in the Acquired Assets and its ability to conduct the IPA Business. CFC IPA and IPA Equityholder shall refer all inquiries by providers, payors or other commercial relations with respect to the IPA Business to PC Buyer.

6.20 <u>Use of Names</u>. CFC IPA and IPA Equityholder each agree that from and after the Initial Closing Date it will not use (and it will prevent its Affiliates (other than CFCHP) from using), (a) the name "Community Family Care" or any other name used by the IPA Business or included in the Acquired Assets, (b) any translations, transliterations, adaptations, derivations, acronyms, variations, insignias, designations or combinations of any name referenced in clause (a) or (c) any name likely to cause confusion with any of the names referenced in clause (a) or (b) (collectively, the "Business Names"). As soon as practicable after the Initial Closing, but no later than thirty (30) days following the Initial Closing, CFC IPA shall file to change each of its names and any "d/b/a" of it to a name that does not include any of the Business Names. On the Initial Closing Date, CFC IPA shall deliver a signed Fictitious Name Permit Cancellation Application to the Buyer Parties so that the Buyer Parties may file a new Fictitious Name Permit Application with the Medical Board of California.

6.21 <u>Wrong Pockets; Payments</u>. If, following the Initial Closing, either PC Buyer, on the one hand, or CFC IPA or IPA Equityholder, on the other hand, identifies any Acquired Asset that was not previously transferred by CFC IPA to PC Buyer, then CFC IPA shall execute, acknowledge and deliver all reasonable further documents and take all further reasonable steps in order to transfer, or cause to be transferred, to PC Buyer any such asset for no additional consideration. If, after the Initial Closing, CFC IPA or any of its Affiliates receives any payment arising from the operation of the IPA Business after the Initial Closing or any payment included in the Acquired Assets, CFC IPA shall promptly remit such funds to PC Buyer or its designee by wire transfer of immediately available funds, together with any supporting documentation that accompanies any such deposits or payments used funds to CFC IPA or its designee by wire transfer of immediately available funds, together with any supporting documentation that accompanies any such deposits or payments whether in electronic or physical form. If, after the Initial Closing, PC Buyer or any of its Affiliates receives any payments included in the Excluded Assets, PC Buyer shall promptly remit such funds to CFC IPA or its designee by wire transfer of immediately available funds, together with any supporting documentation that accompanies any such deposits or payments whether in electronic or physical form.

-55-

6.22 <u>Additional Restrictions on Transfers</u>. Neither CFC IPA nor the Buyer Parent Shares Transferees shall, during the six (6) month period following the Closing (the "<u>Additional Restriction Period</u>"), regardless of whether or not CFC IPA or the Buyer Parent Shares Transferees are then an "affiliate" of the Buyer Parent: (i) offer, pledge, sell, contract to sell, grant, lend, or otherwise transfer or dispose of, directly or indirectly, any of the Buyer Parent Shares; (ii) enter into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any of the Buyer Parent Shares, whether any such transaction described in clause (i) or (ii) above is to be settled by delivery of the Buyer Parent Shares, in cash or otherwise; or (iii) publicly disclose the intention to make any offer, sale, pledge or disposition, or to enter into any transaction, swap, hedge or other arrangement relating to any of the Buyer Parent Shares. CFC IPA acknowledges that Buyer Parent is relying upon the agreement set forth in this Section, which shall be irrevocable and shall be binding upon CFC IPA's heirs, legal representatives, successors and assigns, including the Buyer Parent Shares Transferees.

6.23 <u>Rule 144.</u> CFC IPA acknowledges that the Buyer Parent Shares will not be eligible for offer, resale, transfer, pledge or disposition pursuant to Rule 144 promulgated under the Securities Act ("<u>Rule 144</u>") until after the Additional Restriction Period expires. CFC IPA also acknowledges that if a holder of the Buyer Parent Shares is an "affiliate" of Buyer Parent following the Additional Restriction Period, any sales made by such holder also shall be subject to the limitations applicable to sales by an "affiliate" of Buyer Parent will use commercially reasonable efforts to file with the Securities and Exchange Commission all reports required to be filed to meet the requirements of Rule 144, Buyer Parent will use commercially reasonable efforts to file with the Securities and Exchange Commission all reports required to be filed to meet the requirements of Rule 144(c)(1). In addition, after the expiration of the Additional Restriction Period pursuant to Rule 144, if requested by CFC IPA or a Buyer Parent Shares Transferee, Buyer Parent shall reasonably cooperate with such holder to provide its transfer agent with customary documentation, including, if required by the transfer agent, an opinion of Buyer Parent's counsel, as required to remove, as applicable, any restrictive legends set forth on the stock certificates representing the Buyer Parent Shares or related to any book entry account holding such Buyer Parent Shares and make a new, un-legended stock certificate or entry for such book entry Buyer Parent Shares sonably coepsead or disposid or disposid of without restrictive legends, provided that Buyer Parent and the transfer agent have timely received from CFC IPA or the Buyer Parent Shares Sol or disposid or disposid of the Additional reasonably acceptable to Buyer Parent and the transfer agent in connection therewith. Neither Buyer Parent for so long as such holder continues to hold the Buyer Parent Shares.

-56-

Insurance. For a period of eighteen (18) months after the Initial Closing Date, Buyer Parties shall have the right to require CFC IPA to make an insurance 624 claim (to the extent that coverage exists when such claim is made) for any Damages relating to any Acquired Assets or Assumed Liabilities suffered by PC Buyer that are solely caused by or arose from the operation of the IPA Business or Acquired Assets prior to the Initial Closing (each, an "Insurance Claim") under the appropriate occurrence-based liability insurance policies of Seller (each, a "CFC IPA Insurance Policy") that covered CFC IPA at the time of such event or Damages. For the avoidance of doubt, no Buyer Party shall have the right to require that any Insurance Claim be made with respect to any Excluded Assets or Excluded Liabilities. For the purpose of making such Insurance Claim, PC Buyer shall follow the procedures set forth in Section 10.5(d), including without limitation providing written notice to CFC IPA, describing the Insurance Claim in reasonable detail, attaching copies of all material written evidence thereof that the Damages were solely caused by or arose from the operation of the IPA Business or Acquired Assets prior to Initial Closing, and indicating the estimated amount, if reasonably practicable, of the Damages that has been or may be sustained by PC Buyer. After receipt of such written notice from PC Buyer, CFC IPA and the IPA Equityholder shall have the rights set forth in Section 10.5(d). If PC Buyer is entitled to make an Insurance Claim, CFC IPA and the IPA Equityholder shall (a) cooperate with the Buyer Parties by providing information reasonably requested by the Buyer Parties regarding the CFC IPA Insurance Policy coverage, (b) make the Insurance Claim on behalf of the Buyer Parties if the Buyer Parties are not entitled to make such Insurance Claim directly under such CFC IPA Insurance Policy and (c) assist the Buyer Parties in the collection of proceeds in respect of such Insurance Claims. If CFC IPA or the IPA Equityholder receives insurance proceeds in respect of any such Insurance Claim, CFC IPA shall promptly remit such proceeds to PC Buyer or its designee(s). Except to the extent required under Article 10, for the avoidance of doubt, CFC IPA shall not be required to incur or expend additional costs or expenses or commence or file any Actions against any third party as a result of CFC IPA's or the IPA Equityholder's obligations under this Section 6.24.

6.25 <u>Bulk Sales</u>. The Parties hereby waive compliance with the bulk sales Laws of the state of California.

6.26 <u>Letter of Credit Requirements</u>. Prior to the date hereof, CFC IPA had obtained the letters of credit, security agreement and other Indebtedness identified on <u>Schedule 6.26</u> (the "<u>CFC LOC Obligations</u>") for the benefit of the applicable Plan as required under the applicable Contract. On or prior to the Initial Closing Date, (i) the applicable Buyer Party shall seek agreement with the Plans under any Contract with such Plans that is an Assumed Contract as to what credit support arrangements (e.g. notes, letters of credit, capitation withhold reserves, etc.) are required to be in place on and after the Initial Closing Date ("<u>LOC Requirement</u>"); and (ii) the applicable Buyer Party shall satisfy any such LOC Requirements so as to enable the assignment of such Assumed Contracts and to enable CFC IPA to be released from the CFC LOC Obligations.

6.27 <u>Additional Covenants</u>. On or prior to the Initial Closing, CFC IPA and CFCHP shall terminate that certain Independent Physician Association Services Agreement, dated October 1, 2020 (the "Existing CFC Services Agreement"), by and between CFC IPA and CFCHP, and PC Buyer and CFCHP shall enter into a new agreement substantially identical to the Existing CFC Services Agreement (the "New PC Buyer Services Agreement").

6.28 Disclosure Schedule Supplement. From and after the date of this Agreement until the applicable Closing, the Group Companies and the Equityholder Representative shall provide prompt notice to the Buyer Parties if any representation or warranty made by the Equityholders and the Group Companies in this Agreement was, when made, or has subsequently become, untrue in any material respect in the form of an updated Disclosure Schedule (each, a "Schedule Supplement"). Each such Schedule Supplement shall be deemed to be incorporated into and to supplement and amend the Disclosure Schedule solely with respect to any matter arising (or, in the case of matters for which such party's disclosure obligation hereunder is limited to the Knowledge of the applicable Company party, discovered) after the date hereof; provided, that no such amended, supplemental or new disclosure will (a) be deemed to have cured any breach of the representations and warranties set forth in <u>Article 3</u> and <u>Article 4</u> (i) for purposes of indemnification under <u>Article 10</u> or (ii) of determining satisfaction of the closing conditions in <u>Article 8</u> or (b) affect, limit, modify in any manner the indemnification obligations in <u>Article 10</u>, unless in each case, specifically agreed by the Buyer Parties in writing provided, however, that, notwithstanding the foregoing, the Group Companies shall be entitled to update the Disclosure Schedule for Contracts required to be disclosed pursuant to <u>Section 4.7</u> that are entered into between the date hereof and the applicable Closing Date, to the extent such Contracts are entered into in accordance with <u>Section 6.4</u>.

ARTICLE 7 Tax matters

7.1 <u>Tax Covenants</u>.

(a) The Equityholder Representative shall prepare, or cause to be prepared, at the Equityholders' sole expense, all Tax Returns required to be filed by the Plan Companies after the applicable Closing Date with respect to a Pre-Closing Tax Period, other than Tax Returns for Straddle Periods, and Buyers shall cause the Plan Companies to execute and file timely such Tax Returns as so prepared (subject to <u>Section 7.1(c)</u>).

(b) Buyer Parties shall prepare, or cause to be prepared, at Buyer Parties' expense, all Tax Returns of the Plan Companies required to be filed with respect to a Straddle Period, and Buyers and the Equityholders shall pay (or cause to be paid), when due, all Taxes reflected as due and payable on such Tax Returns to the extent allocated to such Party under <u>Section 7.2</u>, provided, however, that the Equityholders shall have no obligation to pay (or cause to be paid) any Taxes to the extent such Taxes are taken into account in the final determination of the Initial Closing Funded Indebtedness, Second Closing Funded Indebtedness, Initial Closing Company Transaction Expenses or Second Closing Company Transaction Expenses.

-58-

Any such Tax Return referred to in Section 7.1(a) or Section 7.1(b) shall be prepared in a manner consistent with past practice (unless otherwise (c) required by Applicable Law) and the provisions of this Agreement, without a change of any election or any accounting method, and shall be submitted by the Party preparing the Tax Return (Buyers or the Equityholder Representative) to the other Party (together with schedules, statements and, to the extent requested by Buyers or the Equityholder Representative, supporting documentation) at least forty-five (45) days prior to the due date (including extensions) of such Tax Return. If the Party reviewing the Tax Return objects to any item on any such Tax Return, it shall, within fifteen (15) days after delivery of such Tax Return, notify the Party that prepared the Tax Return, in writing, that it so objects, specifying with particularity any such item and stating the specific factual or legal basis for any such objection. If a notice of objection shall be duly delivered, Buyers, on the one hand and the Equityholder Representative, shall negotiate in good faith and use their reasonable best efforts to resolve such items. If Buyers and the Equityholder Representative are unable to reach such agreement within ten (10) days after receipt by Buyers of such notice, the disputed items shall be resolved by a Neutral Accounting Firm selected by the Equityholder Representative, and reasonably acceptable to Buyers (the "Accounting Referee") and any determination by the Accounting Referee shall be final. The Accounting Referee shall resolve any disputed items within twenty (20) days of having the item referred to it pursuant to such procedures as it may require. If the Accounting Referee is unable to resolve any disputed items before the due date for such Tax Return, the Tax Return shall be filed as prepared and then amended to reflect the Accounting Referee's resolution. The costs, fees and expenses of the Accounting Referee shall be borne by Buyers, jointly and severally, on the one hand, and the AHMS Equityholders, jointly and severally, on the other hand, in such amount(s) as shall be determined by the Accounting Referee based on the proportion that the aggregate amount of disputed items submitted to the Accounting Referee that is unsuccessfully disputed by Buyers, on the one hand, or the Equityholder Representative, on the other hand, as determined by the Accounting Referee, bears to the total amount of such disputed items so referred to the Accounting Referee for resolution. Notwithstanding the foregoing, the Parties agree that no value shall be ascribed to the payment rights under Section 1.3 hereof. Notwithstanding anything to the contrary contained in this Agreement, CFC IPA shall have sole and absolute control over the preparation and filing of any Tax Returns of CFC IPA.

7.2 <u>Straddle Period</u>. In the case of Taxes that are payable with respect to a Straddle Period, the portion of any such Taxes that are treated as Taxes of the Group Companies for any Pre-Closing Tax Period shall be:

(a) in the case of Taxes based upon, or related to, income or receipts, deemed equal to the amount which would be payable if the taxable year ended with the applicable Closing Date; and

(b) in the case of other Taxes, deemed to be the amount of such Taxes for the entire period multiplied by a fraction the numerator of which is the number of days in the period ending on the applicable Closing Date and the denominator of which is the number of days in the entire period.

(c) To the maximum extent permitted by applicable Law (including through the potential use of the safe harbor procedures of IRS Revenue Procedure 2011-29 with respect to any success-based fees), any deductions attributable to the payment or accrual of the Initial Closing Company Transaction Expenses and the Second Closing Company Transaction Expenses shall be reported and treated as income Tax deductions for the applicable Pre-Closing Tax Period.

7.3 <u>Tax Indemnification</u>.

(a) IPA Equityholder and the AHMS Limited Partners, severally but not jointly, shall indemnify the Buyer Indemnitees and hold each of them harmless from and against each such IPA Equityholder's and AHMS Limited Partner's Allocable Portion of any Damages attributable to (i) all Taxes (or the non-payment thereof) of any Group Company for all taxable periods ending on or before the applicable Closing Date and the portion through the end of the applicable Closing Date for any applicable Pre-Closing Tax Period; (ii) all Taxes of any member of an affiliated, consolidated, combined or unitary group of which any Group Company thereof (or any predecessor of any of the foregoing) is or was a member prior to the applicable Closing Date, including pursuant to Treasury Regulation section 1.1502-6 or any analogous or similar state, local or non-U.S. law or regulation; (iii) any and all Taxes of any Person (other than such Group Company) imposed on any Group Company thereof as a transferee or successor, by contract (other than a contract entered into in the ordinary course of business the primary purpose of which does not relate to Taxes) or pursuant to any law, which Taxes relate to an event or transaction occurring on or before the applicable Closing Date; (iv) all Taxes imposed on any Group Company as a result of an election made by such Company or Subsidiary under Code section 108(i); (v) failure to make a Push-Out Election with respect to AHMS; and (vi) reasonable third party costs and expenses associated with preparing any Tax Return for Pre-Closing Tax Periods.

(b) Equityholders shall reimburse Buyer Parties for any Taxes of any Group Company thereof that are the responsibility of Equityholders pursuant to this <u>Section 7.3</u> within fifteen (15) Business Days after payment of such Taxes by a Buyer Party or such Company.

(c) The indemnification under this Section 7.3 shall not be subject to the other limitations set forth in Article 10 (other than Sections 10.4(b)-(j)).

(d) Notwithstanding any other provision of this Agreement to the contrary, no Buyer Indemnitee shall have any right to indemnification or payment under this Agreement with respect to, or based on, Taxes to the extent such Taxes (i) are attributable to any Tax period other than a Tax period (or portion of a Straddle Period) ending on or before the applicable Closing Date, (ii) are due to the unavailability in any Tax periods (or portions thereof) beginning after the applicable Closing Date of any net operating losses, credits or other Tax attributes from a Tax period (or portion thereof) ending on or before the applicable Closing Date, (iii) result from any transactions or actions taken by, or omissions by, the Buyer Indemnitees or any of their Affiliates (including the Companies) after the applicable Closing that are not specifically contemplated by this Agreement, (iv) were already taken into account in the calculation of the Initial Closing Company Transaction Expenses, the Second Closing Funded Indebtedness, in each case as finally determined hereunder, or (v) are not attributable to a determination (within the meaning of Section 1313(a) of the Code or corresponding provisions of state or local Tax law) resulting from a Tax claim by a Tax Authority conducted in accordance with the procedures set forth in <u>Section 7.5</u> (other than with respect to any Taxes reflected on any Tax Return of a Company for a taxable period (or portion thereof) ending on or prior to the applicable Closing Date that are first due after the applicable Closing).

-60-

7.4 <u>Refunds</u>. Any credits or refunds of Taxes of the Group Companies with respect to Pre-Closing Tax Periods shall belong to the Equityholders and shall be paid to the Equityholder Representative for distribution to the Equityholders promptly upon actual receipt of such refund in cash or application of such credit actually to reduce other Taxes due and owing by Buyers, the Group Companies or any of their Affiliates. The amount of any refund or credit which is required to be paid to the Equityholder Representative shall be reduced by the amount of any Taxes on such refund or credit and reasonable out-of-pocket expenses that Buyers, the Group Companies or any of their respective Affiliates incur (or will incur) with respect to such refund or credit.

7.5 Contests. Buyers shall promptly notify the Equityholder Representative following receipt of any notice of audit or other proceeding relating to any Tax Return of any Group Company that includes a Pre-Closing Tax Period (the "Prior Period Returns"). The Equityholder Representative shall control any and all audits or other proceedings and litigation (a "Tax Contest") relating to any Prior Period Return, including the filing of an amended Tax Return, other than Prior Period Returns relating to any Straddle Period. With respect to any Tax Contest that it controls, the Equityholder Representative (i) shall give prompt notice to Buyers of any Tax adjustment proposed in writing pursuant to such Tax Contest, (ii) shall afford Buyers the opportunity to participate in the conduct of such Tax Contest, including without limitation the right to participate in conferences with Tax Authorities and submit pertinent material in support of Buyers' position, and (iii) shall not settle such Tax Contest without Buyers' consent, which shall not be unreasonably withheld, conditioned or delayed. MSO Buyers and the Equityholder Representative shall have joint control of any and all Tax Contests relating to a Tax Return for a Straddle Period, including the filing of an amended Tax Return for such Straddle Period, that neither the Equityholder Representative nor Buyers shall settle or compromise any Tax Contest relating to a Tax Return for a Straddle Period, including the filing of an amended Tax Return for such Straddle Period, the other Party, which consent shall not be unreasonably withheld, conditioned or delayed. To the extent permitted under Applicable Law, the AHMS Equityholders shall make, or cause to be made, an election under Code section 6226 and Treasury Regulations thereunder with respect to any Tax Contest in respect of AHMS in respect of any Pre-Closing Tax Period (a "<u>Push-Out Election</u>"). Notwithstanding anything to the contrary contained in this Agreement, CFC IPA shall have sole and absolute control over any Tax Co

7.6 Cooperation and Exchange of Information. The Equityholder Representative and Buyers shall provide each other with such cooperation and information as either of them reasonably may request of the other in filing any Tax Return pursuant to this <u>Article 7</u> or in connection with any audit or other proceeding in respect of Taxes of the Group Companies. Such cooperation and information shall include providing copies of relevant Tax Returns or portions thereof, together with accompanying schedules, related work papers and documents relating to rulings or other determinations by Tax Authorities, making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder, and the provision of any powers of attorney or similar authorizations, or other designations, as may be necessary to effectuate the intent of this <u>Article 7</u>. The Equityholder Representative and Buyers shall retain, or cause to be retained, all Tax Returns, schedules and work papers, records and other documents in their possession relating to Tax matters of the Group Companies for any taxable period beginning before the applicable Closing Date until the expiration of the statute of limitations of the respective Tax periods. Prior to transferring, destroying or discarding any Tax Returns, schedules and work papers, records and other documents in its possession relating to Tax matters of the Group Companies for any taxable period beginning before the applicable Closing Date, the Equityholders or Buyers (as the case may be) shall provide the other Party with reasonable written notice and offer the other Party the opportunity to take custody of such materials.

-61-

7.7 <u>Post-Closing Actions</u>. None of Buyers, Buyer Parent, the Plan Companies or any of their Affiliates shall (a) make any election under Section 338 or Section 336 of the Code (or any similar election pursuant to any provision of federal, state or local Applicable Law) with respect to any Plan Company in connection with the Transactions, (b) amend any Tax Return of any Plan Company or change or revoke any election, relating to a Tax period (or portion thereof) of any Plan Company ending on or before the applicable Closing Date, (c) initiate any voluntary disclosure or similar proceeding with any Tax Authority relating to a Tax period (or portion thereof) of any Plan Company ending on or before the applicable Closing Date, or (d) make any other Tax election or take any other action that has the effect of increasing the amount of Taxes for which any Equityholder may be liable under the terms of this Agreement or under applicable law, in each case without the prior written consent of the Equityholder Representative.

7.8 Intended Tax Treatment; Purchase Price Allocation.

(a) Unless otherwise required by a "determination" within the meaning of Section 1313 of the Code (or any similar applicable provision of state or local income Tax law), the Parties agree to treat for U.S. federal, state and local income Tax purposes, and so acknowledge, the purchase and sale of the AHMS GP Interests and the AHMS LP Interests as (i) a sale of partnership interests in AHMS by the AHMS General Partner and the AHMS Limited Partner under Section 741 of the Code and the purchase of the assets of AHMS as described in Revenue Ruling 99-6, Situation 2, and (ii) a termination of AHMS as a partnership for Tax purposes under Code section 708(b) and conversion to a "disregarded entity" with respect to MSO LP Buyer as a result of MSO GP Buyer being a "disregarded entity" with respect to MSO LP Buyer for U.S. federal and applicable state and local income Tax purposes.

(b) Unless otherwise required by a "determination" within the meaning of Section 1313 of the Code (or any similar applicable provision of state or local income Tax law), the Parties agree to treat for U.S. federal, state and local income Tax purposes, and so acknowledge, (i) PC Buyer's acquisition of the Acquired Assets from CFC IPA as a taxable sale by CFC IPA to PC Buyer of the Acquired Assets in exchange for the CFC IPA Cash Purchase Price, the Buyer Parent Shares and the CFC IPA Deferred Payments the under Section 1001 of the Code (and similar applicable provisions of state or local income Tax law), and (ii) that a portion of the CFC IPA Deferred Payments shall be characterized as interest to the extent required by Code section 1275 and determined under Treasury Regulation section 1.1275-4(c).

(c) Unless otherwise required by Applicable Law, the Parties agree to treat for California sales and use tax purposes, and so acknowledge, PC Buyer's acquisition the Acquired Assets from CFC IPA, if and to the extent the same consist of tangible personal property, as an "occasional sale" under sections 6006.5(a) and 6367 of the California Revenue and Taxation Code.

(d) Within sixty (60) days after the final determination of the Initial Closing Purchase Price, the Equityholder Representative shall provide Buyer Parent with an allocation of the Initial Closing Purchase Price (together with any assumed liabilities and any other amounts required to be taken into account for U.S. federal income Tax purposes) (the "<u>Initial Closing Tax Purchase Price</u>") among the Acquired Assets in a manner consistent with Section 1060 of the Code, the Treasury Regulations thereunder, the principles set forth on <u>Schedule 7.8</u>, and the provisions of this Agreement (the "<u>Draft CFC IPA Purchase Price Allocation</u>") for the Buyer Parent's review, comment, and consent. If the Buyer Parent disagrees with any item set forth in the Draft CFC IPA Purchase Price Allocation, Buyer Parent shall, within thirty (30) days of delivery of the Draft CFC IPA Purchase Price Allocation to Buyer Parent, provide to the Equityholder Representative written notice of its objection to the Draft CFC IPA Purchase Price Allocation and the reasons therefor, in which event Buyer Parent and the Equityholder Representative shall endeavor in good faith to agree upon an allocation schedule. If the parties cannot resolve such objection, the item in question shall be resolved by the Accounting Referee in accordance with the provisions of <u>Section 7.1(c)</u>, *mutatis mutandis*. The final allocation schedule, determined pursuant to the previous two sentences, as applicable, shall be the "<u>Final CFC IPA Purchase Price Allocation</u>."

(c) Within sixty (60) days after the final determination of the Second Closing Purchase Price, the Equityholder Representative shall provide Buyer Parent with an allocation of the Second Closing Purchase Price (together with any assumed liabilities and any other amounts required to be taken into account for U.S. federal income Tax purposes) among the assets of AHMS in a manner consistent with Section 1060 of the Code, the Treasury Regulations thereunder, the principles set forth on <u>Schedule 7.8</u>, and the provisions of this Agreement (the "<u>Draft AHMS Purchase Price Allocation</u>") for the Buyer Parent's review, comment, and consent. If the Buyer Parent disagrees with any item set forth in the Draft AHMS Purchase Price Allocation, Buyer Parent shall, within thirty (30) days of delivery of the Draft AHMS Purchase Price Allocation to Buyer Parent, provide to the Equityholder Representative written notice of its objection to the Draft AHMS Purchase Price Allocation, Buyer Parent and the Equityholder Representative shall endeavor in good faith to agree upon an allocation schedule. If the parties cannot resolve such objection, the item in question shall be resolved by the Accounting Referee in accordance with the provisions of <u>Section 7.1(c)</u>, *mutatis mutandis*. The final allocation schedule, determined pursuant to the previous two sentences, as applicable, shall be the "<u>Final AHMS Purchase Price Allocation</u>."

(f) Buyers, Buyer Parent, the Companies, the Equityholders and their respective Affiliates shall, unless otherwise required by a "determination" within the meaning of Section 1313 of the Code (i) file all Tax Returns (including amended returns and claims for refund) and information reports in a manner consistent with the Final CFC IPA Purchase Price Allocation and Final AHMS Purchase Price Allocation and the intended tax treatment in <u>Section 7.8(a)</u> through <u>Section 7.8(c)</u> hereof, inclusive; (ii) treat and report the transactions described above in all respects consistently with the provisions of this Agreement for purposes of any federal, state, local or foreign Tax; and (iii) not take any actions or positions inconsistent with the obligations of the parties set forth herein

-63-

7.9 <u>Transfer Taxes</u>. Notwithstanding anything to the contrary herein, any transfer, sales, stamp, use documentary, property, or similar Taxes applicable to, imposed upon or arising out of the Transactions (such Taxes, "<u>Transfer Taxes</u>") will be borne entirely by Buyers. Buyers, the Companies, and the Equityholder Representative shall cooperate with each other in preparing and filing all filings, Tax Returns, reports and forms as necessary or appropriate to comply with the provisions of all applicable law in connection with the payment of such Transfer Taxes, and shall cooperate in good faith to minimize the amount of any such Transfer Taxes payable in connection therewith, provided that Buyers shall be responsible for preparing all such Tax Returns and related forms or filings, shall provide each such Tax Return or form or filing to the Equityholder Representative for its review and approval at least five (5) Business Days prior to the due date thereof, and no Equityholder shall have any obligation to execute any such form or filing unless Buyers revise such form or filing for any reasonable comments provided by the Equityholder Representative in respect thereof.

7.10 <u>Overlap</u>. To the extent that any obligation or responsibility pursuant to <u>Article 10</u> may overlap with an obligation or responsibility pursuant to this <u>Article 7</u>, the provisions of this <u>Article 7</u> shall govern.

ARTICLE 8 Conditions to Closing

The conditions to the obligations of CFC IPA, and the applicable Buyer Parties with respect to the Initial Closing shall be set forth in <u>Sections 8.1, 8.2</u>, and <u>8.3</u> and the conditions to the obligations of AHMS, the Equityholders and the relevant Buyer Parties with respect to the Second Closing shall be set forth in <u>Sections 8.4, 8.5</u>, and <u>8.6</u>.

8.1 <u>Conditions to the Obligations of All Parties with Respect to the Initial Closing</u> The obligations of each Party to effect the closing of the Transactions contemplated to take place at the Initial Closing shall be subject to the fulfillment, at or prior to the Initial Closing, of each of the following conditions:

(a) any applicable waiting period under the HSR Act relating to the Transactions shall have expired or been terminated.

(b) No Applicable Law or Judgment enacted, entered, promulgated, enforced or issued by any Governmental Authority or other legal restraint or prohibition preventing the consummation of the Transactions contemplated to take place at the Initial Closing shall be in effect.

(c) No Action shall be pending or threatened which (i) prevents or challenges consummation of the Transactions contemplated to take place at the Initial Closing; or (ii) causes such Transactions to be rescinded following consummation.

8.2 <u>Initial Closing Conditions to Obligations of Buyer Parties</u> The obligations of Buyer Parties to effect the closing of the Transactions contemplated to take place at the Initial Closing shall be subject to the fulfillment or Buyer Parties' waiver, at or prior to the Initial Closing, of each of the following conditions:

(a) The representations and warranties of CFC IPA and IPA Equityholder set forth in this Agreement, other than the representations set forth in <u>Sections</u> <u>3.1</u>(Capacity), 3.2 (Authority), 3.3 (Title to Equity Interests), 4.1 (Organization), 4.2 (Authorization) and 4.3 (Capitalization) (the <u>Specified Representations</u>"), and each Transaction Document and any other writing delivered pursuant hereto or thereto, shall be true and correct in all respects (in the case of any representation or warranty qualified by materiality or "Material Adverse Effect") or in all material respects (in the case of any representations and warranties that are made as of a specific date need be true and correct only as of such date. The Specified Representations shall be true and correct in all respects as of the Initial Closing Date as though made on and as of the Initial Closing Date, except that the Specified Representations that are made as of a specific date need be true and correct only as of such date. (b) The Group Companies, the Equityholders and the Equityholder Representative shall have duly performed and complied in all material respects with all agreements, conditions and covenants required by this Agreement and each of the other Transaction Documents to be performed or complied with by them prior to or on the Initial Closing Date; <u>provided</u>, that with respect to agreements, covenants and conditions qualified by materiality or "Material Adverse Effect", the Group Companies, the Equityholders and the Equityholder Representative shall have duly performed and complied with all agreements, conditions and covenants, as so qualified, in all respects.

(c) From the date of this Agreement, there shall not have occurred any Material Adverse Effect.

(d) At or prior to the Initial Closing, the IPA Equityholder shall cause to be delivered to the applicable Buyer Party the following, in each case, in form and substance reasonably satisfactory to the applicable Buyer Party:

(i) evidence of termination of the Company 401(k) Plans in accordance with <u>Section 6.15</u>;

(ii) evidence of the filing under the Department of Labor's Delinquent Filer Voluntary Compliance Program and the Internal Revenue Services' Late Filer Penalty Relief Program, as necessary, with respect to the delinquent Form 5500 series annual return for the 2020 plan year for The Community Family Care Defined Benefit Pension Plan;

(iii) a certificate, dated as of the Initial Closing Date and signed by a duly authorized officer of CFC IPA, that each of the conditions set forth in Section 8.2(a), Section 8.2(b) and Section 8.2(c) have been satisfied;

(iv) a certificate of the Secretary (or equivalent officer) of CFC IPA certifying in their corporate capacity (i) that attached thereto are true and complete copies of all resolutions adopted by the board of directors/managers/members or general partner, as applicable, of CFC IPA authorizing the execution, delivery and performance of this Agreement and the Transaction Documents and the consummation of the Transactions, (ii) that all such resolutions are in full force and effect and are all the resolutions adopted in connection with the Transactions, (iii) the names and signatures of the officers of CFC IPA authorized to sign this Agreement and the Transaction Documents, (iv) that there have been no amendments to the incorporation or formation documents since the date such documents were delivered pursuant to <u>Section 8.2(d)(v)</u>, and (v) that attached thereto are true and complete copies of all Governing Documents of CFC IPA;

-65-

(v) the incorporation or formation document and all amendments thereto of CFC IPA, duly certified as of a recent date by the Secretary of State of the State of incorporation or formation, as the case may be, of such Entity;

(vi) a good standing certificate (or its equivalent) as of a recent date from the secretary of state or similar Governmental Authority of the jurisdiction in which CFC IPA is organized;

(vii) the Escrow Agreement duly executed by the Equityholder Representative;

(viii) a bill of sale and assignment and assumption agreement in the form of <u>Exhibit D</u> (the "<u>Bill of Sale</u>") and such other instruments of sale, assignment, transfer and conveyance as may be necessary to evidence and effect the sale and transfer to PC Buyer of the Acquired Assets and the assumption by PC Buyer of the Assumed Liabilities, in each case, duly executed by CFC IPA;

(ix) the Management Services Agreement, duly executed by AHMS;

(x) an IRS Form W-9 executed by CFC IPA;

(xi) evidence of termination of the CFC LOC Obligations pursuant to <u>Section 6.26</u>;

(xii) the New PC Buyer Services Agreement duly executed by CFCHP; and

(xiii) such other documents or instruments as Buyer Parties reasonably request and are reasonably necessary to consummate the Transactions.

(e) The (i) third party consents, notices and filings and (ii) notices, filings, consents and authorizations identified on <u>Schedule 8.2(e)</u> in form and substance reasonably satisfactory to the Buyer Parties shall have been duly delivered and obtained (collectively, the "<u>Initial Closing Required Consents</u>").

(f) CFC IPA shall have delivered to PC Buyer evidence of termination of the Contracts between CFC IPA, on the one hand, and any of its Affiliates, Related Parties, or any Affiliate of any Related Party, on the other hand, as set forth on <u>Schedule 8.2(f)</u>, in each case, in form and substance reasonably satisfactory to PC Buyer.

-66-

To the extent that, at or prior to the Initial Closing, the Equityholder Representative delivers to the Buyer Parties a written notice specifying the failure of any of the foregoing conditions and the Buyer Parties nevertheless proceed with the Initial Closing, the Buyer Parties shall be deemed to have waived for all purposes any rights or remedies it may have against CFC IPA, IPA Equityholder, IPA Beneficial Owner or Equityholder Representative by reason of the failure of any such conditions.

8.3 <u>Initial Closing Conditions to Obligations of CFC IPA and the IPA Equityholder</u>. The obligations of CFC IPA and the IPA Equityholder to consummate the Transactions shall be subject to the fulfillment or the Equityholder's waiver, at or prior to the Initial Closing, of each of the following conditions:

(a) The representations and warranties of the Buyer Parties set forth in this Agreement, other than the Buyer Fundamental Representations, and each Transaction Document and any other writing delivered pursuant hereto or thereto, shall be true and correct in all respects (in the case of any representation or warranty qualified by material Adverse Effect") or in all material respects (in the case of any representations and warranties that are made as of a specific date need be true and correct only as of such date. The Buyer Fundamental Representations shall be true and correct in all respects as of the Initial Closing Date as though made on and as of the Initial Representations shall be true and correct in all respects as of the Initial Closing Date as though made on and as of the Initial Closing Date, except that representations and warranties that are made as of a specific date need be true and correct only as of such date.

(b) The Buyer Parties shall have duly performed and complied in all material respects with all agreements, conditions and covenants required by this Agreement and each of the other Transaction Documents to be performed or complied with by them prior to or on the Initial Closing Date; <u>provided</u>, that with respect to agreements, covenants and conditions qualified by materiality or "Material Adverse Effect", the Buyer Parties shall have duly performed and complied with all agreements, conditions and covenants, as so qualified, in all respects.

(c) At or prior to Initial Closing, the Buyer Parties shall each deliver to the Equityholder Representative the following, each case, in form and substance reasonably satisfactory to the Equityholder Representative:

(i) a good standing certificate (or its equivalent) as of a recent date from the secretary of state or similar Governmental Authority of the jurisdiction under the Applicable Laws in which the Buyer Parties are organized;

(ii) a certificate, dated as of the Initial Closing Date and signed by a duly authorized officer of the applicable Buyer Party, that each of the conditions set forth in Section 8.3(a), Section 8.3(b) and Section 8.2(c) have been satisfied;

(iii) a certificate of the Secretary (or equivalent officer) of each Buyer Party certifying in their corporate capacity (i) that attached thereto are true and complete copies of all resolutions adopted by the board of directors/manager/member of such the Buyer Party authorizing the execution, delivery and performance of this Agreement and the Transaction Documents and the consummation of the Transactions, (ii) that all such resolutions are in full force and effect and are all the resolutions adopted in connection with the Transactions, and (iii) the names and signatures of the officers of such the Buyer Party authorized to sign this Agreement and the Transaction Documents;

-67-

(iv) the Escrow Agreement duly executed by Buyer Parent and the Escrow Agent;

(v) the Management Services Agreement, duly executed by PC Buyer and MSO LP Buyer;

(vi) evidence that the LOC Requirements are satisfied;

(vii) the New PC Buyer Services Agreement duly executed by PC Buyer; and

(viii) the payments described in <u>Section 1.4(a)</u> above.

(d) The R&W Insurance Policy shall be in full force and effect and the Buyer Parties shall have delivered evidence reasonably satisfactory of the same

to CFC IPA.

(e) To the extent that, at the Initial Closing, the Buyer Parties deliver to the Equityholder Representative a written notice specifying the failure of any of the foregoing conditions and CFC IPA nevertheless proceeds with the Initial Closing, CFC IPA, the IPA Equityholder and the Equityholder Representative shall be deemed to have waived for all purposes any rights or remedies they may have against the Buyer Parties by reason of the failure of any such conditions.

8.4 <u>Second Closing Conditions to Obligations of All Parties</u>. The obligations of each Plan Company, the AHMS Equityholders and applicable Buyer Party to consummate the Transactions contemplated herein to be consummated at the Second Closing shall be subject to the fulfillment, at or prior to the Second Closing, of each of the following conditions:

(a) No Applicable Law or Judgment enacted, entered, promulgated, enforced or issued by any Governmental Authority or other legal restraint or prohibition preventing the consummation of the Transactions contemplated herein to be consummated at the Second Closing shall be in effect.

(b) No Action shall be pending or threatened which (i) prevents or challenges consummation of the Transactions contemplated herein to be consummated at the Second Closing; or (ii) causes such Transactions to be rescinded following consummation.

(c) the DMHC shall have issued an order approving the Notice of Material Modification prior to the Second Closing and such order shall be in full force.

-68-

8.5 <u>Second Closing Conditions to Obligations of Buyer Parties</u> The obligations of Buyer Parties to effect the closing of the Transactions contemplated to take place at the Second Closing shall be subject to the fulfillment or Buyer Parties' waiver, at or prior to the Second Closing, of each of the following conditions:

(a) The representations and warranties of the Plan Companies and the AHMS Equityholders set forth in this Agreement, other than the Specified Representations, and each Transaction Document and any other writing delivered pursuant hereto or thereto, shall be true and correct in all respects (in the case of any representation or warranty qualified by materiality or "Material Adverse Effect") or in all material respects (in the case of any representation or warranty not qualified by materiality or "Material Adverse Effect") or in all material respects (in the case of any representation or warranty not qualified by materiality or "Material Adverse Effect") or and as of the Second Closing Date as though made on and as of the Second Closing Date, except that representations and warranties that are made as of a specific date need be true and correct only as of such date. The Specified Representations that are made as of a specific date need be true and correct only as of such date.

(b) The Plan Companies, the AHMS Equityholders, and the Equityholder Representative shall have duly performed and complied in all material respects with all agreements, conditions and covenants required by this Agreement and each of the other Transaction Documents to be performed or complied with by them prior to or on the Second Closing Date; provided, that with respect to agreements, covenants and conditions qualified by materiality or "Material Adverse Effect", the Plan Companies, the AHMS Equityholders, and the Equityholder Representative shall have duly performed and complied with all agreements, conditions and covenants, as so qualified, in all respects.

(c) From the date of this Agreement, there shall not have occurred any Material Adverse Effect.

(d) At or prior to the Second Closing, the Equityholder Representative shall cause to be delivered to the applicable Buyer Party the following, each case, in form and substance reasonably satisfactory to the applicable Buyer Party:

(i) for uncertificated AHMS Interests, an assignment of equity interest in a form reasonably acceptable to such Buyer Party;

- (ii) an IRS Form W-9 executed by the AHMS General Partner and each AHMS Limited Partner; and
- (iii) such other documents or instruments as the Buyer Parties reasonably request and are reasonably necessary to consummate the

Transactions.

(e) At or prior to the Second Closing, the Plan Companies shall cause to be delivered to the applicable Buyer Parties the following, each case, in form and substance reasonably satisfactory to the applicable Buyer Parties:

(i) resignations of the directors, managers and officers of each of the Plan Companies;

-69-

(ii) a certificate, dated as of the Second Closing Date and signed by a duly authorized officer of each Plan Company, that each of the conditions set forth in Section 8.5(a), Section 8.5(b) and Section 8.5(c) have been satisfied;

(iii) a certificate of the Secretary (or equivalent officer) of each Plan Company certifying in their corporate capacity (i) that attached thereto are true and complete copies of all resolutions adopted by the board of directors/managers/members or general partner, as applicable, of such Plan Company authorizing the execution, delivery and performance of this Agreement and the Transaction Documents and the consummation of the Transactions, (ii) that all such resolutions are in full force and effect and are all the resolutions adopted in connection with the Transactions, (iii) the names and signatures of the officers of each Plan Company authorized to sign this Agreement and the Transaction Documents, (iv) that there have been no amendments to the incorporation or formation documents since the date such documents were delivered pursuant to Section 8.5(e)(iv), and (v) that attached thereto are true and complete copies of all Governing Documents of each Plan Company;

(iv) the incorporation or formation document and all amendments thereto of each Plan Company, duly certified as of a recent date by the Secretary of State of the State of incorporation or formation, as the case may be, of such Plan Company;

(v) a good standing certificate (or its equivalent) as of a recent date from the secretary of state or similar Governmental Authority of the jurisdiction in which each Plan Company is organized; and

(vi) evidence of termination of employment with each of the employees and independent contractors of the Plan Companies;

(vii) such other documents or instruments as Buyer Parties reasonably request and are reasonably necessary to consummate the Transactions.

(f) The (i) third party consents, notices and filings and (ii) notices, filings consents and authorizations identified on <u>Schedule 8.5(f)</u> in form and substance reasonably satisfactory to the applicable Buyer Parties shall have been duly delivered and obtained (collectively, the <u>"Second Closing Required Consents</u>"); and

(g) The Equityholder Representative shall have delivered to the Buyer Parties evidence of termination of the Contracts between any of AHMS, CFCH and CFCHP, on the one hand, and any of their respective Affiliates, Related Parties, or any Affiliate of any Related Party, on the other hand, including those set forth on <u>Schedule 8.5(g)</u>, in each case, in form and substance reasonably satisfactory to the Buyer Parties.

(h) CFCHP shall be licensed by the DMHC as a Restricted Knox-Keene health care service plan.

To the extent that, at or prior to the Second Closing, the Equityholder Representative delivers to the Buyer Parties a written notice specifying the failure of any of the foregoing conditions and the Buyer Parties nevertheless proceed with the Second Closing, the Buyer Parties shall be deemed to have waived for all purposes any rights or remedies it may have against the Equityholders by reason of the failure of any such conditions.

-70-

8.6 <u>Second Closing Conditions to Obligations of the Plan Companies and the AHMS Equityholders</u> The obligations of the Plan Companies and the AHMS Equityholders to consummate the Transactions shall be subject to the fulfillment or the Equityholder Representative's waiver, at or prior to the Second Closing, of each of the following conditions:

(a) The representations and warranties of the Buyer Parties set forth in this Agreement, other than the Buyer Fundamental Representations, and each Transaction Document and any other writing delivered pursuant hereto or thereto, shall be true and correct in all respects (in the case of any representation or warranty qualified by material Adverse Effect") or in all material respects (in the case of any representation or warranty not qualified by materiality or "Material Adverse Effect") on and as of the Second Closing Date as though made on and as of the Second Closing Date, except that representations and warranties that are made as of a specific date need be true and correct only as of such date. The Buyer Fundamental Representations that are made as of a specific date need be true and correct only as of such date.

(b) The Buyer Parties shall have duly performed and complied in all material respects with all agreements, conditions and covenants required by this Agreement and each of the other Transaction Documents to be performed or complied with by them prior to or on the Second Closing Date; <u>provided</u>, that with respect to agreements, covenants and conditions qualified by materiality or "Material Adverse Effect", the Buyer Parties shall have duly performed and complied with all agreements, conditions and covenants, as so qualified, in all respects.

(c) At or prior to Second Closing, the Buyer Parties shall each deliver to the Equityholder Representative the following, each case, in form and substance reasonably satisfactory to the Equityholder Representative:

(i) a good standing certificate (or its equivalent) as of a recent date from the secretary of state or similar Governmental Authority of the jurisdiction under the Applicable Laws in which the Buyer Parties are organized;

(ii) a certificate, dated as of the Second Closing Date and signed by a duly authorized officer of the applicable Buyer Party, that each of the conditions set forth in Section 8.6(a), Section 8.6(b) and Section 8.6(c) have been satisfied; and

(iii) a certificate of the Secretary (or equivalent officer) of each Buyer Party certifying in their corporate capacity (i) that attached thereto are true and complete copies of all resolutions adopted by the board of directors/manager/member of such the Buyer Party authorizing the execution, delivery and performance of this Agreement and the Transaction Documents and the consummation of the Transactions, (ii) that all such resolutions are in full force and effect and are all the resolutions adopted in connection with the Transactions, and (iii) the names and signatures of the officers of such the Buyer Party authorized to sign this Agreement and the Transaction Documents.
(d) The R&W Insurance Policy shall be in full force and effect and the Buyer Parties shall have delivered evidence reasonably satisfactory of the same to the Equityholder Representative.

(e) The applicable Buyer Parties shall make the payments described in <u>Section 1.4(b)</u> above relating to the Second Closing.

To the extent that, at the Second Closing, the Buyer Parties deliver to the Equityholder Representative a written notice specifying the failure of any of the foregoing conditions and the Equityholder Representative nevertheless proceeds with the Second Closing, the Equityholders shall be deemed to have waived for all purposes any rights or remedies they may have against the Buyer Parties by reason of the failure of any such conditions.

ARTICLE 9 Termination

9.1 <u>Termination of the Agreement</u>. Notwithstanding anything to the contrary in this Agreement, this Agreement may be terminated and the Transactions abandoned at any time prior to the Initial Closing:

(a) by the mutual written consent of the Equityholder Representative and Buyer Parties;

(b) by Buyer Parties, in the event of (i) any material breach by the Equityholders or Group Companies of any of the Equityholders' or Group Companies' covenants, obligations, agreements, representations or warranties contained herein and the failure of the Equityholders or Group Companies to cure, if curable, such breach within thirty (30) days after receipt of written notice from Buyers requesting such breach to be cured and, if not cured (if capable of cure), such breach would result in the failure of the conditions set forth in Section 8.1 or 8.2 not being satisfied as of the Initial Closing, or (ii) receipt of a termination or breach notice pursuant to Section 6.6(b); provided, however, that the right to terminate this Agreement under this Section 9.1(b) will not be available to Buyer Parties if Buyer Parties are in material breach of any of their respective covenants, obligations or agreements under this Agreement;

(c) by the Equityholder Representative, in the event of any material breach by any Buyer Parties of any of such Buyer Party's covenants, obligations, agreements, representations or warranties contained herein and the failure of Buyer Parties to cure, if curable, such breach within thirty (30) days after receipt of written notice from the Equityholder Representative requesting such breach to be cured and, if not cured (if capable of cure), such breach would result the failure of the conditions set forth in <u>Sections 8.1</u> or <u>8.3</u> not being satisfied as of the Initial Closing; provided, however, that the right to terminate this Agreement under this <u>Section 9.1(c)</u> will not be available to the Equityholder Representative if any Seller or Plan Company is in material breach of any of its respective covenants, obligations or agreements under this Agreement;

(d)by the Equityholder Representative, on the one hand, or Buyer Parties, on the other hand, if the Initial Closing and Second Closing does not occur on or prior to the date that is March 6, 2024 and July 24, 2024 days, respectively, after (but not including) the date of this Agreement (such dates, the "<u>Outside Dates</u>"); provided, <u>however</u>, that the right to terminate this Agreement under this <u>Section 9.1(d)</u> will not be available to any Party whose material breach of any of its covenants, obligations or agreements under this Agreement has been the cause of or resulted in the failure of the applicable Closing to occur on or prior to such date; or

(e) by Buyer Parties, on the one hand or the Equityholder Representative, on the other hand, if there will be in effect a final, non-appealable Judgment of a court of competent jurisdiction in effect precluding consummation of the Transactions; <u>provided</u>, <u>however</u>, that the right to terminate this Agreement under this <u>Section 9.1(e)</u> will not be available to any Party whose failure to fulfill any of its covenants, obligations or agreements under this Agreement has been the cause of or resulted in such Judgment.

9.2 Effect of Termination

(a) If this Agreement is terminated for any reason, the provisions of this Section 9.2 and Article 12 shall remain in full force and effect.

(b) If this Agreement is terminated by any Party for any reason, the Deposit Escrow Amount shall be released to CFC IPA. Simultaneously with the delivery of this Agreement, the applicable Parties shall sign and deliver the joint instructions attached hereto as Exhibit C instructing the Deposit Escrow Agent to release the Deposit Escrow Amount to CFC IPA, which shall be delivered to the Escrow Agent upon the termination of this Agreement.

(c) If this Agreement is terminated as provided in this <u>Section 9.1(a)</u>, this Agreement shall forthwith become void (except as stated in <u>Section 9.2(a)</u>) and there shall be no Liability or obligation hereunder on the part of any Party hereto or their respective Representatives.

(d) If this Agreement is terminated as provided in subsections (a), (b), (c) or (e) of Section 9.1, such termination shall be without prejudice to any rights that the terminating Party may have against any breaching Party or any other Person under the terms of this Agreement or otherwise.

ARTICLE 10 Indemnification

10.1 <u>Survival</u>.

(a) Each representation and warranty contained herein and in any Transaction Document shall survive the applicable Closing Date until, and will expire and be of no force and effect on the three (3) year anniversary of the Second Closing Date (provided that, if the Second Closing does not occur, of the Initial Closing Date), except that the representations and warranties in <u>Sections 3.1</u> (Capacity); <u>3.2</u> (Authority and Enforceability); <u>3.3</u> (Title to Equity Interests); <u>3.4</u> (No Conflict); <u>3.5</u> (Legal Proceedings); <u>3.6</u> (No Brokerage); <u>4.1</u> (Organization and Good Standing; Subsidiaries); <u>4.2</u> (Authorization and Effect); <u>4.3</u> (Capitalization); <u>4.4</u> (Non-Contravention; Consents); <u>4.13</u> (Healthcare Regulatory Matters); <u>4.19</u> (Employee Benefit Matters) <u>4.20</u> (Brokers) (collectively, the "<u>Seller Fundamental Representations</u>") and the Buyer Fundamental Representations shall survive the applicable Closing until the date that is six (6) years from the Second Closing Date with respect to the subject matter of any such representation or warranty. All covenants and agreements contained in this Agreement and the Transaction Documents, shall survive the applicable Closing for the period contemplated by its terms.

(b) No Indemnified Person shall be entitled to make any claim in respect of any representation, warranty, covenant or agreement contained in this Agreement or any Transaction Document after the expiration of its applicable survival date as set forth in <u>Section 10.1(b)</u> (each, a "<u>Survival Date</u>"), except that any bona fide

claim initiated by an Indemnified Person prior to the expiration of the applicable Survival Date in accordance with the provisions hereof shall survive until it is settled or resolved pursuant to this Agreement to the extent that an Indemnified Person provides written notice of such breach or inaccuracy (which notice shall describe the applicable breach or inaccuracy in reasonable detail, the amount or the estimated amount of damages sought thereunder to the extent then ascertainable, any relevant time constraints relating thereto, and any other material details pertaining thereto) to the party to provide indemnity prior to the applicable Survival Date. The Parties specifically and unambiguously intend that the survival periods that are set forth in this <u>Section 10.1</u> shall replace any statute of limitations that would otherwise be applicable.

(c) The representations and warranties of each of the Parties set forth in this Agreement are material and shall be deemed to have been relied upon by the Party or Parties to whom they are made and shall survive as set forth above, regardless of any investigation on the part of such Party or its Representatives.

10.2 Indemnification by the Equityholders.

(a) Subject to the other terms and conditions of this <u>Article 10</u>, CFC IPA, IPA Equityholder and IPA Beneficial Owner shall indemnify and defend the Buyer Parties and their respective Affiliates and their respective successors, assigns, heirs and Representatives (collectively, the "<u>PC Buyer Indemnitees</u>") against, and shall hold each of them harmless from and against, and shall pay and reimburse each of them for, any and all Damages incurred or sustained by, or imposed upon, the PC Buyer Indemnitees based upon, arising out of, with respect to or by reason of:

(i) any breach or inaccuracy of any of the representations or warranties of the IPA Equityholder or CFC IPA as set forth in <u>Article 3</u> or <u>Article 4</u> of this Agreement or any Transaction Documents delivered by the IPA Equityholder or CFC IPA;

(ii) any failure to perform or breach of any covenant or agreement of IPA Equityholder or CFC IPA contained in this Agreement or any Transaction Document delivered by the IPA Equityholder or CFC IPA;

(iii) any Indebtedness of CFC IPA incurred or owed by CFC IPA prior to the Initial Closing but not discharged at or prior to the Initial Closing (except, in each case, to the extent deducted from the Initial Closing Payment in the calculation thereof or taken into account in the determination of any Final Initial Closing Adjustment and/or Final Calculation of the Initial Closing Payment);

(iv) any Excluded Liabilities;

-74-

(v) any Company Transaction Expenses of IPA Equityholder, CFC IPA, IPA Beneficial Owner or their respective Affiliates (other than AHMS or its subsidiaries or any of the AHMS Equityholders) incurred by such Person prior to the Initial Closing or owed by such Person but not discharged at or prior to the Initial Closing (except, in each case, to the extent deducted from the Initial Closing Payment in the calculation thereof or taken into account in the determination of any Final Initial Closing Adjustment and/or Final Calculation of the Initial Closing Payment);

(vi) any Fraud committed by or on behalf of the IPA Equityholder, the IPA Beneficial Owner or CFC IPA; or

(vii) the (i) Option Agreement, dated December 27, 2022, with Marc Mitchell, as Trustee of the Marc Mitchell Family Irrevocable Trust, dated December 6, 2022, or (ii) Option Agreement, dated December 27, 2022, with Alex M. Mitchell, as Trustee of the Alex Mitchell Family Irrevocable Trust.

(b) Subject to the other terms and conditions of this <u>Article 10</u>, the AHMS Equityholders shall, on a joint and several basis, indemnify and defend the PC Buyer Indemnitees, the Plan Companies after the Second Closing, and their respective Affiliates and their respective successors, assigns, heirs and Representatives (together with the PC Buyer Indemnitees, the "<u>Buyer Indemnitees</u>") against, and shall hold each of them harmless from and against, and shall pay and reimburse each of them for, any and all Damages incurred or sustained by, or imposed upon, the Buyer Indemnitees based upon, arising out of, with respect to or by reason of:

(i) any breach or inaccuracy of any of the representations or warranties concerning any Plan Company set forth in<u>Article 4</u> of this Agreement or any Transaction Document;

(ii)

any failure to perform or breach of any covenant or agreement of any Plan Company contained in this Agreement or any Transaction

Document;

(iii) any Indebtedness of a Plan Company incurred prior to the Second Closing or owed by such Person but not discharged at or prior to the Second Closing (except, in each case, to the extent deducted from the Second Closing Payment in the calculation thereof or taken into account in the determination of any Final Calculation of the Second Closing Payment);

(iv) any Company Transaction Expenses of any Plan Company incurred prior to the Second Closing or owed by such Persons but not discharged at or prior to the Second Closing (except, in each case, to the extent deducted from the Second Closing Payment in the calculation thereof or taken into account in the determination of any Final Calculation of the Second Closing Payment);

(v) any inaccuracy in the Closing Payments Schedule; or

(vi) any Fraud committed by or on behalf of any Plan Company or any AHMS Equityholder, in each instance, solely with respect to the Plan Companies and Plan Business.

(c) Subject to the other terms and conditions of this Article 10, each of the AHMS Equityholders shall, severally and not jointly, indemnify and defend the Buyer Indemnitees against, and shall hold each of them harmless from and against, and shall pay and reimburse each of them for, any and all Damages incurred or sustained by, or imposed upon, the Buyer Indemnitees based upon, arising out of, with respect to or by reason of:

(

(ii)

(i) any breach or inaccuracy of any of the representations or warranties set forth in <u>Article 3</u> of this Agreement by such AHMS Equityholder;

any failure to perform or breach of any covenant or agreement of such AHMS Equityholder contained in this Agreement or any Transaction

Document;

(iii) any Indebtedness of such Equityholder or any of its respective Affiliates, in each instance arising from or relating to a Plan Company, incurred prior to the Second Closing or owed by such Persons prior to the Second Closing but not discharged at or prior to the Second Closing (except, in each case, to the

extent deducted from the Second Closing Payment in the calculation thereof or taken into account in the determination of any Final Calculation of the Second Closing Payment); or

(iv) any Company Transaction Expenses of such Equityholder or its respective Affiliates incurred prior to the Second Closing or owed by such Persons prior to the Second Closing but not discharged at or prior to the Second Closing (except, in each case, to the extent deducted from the Second Closing Payment in the calculation thereof or taken into account in the determination of any Final Calculation of the Second Closing Payment).

-76-

(d) The amount of any Damages that are subject to indemnification under this <u>Article 10</u> shall be calculated net of any amounts actually recovered by the Buyer Indemnitee under insurance policies in connection with such Damages (including the R&W Insurance Policy), net of any insurance deductibles. The Buyer Parties shall use their reasonable commercial efforts to recover all amounts payable from an insurer or other third party under any such insurance policy or Contract (including the R&W Insurance Policy) prior to seeking indemnification hereunder. In the event that a recovery is made by a Buyer Indemnitee with respect to any Damages for which such Buyer Indemnitee has already been indemnified hereunder, then a refund equal to the amount of the recovery shall be made promptly (as applicable) to CFC IPA or the AHMS Equityholders in accordance with the Pro Rata Portion, as applicable.

(c) All of the representations and warranties set forth in this Agreement or any certificate or schedule that are so qualified as to "material," "materiality," "material respects," "Material Adverse Effect" or words of similar import or effect shall be deemed to have been made without such qualification for purposes of determining (i) whether a breach of such representation or warranty has occurred and (ii) the amount of Damages resulting from, arising out of or relating to any such breach of such representation or warranty; provided, however, that (i) such qualifications shall not be disregarded (x) in determining whether Fraud exists, or (y) with respect to references of the defined terms "Material Contract" or "Material Permits", and (ii) the term "Material Adverse Effect" contained in <u>clause (a) of Section 4.6</u> shall not be ignored for purposes of determining any such inaccuracy or breach thereof.

10.3 <u>Indemnification by the Buyer Parties</u>. Subject to the other terms and conditions of this <u>Article 10</u>, the Buyer Parties shall, jointly and severally, indemnify and defend each of the Equityholders and their respective successors, assigns, heirs and Representatives (collectively, the "<u>Equityholder Indemnitees</u>") against, and shall hold each of them harmless from and against, and shall pay and reimburse each of them for, any and all Damages incurred or sustained by, or imposed upon, the Equityholder Indemnitees based upon, arising out of, with respect to or by reason of:

(a) any breach or inaccuracy of any of the representations or warranties made by any Buyer Party in this Agreement or any Transaction Document;

(b) any breach of any covenant or agreement of any Buyer Party contained in this Agreement or any Transaction Document;

(c) any Assumed Liabilities;

-77-

(d) any Liabilities arising on or after the Second Closing Date with respect to any Continuing Employee, solely to the extent any such Liabilities relate to acts or omissions that occurred on or following the Second Closing Date; or

- (e) any Fraud committed by or on behalf of any Buyer Party.
- 10.4 Certain Limitations. The indemnification provided for in Section 10.2 and Section 10.3 shall be subject to the following additional limitations:

(a) No Equityholder shall have any liability for indemnification in respect of any claim for indemnification under<u>Section 10.2(a)(i)</u>, 10.2(b)(i) or 10.2(c) (i) until the aggregate amount of Damages indemnifiable pursuant to such Sections exceeds Six Hundred Thirty Thousand Dollars (\$630,000) (<u>Basket</u>"), and then the Equityholders shall only be liable for such amounts in excess of the Basket; <u>provided</u>, <u>however</u>, that the aggregate amount of all Damages under<u>Sections 10.2(a)(i)</u>, 10.2(b)(i) or 10.2(c) (i) and or 10.2(c)(i) for which the Equityholders may be liable to the Buyer Indemnitees shall not exceed Six Hundred Thirty Thousand Dollars (\$630,000) (the <u>Cap</u>"), and thereafter the Buyer Indemnitees' only recourse in respect of any Damages exceeding the Cap shall be with respect to the R&W Insurance Policy; <u>provided</u>, that the Cap shall be reduced to Three Hundred Sixty Thousand Dollars (\$360,000), on the twelve (12)-month anniversary of the Initial Closing Date; <u>provided</u>, further, <u>however</u>, that neither the Basket nor the Cap shall apply to any claim for or based on Fraud. Notwithstanding anything to the contrary in this Agreement, in the event a Buyer Indemnitee's right to indemnification pursuant to this <u>Article 10(i)</u> results from or arises under both <u>Section 10.2(a)(i)</u> and <u>Section 10.2(a)(i)</u> ("<u>Dual Claims</u>"), such Damages shall be deemed to result from or arise solely under <u>Section 10.2(a)(i)</u> and be subject to the limitations on liability applicable thereto (including the Basket and Cap).

(b) Notwithstanding anything to the contrary in this Agreement, other than any claim for or based on Fraud, no Equityholder shall have liability for any reason under this Agreement in an aggregate amount greater than, in the case of the AHMS Equityholders, the Pro Rata Portion of the cash Purchase Price such Equityholder actually receives in connection with the sale of the AHMS Interests, and in the case of the IPA Equityholder and IPA Beneficial Owner, the sum of the CFC IPA Cash Purchase Price plus the value of any CFC IPA Deferred Payments actually paid by PC Buyer.

(c) Buyer Parties shall not (nor permit any Affiliate thereof to) solicit claims from any Persons for the primary purpose of creating claims for indemnification for which any Equityholder would be responsible under this Agreement.

(d) For all purposes of this <u>Article 10</u>, "Damages" shall be net of any amounts paid or payable to an Indemnified Person under any insurance policy or Contract (including the R&W Insurance Policy) in connection with the facts giving rise to the right of indemnification hereunder, and each Indemnified Person shall use its reasonable commercial efforts to recover all amounts payable from other third party under any such Contract (including the R&W Insurance Policy) prior to seeking indemnification hereunder; <u>provided</u>, <u>however</u>, that the amount deemed to be paid under such insurance policies shall be net of the deductible for such policies; and<u>provided</u>, <u>further</u>, that the Equityholder Representative shall be subrogated (and Buyer Parties shall and shall cause Buyer Indemnitees to cause the Equityholder Representative to be subrogated) to the rights of Buyer Indemnitees under the R&W Insurance Policy.

(e) In calculating any Damages, there shall be deducted any Tax benefit, credit or refund to which the applicable Indemnified Person actually realizes or receives as a result of such Damages.

(f) If the amount to be netted hereunder from any indemnification payment required hereunder is determined after payment by an Indemnifying Person to an Indemnified Person of any amount otherwise required to be paid as indemnification pursuant hereto, the Indemnified Person shall repay, promptly after such determination, any amount that the Indemnifying Person would not have had to pay pursuant hereto had such determination been made at the time of such payment.

(g) Each Indemnified Person shall use its Commercially Reasonable Efforts to mitigate any indemnifiable Damage.

(h) The covenants of the Equityholders and the Group Companies contained in this Agreement may not be used to circumvent the negotiated limitations (e.g., knowledge qualifiers, materiality standards, dollar thresholds, survival periods and the like) contained in the representations and warranties and procedures with respect to the recovery by Buyer Indemnitees on account of the breach by any Equityholder or any Group Company or any of their respective representations or warranties.

(i) Upon making any payment to an Indemnified Person for any indemnification claim pursuant to this<u>Article 10</u>, the Indemnifying Person shall be subrogated, to the extent of such payment, to any rights which the Indemnified Person may have against any third parties with respect to the subject matter underlying such indemnification claim and the Indemnified Person shall assign any such rights to the Indemnifying Person.

(j) Notwithstanding the fact that any Indemnified Person may have the right to assert claims for indemnification under or in respect of more than one provision of this Agreement in respect of any fact, event, condition or circumstance, no Indemnified Person shall be entitled to recover the amount of any Damage suffered by such Indemnified Person more than once, regardless of whether such Damage may be as a result of a breach of more than one representation, warranty, obligation or covenant or otherwise. In addition, any liability for indemnification hereunder shall be determined without duplication of recovery by reason of the state of facts giving rise to such liability, or a breach of more than one representation, warranty, covenant or agreement, as applicable.

10.5 <u>Indemnification Procedures</u>. The Party making a claim under this <u>Article 10</u> is referred to as the "<u>Indemnified Person</u>", and the Party against whom such claims are asserted under this <u>Article 10</u> is referred to as the "<u>Indemnifying Person</u>". The Equityholder Representative shall act on behalf of the Sellers in connection with any claim made under this <u>Article 10</u>.

-79-

Third Party Claims. If, on or after the applicable Closing Date, any Indemnified Person receives notice of the assertion or commencement of any (a) Action made or brought by any Person who is not a Party or an Affiliate of a Party or a Representative of the foregoing (a "Third Party Claim") against such Indemnified Person with respect to which the Indemnifying Person is obligated to provide indemnification under this Agreement, the Indemnified Person shall give the Indemnifying Person reasonably prompt written notice thereof, but in any event not later than thirty (30) days after receipt of such notice of such Third Party Claim. The failure to give such prompt written notice shall not, however, relieve the Indemnifying Person of its indemnification obligations, except and only to the extent that the Indemnifying Person forfeits rights or defenses by reason of such failure. Such notice by the Indemnified Person shall describe the Third Party Claim in reasonable detail, shall include copies of all material written evidence thereof and shall indicate the estimated amount, if reasonably practicable, of the loss that has been or may be sustained by the Indemnified Person. Subject in all respects to the R&W Insurance Policy, if an Indemnified Person gives notice to the Indemnifying Person pursuant of the assertion of a Third Party Claim, then the Indemnifying Person shall be entitled to participate in the defense of such Third Party Claim and, to the extent that it wishes to assume the defense of such Third Party Claim, with counsel of its choosing. After notice from the Indemnifying Person to the Indemnified Person of its election to assume the defense of such Third Party Claim, the Indemnifying Person shall not, so long as it diligently conducts such defense, be liable to the Indemnified Person under this Article 10 for any fees of other counsel or any other expenses with respect to the defense of such Third Party Claim. The assumption of the defense of a Third Party Claim shall not be an admission as to the liability of the Indemnifying Person in respect of such Third Party Claim or be deemed an acceptance or acknowledgment of the Indemnifying Person's obligation to indemnify the Indemnified Person in respect of such Third Party Claim. CFC IPA and/or the Equityholder Representative, as applicable, and Buyer Parties shall cooperate with each other in all reasonable respects in connection with the defense of any Third Party Claim.

(b) <u>Cooperation regarding Third Party Claims</u>. With respect to any Third Party Claim, both the Indemnified Person and the Indemnifying Person, as the case may be, shall keep the other Person fully informed of the status of such Third Party Claim and any related Actions at all stages thereof where such Person is not represented by its own counsel. The Parties agree to provide reasonable access to the other Parties to such documents and information as may be reasonably requested in connection with the defense, negotiation or settlement of any such Third Party Claim; <u>provided, however</u>, that the Parties shall cooperate in such a manner as to preserve in full (to the extent possible) the confidential information and the attorney-client and work-product privileges of the other Parties. In connection therewith, each Party agrees that: (i) it will use Commercially Reasonable Efforts, in respect of any Third Party Claim in which it has assumed or participated in the defense, to avoid production of confidential information (consistent with Applicable Law and rules of procedure); and (ii) all communications between any Party and counsel responsible for or participating in the defense of any Third Party Claim shall, to the extent possible, be made so as to preserve any applicable attorney-client or work-product privilege.

(c) <u>Settlement of Third Party Claims</u>. If a firm offer is made to settle a Third Party Claim without leading to liability or the creation of a financial or other obligation on the part of the Indemnified Person and provides, in customary form, for the unconditional release of each Indemnified Person from all liabilities and obligations in connection with such Third Party Claim and the Indemnifying Person desires to accept and agree to such offer, the Indemnifying Person shall give written notice to that effect to the Indemnified Person. If the Indemnified Person has assumed the defense pursuant to <u>Section 10.5(a)</u>, it shall not agree to any settlement without the written consent of the Indemnifying Person (which consent shall not be unreasonably withheld or delayed).

-80-

(d) Direct Claims. Any Action by an Indemnified Person on account of any Damages which does not result from a Third Party Claim (a 'Direct Claim') shall be asserted by the Indemnified Person giving the Indemnifying Person reasonably prompt written notice thereof, but in any event not later than sixty (60) days after the Indemnified Person becomes aware of such Direct Claim. The failure to give such prompt written notice shall not, however, relieve the Indemnifying Person of its indemnification obligations, except and only to the extent that the Indemnifying Person forfeits rights or defenses by reason of such failure. Such notice by the Indemnified Person shall describe the Direct Claim in reasonable detail, shall include copies of all material written evidence thereof and shall indicate the estimated amount, if reasonably practicable, of the loss that has been or may be sustained by the Indemnified Person. The Indemnifying Person shall have sixty (60) days after its receipt of such notice to respond in writing to such Direct Claim, and whether and to what extent any amount is payable in respect of the Direct Claim and the Indemnified Person shall allow the Indemnifying Person and its professional advisors to investigate the matter or circumstance alleged to give rise to the Direct Claim, and whether and to what extent any amount is payable in respect of the Direct Claim and the Indemnified Person shall assist the Indemnifying Person's investigation by giving such information and assistance (including access to the Companies' premises and personnel and the right to examine and copy any accounts, documents or records) as the Indemnifying Person or any of its professional advisors may reasonably request. If the Indemnified Person shall be deemed to have rejected such claim, in which case the Indemnified Person shall be free to pursue such remedies as may be available to the Indemnifying Person on the terms and subject to the provisions of this Agreement. If there is a dispute regarding a Direct Claim the dispute

10.6 <u>Tax Treatment of Indemnification Payments</u>. All indemnification payments made under this Agreement shall be treated for Tax purposes by the Parties as an adjustment to the Purchase Price, unless otherwise required by Applicable Law.

10.7 <u>Recourse</u>.

(a) Notwithstanding anything to the contrary contained in this Agreement but subject to the limitations set forth in Section 10.4, other than any claim for or based on Fraud, the sole and exclusive source of payment for Damages resulting or arising from claims for indemnification pursuant to Sections 10.2(a)(i), and 10.2(b)(i) for breaches of representations will be sought only from the following sources and only in the following sequence: (i) first, from the Escrow Amount up to the Cap (as may be adjusted pursuant to Section 10.4) pursuant to Section 10.7(b), and (ii) second, from the R&W Insurance Policy to the full extent of coverage available therefor under the R&W Insurance Policy (provided that, (x) to the extent that the R&W Insurance Policy coverage limits have been exhausted, then Buyer Parties shall not have any rights of recourse against the Equityholders, and (y) to the extent that any breaches of representation and warranties claim has been denied or rejected by the R&W Insurance Policy insurer, or the R&W Insurance Policy does not cover the full amount of any Damages, Buyer Parties shall not have any rights of recourse against any Equityholders with respect thereto).

(b) If any Buyer Indemnitee is entitled to any payment of an Escrow Amount pursuant to this <u>Article 10</u> other than pursuant to <u>Sections 10.2(a)(i)</u>, <u>10.2(a)(iv)</u> with respect to Dual Claims, <u>10.2(b)(i)</u> and <u>10.2(c)(i)</u> for breaches of representations and warranties, then the following shall occur:

(i) In the case of any amount owed therefrom in respect of an indemnification obligation contained in<u>Section 10.2(a)</u>, (x) the Equityholder Representative and Buyer Parent shall promptly, in no event later than three (3) Business Days after the determination of such Buyer Indemnitee's entitlement to such amount under this <u>Article 10</u>, execute and deliver to Escrow Agent joint written instructions directing the release to such Buyer Indemnitee from the Escrow Account of an amount equal to the lesser of the then remaining balance of the CFC Indemnity Escrow Amount and such amount, and (y) the Buyer Indemnitee may seek the amount of any unpaid Damages from CFC IPA, IPA Equityholder, IPA Beneficial Owner and/or the applicable Indemnifying Person directly.

(ii) In the case of any amount owed therefrom in respect of an indemnification obligation contained in<u>Section 10.2(b)</u>, (x) Equityholder Representative and Buyer Parent shall promptly, in no event later than three (3) Business Days after the determination of such Buyer Indemnitee's entitlement to such amount under this <u>Article 10</u>, execute and deliver to Escrow Agent joint written instructions directing the release to such Buyer Indemnitee from the Escrow Account of an amount equal to the lesser of the then remaining balance of the AHMS Indemnity Escrow Amount and such amount, and (y) the Buyer Indemnitee may seek the amount of any unpaid Damages from the AHMS Equityholders.

(iii) In the case of any amount owed therefrom in respect of an indemnification obligation contained in<u>Section 10.2(c)</u>, the Buyer Indemnitee may seek the amount of such Damages from (i) the R&W Insurance Policy to the full extent of coverage available therefor under the R&W Insurance Policy subject to <u>Section 10.9</u>, and (ii) such AHMS Equityholder responsible for such breach directly.

(c) On the eighteen (18) month anniversary of the Closing Date, Equityholder Representative and Buyer Parent shall execute and deliver to Escrow Agent joint written instructions directing the release to CFC IPA from the Escrow Account of the then remaining balance of the CFC Indemnity Escrow Amount and to the AHMS Equityholders in accordance with the Pro Rata Portion from the Escrow Account of the then remaining balance of the AHMS Indemnity Escrow Amount, in each case, subject to any pending claims and pursuant to the terms and conditions of the Escrow Agreement.

10.8 Exclusive Remedies. Notwithstanding anything to the contrary herein, except as provided in<u>Section 2.2</u> (Post-Closing Determination) and <u>Article 7</u> (Tax Matters), and except for in respect of any Fraud from and after the applicable Closing, the rights and remedies of the Buyer Parties, the Group Companies, the Equityholders, and any Buyer Indemnified Person under this <u>Article 10</u> are exclusive and in lieu of any and all other rights and remedies which the Buyer Parties, the Group Companies, the Equityholders, any Buyer Indemnified Person under this <u>Article 10</u> are exclusive and in lieu of any and all other rights and remedies which the Buyer Parties, the Group Companies, the Equityholders, any Buyer Indemnified Person under this <u>Article 10</u> are exclusive and in lieu of any and all other rights and remedies which the Buyer Parties, the Group Companies, the Equityholders, any Buyer Indemnitee or any Indemnified Person, may have under this <u>Agreement</u> or any Transaction Document or otherwise against each other with respect to this <u>Agreement</u> or any Transaction Document and with respect to the transactions contemplated hereby or thereby, and from and after the applicable Closing each party expressly waives and releases and agrees to waive and release any and all other rights or causes of action it or its Affiliates may have against the other Party or its Affiliates now or in the future under any Applicable Law (regardless of the theory of recourse) with respect to the preceding matters. IN FURTHERANCE OF THE FOREGOING, FROM AND AFTER THE APPLICABLE CLOSING, EACH PARTY HEREBY WAIVES, WITH RESPECT TO THIS AGREEMENT AND THE TRANSACTIONS, ALL OTHER RIGHTS AND REMEDIES ARISING UNDER OR BASED UPON ANY STATUTORY OR COMMON LAW OR OTHERWISE, AND AGREES NOT TO BRING ANY ACTIONS OR PROCEEDINGS AT LAW, IN EQUITY, IN TORT OR OTHERWISE, INCLUDING RESCINDING THE AGREEMENT, IN RESPECT OF ANY BREACHES OF REPRESENTATIONS, WARRANTIES OR OTHER PROVISIONS OF THIS AGREEMENT OR IN CONNECTION WITH THE TRANSACTIONS.

10.9 <u>R&W Insurance Policy</u>. Notwithstanding any provision in this Agreement to the contrary, the Buyer Indemnitees acknowledge and agree that: (i) the Equityholder shall not be required to pay the Buyer Indemnitees any amounts with respect to Damages arising under <u>Sections 10.2(a)(i)</u>, 10.2(b)(i) and 10.2(c)(i) for breaches of representations in excess of the Cap, other than for claims for indemnification with respect to Fraud on behalf of the Equityholders or the Companies, and accordingly once the Cap is exhausted the sole recourse and remedy of the Buyer Indemnitees for indemnification with respect to breaches of representations and warranties, other than for claims for indemnification with respect to breaches of representations and warranties, other than for claims for indemnification with respect to breaches of representations and warranties, other than for claims for indemnification with respect to breaches of representations and warranties, other than for claims for indemnification with respect to breaches of representations and warranties, other than for claims for indemnification with respect to breaches of representations and warranties, other than for claims for indemnification with respect to breaches of representations and warranties, other than for claims for indemnification with respect to Fraud on behalf of the Equityholders or the Group Companies, shall be made against and, to the extent of, the R&W Insurance Policy. NOTWITHSTANDING ANY PROVISION IN THIS AGREEMENT TO THE CONTRARY, BUYER PARTIES, ON BEHALF OF THEMSELVES AND THE OTHER BUYER INDEMNITEES, ACKNOWLEDGE AND AGREE THAT THE FOREGOING SHALL CONTINUE TO APPLY EVEN IF (i) THE R&W INSURANCE POLICY IS REVOKED, CANCELLED OR MODIFIED, OR EXPIRES, IN ANY MANNER (AND EVEN IF THE R&W INSURANCE POLICY IS NOT ISSUED); (ii) ANY CLAIM MADE AGAINST THE R&W INSURANCE POLICY IS DENIED BY THE INSURER; OR (iii) ALL AMOUNTS PERMITTED TO BE RECOVERED AGAINST THE R&W INSURANCE POLICY HAVE BEEN RECOVERED.

10.10 <u>No Contribution</u>. Anything to the contrary herein notwithstanding, the Equityholders shall not have any right to seek any indemnification or contribution from or remedy against the Group Companies whether arising prior to or after the applicable Closing Date in respect of any breach of any representation or warranty by the Group Companies or the failure of the Companies to comply with any covenant or agreement to be performed by the Group Companies on or prior to the applicable Closing Date and the Equityholders hereby waive any such claim they may have against the Group Companies with respect thereto whether at law, in equity or otherwise.

ARTICLE 11 Definitions

11.1 Certain Definitions. In addition to the other definitions contained in this Agreement, the following terms will, when used in this Agreement, have the following respective meanings:

"Action" means any governmental, judicial, administrative or adversarial proceeding (public or private), any action, complaint, claim, demand, lawsuit, legal proceeding, whistleblower complaint, litigation, arbitration or mediation, any hearing, any inquiry, any notice of violation, summons, subpoena, audit, investigation or demand or threat of a civil, criminal, administrative, or regulatory nature, whether at law or in equity, conducted or heard by or before any Governmental Authority, or any other dispute, including any adversarial proceeding arising out of this Agreement.

"Actual Member Month Amount" means the total number of Member Months, which service is attributed to or related to PC Buyer and CFCHP, measured as of the last day of each month of each Measurement Period as evidenced by membership eligibility reports (inclusive of subsequent corrections made within six (6) months following the date of such report) received by CFCHP and PC Buyer from the Plans, divided by twelve (12); provided, that the Actual Member Month Amount shall be increased by the subsequent addition of Members to either PC Buyer or CFCHP from other Plans managed by MSO LP Buyer during the first month in which such Members join PC Buyer or CFCHP and are managed by MSO LP Buyer. For the avoidance of doubt, any eligible individual who is enrolled in a Plan that contracts with CFCHP or PC Buyer at any point during a given month shall be considered a Member for such month. A hypothetical example of calculating the Target Member Month Amount and Actual Member Month Amount is attached hereto as <u>Schedule 1.3(a)</u>.

"Affiliate" means, with respect to any Person, any Person who directly or indirectly owns or controls, is owned or controlled by, or is under direct or indirect common ownership or control with such other Person. Without limiting the generality of the foregoing, a Person shall be deemed to "own" another person if it owns, directly or indirectly, more than fifty percent (50%) of the capital stock or other equity interest of such other Person and a Person shall be deemed to "control" another Person if such Person possesses, directly or indirectly, the power to direct or cause the direction of the management and policies of such specified Person, and/or if such other Person is at such time a, direct or indirect, beneficial holder of at least fifty percent (50%) of any class of the equity interests of such specified Person.

"AHMS Adjustment Escrow Amount" means Six Hundred Thousand Dollars (\$600,000).

"AHMS Indemnity Escrow Amount" means Two Hundred Sixteen Thousand Dollars (\$216,000).

"<u>Allocable Portion</u>" means (a) with respect to IPA Equityholder with respect to a breach of a representation, warranty, or covenant by or Damages with respect to CFC IPA, one hundred percent (100%); (b) with respect to the AHMS Equityholders with respect to a breach of a representation, warranty, or covenant by or Damages with respect to a Plan Company, the percentage by which the Second Closing Payment is allocated to the AHMS General Partner and AHMS Limited Partners under <u>Section 1.5</u>; (c) with respect to a AHMS Limited Partner with respect to a breach of a representation, warranty, or Covenant by or Damages with respect to a Plan Company, the percentage by which the Second Closing Payment is allocated to the AHMS General Partner and AHMS Limited Partners under <u>Section 1.5</u>.

-84-

"Antitrust Laws" means the HSR Act, the Sherman Act, as amended, the Clayton Act, the Federal Trade Commission Act and any other applicable federal, state or foreign law, regulation or decree designed to prohibit, restrict or regulate actions for the purpose or effect of monopolization or restraint of trade.

"Antitrust Order" means any judgment, injunction, order (whether temporary, preliminary or permanent) or decree issued under or with respect to any Antitrust Laws.

"<u>Applicable Law</u>" means any federal, state, local, municipal, foreign or other law, statute, legislation, constitution, principle of common law, resolution, ordinance, code, edict, decree, proclamation, treaty, convention, rule, regulation, proposed regulation, listing standard, ruling, directive, pronouncement, requirement, specification, determination, decision, opinion or interpretation that is, or has been issued, enacted, adopted, passed, approved, promulgated, made, implemented or otherwise put into effect by or under the authority of any Governmental Authority or the governing body of any national securities exchange.

"Assumed Contracts" means the Contracts listed on the Schedule 11.1(a).

"Assumed Intellectual Property" means the Company Intellectual Property owned or purported to be owned by CFC IPA.

"AV Escrow Amount" means Fourteen Million Dollars (\$14,000,000).

"Business" means (i) the IPA Business, (ii) the Plan Business, and (iii) any other business activity conducted by the Group Companies during the twelve (12) month period immediately preceding the Closing or currently contemplated to be conducted; provided, that the term "Business" shall not include the ownership or operation of the Excluded Real Property.

-85-

"Business Day" means each day other than a Saturday, Sunday or other day on which commercial banks in California are authorized or required by Applicable Law to

close.

"Buyer Fundamental Representations" means the representations and warranties in Section 5.1 (Organization and Authority of Buyer), Section 5.3 (Investment Intention), Section 5.6 (Title) and Section 5.7 (Brokers) and Section 5.9 (Issuance of the Buyer Parent Shares).

"Buyer Parent Share Consideration" means Twenty Million Dollars (\$20,000,000).

"Buyer Parent Shares" means 631,712 shares of common stock in Buyer Parent with a value equal to the quotient of the (i) Buyer Parent Share Consideration divided by the volume weighted average closing price of one (1) share of common stock of Buyer Parent reported by Nasdaq for the twenty (20) most recent days such shares were traded on the Nasdaq prior to five (5) Business Days to the date hereof (as adjusted as appropriate to reflect any stock splits, stock dividends, combinations, reorganizations, reclassifications or similar events occurring after the date of this Agreement and prior to the Initial Closing).

"<u>Cash-to-Claims Ratio</u>" has the meaning set forth in Title 28, Article 9, Section 1300.75.4(f) of the California Code of Regulations. An example calculation of the Cash-to-Claims Ratio applicable to CFC IPA is set forth in <u>Exhibit B-1</u>.

"Catch-Up Payment" means a payment of \$5,000,000 for each Measurement Period in which the Target Member Month Amount is achieved pursuant to Section 1.3(b).

"CERCLA" means the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended from time to time prior to the applicable Closing Date, and any rules or regulations promulgated thereunder.

"CFCHP TNE" means, with respect to CFCHP only, the "Tangible Net Equity" (as such term is defined in Section 1300.76(c) of Title 28 of the California Code of

Regulations) calculated as of a specific point in time, as applicable, and using the accounting practices, assumptions, policies and methodologies as set forth on, and used in the calculation of CFCHP's related risk-bearing organization filings on the unaudited balance sheet of CFCHP attached to <u>Section 4.5</u> of the Disclosure Schedule, and shall consist of the line items set forth on <u>Exhibit B-2</u> attached hereto, which is consistent with the requirements of the DMHC, determined as of the Second Effective Time. Notwithstanding the foregoing, for purposes of calculating the CFCHP TNE, the CFCHP TNE shall exclude (a) Funded Indebtedness, (b) Company Transaction Expenses, (c) any current or deferred Tax assets or liabilities (including any Tax assets or Tax receivables), (d) the impact of purchase price adjustments related to cash, and (e) deferred revenue at the applicable Closing.

"<u>CFC IPA TNE</u>" means, with respect to CFC IPA only (and not any other Group Company, including CFCHP) the "Tangible Net Equity" (as such term is defined in Section 1300.76(c) of Title 28 of the California Code of Regulations) calculated as of a specific point in time, as applicable, and using the accounting practices, assumptions, policies and methodologies as set forth on, and used in the calculation of CFC IPA's related risk-bearing organization filings on the unaudited balance sheet of CFC IPA attached to <u>Section 4.5</u> of the Disclosure Schedule, and shall consist of the line items set forth on<u>Exhibit B-1</u> attached hereto, which is consistent with the requirements of the DMHC, determined as of the Initial Effective Time. Notwithstanding the foregoing, for purposes of calculating the CFC IPA TNE, the CFC IPA TNE shall exclude (a) Funded Indebtedness, (b) Company Transaction Expenses, (c) intercompany balances, (d) any current or deferred Tax assets or liabilities (including any Tax assets or Tax receivables), (e) the impact of purchase price adjustments related to cash, and (f) deferred revenue at the applicable Closing.

-86-

"<u>Closing</u>" means (i) the Initial Closing, with respect to the Transactions relating to the transfer of the Acquired Assets, and (ii) the Second Closing, with respect to the Transactions relating to the transfer of the of the Purchased Interests.

"<u>Closing Actual Adjustment</u>" means (i) with respect to the Initial Closing, an amount (expressed as a positive or a negative number) equal to the First Calculation of the Initial Closing Payment minus the Initial Closing Payment, and (ii) with respect to the Second Closing, an amount (expressed as a positive or a negative number) equal to the Final Calculation of the Second Closing Payment minus the Second Closing Payment.

"<u>Closing Cash</u>" means the aggregate amount of the following as of the Effective Time (without duplication), but without giving effect to any changes thereto (at the direction of Buyer or any Affiliate thereof, purchase accounting adjustments or other changes) arising from or resulting as a consequence of the transactions contemplated hereby: (i) all cash, commercial paper, certificates of deposit and other bank deposits, treasury bills, short term investments and all other cash equivalents in the accounts of the applicable Group Companies (including security deposits held by third parties), plus (ii) third-party checks deposited or held in the accounts of the applicable Group Companies that have not yet cleared (but only to the extent not counted as a current asset), plus (iii) any and all credit card receivables and credit card deposits in transit of the applicable Group Companies, plus (iv) all security or rent or other similar deposits held for the applicable Group Companies' account, minus (v) issued but uncleared checks, drafts and wire transfers of the applicable Group Companies.

"Closing Date" means (i) the Initial Closing Date, with respect to the Transactions relating to the transfer of the Acquired Assets, and (ii) the Second Closing Date, with respect to the Transactions relating to the transfer of the of the Purchased Interests.

"Closing IPA Cash-to-Claims Amount" means the amount, which could be positive or negative, equal to the Closing Cash of CFC IPA as of the Initial Effective Time minus the Minimum IPA Cash-to-Claim Ratio Amount.

"COBRA" means Part 6 of Subtitle B of Title I of ERISA, Section 4980B of the Code, and any similar Applicable Law.

"Code" means the Internal Revenue Code of 1986, as amended.

"<u>Commercially Reasonable Efforts</u>" means, where applied to carrying out specific tasks and obligations of a Party, expending reasonable, diligent, good faith efforts and resources to accomplish such task or obligation as such party (on its own and/or acting through any of its Affiliates, agents or subcontractors) would normally use to accomplish a similar task or obligation under similar circumstances.

-87-

"Company 401(k) Plan" means any 401(k) plan sponsored or maintained by a Group Company.

"Company Intellectual Property" means all Intellectual Property owned or purported to be owned by any Group Company.

"<u>Company Licensed Intellectual Property</u>" means all Intellectual Property subject to any licenses, sublicenses or other agreements under which any Group Company: (i) is granted rights by others in Intellectual Property (other than commercial off the shelf software that is made available to one of the Group Companies for a one-time or annual license fee of \$50,000 or less) or (ii) has granted material rights to others in Company Intellectual Property.

"<u>Company Registered Intellectual Property</u>" means all Company Intellectual Property actively registered, filed, or issued under the authority of any domain name registrar, any state, federal, local or international government or other Governmental Authority, including all active patents, registered copyrights, registered trademarks, and domain names, and all pending applications for any of the foregoing.

"<u>Company Transaction Expenses</u>" means all: (a) Transaction Expenses of the applicable Group Companies, any of the applicable Equityholders or any of their respective Affiliates; (b) any change of control, severance, sale, retention, success or similar payments, bonuses, deferred compensation or promises to payments made by any of the applicable Group Company to any Person which are triggered, whether in part or whole, by the Transactions; (c) 50% of the filing fees paid by the Buyer Parties under the HSR Act and other applicable Antitrust Laws; and (d) 50% of the costs and expenses of the Tail Policy.

"<u>Contracts</u>" means any contract, agreement, understanding, arrangement, warranty, deed, power of attorney, purchase order, work order, commitment, assignment, certificate, covenant, assurance, undertaking, option, lease, license, insurance policy, sale and purchase order, commitment and other instruments of any kind, whether written or oral, to which a Person is a party, is otherwise bound or has any current or future obligation or Liability.

"Damages" shall include any loss, damage, Liability, claim, demand, settlement, Judgment, award, fine, penalty, Tax, fee (including any legal fee, expert fee, accounting fee or advisory fee), charge, cost or expense (including any cost of investigation, penalty, fee and disbursement of counsel); provided, that "Damages" shall not include (a) any amounts attributable to exemplary or punitive damages (unless payable to third parties), (b) loss of revenue or diminution of value (except, in each, to the extent case constituting direct, market-measured or general damages and/or to the extent constituting a part of any Damages payable to third parties), or (c) civil or criminal fines or penalties except to the extent (i) insurable under the applicable law; and, (ii) awarded or assessed in connection with a third-party demand pursuant to (A) a final settlement consented to in writing by the Indemnifying Person (not to be unreasonably withheld, conditioned or delayed) or (B) a final and non-appealable (x) order of a governmental or regulatory agency, (y) judgment of a court of competent jurisdiction or (z) award of an arbitrator, arbitration panel or similar adjudicative body.

"Disclosure Schedule" means the disclosure schedule prepared by the Equityholders and Group Companies and delivered to the Buyer Parties concurrently with the execution and delivery of this Agreement and as may be supplemented after the date of this Agreement in accordance with the terms and conditions set forth herein.

"Effective Time" means (i) the Initial Effective Time, with respect to the Transactions relating to the transfer of the Acquired Assets, and (ii) the Second Effective Time, with respect to the Transactions relating to the transfer of the of the Purchased Interests.

"Entity" means any firm, corporation, partnership, joint venture, limited liability company, Governmental Authority, unincorporated organization, trust, association or other entity.

"Enumerated Assets" means all of the following assets of CFC IPA, in each case, other than the Excluded Assets: (a) the Assumed Contracts; (b) all Assumed Intellectual Property and all income, royalties, damages and payments accrued, due or payable as of the Initial Closing Date or thereafter (including damages and payments for past, present or future infringement, misappropriation or dilution thereof or other conflict therewith, the right to sue and recover for past, present or future infringement, misappropriation or dilution thereof or other conflict therewith, and any and all corresponding rights that, now or hereafter may be secured throughout the world), and all trademarks, service marks, trade dress, trade names, logos, dba's, fictitious names, and corporate names and registrations and applications (including intents to use) for registration thereof, together with all of the goodwill associated therewith used by CFC IPA; (c) all claims, refunds, credits, causes of action, rights of recovery and rights of setoff; (d) all documents, books, ledgers, files and records (excluding Personnel Records); (e) all advertising, marketing and promotional materials and the right to receive mail and other communications of the IPA Business; (f) all Permits; (g) all goodwill (including goodwill related to the Assumed Intellectual Property); (h) all proceeds under warranty, indemnity and similar rights against third parties, rights of recovery relating to any Insurance Claims (except those relating to the Excluded Assets and Excluded Liabilities); (i) all equipment used in the IPA Business that is owed or leased by CFC IPA; (j) all tangible assets used in the IPA Business, including, without limitation, all inventory, furniture, furnishings, fixtures and supplies, (k) the right to receive all payments in respect of accounts and notes receivable outstanding as of the Initial Closing; (l) all prepayments and prepaid expenses, advance payments, employee advances, surety accounts, bonds and deposits and similar assets; (m)

"Environmental Requirements" means all Applicable Laws and any judicial or administrative interpretation thereof, including all judicial and administrative orders and determinations, and all contractual obligations, in each case concerning public health and pollution or protection of the environment, including all those relating to the presence, use, production, generation, handling, transport, treatment, storage, disposal, distribution, labeling, testing, processing, discharge, Release, threatened Release, control or cleanup of any Hazardous Substances or hazardous or otherwise regulated materials, substances or wastes, chemical substances or mixtures, pesticides, pollutants, contaminants, toxic chemicals, petroleum products or byproducts, asbestos, polychlorinated biphenyls, noise or radiation, as the foregoing are enacted and in effect prior to or on the Closing Date.

-89-

"Equity Interests" means, with respect to any Person, (a) any capital stock, partnership or membership interest, joint venture interest, unit of participation or other similar interest (however designated) in such Person and (b) any option, warrant, call, right (including purchase rights, conversion rights, exchange rights, preemptive rights, co-sale rights, rights of first refusal and similar rights) or other Contract which would entitle any other Person to acquire any such interest in such Person.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" means any Person that at any relevant time is or was considered a single employer with any Group Company under Section 414 of the Code.

"Escrow Agent" means JPMorgan Chase Bank, NA.

"Escrow Amount" means the CFC Adjustment Escrow Amount, the AHMS Adjustment Escrow Amount, the CFC Indemnity Escrow Amount the AHMS Indemnity Escrow Amount, and the AV Escrow Amount, as applicable.

"Estimated Closing CFCHP TNE Deficiency" means the amount, if any, by which the Estimated Closing CFCHP TNE is less than the Target CFCHP TNE.

"Estimated Closing CFCHP TNE Surplus" means the amount, if any, by which the Estimated Closing CFCHP TNE exceeds the Target CFCHP TNE.

"Estimated Closing IPA TNE Deficiency" means the amount, if any, by which the Estimated Closing IPA TNE is less than the Target IPA TNE.

"Estimated Closing IPA TNE Surplus" means the amount, if any, by which the Estimated Closing IPA TNE exceeds the Target IPA TNE.

"Exchange Act" means Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

"Excluded Assets" means, collectively, (a) all capital stock or other equity interests of CFC IPA; (b) all organizational documents, qualifications to conduct business as a foreign corporation or entity, arrangements with registered agents relating to foreign qualifications, and taxpayer and other identification numbers of CFC IPA; (c) all Employee Benefit Plans sponsored or maintained by CFC IPA, as indicated in <u>Section 4.19(a)</u> of the Disclosure Schedule; (d) all Contracts of CFC IPA other than the Assumed Contracts; (e) all real property including the Leased Real Property; (f) any Permits held by CFC IPA that are not transferable to PC Buyer under Applicable Law; (g) any insurance policies of Sellers including any refunds paid or payable in connection with the cancellation or discontinuance of any such insurance policies, and any claims for refund of Tax, Tax assets and any other such items relating to the payment of Taxes (including any such assets or items relating to the Acquired Assets, the IPA Business, or the Assumed Liabilities) for taxable periods (or portions thereof) ending on or prior to the Initial Closing Date; (i) any Tax Returns of CFC IPA; (j) all Actions, demands, rights and privileges against third parties that relate to any of the Excluded Liabilities (including any claims for refund of Tax, Tax assets and any other such items relating to the payment of Taxes (or portions thereof) ending on or prior to the Initial Closing Date; (i) any Tax refunds, credits, causes of action, rights of recovery and rights of set off with respect to the Excluded Assets and the Excluded Liabilities (including any claims for refund of Tax, Tax assets and any other such items relating to the payment of Taxes for taxable periods (or portions thereof) ending on or prior to the Initial Closing Date; (i) all insurance policies or Tail Policies, (m) cash and cash equivalents (subject to Tangible Net Equity and Cash-to-Claims Ratio requirements under this Agreement); (n) all rights of CFC IPA or its Affiliates under

-90-

"Final Initial Closing Adjustment" means an amount (expressed as a positive or a negative number) equal to the Final Calculation of the Initial Closing Payment minus the First Calculation of the Initial Closing Payment.

"First Measurement Period" means the period commencing on the first day of the month immediately following the Second Closing Date and ending on the one (1) year anniversary thereof.

"Fraud" means actual and intentional common law fraud under Applicable Laws of the State of Delaware solely with respect to a representation or warranty contained in this Agreement. For the avoidance of doubt, "Fraud" shall not include equitable fraud, constructive fraud, negligent misrepresentation or omission, or any form of fraud premised on recklessness or negligence.

"Funded Indebtedness" means, without duplication, any Indebtedness of the Group Companies. Notwithstanding the foregoing, in no event will any Tax liabilities or any item included in Initial Closing Company Transaction Expenses, Second Closing Company Transaction Expenses be included in Funded Indebtedness.

"GAAP" means generally accepted accounting principles in the United States, as in effect as of the date of this Agreement.

"Governing Documents" means, with respect to any particular entity: (a) if a corporation, the articles or certificate of incorporation and the bylaws; (b) if a general partnership, the partnership agreement and any statement of partnership; (c) if a limited partnership, the limited partnership agreement and the articles or certificate of limited partnership; (d) if a limited liability company, the articles of organization or certificate of formation and operating agreement, regulations, limited liability company agreement, or company agreement; I if another type of Person, any other charter or similar document adopted or filed in connection with the creation, formation or organization of the Person; (e) all equityholders' agreements, voting agreements, joint venture agreements, registration rights agreements or other agreements or documents relating to the organization, management or operation of any Person or relating to the rights, duties and obligations of the equityholders of any Person; and (f) any amendment or supplement to any of the foregoing.

-91-

"Governmental Authority" means any federal, state, local or foreign government or political subdivision thereof, or any agency or instrumentality of such government or political subdivision, or any self-regulated organization or other non-governmental regulatory authority or quasi-governmental authority (to the extent that the rules, regulations or orders of such organization or authority have the force of Applicable Law), or any arbitrator, court or tribunal of competent jurisdiction.

"Governmental Order" means any order, writ, judgment, injunction, decree, stipulation, determination or award entered by or with any Governmental Authority.

"Hazardous Substance" means (a) substances that are defined or listed, in, or otherwise regulated pursuant to any applicable Laws as "hazardous substances," "hazardous materials," "hazardous wastes," "toxic substances," "pollutants," "contaminants," or any other similar term intended to define, list, or classify a substance by reason of such substance's ignitability, corrosivity, reactivity, carcinogenicity, reproductive toxicity, "EP toxicity" or adverse effect on human health or the environment, (b) oil, petroleum, natural gas, natural gas liquids, synthetic gas, drilling fluids and other wastes associated with the exploration, development or production of crude oil, natural gas or geothermal resources, (c) any explosives or any radioactive materials, (d) asbestos in any form, (e) polychlorinated biphenyls, (f) toxic mold, mycotoxins or microbial matter (naturally occurring or otherwise) and (g) infectious waste.

"Healthcare Laws" means, to the extent applicable, (a) any state and federal civil or criminal healthcare Laws applicable to the Business, local fraud and abuse laws, including Medicaid, Medicare, all applicable anti-kickback, false claims, self-referral, and insurance fraud laws, such as (i) the federal Anti-Kickback Statute (42 U.S.C. § 1320a-7b(b)), (ii) the Stark Law (42 U.S.C. § 1395nn), (iii) the civil False Claims Act (31 U.S.C. § 3729 et seq.), the administrative False Claims Law and (iv) criminal laws relating to health care fraud and abuse, including but not limited to 18 U.S.C. Sections 286, 287, 1347 and 1349 and the health care fraud criminal provisions under HIPAA and the regulations promulgated pursuant to such statutes; (b) laws, rules and regulations governing the health insurance program for the aged and disabled established by Title XVIII of the Social Security Act (42 U.S.C. § 1320a-7b(a)), the Civil Monetary Penalties Law (42 U.S.C. § 1320a-7a) and any applicable statutes succeeding thereto (c) any applicable laws, rules and regulations governing the healthcare assistance program established by Title XIX of the Social Security Act (42 U.S.C. § 1320a-7a; 42 U.S.C. § 1320c-8(a)), HIPAA, the exclusion laws (42 U.S.C. § 1320a-7), or § 1396 et seq.) and any applicable statutes succeeding thereto; (d) the Medicare Prescription Drug, Improvement, and Modernization Act of 2003 (Pub. L. No. 108-173) and the regulations promulgated pursuant thereto; (e) HIPAA and applicable state health information privacy and security laws; (f) laws and regulations relating to the billing or submission of claims, collection of accounts receivable, underwriting the cost of, or provision of management or administrative services in connection with, any and all of the foregoing; (g) applicable laws and regulations regarding professional licensure, credentialing, insurance entity licensure, scope of practice, fee-splitting and laws relating to the corporate practice of a licensed profession and the maintenance of a provider network, including any willing or authorized provider laws; (h) applicable laws and regulations (including without limitation the Knox Keene Act and its implementing regulations) regarding the licensure, qualification, or authority to conduct third party administrator services or similar services or to take on reimbursement or expense risk, under agreements with health plans, Medicare Advantage or managed Medicaid health plans, other insurers, or self-insured plans; and (i) any comparable state or federal laws in respect of or similar to any of the foregoing, each of (a) through (i) as may be amended from time to time and the regulations promulgated pursuant to such Applicable Laws, and comparable state laws.

-92-

"HHS" means the U.S. Department of Health and Human Services.

"HSR Act" means the Hart Scott Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder.

"Indebtedness" means, without duplication: (a) with respect to any Person, including a Group Company, all obligations (including all obligations in respect of principal, accrued interest, penalties, fees, costs and premiums, whether or not contingent and including any prepayment penalties, premiums and any other fees, expense, breakages costs, indemnities, "make-whole amounts", expense reimbursement obligations and other monetary obligations in connection therewith) of such Person for any (i) indebtedness for borrowed money (including overdraft facilities), whether accrued, absolute, contingent, unliquidated or otherwise, known or unknown, whether due or to become due; (ii) deferred price of property, goods or services (other than trade payables incurred in the Ordinary Course of Business that are less than ninety (90) days past due); (iii) overpayments of capitation revenue received in March, April, and July 2023 and due to L.A. Care in the approximate amount of \$1,666,000; (iv) reimbursement and other obligations for surety bonds and letters of credit and similar transactions to the extent drawn; (v) obligations evidenced by notes, bonds, debentures or similar Contracts; (vi) breakage fees with respect to Contracts relating to interest rate protection, swap agreements and collar agreements; and (vii) guaranties of any of the foregoing; and (b) with respect to a Group Company, all obligations (including all obligations in respect of principal, accrued interest, penalties, fees, costs and premiums, whether or not contingent and including any prepayment penalties, premiums and any other fees, expense, breakages costs, indemnities, "make-whole amounts", expense reimbursement obligations and other monetary obligations in connection therewith) of such Group Company for any (i) payable or other indebtedness owing to the Equityholders, any of their respective Affiliates or any Related Party of the Equityholders or any of their respective Affiliates, whether accrued, absolute, contingent, unliquidated or otherwise, kn

"Initial Closing Funded Indebtedness" means without duplication, the obligations of CFC IPA outstanding and unpaid immediately prior to the Initial Closing (excluding any undrawn amounts under credit lines or revolving (or similar) credit facilities, and excluding any indemnification or contingent obligations then owning) under any Indebtedness.

"Initial Closing Payment" means (A) an amount equal to the CFC IPA Cash Purchase Price, (B) plus the Deposit Escrow Amount minus \$21,000, (C) plus or minus, as applicable, the Estimated Closing IPA Cash-to-Claims Amount, (D) plus the Estimated Closing IPA TNE Surplus or minus the Estimated Closing IPA TNE Deficiency, as applicable, (E) minus the CFC Indemnity Escrow Amount, (F) minus the CFC Adjustment Escrow Amount, (G) minus the Estimated Initial Closing Funded Indebtedness, and (H) minus the Initial Closing Estimated Transaction Expenses as of the Initial Closing.

"Intellectual Property" means all: (a) patent registrations, patent applications, patent disclosures, improvements and inventions, (b) trademarks, service marks, trade dress, trade names, logos, dba's, fictitious names, and corporate names and registrations and applications (including intents to use) for registration thereof, together with all of the goodwill associated therewith, (c) copyrights (registered or unregistered) and copyrightable works and registrations and applications for registration thereof, (d) mask works and registrations and applications for registration thereof, (e) computer software, data, data bases and documentation thereof, (f) trade secrets, trade compositions, technologies, inventions (whether patentable or unpatentable and whether or not reduced to practice), know-how, manufacturing and production processes and techniques, research and development information, drawings, specifications, designs, plans, manuals, proposals, technical data, copyrightable works, financial, marketing and business plans and customer and supplier lists and other proprietary information, (g) domain names, (h) Software, (i) other intellectual property rights, (j) license agreements or other rights pertaining to the foregoing, (k) rights of enforcement of any of the foregoing, and (l) copies and tangible embodiments therefore in whatever form or medium.

"IT Systems" means computers, software, hardware, servers, websites, workstations, routers, hubs, switches, data communications lines, firmware, networks, and all other information technology equipment owned or used by any of the Group Companies.

"Judgment" means any: (a) order, judgment, injunction, edict, decree, ruling, pronouncement, determination, decision, opinion, verdict, sentence, subpoena, writ or award issued, made, entered, rendered or otherwise put into effect by or under the authority of any court, administrative agency or other Governmental Authority or any arbitrator or arbitration panel; or (b) Contract with any Governmental Authority entered into in connection with any Action.

"Knowledge of the applicable Company" or any other similar knowledge qualification, means the actual knowledge of any of the following persons and the knowledge that each of such persons should have obtained after reasonable inquiry: Accie Mitchell, M.D., Ron Brandt and Avanthi Wadu.

-94	
-----	--

"Liability" or "Liabilities" means, with respect to any Person, any liability or obligation of such Person of any kind, character or description, whether known or unknown, absolute or contingent, accrued or unaccrued, disputed or undisputed, liquidated or unliquidated, secured or unsecured, joint or several, vested or unvested, executory, determined, determinable or otherwise and whether or not the same is required to be accrued on the financial statements of such Person.

"Lien" means any charge, mortgage, pledge, security interest, lien, or Lien, other than those that customarily arise under securities Applicable Laws in private transactions.

"Management Services Agreement" means that certain Management Services Agreement, to be entered into by and between AHMS, PC Buyer and MSO LP Buyer, in the form mutually agreed by the Parties on the date hereof.

"Material Adverse Effect" means any change, event or effect that, when taken individually or together with all other related changes, events or effects, has had or would reasonably be expected to have a material adverse effect on (x) the Business, properties, assets, results of operations or condition, financially or otherwise, taken as a whole, of the Group Companies, or (y) the ability of the Group Companies or Equityholders to consummate the Transactions; provided, however, that a "Material Adverse Effect" shall not include any change or effect: (a) in general economic or business conditions; (b) financial, banking or securities markets of the United States in general (including any disruption thereof and any decline in the price of any security or any market index); (c) in national or international political or social conditions, including the engagement by the United States in hostilities, whether or not pursuant to the declaration of a national emergency or war, or the occurrence of any military or terrorist attack upon the United States or any of its territories, possessions or diplomatic or consular offices or upon any military installation, equipment or personnel of the United States; (d) affecting generally the industries or markets in which the Group Companies operate; (e) related to the announcement taken at the written request, or with the express written permission, of the Buyer Parties or their Representatives or consummation of any of the Transactions; (f) resulting from any act taken by the Equityholders or the Group Companies pursuant to or in accordance with this Agreement taken at the written request, or with the express written permission, of the Buyer Parties or their Representatives; (g) resulting from changes in Applicable Law; (h) resulting from changes in reimbursement rates, including but not limited to changes in any reimbursement policies or fee schedules by Governmental Authority; (i) adverse effects related to the COVID-19 pandemic; or (j) items which are cured prior to the earlier of the applicable Closing Date or termination of this Agreement in accordance with Article 9, so long as, in the case of each of the foregoing clauses (a) through (j), any such effect, change, event or condition does not affect the Group Companies in a materially disproportionate manner relative to other participants in similar industries in which the Group Companies operate. A failure by the Group Companies to meet any projections, estimates or budgets for any period prior to, on or after the date of this Agreement shall not in itself constitute a Material Adverse Effect; provided, that this sentence shall not prevent a determination that any change, event or effect underlying such failure to meet projections or forecasts has resulted in a Material Adverse Effect (to the extent such change or effect is not otherwise excluded from this definition of Material Adverse Effect).

"Measurement Period" means, with respect to the CFC IPA Deferred Payment or any Catch-Up Payment, the First Measurement Period, the Second Measurement Period and the Third Measurement Period.

"Member" means eligible individuals who are enrolled in a Plan that contracts with CFC IPA (if before the Initial Closing) or PC Buyer (if after the Initial Closing), with respect to whom CFCHP (whether before or after the Second Closing) or CFC IPA (if before the Initial Closing) or PC Buyer (if after the Initial Closing) assumes financial risk or is entitled to receive payment (whether on a capitated, percentage of premium, or other basis) from, as evidenced by membership eligibility files received by CFCHP and PC Buyer from the Plans. For the avoidance of doubt, any eligible individual who is enrolled in a Plan that contracts with CFCHP or PC Buyer at any point during a given month shall be considered a Member for such month.

"<u>Member Month</u>" means the quotient of the (i) total number of Members per month during a Measurement Period, divided by (ii) twelve (12), as evidenced by membership eligibility files received by CFCHP and PC Buyer from the Plans. By way of example only, a Member who is a Member for each month of a twelve (12) month Measurement Period will be counted twelve (12) times; a Member who is a Member for only six (6) of the twelve (12) months will be counted six (6) times.

"Minimum IPA Cash-to-Claim Ratio Amount" means, with respect to CFC IPA only (and not any other Group Company, including CFCHP), 0.75 cash-to-claims ratio as defined in California Code Regulations Title 28, 1300.75.4(f), but subject to the interpretations set forth in Exhibit B-1.

"Neutral Accounting Firm" means an independent accounting firm of nationally recognized standing that is not at the time it is to be engaged hereunder rendering services to any Party, or any Affiliate of any Party, and has not done so within the two (2) year period prior thereto.

"Notice of Material Modification" means a health care service plan regulatory filing, including any amendments thereto filed with DMHC, required to be filed by AHMS pursuant to the Knox-Keene Act to obtain approval from the DMHC for an indirect change in ownership or control relating to CFCHP resulting from the sale of the AHMS Interests.

"<u>Ordinary Course of Business</u>" means an action taken by a Person if that action (a) is consistent in nature, scope and magnitude with the past practices of such Person and is taken in the ordinary course of the normal, day-to-day operations of such Person; (b) does not require special or separate authorization by the board of directors or shareholders of such Person (or by any Person or group of Persons exercising similar authority) and does not require any other separate or special authorization of any nature; and (c) is similar in nature, scope and magnitude to actions customarily taken, without any separate or special authorization, in the ordinary course of the normal, day-to-day operations of other Persons that are in the same line of business as such Person.

"Permits" means all approvals, permits, consents, certificates and licenses, if any, of all Governmental Authorities required for the operation of the Business.

-96-

"Permitted Liens" means (a) landlords', lessors', mechanics', materialmen's, warehousemen's, carriers', workers', manufacturer's or repairmen's Liens or other similar Liens arising or incurred in the Ordinary Course of Business; (b) Liens for Taxes, assessments and other governmental charges not yet due and payable or due but not delinquent or being contested in good faith by appropriate legal proceedings; (c) with respect to real property, (i) easements, quasi-easements, licenses, covenants, rights-of-way, rights of re-entry or other similar restrictions, including any other agreements, conditions or restrictions that would be shown by a current title report or other similar report or listing, which do not materially impair the occupancy or use of the real property for the purposes for which it is currently used in connection with the Companies' businesses, (ii) any conditions that may be shown by a current survey or physical inspection, which do not materially impair the occupancy or use of the applicable real property (except for any violations that would not materially and occupied) and (iv) any and all service and/or maintenance Contract or other Contract affecting any real property with any Person which do not impair in any material respect the current use or occupancy of its Representatives) or made available in the Electronic Data Room; (e) pledges to secure the payment of workers' compensation, unemployment insurance, social security benefits or obligations arising under similar Applicable Laws, or to secure the performance of public or statutory obligations, surety or appeal bonds, and other obligations of a like nature; (f) Liens incurred in the Ordinary Course of Business that would not materially impair the value of the assets of the Companies; and (g) the Liens set forth on <u>Schedule 11.1(c)</u>.

"Person" means any individual or Entity.

"<u>Personnel Records</u>" means the following current employment and current personnel information with respect to each Continuing Employee, in each case, to the extent permitted by applicable law: salary, wage grade, job function, variable compensation targets, performance documentation, and business and personal mailing addresses and telephone numbers, including as applicable, any employment-related agreements, Family and Medical Leave Act (or similar) records, Forms I-9 (Employment Eligibility Verification), employee development plans related to such Continuing Employee, and Continuing Employee medical restrictions, exposure and occupational health records.

"Post-Closing Tax Period" means (i) with respect to CFC IPA, any taxable period beginning after the Initial Closing Date and, with respect to any taxable period beginning on or before and ending after the Initial Closing Date, the portion of such taxable period beginning after the Initial Closing Date, and (ii) with respect to the Plan Companies, any taxable period beginning after the Second Closing Date and, with respect to any taxable period beginning on or before and ending after the Second Closing Date and, with respect to any taxable period beginning on or before and ending after the Second Closing Date and, with respect to any taxable period beginning on or before and ending after the Second Closing Date, the portion of such taxable period beginning after the Second Closing Date.

"Pre-Closing Tax Period" means (i) with respect to CFC IPA, any taxable period ending on or before the Initial Closing Date and, with respect to any taxable period beginning on or before and ending after the Initial Closing Date, the portion of such taxable period ending on and including the Initial Closing Date, and (ii) with respect to the Plan Companies, any taxable period ending on or before the Second Closing Date and, with respect to any taxable period beginning on or before and ending after the Second Closing Date and, with respect to any taxable period beginning on or before and ending after the Second Closing Date and, with respect to any taxable period beginning on or before and ending after the Second Closing Date.

-97-

"Privacy Laws" means all Applicable Laws concerning the access, collection, use, processing, analysis, retention, storage, protection, transfer, disclosure and/or disposal of individually identifiable information including, but not limited to state consumer protection laws, state breach notification laws, and state social security number protection laws. Such Privacy Laws include but are not limited to, the California Consumer Privacy Act of 2018, the California Confidentiality of Medical Information Act (CMIA), the Federal Trade Commission Act, the federal Privacy Act of 1974, the Telephone Consumer Protection Act, the CAN-SPAM Act, the Fair Credit Reporting Act and its state law equivalents, the California Online Privacy Protection Act, the California Consumer Privacy Act, and all other U.S. state laws concerning privacy, data protection, and/or data security, each as amended from time to time, the Payment Card Industry Data Security Standard and all other requirements of the payment card brands and any state or federal law or regulation binding upon the Group Companies the purpose of which is to protect the privacy and/or security of individually identifiable information.

"Pro Rata Portion" means, with respect to each AMHS Equityholder, a fraction, the numerator of which is the aggregate portion of the Purchase Price received by such AMHS Equityholder, and the denominator of which is the aggregate Purchase Price received by all such AHMS Equityholders, in each case, in connection with the sale of the AHMS Interests.

"Protected Business" means (i) (A) the IPA Business and (B) any other business activity conducted by CFC IPA during the twelve (12) month period immediately preceding the Initial Closing or currently contemplated to be conducted if and only if the Initial Closing occurs, and (ii) (A) the Plan Business and (B) any other business activity conducted by the Plan Companies during the twelve (12) month period immediately preceding the Second Closing or currently contemplated to be conducted if and only if the Second Closing occurs; provided, that the term "Protected Business" shall not include the ownership or operation of the Excluded Real Property: provided, further that the term "Protected Business" shall include, if and only if the applicable Closing has occurred, all offerings and products of the applicable Business as of the applicable Closing Date, including but not limited to any independent practice association, arranging for the provision of healthcare services to subscribers or enrollees in health care service plans, or managed care services business.

"Related Party" means, with respect to any Person, any shareholder, member, manager, partner, employee, officer, director of, or consultant to, such Person or any individual related by blood, marriage or adoption to any such Person or any Entity in which any such Person owns any beneficial interest.

"Release" shall have the meaning set forth in CERCLA.

-98-

"Representative" means, with respect to any Person, any and all directors, managers, members, partners, employees, attorneys, accountants, financial advisors, officers and independent contractors, and other agents of such Person.

"<u>Review Period</u>" means a period of thirty (30) calendar days after the date that the Equityholder Representative receives the Initial Closing Statement, the Second Closing Statement or the 12-Month Closing Statement, as the case may be; <u>provided</u> that such period of time shall be tolled and extended to account for any delay or failure by Buyers of their obligations under <u>Section 2.2(f)</u>.

"Second Closing Funded Indebtedness" means without duplication, the obligations of the Plan Companies outstanding and unpaid immediately prior to the Second Closing (excluding any undrawn amounts under credit lines or revolving (or similar) credit facilities, and excluding any indemnification or contingent obligations then owning) under any Indebtedness.

"Second Closing Payment" means (A) an amount equal to the Second Closing Purchase Price, (B) plus the Closing Cash for the Plan Companies, (C) plus the Estimated Closing CFCHP TNE Surplus or minus the Estimated Closing CFCHP TNE Deficiency, as applicable, (D) minus the AHMS Indemnity Escrow Amount, (E) minus the AHMS Adjustment Escrow Amount, (F) minus the AV Escrow Amount, (G) minus the Estimated Second Closing Funded Indebtedness, (H) minus the Second Closing Estimated Transaction Expenses as of the Second Closing, and (I) minus the Equityholder Representative Fund.

"Second Measurement Period" means the period commencing on the date following the expiration of the First Measurement Period and ending on the one year anniversary thereof.

"Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

"Software" means any and all (a) computer programs and other software, including software implementations of algorithms, models, and methodologies, whether in source code, object code or other form, including libraries, subroutines and other components thereof; (b) computerized databases and other computerized compilations and collections of data or information, including all data and information included in such databases, compilations or collections (whether machine readable or otherwise) and rights therein; (c) screens, user interfaces, command structures, report formats, templates, menus, buttons and icons; and (d) all documentation, including development, diagnostic, support, user and training documentation related to any of the foregoing.

"Straddle Period" means (i) with respect to CFC IPA, any taxable period that begins on or before and ends after the Initial Closing Date, and (ii) with respect to the Plan Companies, any taxable period that begins on or before and ends after the Second Closing Date.

"Subsidiary" means with respect to any Person, any Entity of which fifty percent (50%) or more of the outstanding voting securities or other voting Equity Interests are owned, directly or indirectly, by the referenced Person.

-99-

"<u>Tangible Net Equity</u>" has the meaning set forth in Title 28, Article 9, Section 1300.76(c) of the California Code of Regulations. An example calculation of Tangible Net Equity for CFC IPA is set forth in <u>Exhibit B-1</u> and for CFCHP is set forth in <u>Exhibit B-2</u>.

"Target CFCHP TNE" means, with respect to CFCHP only, two hundred percent (200%) of the minimum Tangible Net Equity that is required to be maintained under the Knox-Keene Act as of immediately prior to the Second Closing (but expressly excluding any additional amounts required by the DMHC in connection with the transactions contemplated hereby). Additionally, for purposes of clarity, Target CFCHP TNE shall be calculated as presented in Exhibit B-2 attached hereto (i) based on annualized revenues and expenses incurred between September 1, 2023 and the Second Closing Date; to the extent the Second Closing Date is after August 31, 2024, it will be calculated based on the most recent trailing twelve (12) months at the Second Closing Date, and (ii) so as to avoid "double-counting" of assets or amounts otherwise taken into account in determining the Purchase Price and/or any adjustments thereto.

"Target IPA TNE" means, with respect to CFC IPA only, 100% of the Tangible Net Equity that is required to be maintained under the Knox-Keene Act as of immediately prior to the Initial Closing. Additionally, for purposes of clarity, Target IPA TNE shall be calculated as presented in Exhibit B-1 attached hereto (i) based on trailing twelve (12) month revenues and expenses at the Initial Closing Date and (ii) so as to avoid "double-counting" of assets or amounts otherwise taken into account in determining the Purchase Price and/or any adjustments thereto.

"Target Member Month Amount" means eighty-seven and fifty hundredths percent (87.5%) of the Member Months as of the last day of the first month of the Initial Closing as evidenced by membership eligibility reports received by CFCHP and PC Buyer from the Plans; <u>provided</u>, that the Target Member Month Amount shall be increased by the subsequent addition of Members to either PC Buyer or CFCHP from other Plans managed by MSO LP Buyer during the first month in which such Members join PC Buyer or CFCHP and are managed by MSO LP Buyer. For the avoidance of doubt, any eligible individual who is enrolled in a Plan that contracts with CFCHP or PC Buyer at any point during a given month shall be considered a Member for such month. By way of example only, if the Initial Closing is January 15, 2023, and there are 100 Members as of January 31, 2023, and on April 15, 2023, 100 Members are added, then (i) for the First Measurement Period, the Target Member Month Amount would be 87.5% of 175 Member Months, and (ii) for the Second Measurement Period, the Target Member Month Amount and Actual Member Month Amount is attached hereto as <u>Schedule 1.3(a)</u>.

"<u>Tax</u>" or "<u>Taxes</u>" (and with correlative meaning, "<u>Taxable</u>") means any United States federal, state or local, or non-United States, income, gross receipts, franchise, estimated, alternative minimum, add-on minimum, sales, use, transfer, registration, value added, excise, natural resources, severance, stamp, withholding, occupation, premium, windfall profit, environmental, customs, duties, real property, personal property, capital stock, net worth, intangibles, social security, unemployment, disability, payroll, license, employee or other tax or similar levy, of any kind whatsoever, including any interest, penalties, fines or additions to tax in respect of the foregoing.

"Tax Authority" means any governmental entity or any subdivision, agency, commission or authority thereof having jurisdiction over the assessment, determination, collection or imposition of any Tax.

"Tax Return" means any return, declaration, report, claim for refund, information return or other document (including any related or supporting estimates, elections, schedules, statements or information), including any amendment thereof, filed or required to be filed in connection with the determination, assessment or collection of any Tax or the administration of any laws, regulations or administrative requirements relating to any Tax.

"Third Measurement Period" means the period commencing on the date following the expiration of the Second Measurement Period and ending on the one year anniversary thereof.

"Transaction Documents" means, with respect to a Party, all agreements, certificates and other instruments to be delivered by such Party pursuant to this Agreement and any documents, certificates or instruments executed or delivered in connection with the transactions contemplated hereby, including without limitation, the Bill of Sale and the Escrow Agreement.

"Transaction Expenses" means, without duplication, with respect to any Person, all fees, costs and expenses (including all legal fees and expenses, all fees and expenses payable to any investment banker, broker or finder, and all fees and expenses of any audit firm or accountants) that have been incurred in connection with the Transactions on behalf of or for the benefit of such Person.

"Transactions" means the purchase and sale of the Purchased Interests and Acquired Assets and the other transactions contemplated hereby.

"Treasury Regulations" means any proposed, temporary or other applicable Treasury Regulations promulgated under the Code.

The following terms are not defined above but are defined in the sections of this Agreement indicated below:

Term Section 12-Month Closing Statement Accounting Expert Accounting Referee Acquisition Engagement Additional Restriction Period Section 6.22 Agreement Preamble AHMS Preamble AHMS Equityholders Preamble AHMS Escrow Amount AHMS General Partner Preamble AHMS GP Interests Recitals AHMS Interests Recitals AHMS Limited Partner Preamble

Section 2.2(d) Section 2.2(g) Section 7.1(c) Section 12.17(a) Section 1.4(b)(iii)

-101-

Term	Section
AHMS LP Interests	Recitals
AHMS Representative Fund	Section 1.4(b)(ii)
AMFIT	Section 1.1
Acquired Assets	Section 1.1(a)
AV Escrow	Section $1.4(b)(v)$
Bill of Sale	Section 8.2(d)(viii)
Business	Recitals
Buyer	Preamble
Buyer Indemnitees	Section 10.2(b)
Buyer Parent	Preamble
Buyer Parent Shares Transferees	Section 1.1(a)
Buyer Parties	Preamble
Buyer Party	Preamble
California CPOM	Recitals
Cap	Section 10.4(a)
Causes of Action	Section 12.18(a)
CFC Adjustment Escrow Amount	Section 1.4(a)(iii)
CFC Indemnity Escrow Amount	Section 1.4(a)(iii)
CFC IPA	Preamble
CFC IPA Cash Purchase Price	Section 1.1(b)
CFC IPA Insurance Policy	Section 6.24
CFC IPA Shares	Recitals
CFC LOC Obligations	Section 6.26
CFC Management Services	Recitals
CFCH	Recitals
CFCHP	Recitals
Closing IPA TNE	Section 2.2(a)
Closing Payments Schedule	Section 1.5
Companies	Preamble
Company	Preamble
Company Engagements	Section 12.17(a)
CFC IPA Deferred Payments	Section 1.1(d)
CFC IPA Deferred Payment Condition	Section 1.3(a)
CFC IPA Deferred Payment Review Period	Section 1.3(d)
Company Insurance Agreements	Section 4.16
Confidentiality Agreement	Section 6.1
Continuation Period	Section 6.9(a)
Continuing Employees	Section 6.9(a)

Term

Initial Deferred Review Period Deposit Escrow Account Deposit Escrow Agent Deposit Escrow Agreement Direct Claim DMHC Dual Claims Employee Benefit Plan Equityholder Indemnitees Equityholder Representative Fund Equityholder Representative Equityholders Escrow Account Escrow Agreement Estimated Balance Sheet Estimated Calculations Estimated Closing Cash Estimated Closing Consideration Estimated Closing Funded Indebtedness Estimated Closing IPA Cash-to-Claims Estimated Closing IPA TNE Estimated Initial Closing Statement Estimated Initial Closing Indebtedness Estimated Second Closing Statement Estimated Second Closing Indebtedness Estimated Transaction Expenses Excluded Real Property Existing CFC Services Agreement Final Calculation of the Initial Closing Payment Final Calculation of the Second Closing Payment Final Purchase Price Allocation Financial Statements Financing Final Calculation of the Second Closing Payment First Calculation of the Initial Closing Payment FLSA Group Companies HIPAA HITECH I Health I Health Plans Indemnified Person Indemnifying Person Initial CFC IPA Deferred Payment Statement Initial Closing

-102-

Section
Section 1.3(d)
Recital
Recital
Recitals
Section 10.5(d)
Recitals
Section 10.4(a) Section 4.19(a)
Section 4.19(a)
Section 10.3 Section 1.4(b)(ii)
Section 1.4(b)(ii)
Preamble
Preamble
Section 1.4(a)(iii)
Section 1.4(a)(iii)
Section 2.1(a)
Section 2.1(a)
Section 2.1(a) Section 2.1(a)
Section 2.1(a)
Section 2.1(a)
Section 2.1(a)
Section 2.1(a)
Section 2.1(a)
Section 2.1(a)
Section 2.1(a) Section 2.1(b)
Section 2.1(b)
Section 2.1(b)
Section 4.9(a)
Section 6.22
Section 2.2(d)
Section 2.2(d)
Section 2.2(d) Section 7.8
Section 4.5(a)
Section 6.17
Section 2.2(a)
Section 2.2(a)
Section 4.18
Recitals
Section 4.14
Section 4.14
Preamble
Section 6.9(b)
Section 10.5
Section 10.5
Section 1.3(d) Section 1.6(a)
Section 1.6(a)

-103-

Term	Section
Initial Closing Company Transaction Expenses	Section 2.1(a)
Initial Closing Date	Section 1.6(a)
Initial Closing Estimated Transaction Expenses	Section 2.1(a)
Initial Closing Purchase Price	Section 1.1(d)
Initial Closing Required Consents	Section 8.2(e)
Initial Closing Statement	Section 2.2(a)
Initial Closing Tax Purchase Price	Section 7.8(d)
Initial Effective Time	Section 1.6(a)
Interim Balance Sheet	Section 4.5(a)
Interim Balance Sheet Date	Section 4.5(a)
Interim Financial Statements	Section 4.5(a)
IPA Beneficial Owner	Preamble
IPA Business	Recitals
IPA Equityholder	Preamble
IPA Employees	Section 6.9(a)
IPA Representative Fund	Section 1.4(a)(iii)
JAMS	Section 12.9(d)
Leased Real Property	Section 4.9(b)
Leased Real Property Documents	Section 4.7(a)(viii)
Leases	Section 4.9(b)
LOC Requirement	Section 6.26
Material Adverse Effect	Section 8.2(a)

Material Contract MMFIT MSO Buyer Indemnitees MSO Business MSO GP Buyer MSO LP Buyer NMM MSA Nasdaq New PC Buyer Services Agreement Outside Date Parties Party PC Buyer PC Buyer Indemnitees Plan Business Plan Companies Plans Pre-Closing Incentive Payments Prior Period Returns Protected Communications Purchase Price Purchased Interests

Section 4.7(a) Section 1.1 Section 10.2(b) Recitals Preamble Preamble Section 5.1 Section 5.6 Section 6.22 Section 9.1(d) Preamble Preamble Preamble Section 10.2(a) Recitals Recitals Recitals Section 6.12 Section 7.5 Section 12.17(c) Section 1.4 Section 1.2(a)(ii)

-104-

Term Section Push-Out Election Section 7.5 **R&W Insurance Policy** Section 5.8 Released Parties Section 12.18(a) **Releasing Parties** Section 12.18(a) Representative Section 12.1(a) Section 6.16(a) Restricted Period Restricted Territory Section 6.16(a) Rule 144 Section 6.23 Schedule Supplement Section 6.28 Second Closing Section 1.6(b) Second Closing Date Section 1.6(b) Second Closing Estimated Transaction Expenses Section 2.1(b) Second Closing Purchase Price Section 1.2(b) Second Closing Required Consents Section 8.5(e)(vii) Second Closing Statement Section 2.2(a) Second Effective Time Section 1.6(b) Preamble Sellers Seller Taxes Section 1.1(b) SMRH Section 12.17(a) Specified Representations Section 8.2(a) Survival Date Section 10.1(b) Section 6.11 Tail Policy Tax Contest Section 7.5 Third Party Claim Section 10.5(a) Unpaid CFC IPA Deferred Payments Section 1.3(c) Year-End Balance Sheet Section 4.5(a) Year-End Balance Sheet Date Section 4.5(a) Year-End Financial Statements Section 4.5(a)

-105-

11.2 Interpretation. As used in this Agreement, except as otherwise indicated in this Agreement or as the context may otherwise require: (a) the words "include," "includes," and "including" are deemed to be followed by "without limitation" whether or not they are in fact followed by such words or words of similar import; (b) the word "or" is not exclusive; (c) references to an "Article," "Section," "preamble," "recital," or any other subdivision, or to an "Appendix," "Exhibit," "Schedule," or "Disclosure Schedule" are to an article, section, preamble, recital, or subdivision of this Agreement, or to an appendix, exhibit, schedule, or disclosure schedule to this Agreement; (d) the words "this Agreement," "hereby," "hereof," "herein," "hereinder," and comparable words refer to all of this Agreement, including the Appendices, Exhibits, Schedules, and Disclosure Schedule to this Agreement, and not to any particular Article, Section, preamble, recital, or other subdivision of this Agreement or Appendix, Exhibit, Schedule, or Disclosure Schedule to this Agreement; (e) any pronoun in masculine, feminine, or neuter form shall include any other gender; (f) any word in the singular form includes the plural and vice versa; (g) references to any Applicable Law are to it as amended, modified, supplemented, and restated now or from time to time after the date of this Agreement, and to any corresponding provisions of successor Applicable Laws, and, unless the context requires otherwise, any reference to any statute shall be deemed also to refer to all rules and regulations promulgated thereunder; (h) references to any Person include such Person's respective successors and permitted assigns (and in the case of a natural person, such Person's heirs, estate, and personal representatives); (i) references to a "day" or number of "days" (without the explicit qualification of "Business") refer to a calendar day or number of calendar days; (j) references to a "Company" or the "Companies" shall include, unless the context requires otherwise, reference to such Company or the Companies and its/their, direct and indirect, Subsidiaries; and (k) the term "made available" means posted to the "Community Family Care Medical Group IPA, Inc." virtual data room hosted at https://collaborate.sheppardmullin.com/.com and continuously viewable by Buyer and its Representatives at least 48 hours prior to the execution hereof. If interest is to be computed under this Agreement, it shall be computed on the basis of a 360-day year of thirty (30)-day months. If any action or notice is to be taken or given on or by a particular calendar day, and such calendar day is not a Business Day, then such action or notice may be taken or given on the next succeeding Business Day. Any financial or accounting term that is not otherwise defined in this Agreement shall have the meaning given such term under GAAP.

12.1 Equityholder Representative.

By executing this Agreement, each Seller hereby appoints Marc Mitchell as the Equityholder Representative under and pursuant to the terms of this (a) Agreement. The Equityholder Representative shall not be compensated for his services as the Equityholder Representative; provided, however, that the Equityholder Representative shall be entitled to reimbursement from the Sellers for all fees, costs and expenses incurred by the Equityholder Representative in discharging the duties hereunder in accordance with Section 12.1(f). The Equityholder Representative is hereby vested with the full power, authority, duty and responsibility to represent the interests of the Sellers as set forth in this Agreement, and the Equityholder Representative shall be authorized to (i) execute and deliver all documents necessary or desirable to carry out the intent of this Agreement, the Escrow Agreement, and any other Transaction Document, (ii) serve as the named party with respect to any claims hereunder on behalf of each of the Sellers, (iii) grant any consent or approval on behalf of the Sellers under this Agreement and any Transaction Document and make all other elections or decisions contemplated by this Agreement and any Transaction Document, (iv) defend, negotiate, agree to, enter into settlements and compromises of, and comply with orders and awards of courts with respect to, claims against a Seller, (v) defend, negotiate, agree to, enter into settlements and compromises of, and comply with orders and awards of courts with respect to, or any claims or disputes related to this Agreement on behalf of a Seller, (vi) give and receive on behalf of the Sellers any and all notices from or to any Seller pursuant to this Agreement or any Transaction Document and (vii) amend, modify or supplement this Agreement and any Transaction Document, in each such Seller's name, place and stead, as if such Seller had personally done such act, and the Equityholder Representative hereby accept such appointment. All actions taken by the Equityholder Representative shall be binding upon the Sellers, their successors, heirs, Representatives and assigns as if expressly confirmed and ratified in writing by each of them. In discharging the duties and responsibilities of the Equityholder Representative hereunder, the Equityholder Representative shall have all rights and powers necessary and incident to the proper discharge thereof, including the right and power to engage and pay for professional or other services. The death, incapacity, insolvency or bankruptcy of any Sellers shall not terminate such appointment or the authority and agency of the Equityholder Representative. The power-of-attorney granted in this Section is coupled with an interest and is irrevocable.

-106-

(b) Neither the Equityholder Representative nor any agent employed by the Equityholder Representative shall be liable to any Seller relating to the performance of the Equityholder Representative's duties under this Agreement for any errors in judgment, negligence, oversight, breach of duty or otherwise except to the extent it is finally determined in a court of competent jurisdiction by clear and convincing evidence that the actions taken or not taken by the Equityholder Representative constituted Fraud or were taken or not taken in bad faith. The Equityholder Representative shall be indemnified and held harmless by the Sellers against all losses, including costs of defense, paid or incurred in connection with any action, suit, proceeding or claim to which the Equityholder Representative is made a party by reason of the fact that the Equityholder Representative was acting as the Equityholder Representative pursuant to this Agreement; provided, however, that the Equityholder Representative shall be protected in acting upon any notice, statement or certificate believed by the Equityholder Representative to be genuine and to have been furnished by the appropriate Person and in acting or refusing to act in good faith on any matter. Neither the Equityholder Representative nor any agent employed by the Equityholder Representative shall be liable to any Buyer Party or any Affiliate of any Buyer Party solely relating to the performance of Equityholder Representative's duties pursuant to this Agreement.

(c) All reasonable fees and expenses (including attorneys' fees and other professional fees) incurred by the Equityholder Representative in connection with the performance of his duties under this Agreement and the Escrow Agreement shall be reimbursed by CFC IPA and/or the AHMS Equityholders based on their Pro Rata Portion, as applicable. In the event the Equityholder Representative is entitled to reimbursement, the Equityholder Representative shall first make a written demand for such fees and expenses upon CFC IPA and/or the AHMS Equityholders based on their Pro Rata Portion, as applicable. In the event the Equityholders based on their Pro Rata Portion, as applicable. In the event the Equityholder Representative is not reimbursed by the applicable Sellers within fifteen (15) Business Days of such demand, the Equityholder Representative may pay such fees and expenses by deducting the full amount of such fees and expenses from the Escrow Account otherwise payable to such Sellers upon termination of the Escrow Account or, prior to termination, from any interest earned on the applicable Escrow Amount.

(d) Any advanced amounts (including amounts in the Equityholder Representative Fund) not expended by the Equityholder Representative on any fees or expenses incurred by the Equityholder Representative pursuant to this Agreement or otherwise on the Sellers' behalf shall be released to CFC IPA and/or the AHMS Equityholders based on their Pro Rata Portion thereof, as applicable, following final resolution of any indemnification claims or other proceedings relating to this Agreement.

-107-

(c) On the date which is eighteen (18) months following the Closing Date, the Equityholder Representative shall release to CFC IPA all of the thenremaining IPA Representative Fund and release to the AHMS Equityholders, in accordance with their Pro Rata Portions, all of the then-remaining AHMS Representative Fund; <u>provided</u>, <u>however</u>, that to the extent there remains any outstanding pending indemnification claims or other proceedings relating to this Agreement, the Equityholder Representative shall have the right to continue to hold any amount remaining in the IPA Representative Fund or the Equityholder Representative Fund, as applicable, until the Equityholder Representative determines in his sole discretion to release such remaining amount to CFC IPA or the AHMS Equityholders, in accordance with their Pro Rata Portions, as applicable.

(f) The Equityholder Representative shall be each Seller's agent for service of process in connection with any dispute or claim arising under this Agreement or any Transaction Document.

(g) The service by the Equityholder Representative shall be without compensation.

(h) As the Representative of the Sellers, the Equityholder Representative shall act as the agent for all such Persons, shall have authority to bind each such Person in accordance with this Agreement, and the Buyer Parties may rely on such appointment and authority until the receipt of notice of the appointment of a successor upon thirty (30) days' prior written notice to the Buyer Parties.

12.2 <u>Notices</u>. All notices, requests, consents, claims, demands, waivers and other communications hereunder shall be in writing and shall be deemed to have been given (a) when delivered by hand (with written confirmation of receipt); (b) when received by the addressee if sent by a nationally recognized overnight courier (receipt requested); (c) on the date sent by facsimile or e-mail of a PDF document (with confirmation of transmission) if sent during normal business hours of the recipient, and on the next Business Day if sent after normal business hours of the recipient or (d) on the third (3rd) day after the date mailed, by certified or registered mail, return receipt requested, postage prepaid. Such communications must be sent to the respective Parties at the following addresses (or at such other address for a Party as shall be specified in a notice given in accordance with this <u>Section 12.2</u>):

If to AHMS (prior to the Second Closing):

Advanced Health Management Systems, L.P. 19210 S. Vermont Ave, Bldg D Ste 400 Gardena, CA 90248 Attn: Ron Brandt Email: brandt@cfc-sys.com

with a copy (which shall not constitute notice) to:	Sheppard, Mullin, Richter & Hampton, LLP 1901 Avenue of the Stars, Suite 1600 Los Angeles, CA 90067-6017 Attention: Eric Klein, Esq. and Chi Huynh, Esq. Email: <u>eklein@sheppardmullin.com</u> <u>chuynh@sheppardmullin.com</u>
If to the Sellers, to the Equityholder Representative:	8033 Sunset Blvd. #895 Los Angeles, CA 90046 Attention: Marc Mitchell Email: <u>mmitchell@dominion-media.com</u>
with a copy (which shall not constitute notice) to:	Ronald L. Brandt 19210 S. Vermont Ave, Bldg D Ste 400 Gardena, CA 90248 Email: <u>brandt@cfc-sys.com</u>
with a copy (which shall not constitute notice) to:	Sheppard, Mullin, Richter & Hampton, LLP 1901 Avenue of the Stars, Suite 1600 Los Angeles, CA 90067-6017 Attention: Eric Klein, Esq. and Chi Huynh, Esq. Email: <u>eklein@sheppardmullin.com</u> <u>chuynh@sheppardmullin.com</u>
If to Buyer:	Apollo Medical Holdings, Inc. 1668 S. Garfield Avenue, 2nd Floor Alhambra, CA 91801 Attention: Chandan Basho Email: <u>cbasho@apollomed.net</u>
with a copy (which shall not constitute notice) to:	Russ August & Kabat, LLP 12424 Wilshire Blvd., 12th Floor Los Angeles, CA 90025 Attention: Lauren Geisser and Christine Shin Email: lgeisser@raklaw.com and cshin@raklaw.com

12.3 Disclosure Schedule. The information in the Disclosure Schedule constitutes (a) exceptions to particular representations, warranties, covenants and obligations of the Group Companies as set forth in this Agreement or (b) descriptions or lists of the equity interests, assets and Liabilities and other items referred to in this Agreement. If there is any inconsistency between the statements in this Agreement and those in the Disclosure Schedule (other than an exception expressly set forth as such in the Disclosure Schedule with respect to (i) a specifically identified representation or warranty or (ii) Sections or subsections of this Agreement to which such exception is reasonably apparent on its face), the statements in this Agreement will control. The statements in the Disclosure Schedule, and those in any supplement to the Disclosure Schedule, shall relate to the numerically corresponding Section or subsection of this Agreement identified in the Disclosure Redule (or supplement to the Disclosure Schedule) and to those Sections or subsections of this Agreement to which it is reasonably apparent on its face that such disclosure relates and not to any other Section or subsection of this Agreement.

12.4 <u>Assignment</u>. This Agreement shall be binding upon and inure to the benefit of the Parties hereto and their respective heirs, successors and permitted assigns. Neither this Agreement nor the obligations of any Party hereunder shall be assignable or transferable by such Party without the prior written consent of the other Parties hereto. Subject to the preceding sentence, this Agreement shall be binding upon, inure to the benefit of and be enforceable by the parties to this Agreement and their respective successors and assigns. Notwithstanding anything to the contrary, (a) any Buyer Party may transfer and assign its rights and obligations under this Agreement and any Transaction Document without the prior written consent of any other Party by way of collateral assignment to any bank, financing institution or other lender providing financing to such Buyer Party now or in the future or to an Affiliate of a Buyer Party or to any successor to such Buyer Party by merger, consolidation or reorganization, and (b) CFC IPA may transfer and assign its rights to receive any consideration for the Acquired Assets, including all or any portion of the CFC IPA Cash Purchase Price, the Buyer Parent Share Consideration and/or any CFC IPA Deferred Payments or Catch-Up Payments, to the Buyer Parent Shares Transferees without prior written consent of the Buyer Parent Shares Transferees shall have delivered to the Buyer Parties (i) a joinder to this Agreement substantially in the form of Exhibit <u>F</u> attached hereto by each of the Buyer Parent Shares Transferees, and (ii) a legal opinion to the Buyer Party of any of its obligations hereunder.

12.5 <u>No Third Party Beneficiaries</u>. This Agreement is for the sole benefit of the Parties hereto and their successors and permitted assigns and nothing herein expressed or implied shall give or be construed to give to any Person any legal or equitable rights hereunder, other than (i) the Parties hereto and such successors and assigns, (ii) the Indemnified Persons and (iii) the Released Parties. Without limiting the generality of the foregoing, (a) no employee of the Companies shall have any rights, as an employee, under this Agreement or under any of the Transaction Documents to which he or she is not personally a party and (b) other than as an express party to this Agreement or the other Transaction Documents, if applicable, no creditor of any of the Parties shall have any rights under this Agreement or any of the Transaction Documents.

12.6 <u>Counterparts</u>. This Agreement may be executed and delivered in one or more counterparts, each of which when executed and delivered shall be an original, and all of which when executed shall constitute one and the same instrument. The exchange of copies of this Agreement and of signature pages by facsimile or other electronic transmission shall constitute effective execution and delivery of this Agreement as to the Parties and may be used in lieu of the original Agreement for all purposes. Signatures of the Parties transmitted by facsimile or other electronic means shall be deemed to be their original signatures for all purposes.

12.7 <u>Amendment and Modification; Waiver</u>. This Agreement may only be amended, modified or supplemented by an agreement in writing signed by each Party hereto. No waiver by any Party hereto of any of the provisions hereof shall be effective unless explicitly set forth in writing and signed by the Party so waiving. No waiver by any Party shall operate or be construed as a waiver in respect of any failure, breach or default not expressly identified by such written waiver, whether of a similar or different character, and whether occurring before or after that waiver. No failure to exercise, or delay in exercising, any right, remedy, power or privilege arising from this Agreement shall operate or be construed as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege.

12.8 <u>Expenses</u>. Except as otherwise expressly provided by this Agreement, whether or not either Closing is effected, all Transaction Expenses shall be borne by the Party incurring the same; <u>provided</u>, that, for the avoidance of doubt, (a) the costs and expenses relating to obtaining the Tail Policy shall be shared equally between the applicable Buyer Party and the applicable Group Company, (b) all filing fees associated with the filings required by this <u>Section 6.6</u> shall be shared equally between the applicable Buyer Parties, on the one hand, and the applicable Group Companies, on the other hand, and (c) all costs and expenses related to the R&W Insurance Policy, including the total premium, underwriting costs, brokerage commissions for the Buyer Party's broker, Taxes related to such policy and other fees and expenses of such policy shall be borne by the Buyer Parties.

12.9 Governing Law; Submission to Jurisdiction.

(a) The validity and construction of this Agreement shall be governed by the internal laws of the State of Delaware, regardless of the laws that might otherwise govern under applicable principles of conflict of laws thereof.

(b) Except as otherwise provided herein, the successful or prevailing party shall be entitled to recover reasonable attorneys' fees and other costs incurred in connection with that Action.

(c) The Parties shall attempt in good faith to resolve any dispute, controversy or claim between the Parties that relates to the interpretation, carrying out of obligations, breach, termination or enforcement of this Agreement or in any way arising out of or connected with this Agreement ("Dispute").

(d) In the event of a Dispute among the Parties, prior to referring to arbitration as described in<u>Section 12.9(e)</u> the Parties must engage in a mediation under the then current mediation rules of JAMS, Inc. ("JAMS"). A single disinterested third party mediator familiar with Delaware law shall be selected by JAMS in accordance with its then current arbitration rules. No Party shall commence arbitration as described in <u>Section 12.9(e)</u> without first attempting in good faith to engage in a mediation as set forth in this <u>Section 12.9(d)</u>. The parties to the Dispute shall share the expenses of the mediator and the other costs of mediation 50-50 between the applicable Buyer Parties, on the one hand, the Companies and their Affiliates, on the other hand.

-111-

(e) If any Dispute cannot be resolved by the Parties pursuant to this <u>Section 12.9</u> or otherwise, such Dispute shall be exclusively referred to and finally settled by arbitration in accordance with the Rules of Arbitration of JAMS. The arbitration proceedings shall take place in Los Angeles, California and shall be conducted in the English language. The number of arbitrators shall be three (3), and the presiding arbitrator shall be chosen by the two arbitrators nominated by the Parties with input from the parties. If the presiding arbitrator is not chosen by the two (2) arbitrators within 30 days of the date of confirmation of the later of the two (2) party nominated arbitrators, he or she shall be appointed by JAMS. The arbitral tribunal shall be final and binding upon the Parties and shall be the sole and exclusive remedy between the parties regarding any claims, counterclaims, issues, or requests for declaratory, accounting, or other relief presented to the arbitral tribunal. The costs of arbitration shall be apportioned by the arbitrat tribunal in the award.

(f) EACH OF THE PARTIES TO THIS AGREEMENT HEREBY IRREVOCABLY WAIVES ALL RIGHT TO A TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

12.10 Specific Performance. The Parties agree that irreparable damage may occur and that the Parties would not have any adequate remedy at law in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that prior to the valid termination of this Agreement pursuant to Section 9.1, the Parties shall be entitled to seek an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement in any federal or state courts located in the State of California without proof of actual damages, this being in addition to any other remedy to which they are entitled at law or in equity.

12.11 <u>Remedies Cumulative</u>. Except as otherwise set forth herein, the various rights and remedies herein provided will be cumulative and not exclusive of any other rights or remedies herein provided or any rights or remedies provided by Applicable Law.

12.12 Construction.

(a) Inasmuch as this Agreement is the result of negotiations between sophisticated Parties of equal bargaining power represented by counsel, no inference in favor of or against any Party will be drawn from the fact that any portion of this Agreement has been drafted by or on behalf of such Party.

(b) Any capitalized terms used in any Exhibit or Schedule but not otherwise defined therein, shall have the meaning as defined in this Agreement.

(c) The heading references herein are for convenience purposes only, do not constitute a part of this Agreement and will not be deemed to limit or affect any of the provisions hereof.

12.13 <u>Severability</u>. The provisions of this Agreement will be deemed severable and the invalidity or unenforceability of any provision will not affect the validity or enforceability of the other provisions hereof. If any provision of this Agreement, or the application thereof to any Person or any circumstance, is invalid or unenforceable, (a) a suitable and equitable provision will be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision, and (b) the remainder of this Agreement and the application of such provision to other Persons or circumstances will not be affected by such invalidity or unenforceability, nor will such invalidity or unenforceability affect the validity or enforceability of such provision, or the application thereof, in any other jurisdiction.

12.14 <u>Entire Agreement</u>. This Agreement, the Disclosure Schedules and the Transaction Documents contain the entire understanding of the Parties with respect to the subject matter hereof and thereof and thereof and supersede all prior oral or written agreements, negotiations, understandings, statements or proposals with respect to the subject matter hereof and thereof.

12.16 <u>No Rescission</u>. No party shall be entitled to rescind the transactions contemplated hereby by virtue of any failure of any party's representations and warranties herein to have been true or any failure by any party to perform its obligations hereunder.

12.17 Certain Matters Regarding Representation of the Equityholders and the Companies

(a) Each of the Parties to this Agreement hereby acknowledges and agrees that (i) Sheppard, Mullin, Richter & Hampton, LLP ('<u>SMRH</u>") has served as counsel for the Equityholders and the Group Companies in connection with this Agreement (the "<u>Acquisition Engagement</u>") and in that connection not as counsel for any other Person, including without limitation, the Buyer Parties, (ii) SMRH may also represent the Group Companies in respect of other matters ("<u>Company Engagements</u>"), and (iii) following consummation of the Transactions, SMRH (or any successor) may serve as counsel to any Equityholder or any director, members, partner, officer, employee or affiliate of any Equityholder, in connection with any litigation, claim or obligation arising out of or relating to this Agreement or the Transactions notwithstanding such representation. Each of the Parties to this Agreement hereby consents thereto and waives any conflict of interest arising thereform, and each of such Parties shall cause any Affiliate thereof to consent to waive any conflict of interest arising from such representation. If the Equityholders of desire, and without limitation anything related to the Transactions or any disagreement or dispute relating thereto; <u>provided</u>, further; that if any Buyer Party or Group Company so desire, and without the need for any consent or waiver by any Equityholder, SMRH shall be permitted to represent the any Group Company or any Buyer Party after the Closing in connection with any matter unrelated to the transactions contemplated by this Agreement or dispute relating thereto; provided, further; that if any Buyer Party after the Closing in connection with any matter unrelated to the transactions contemplated by this Agreement or any disagreement or dispute relating thereto; the any Group Company or any Buyer Party after the Closing in connection with any matter unrelated to the transactions contemplated by this Agreement or any disagreement or dispute relating thereto.

(b) Upon and after the Closing, the Group Companies shall cease to have any attorney-client relationship with SMRH, including with respect to any Company Engagements, unless after the Closing, SMRH is subsequently engaged in writing by any Group Company to represent such Group Company and either such engagement involves no conflict of interest with respect to the Equityholders or the Equityholder Representative consents in writing to such engagement. Any such representation of a Group Company by SMRH after Closing shall not affect the provisions of this Section. For example, and not by way of limitation, even if SMRH is representing a Group Company after the Closing, SMRH shall be permitted simultaneously to represent the Equityholders in any matter, including, without limitation, any disagreement or dispute relating to this Agreement. Furthermore, SMRH shall be permitted to withdraw from any Company Engagement in order to be able to represent or continue so representing the Equityholders, even if such withdrawal causes a Group Company or the Buyer Parties additional legal expense (such as to bring new counsel "up to speed"), delay or other prejudice. In the event that there has been an overpayment of legal fees to SMRH by the Group Companies after the Closing, SMRH shall release the amount of such overpayments to the Equityholder Representative.

-113-

(c) Notwithstanding anything to the contrary contained herein, the parties intend that all communications prior to the Closing between the Group Companies and Equityholders, on the one hand, and any of their attorneys, on the other hand, relating to the negotiation of the Transactions are eligible for protection as attorney-client privileged communications (collectively, the "<u>Protected Communications</u>"), and all associated rights to assert, waive and otherwise administer the attorney-client privilege in such Protected Communications, will, from and after the Closing, rest exclusively with the Equityholders and will not be transferred, assigned, conveyed or deliver, by operation of law or otherwise, to any of the Buyer Parties or any of their respective Affiliates (including, after the Closing, the Group Companies). Notwithstanding the foregoing, in the event that a dispute arises between any Buyer Party, Group Company, or any of their Subsidiaries, on the one hand, and a third party other than any Equityholder, on the other hand, the applicable Group Company may assert the attorney-client privilege to prevent the disclosure of Protected Communications to such third party.

(d) The Equityholders, the Group Companies, and the Buyer Parties consent to the arrangements in this Section and waive any actual or potential conflict of interest that may be involved in connection with any representation by SMRH permitted hereunder.

12.18 <u>Release</u>.

(a) For and in consideration of the amount to be paid to the Sellers under this Agreement and the Transaction Documents, and the additional covenants and promises set forth in this Agreement, each Seller, effective as of such Equityholder's applicable Closing Date, on behalf of itself, himself or herself and its Affiliates (the "Releasing Parties"), hereby fully, finally and irrevocably releases, acquits and forever discharges each of the Buyer Parties, the Group Companies, their respective Subsidiaries and their respective Representatives and Affiliates (collectively, the "Released Parties") from any and all claims, counterclaims, suits, causes of action, demands, damages, liabilities, obligations, costs, expenses and compensation of every kind and nature whatsoever, past, present or future, at law or in equity, whether known or unknown, contingent or otherwise, relating to or arising out of the dealings of any Released Party and such Releasing Party had, has or may have had at any time in the past until and including the date of this Agreement, including but not limited to any claims which relate to or arise out of such Releasing Party's prior relationship with any Group Company or I Health (including their respective present and former subsidiaries, parent entities or any predecessors in interest) (collectively, for the purposes of this <u>Section 12.18(a)</u>. "<u>Causes of Action</u>"); provided, that this release shall not affect or impair: (i) any Causes of Action, rights of the Releasing Parties or any obligations of the Released Parties to the Releasing Parties arising under this Agreement or in any Transaction Document, including without limitation, the Management Services Agreement, (ii) any Causes of Action or rights to reimbursement, indemnification or contribution of any Releasing Party in his, her or its capacity as an officer, director, manager, stockholder, unitholder or employee of a Released Party (whenever arising) under the Governing Documents of the applicable Releasing Party, the Tail Policy or App

-114-

(b) The Releasing Parties hereby represent to the applicable Released Parties that the Releasing Parties: (i) have not assigned any Causes of Action or possible Causes of Action against any applicable Released Party, (ii) fully intend to release all applicable Causes of Action against the applicable Released Parties including without limitation, unknown and contingent Causes of Action (other than those specifically reserved above) and (iii) have consulted with counsel with respect to the execution and delivery of this general release and has been fully apprised of the consequences hereof. The Releasing Parties hereby covenant and agree not to make any claim or cross-claim or to take any Action against any other Person who or which might claim contribution from or indemnification by any Affiliate of the Group Companies under the provisions of any Applicable Law or otherwise except claims relating to rights arising from this Agreement or in any Transaction Document.

(c) Each Releasing Party acknowledges that such Releasing Party has been advised to consult with legal counsel and is familiar with the provisions of California Civil Code Section 1542, a statute that otherwise prohibits the release of unknown claims, which provides as follows:

A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS WHICH THE CREDITOR OR RELEASING PARTY DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE, WHICH IF KNOWN BY HIM OR HER WOULD HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR OR RELEASED PARTY.

(d) Each Releasing Party being aware of said code section agrees to expressly, effective as of the applicable Closing Date, waive any rights such Releasing Party may have thereunder, as well as under any other statute or common law principles of similar effect.

(c) The Releasing Parties hereby irrevocably covenant to, effective as of the applicable Closing Date, refrain from, directly or indirectly (i) asserting any applicable Causes of Action, or commencing, instituting or causing to be commenced, or continuing with any Action for a Cause of Action, and this Agreement may be raised by any applicable Released Party as an estoppel to any such Actions and (ii) making any claim or commencing any Action against any Person (or assisting or encouraging any other Person in connection therewith) in which any Action would arise against any applicable Released Party for contribution or indemnity or other relief from, over and against any applicable Released Party or which otherwise results in an applicable Released Party suffering or incurring any Damages, whether under Applicable Laws or Contract or otherwise, with respect to an applicable Cause of Action. Without in any way limiting any of the rights and remedies otherwise available to any applicable Released Party, the Releasing Parties shall indemnify and hold harmless each applicable Released Party from and against all applicable Causes of Action whether or not involving third-party claims, actions or proceedings, arising directly or indirectly from or in connection with the assertion by or on behalf of each applicable Releasing Party or any of its Affiliates of any Action which is, or is purported to be, an applicable Cause of Action. It is the intention of the Releasing Parties hereby expressly waive any and all rights and benefits conferred upon it by the provisions of Applicable Law with respect to any applicable Cause of Action hereinabove specified. In furtherance of this intention, the Releasing Parties hereby expressly waive any and all rights and benefits shall be given full force and effect according to each and all of its express terms and provisions.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY LEFT BLANK]

-115-

IN WITNESS WHEREOF, each Party has duly executed and delivered this Agreement as of the date first above written.

BUYER PARTIES:

"PC Buyer"

METROPOLITAN IPA

By: <u>/s/ Thomas S. Lam</u> Name: Thomas S. Lam Title: Chief Executive Officer

"MSO LP Buyer"

NETWORK MEDICAL MANAGEMENT, INC.

By:	/s/ Thomas S. Lam
Name:	Thomas S. Lam
Title:	Chief Executive Officer

"MSO GP Buyer"

APOLLOCARE ENABLEMENT OF CA, LLC

By: <u>/s/ Brandon Sim</u> Name: Brandon Sim Title: Manager

"Buyer Parent"

APOLLO MEDICAL HOLDINGS, INC.

By:	/s/ Brandon Sim
Name:	Brandon Sim
Title:	Co-Chief Executive Officer

[Signature Page to Asset and Equity Purchase Agreement]

IN WITNESS WHEREOF, each Party has duly executed and delivered this Agreement as of the date first above written.

COMPANIES:

"CFC IPA"

Community Family Care Medical Group IPA, Inc.

By: /s/ Accie Mitchell Name: Accie Mitchell, M.D. Title: Chief Executive Officer Advanced Health Management Systems, L.P.

By: CFC Management, LLC Its: General Partner

By: /s/ Marc Mitchell Name: Marc Mitchell Title: Manager

[Signature Page to Asset and Equity Purchase Agreement]

IN WITNESS WHEREOF, each Party has duly executed and delivered this Agreement as of the date first above written.

EQUITYHOLDERS:

"IPA Equityholder"

Accie M. Mitchell and Gloria C. Mitchell, as co-trustees of the Mitchell Family Trust dated July 2, 2003 $\,$

/s/ Accie Mitchell Accie Mitchell, M.D. Co-Trustee

/s/ Gloria Mitchell Gloria Mitchell Co-Trustee

"IPA Beneficial Owner"

/s/ Accie M. Mitchell Accie M. Mitchell, M.D.

"AHMS General Partner"

By:	CFC Management, LLC
Its:	Manager

By: <u>/s/ Marc L. Mitchell</u> Name: Marc L. Mitchell Title: Manager

"AHMS Limited Partners"

Mitchell Family Trust

/s/ Accie M. Mitchell Accie M. Mitchell, M.D. Co-Trustee

[Signature Page to Asset and Equity Purchase Agreement]

/s/ Gloria C. Mitchell Gloria C. Mitchell Co-Trustee

Mitchell Children's Trust

/s/ Marc L. Mitchell Marc L. Mitchell Co-Trustee

/s/ Alex M. Mitchell

Alex M. Mitchell Co-Trustee

Marc Mitchell Irrevocable Trust

/s/ Marc L. Mitchell Marc L. Mitchell Trustee

Alex Mitchell Irrevocable Trust

/s/ Alex M. Mitchell Alex M. Mitchell

Trustee

Cynthia Heard Irrevocable Trust

/s/ Marc L. Mitchell Marc L. Mitchell Co-Trustee

/s/ Alex M. Mitchell Alex M. Mitchell Co-Trustee

[Signature Page to Asset and Equity Purchase Agreement]

Tracy Mitchell Irrevocable Trust

/s/ Marc L. Mitchell Marc L. Mitchell Co-Trustee

/s/ Alex M. Mitchell Alex M. Mitchell Co-Trustee

Lori Konsker Irrevocable Trust

/s/ Marc L. Mitchell Marc L. Mitchell Co-Trustee

/s/ Alex M. Mitchell Alex M. Mitchell Co-Trustee

Jason Heard Irrevocable Trust

/s/ Marc L. Mitchell Marc L. Mitchell Co-Trustee

/s/ Alex M. Mitchell Alex M. Mitchell Co-Trustee

Briella Konsker Irrevocable Trust

/s/ Marc L. Mitchell Marc L. Mitchell Co-Trustee

/s/ Alex M. Mitchell Alex M. Mitchell Co-Trustee

[Signature Page to Asset and Equity Purchase Agreement]

/s/ Marc L. Mitchell Marc L. Mitchell Co-Trustee

/s/ Alex M. Mitchell Alex M. Mitchell Co-Trustee

Harrison Konsker Irrevocable Trust

/s/ Marc L. Mitchell Marc L. Mitchell Co-Trustee

/s/ Alex M. Mitchell Alex M. Mitchell Co-Trustee

Bennet Mitchell Irrevocable Trust

/s/ Marc L. Mitchell Marc L. Mitchell Co-Trustee

/s/ Alex M. Mitchell Alex M. Mitchell Co-Trustee

August Mitchell Irrevocable Trust

/s/ Marc L. Mitchell Marc L. Mitchell Co-Trustee

/s/ Alex M. Mitchell Alex M. Mitchell Co-Trustee

[Signature Page to Asset and Equity Purchase Agreement]

Langston Mitchell Irrevocable Trust

/s/ Marc L. Mitchell Marc L. Mitchell Co-Trustee

/s/ Alex M. Mitchell Alex M. Mitchell Co-Trustee

Josephine Mitchell Irrevocable Trust

/s/ Marc L. Mitchell Marc L. Mitchell Trustee

William Calder Mitchell Irrevocable Trust

/s/ Alex M. Mitchell Alex M. Mitchell Trustee

AHMS Trust

/s/ Mr. Dreschler Premier Trust Trustee Mr. Dreschler

/s/ Ronald L. Brandt Name: Ronald L. Brandt IN WITNESS WHEREOF, each Party has duly executed and delivered this Agreement as of the date first above written.

I HEALTH:

I Health Inc.

<u>/s/ Ron Brandt</u> Name: Ron Brandt Title:

[Signature Page to Asset and Equity Purchase Agreement]

IN WITNESS WHEREOF, each Party has duly executed and delivered this Agreement as of the date first above written.

EQUITYHOLDER REPRESENTATIVE:

/s/ Marc Mitchell Name: Marc Mitchell

[Signature Page to Asset and Equity Purchase Agreement]

STOCK PURCHASE AGREEMENT

by and among

NETWORK MEDICAL MANAGEMENT, INC.,

I HEALTH, INC.,

RONALD BRANDT

and

ALLISON BRANDT

DATED AS OF NOVEMBER 7, 2023

TABLE OF CONTENTS

		Page
ARTICLE 1 Pu	urchase and sale; Closing	6
1.1	Purchase and Sale	6
1.2	Closing	6
1.3	Withholding	6
1.4	Disclosure of Excluded Assets	6
ARTICLE 2 Re	epresentations and warranties of the Equityholder and Beneficial Owners	7
2.1	Organization and Qualification	7
2.2	Authority, Power and Enforceability	7
2.3	Capitalization	7
2.4	Non-Contravention; Consents	8
2.5	Financial Statements	9
2.6	Absence of Certain Changes, Events and Conditions	10
2.7	Material Contracts	10
2.8	Tangible Assets	12
2.9	Real Property	12
2.10	Legal Proceedings; Judgments	12
2.11	Compliance with Applicable Laws	13
2.12	Permits	13
2.13	Healthcare Regulatory Matters	13
2.14	Intellectual Property	14
2.15	Privacy and Security Compliance	15
2.16	Insurance	16
2.17	Taxes	16
2.18	Employment Matters	18
2.19	Employee Benefit Matters	20
2.20	Affiliate Transactions	22
2.21	Illegal Payments	22
2.22	Brokers	22
ARTICLE 3 Re	epresentations and warranties of buyer	22
3.1	Organization and Authority of Buyer	23
3.2	Non-Contravention; Consents	23
3.3	Legal Proceedings	23
3.4	Brokers	24
ARTICLE 4 Co	ovenants	24
4.1	Confidentiality	24
4.2	Publicity	24
4.3	Further Assurances	25

- 4.6 Approvals and Consents
- 4.7 No Negotiation
- 4.8 Efforts; Regulatory Filings and Consents

^{4.5} Access to Information

	4.9	Non-Competition; Non-Solicitation; Non-Disparagement	29
	4.10	Termination of 401(k) Plan	30
	4.11	Employee Matters	31
	4.12	PP&E	31
ARTICI	LE 5 TAX	X MATTERS	31
ARTICL	LE 6 Con	ditions to Closing	34
	The con	ditions to the obligations of the Equityholder and the Buyer with respect to the Closing shall be set forth in Sections 6.1 and 6.2, and 6.3.	34
	6.1	Conditions to the Obligations of All Parties with Respect to the Closing	34
	6.2	Closing Conditions to Obligations of Buyer	35
	6.3	Closing Conditions to Obligations of the Equityholder	37
ARTICL	LE 7 Tern	nination	37
	7.1	Termination of the Agreement	37
	7.2	Effect of Termination	38
ARTICL	E 8 Inde	mnification	38
	8.1	Survival	38
	8.2	Indemnification by the Equityholder and Beneficial Owners	39
	8.3	Indemnification by the Buyer	40
	8.4	Certain Limitations	40
	8.5	Indemnification Procedures	41
	8.6	No Contribution	43
ARTICL	.E 9 Defi	nitions	43
	9.1	Certain Definitions	43
	9.2	Interpretation	53
ARTICL	LE 10 Mi	scellaneous	53
	10.1	Notices	53
	10.2	Assignment	54
	10.3	No Third Party Beneficiaries	54
	10.4	Counterparts	54
	10.5	Amendment and Modification; Waiver	55

3

10.0	F	
10.6	Expenses	55
10.7	Governing Law; Submission to Jurisdiction	55
10.8	Specific Performance	56
10.9	Remedies Cumulative	56
10.10	Construction	56
10.11	Severability	56
10.12	Entire Agreement	56
10.13	Time is of the Essence	57
10.14	No Rescission	57
10.15	Release	57
	4	

STOCK PURCHASE AGREEMENT

This STOCK PURCHASE AGREEMENT (this "<u>Agreement</u>"), dated as of November 7, 2023, is entered into by and amongNETWORK MEDICAL MANAGEMENT, INC., a California corporation ("<u>Buyer</u>"); RONALD BRANDT ("<u>Ron Brandt</u>") and ALLISON BRANDT ("<u>Allison Brandt</u>"), each in their individual capacities (as, the "<u>Beneficial Owners</u>") and in their capacities as Co-Trustees of the Ronald Lee Brandt and Allison Leigh Brandt Family Trust dated December 16, 2003 (the "<u>Equityholder</u>"); and I HEALTH, INC., a California corporation (the "<u>Company</u>"). The Buyer, the Company, the Beneficial Owners and the Equityholder are referred to collectively herein as the "<u>Parties</u>" and, each individually, as a "<u>Party</u>").

RECITALS

WHEREAS, the Company is in the business of providing management, consulting, administrative, and other support services to entities that provide or arrange for the provision of professional healthcare services;

WHEREAS, pursuant to that certain Services Agreement, dated as of July 1, 2019 (the 'Services Agreement'), by and between the Company and Advanced Health Management Systems, L.P., a California limited partnership ("<u>AHMS</u>"), the Company provides to AHMS certain management, consulting, administrative, and support services ("<u>Management Services</u>");

WHEREAS, the Equityholder owns 1,000 shares of common stock of the Company constituting one hundred percent (100%) of the issued and outstanding shares of the Company (the "Shares"), and the Beneficial Owners are a married couple who are the sole beneficiaries of the Equityholder;

WHEREAS, pursuant to the terms and conditions set forth herein, the Equityholder wishes to sell to Buyer, and Buyer wishes to purchase from the Equityholder, 250 shares of common stock of the Company (the "Purchased Shares") on the terms specified herein;

WHEREAS, contemporaneously with the execution of this Agreement, Buyer, Ronald Brandt and certain other parties are entering into an Asset and Equity Purchase Agreement (the "<u>CFC Purchase Agreement</u>") whereby Buyer and its Affiliates will be acquiring (i) 100% of the issued and outstanding general partnership interest of AHMS and limited partnership interests of AHMS from the AHMS General Partner and AHMS Limited Partners, respectively, as such terms are defined therein; and (ii) the Acquired Assets and Assumed Liabilities from Community Family Care Medical Group IPA, Inc., a California corporation ("<u>CFC IPA</u>"), all as more specifically provided therein; and

WHEREAS, Ron Brandt is one of the AHMS Limited Partners and, in that capacity, will not only receive (along with the AHMS General Partner and other AHMS Limited Partners) the Second Closing Purchase Price (as defined in the CFC Purchase Agreement) from Buyer but will also be eligible on a stand-alone basis to receive the AHMS Deferred Payments (as defined in the CFC Purchase Agreement) from Buyer, all as more particularly described in and subject to the terms and conditions of the CFC Purchase Agreement.

5

NOW, THEREFORE, in consideration of the premises and the mutual representations, warranties, covenants and agreements hereinafter set forth and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, incorporating the recitals set forth above, the Parties agree as follows:

AGREEMENT

ARTICLE 1

Purchase and sale; Closing

1.1 Purchase and Sale.

(a) <u>Purchase and Sale</u>. On the terms and subject to the conditions of this Agreement, at the Closing the Equityholder shall sell, transfer, convey, assign and deliver to Buyer, and Buyer shall purchase, acquire and accept from Equityholder, all of the right, title and interests of the Equityholder in, to and under the Purchased Shares, free and clear of all Liens.

(b) <u>Purchase Price</u>. In consideration for the sale, transfer, conveyance, assignment and delivery of the Purchased Shares pursuant to <u>Section 1.1(a)</u>, Buyer shall pay to the Equityholder cash in immediately available funds in an aggregate amount equal to (collectively, the "<u>Purchase Price</u>") at the Closing, [*] Dollars (\$[*]]).

1.2 <u>Closing</u>.

(a) The consummation of the sale of the Purchased Shares (the "<u>Closing</u>") shall take place electronically via mutual exchange of facsimile or portable document format (.PDF) signatures commencing at 10:00 a.m. Pacific Standard Time on the third (3rd) Business Day after the last of the conditions set forth in<u>Section 6.1</u>, and <u>Section 6.2</u> have been satisfied or waived (other than conditions which, by their nature, are to be satisfied on the Closing Date), or at such other time as shall be agreed upon in writing by Buyer and the Equityholder. The date on which the Closing is actually held is referred to herein as the "<u>Closing Date</u>." The Closing shall be deemed to be effective at 12:01 A.M. Pacific Standard Time on the Closing Date for all purposes, except as may otherwise be expressly provided herein.

1.3 <u>Withholding</u>. Buyer shall be entitled to deduct and withhold from the consideration or other amounts otherwise payable pursuant to this Agreement to any Person such amounts as they are required to deduct and withhold with respect to such payment under the Code, or any provision of state, local or foreign law. Any amounts withheld by Buyer shall be treated for all purposes of this Agreement as having been paid to the appropriate recipient in respect of which such deduction and withholding was made by Buyer.

1.4 <u>Disclosure of Excluded Assets</u>. Prior to Closing, Equityholder will have transferred out of the Company certain of the assets of Company, as listed on <u>Schedule 1.5</u>, that are not to be a part of the transaction.

6

ARTICLE 2

Representations and warranties of the Equityholder and Beneficial Owners

As a material inducement to Buyer to execute and perform its obligations under this Agreement, the Equityholder, each Beneficial Owner and the Company make the following representations and warranties to Buyer (subject to such exceptions as are set forth in the Disclosure Schedule and delivered herewith by the Equityholder to Buyer with schedules and references qualifying the section or subsection they reference), and agree and acknowledge that Buyer has relied on, among other things, the representations and warranties contained in this Article 2, each of which is true and correct, and each of which shall survive the Closing as herein set forth:

2.1 <u>Organization and Qualification</u>. The Company is a corporation duly incorporated, validly existing and in good standing under the Applicable Laws of its jurisdiction of incorporation. The Company is qualified to do business and is in good standing in the State of California, which jurisdictions constitute all of the jurisdictions in which the ownership of properties or the conduct and nature of its business requires such qualification. The Company has the requisite organizational power and authority to own, lease, operate and use its assets and properties and to carry on its business as now conducted and currently proposed to be conducted. The Company is not in nor has it been in default under or in violation of any provision of its Governing Documents.

2.2 <u>Authority, Power and Enforceability</u>. The Company, the Equityholder and each Beneficial Owner each have all requisite power and authority to execute and deliver this Agreement and the applicable Transaction Documents and to consummate the transactions contemplated hereunder and thereunder and to perform its obligations hereunder and thereunder. This Agreement and each Transaction Document to which each Company, Beneficial Owner or Equityholder is or will be a party has been duly executed and delivered by the Company, each Beneficial Owner and Equityholder, as applicable, and this Agreement and the Transaction Documents constitute legal, valid and binding obligations of the Company, each Beneficial Owner and Equityholder, enforceable against the Company, each Beneficial Owner and Equityholder in accordance with their respective terms. No other proceedings on the part of the Company, each Beneficial Owner or Equityholder are necessary to approve and authorize the execution, delivery and performance of this Agreement and the Transaction Documents to which it is a party and the consummation of the transactions contemplated hereby and thereby.

2.3 <u>Capitalization</u>.

(a) The only authorized, issued and outstanding Equity Interests of the Company are the 1,000 shares of common stock of the Company owned by the Equityholder, including the identity of the Persons that are the record and beneficial owners thereof. All of the Equity Interests of the Company have been duly authorized, were validly issued, fully paid (as applicable), are free and clear of any and all Liens and have been offered, issued, sold and delivered in compliance with applicable federal and state

securities Laws and the organizational documents of the Company. Except for the the 1,000 shares of common stock of the Company owned by the Equityholder, no voting or non-voting units, other Equity Interests or other voting securities of the Company are issued, reserved for issuance or outstanding.

7

(b) (i) No subscription, warrant, option, convertible or exchangeable security, or other right (contingent or otherwise) to purchase or otherwise acquire Equity Interests of the Company is authorized or outstanding, and (ii) there is no commitment by the Company to issue shares, subscriptions, warrants, options, convertible or exchangeable securities, or other such rights or to distribute to holders of any of its Equity Interests any evidence of indebtedness or asset, to repurchase or redeem any securities of the Company or to grant, extend, accelerate the vesting of, change the price of, or otherwise amend any warrant, option, convertible or exchangeable security or other such right. None of the Equity Interests of the Company are subject to any purchase option, call option, right of first refusal, first offer, co-sale or participation, preemptive right, subscription right or any similar right. Except for the Governing Documents of the Company, there are no documents, instruments or agreements relating to the voting of any Equity Interests of the Company. There are no existing, authorized, issued or outstanding securities, options, warrants, calls, rights (including conversion rights, preemptive rights, co-sale rights, rights of first refusal and similar rights), convertible or exchangeable securities or Contracts or obligations of any kind (contingent or otherwise) to which the Company or, to the Knowledge of the Company, any equityholder or beneficial owner of the Company. There are no outstanding obligations of the Company (contingent or otherwise) in, or to provide any guarantee with respect to the obligations of, any Person. There are no equity-appreciation rights, equity-based performance units, "phantom" equity rights, profit participation rights or other contracts or obligations of any character (contingent or otherwise) purchase, equity-based performance units, "phantom" equity rights, profit participation rights or other contracts or obligations of any character (contingent or otherwise) pursuant to

(c) There are no bonds, debentures, notes or other Indebtedness of the Company having the right to vote or consent (or, convertible into, or exercisable or exchangeable for, securities having the right to vote or consent) on any matters on which the equityholders of the Company may vote. There are no voting trusts, irrevocable proxies or other contracts or understandings to which the Company, or, to the Company's Knowledge, any equityholder of the Company is a party or is bound with respect to the voting or consent of any Equity Interests.

(d) All distributions, dividends, repurchases and redemptions of the capital stock (or other equity interests) of the Company were undertaken in compliance with the applicable Governing Documents then in effect, any agreement to which the Company then was a party and in compliance with Applicable Law.

(e) The Company has made available to the Buyer true, correct and complete copies of the Governing Documents for the Company, which Governing Documents are in full force and effect.

2.4 <u>Non-Contravention; Consents</u>. The execution, delivery and performance by the Company of this Agreement and each Transaction Document to which it is a party, and the performance by it of any actions contemplated hereunder or thereunder, does not and will not, directly or indirectly (with or without notice or lapse of time): (a) conflict with or violate any provision of the Governing Documents of such Company; (b) contravene, require notice, consent or approval under, conflict with, violate, result in a breach or violation of, conflict with or constitute or result in a breach or default (whether after the giving of notice, lapse of time or both), accelerate any obligation under, or give rise to a right of acceleration, cancellation or termination of, loss of a benefit or increase in Liabilities or fees under, create in any Person the right to terminate, cancel or modify, or cause a default under or give rise to any rights or penalties under (i) any provision of Applicable Law, (ii) any provision of any Governmental Order to which the Equityholder, any Beneficial Owner or the Company or any of their respective properties are subject, (iii) any Permit, license or authorization to which the Equityholder, any Beneficial Owner or the Company is a party or by which their respective assets are bound or (iii) any provision of any Material Contract; (c) cause any of the assets of the Company to be subject to a Lien; or (d) require a registration, filing, application, notice, consent, approval, order, qualification or waiver with, to or from any Governmental Authority or third party.

8

2.5 Financial Statements.

(a) The Company has made available to the Buyer (i) the unaudited consolidated financial statements consisting of the balance sheet of the Company, dated as of 2021 and 2022 and the related statements of income and retained earnings, and cash flow for the years then ended (collectively, the "<u>Year-End Financial Statements</u>"), and unaudited financial statements consisting of the balance sheet of the Company, dated as of June 30, 2023, and the related statements of income and retained earnings for the 6-month period then ended (collectively, the "<u>Interim Financial Statements</u>"). The Financial Statements (i) were prepared in accordance with GAAP applied on a consistent basis throughout the periods covered, (ii) fairly present in all material respects the financial condition of the Company as of the respective dates they were prepared and the results of operations of the Company for the periods indicated, subject, in the case of the unaudited financial statements, to normal year-end adjustments that are neither individually nor in the aggregate material in amount and a lack of footnotes, and (iii) have been prepared from, and are in accordance with, the books and records of the Company.

(b) The books of account and financial records of the Company are true and correct in all material respects and have been prepared and are maintained on a consistent basis throughout the period involved. The Company has not made any changes in its accounting practice since the date of the Year-End Balance Sheet. The Company maintains a standard system of accounting established and administered on a consistent basis.

(c) There are no Liabilities of the Company of any nature, whether absolute, accrued, contingent or otherwise or whether due or to become due, other than Liabilities reflected on the Year-End Balance Sheet. There are no conditions existing with respect to the properties, assets, personnel of the Company that might materially and adversely affect the Company.

(d) All of the accounts receivable of the Company are valid and enforceable claims, and are subject to no set-off or counterclaim, except for normal course recoupment and related actions undertaken by payors. Since the date of the Year-End Balance Sheet, the Company has collected its accounts receivable in the ordinary course of business and in a manner which is consistent with past practices and has not accelerated any such collections. The Company has no accounts receivable or loans receivable from any Person who is a Related Party.

(e) All accounts payable and notes payable of the Company arose in bona fide arm's length transactions in the ordinary course of business and no such account payable or note payable is delinquent in its payment under the normal practices of the payors. Since the date of the Year-End Balance Sheet, the Company has paid its

accounts payable in the ordinary course of business and in a manner which is consistent with its past practices. The Company does not have any accounts payable or note payable to any Person who is a Related Party.

(f) The Company has not entered into any transactions involving the use of special purpose entities for any off-balance sheet activity other than as specifically described in the Financial Statements.

2.6 <u>Absence of Certain Changes, Events and Conditions</u> Since the date of the Interim Balance Sheet, there has occurred no fact, event or circumstance which, individually or in the aggregate, has had, or could reasonably be expected to have a Material Adverse Effect. Since the Interim Balance Sheet Date, the Company has conducted the Business in the Ordinary Course of Business, and, without limiting the generality of the foregoing, there has not been with respect to the Company any action that would have been prohibited or otherwise restricted under <u>Section 4.4</u>, had such action been taken during the period between the date hereof and the Closing Date.

2.7 <u>Material Contracts</u>.

(ii)

(a) Except as set forth on Section 2.7(a) of the Disclosure Schedule the Company is not a party to or bound by, nor are any of its properties, rights or assets bound by:

(i) any Contract or Judgment that involve obligations (contingent or otherwise) of, or payments to, the Company in excess of Ten Thousand Dollars (\$10,000) per fiscal year, or that may not be terminated without penalty on thirty (30) days' notice or less;

other hand;

(iii) any Contract for employment, independent contractor or consulting services to which the Company is a party;

(iv) any Contract involving payment of a bonus, commissions, severance benefits, nonqualified deferred compensation, profit sharing, employee equity purchase, retirement or other similar agreement to or for any employee, independent contractor or consultant of the Company;

(v) any bond, debenture, mortgage, note or other similar Indebtedness or Liability, or any Contract to create or issue any bonds, debentures, mortgages, notes or other similar Indebtedness;

(vi) any Contract pursuant to which the Company guarantees the performance, Liabilities or obligations of any other Person;

10

(vii) any Contract limiting the freedom of the Company to (A) engage in any line of business or (B) compete with any other Person or (C) solicit any employees, customers or suppliers of any other Person;

(viii) any lease or sublease or similar Contract with any Person under which the Company is a lessee or sublessee of any real property (collectively, the "Leased Real Property Documents");

(ix) any lease, sublease or similar Contract with any Person under which the Company is a lessee or sublessee of any tangible personal property;

any Contract between the Company, on the one hand, and any of its Affiliates, Related Parties, or any Affiliate of any Related Party, on the

(x) any Contract that involves (A) the acquisition, merger or purchase of all or substantially all of the assets or business of a third party, or (B) the purchase or sale of assets, or a series of purchases and sales of assets, involving aggregate consideration of Ten Thousand Dollars (\$10,000) or more outside of the Ordinary Course of Business;

(xi) any Contract pursuant to which the Company has agreed to provide "most favored nation" status, "best pricing" or any other arrangement whereby the Company or Equityholder or any Beneficial Owner has agreed with any Person that such Person will (or could be entitled to) receive at least as favorable terms and conditions that are provided by the Company to any other Person;

(xii) any Contract with any Governmental Authority;

(xiii) any Contract containing any power of attorney or grant of agency, in either case granted by the Company that is currently in effect, except as entered into in the ordinary course of business consistent with past practice;

(xiv) any Contract relating to the settlement of any actions, suits, proceedings, orders, judgments, decrees, claims or investigations, including administrative charge or investigation by a Governmental Authority or other dispute;

(xv) any Contract not executed in the ordinary course of business; or

(xvi) any Contract with any physician or medical group other than CFC IPA;

(xvii) Contract not otherwise referred to in this Section 2.7(a) that if terminated or expired without being renewed would have or be reasonably likely to result in a Material Adverse Effect.

(b) The Contracts required to be disclosed on Section 2.7(a) of the Disclosure Schedule are referred to herein as the 'Material Contracts'. Each Material Contract is valid and binding on the Company in accordance with its terms and is in full force and effect. Neither the Company and to the Knowledge of the Company, no other party thereto is in breach of or default under (or is alleged to be in breach of or default under) in any material respect, or has provided or received any written or, to the Knowledge of the Company, oral notice of any intention to terminate, any Material Contract prior to the expiration of the term (including renewal terms) thereof, or to amend the material terms of any Material Contract outside of the ordinary course of business consistent with past practice. No party to a Material Contract has exercised any termination rights with respect thereto or has given written notice of any dispute with respect thereto. Complete and correct copies of each Material Contract (including all modifications, amendments and supplements thereto and waivers thereunder) have been made available to Buyer.

2.8 <u>Tangible Assets</u>. The Company owns good and valid title to, or a valid leasehold interest in, all of the personal property and assets of the Company, free and clear of all Liens. The assets of the Company constitute all of the assets necessary and sufficient for the conduct of the business of the Company as currently or proposed to be

conducted on the date hereof. The buildings, improvements, machinery, equipment, personal properties, vehicles and other tangible assets of the Company are operated in conformity with all Applicable Laws, are structurally sound (in the case of the buildings and improvements), are in good condition and repair, except for reasonable wear and tear not caused by neglect, and are usable in the ordinary course of business of the Company.

2.9 <u>Real Property</u>. The Company owns no real property. There is real property that is leased or occupied by the Company.

2.10 Legal Proceedings; Judgments.

(a) There is no Action pending or to, the Knowledge of the Company, threatened, (i) against the Company or any officer, manager, director or employee thereof, in his/her capacity as an officer, manager or director or any of the assets or property owned or used by the Company or any Person whose liability the Company has retained or assumed, either contractually or by operation of law, or (ii) that challenge, or that may have the effect of preventing, delaying, making illegal or otherwise interfering with, any of the other transactions contemplated by this Agreement or the Transaction Documents to which the Company, any Beneficial Owner or Equityholder is or will be a party, in each case at law or in equity, or before or by any Governmental Authority, and to the Company's Knowledge, there is no basis for any of the foregoing. Neither the Company nor any of the assets or property owned or used by the Company is or has been subject to any outstanding investigation, order, writ, injunction, judgment or decree issued by any Governmental Authority.

(b) There are no litigation, claims, proceedings or, to the Company's Knowledge, investigations involving the Company or any of its directors, members, managers, officers or employees, in their capacity as such, occurring, arising or existing during the past six (6) years. To the Company's Knowledge, no director, member, manager, officer or employee of the Company is, or during the past six (6) years has been, subject to any order, writ, injunction, judgment or decree that prohibits such director, member, manager, officer or employee from engaging in or continuing any conduct, activity or practice relating to the Company's business.

(c) No Equityholder, Beneficial Owner, officer, director or manager of the Company has been: (i) subject to voluntary or involuntary petition under the federal bankruptcy Laws or any state insolvency Law or the appointment of a receiver, fiscal agent or similar officer by a court for his or her business or property or that of any partnership of which he or she was a general partner or any corporation or business association of which he or she was an executive officer; (ii) convicted in a criminal proceeding or named as a subject of a pending criminal proceeding (excluding traffic violations and other minor offenses) or been otherwise accused of any act of moral turpitude; (iii) the subject of any order, judgment, or decree (not subsequently reversed, suspended or vacated) of any court of competent jurisdiction permanently or temporarily enjoining him or her from, or otherwise imposing limits or conditions on his or her ability to engage in any securities, investment advisory, banking, insurance or other type of business or acting as an officer or director of a public company; or (iv) found by a court of competent jurisdiction to have violated any federal or state commodities, securities or unfair trade practices Law, which judgment or finding has not been subsequently reversed, suspended, or vacated.

12

2.11 Compliance with Applicable Laws. The Company is and has been in compliance with all Applicable Laws (including but not limited to, Health Care Laws) in all material respects. The Company is not and has not been in violation in any material respect of any Applicable Law, Order, policy or guideline of any Governmental Authority, payment network, governmental program or self-regulatory organization, by which the Company is bound, whether by operation of Applicable Law or by Contract, or to which the Company is subject. The Company has not received any written notice of, or to the Company's Knowledge, oral notice, from any Governmental Authority of an investigation with respect to any violation with respect to any such Law, Order, policy or guideline. No regulatory action, investigation or inquiry of any sort (other than non-material routine or periodic inspections or reviews) against or affecting the Company are pending or have been threatened. The Company has not received any written opinion, memorandum or written advice from any attorney or other legal advisor to the effect that it is exposed to any liability or disadvantage that could materially prohibit or materially restrict the Company from, or otherwise adversely affect the Company in, conducting business in any jurisdiction in which it is now conducting business or in which it plans to conduct business. The Company has not received any written notices or other communications regarding any actual or alleged violation of, or any material failure to comply with, any Applicable Laws, or orders or instructions of any Governmental Authority.

2.12 <u>Permits</u>. The Company owns or possesses, and has owned and possessed all right, title and interest in and to all Permits necessary to (i) conduct the Business in the manner currently conducted or contemplated in the business plan and (ii) permit the Company to own and use the assets in the manner in which they are currently owned and used or contemplated in the business. The Company has delivered to Buyer accurate and complete copies of all such Permits including all renewals and all amendments. The Company is in compliance with the terms and requirements of such Permits and all such Permits are valid, in full force and effect and sufficient for the services provided by the Company. All applications required to have been filed for the renewal of such Permits have been duly filed on a timely basis with the appropriate Governmental Authorities, and each other notice or filing required to have been given or made with respect to such Permits has been duly given or made on a timely basis with the appropriate Governmental Authorities.

2.13 <u>Healthcare Regulatory Matters</u>. During the ten (10)-year period ending on the date hereof:

(a) The Company is and has been in compliance in all material respects with all applicable Healthcare Laws and has timely filed all material reports, data and other information required to be filed with Governmental Authorities.

13

(b) The Company has not been charged with, given written notice of, placed under any obligation to take remedial action under, made subject to any Action with respect to, or placed under investigation with respect to, any applicable Healthcare Laws.

(c) Without limiting the generality of the foregoing, neither the Company nor any Beneficial Owner or Equityholder or Representative thereof, has directly or indirectly offered, paid, solicited or received, or made arrangements to offer, pay, solicit or receive, any remuneration, in cash or in kind, to or from any past, present, or potential customers, past, present, or potential suppliers, patients, physicians, healthcare professionals, medical staff members, contractors, referral sources, or third party payors either in violation of any Healthcare Law or in order to obtain business or payments from such Persons that would reasonably be expected to subject the Company or the Business to any material damage or penalty in any civil, criminal or governmental Action.

(d) Neither the Company nor any Equityholder or Beneficial Owner or Representative thereof, has made or is in the process of making a voluntary selfdisclosure under the Self-Referral Disclosure Protocol established by the Secretary of the United States Department of HHS pursuant to Section 6409 of the Patient Protection and Affordable Care Act, or under the self-disclosure protocol established and maintained by the HHS Office of the Inspector General, or any United States Attorney, the U.S. Department of Justice, a state Medicaid program or fraud unit, a state Attorney General, or other Governmental Authority.

(e) The Company is not and has not been (i) a party to a corporate integrity agreement with the HHS Office of Inspector General; (ii) subject to any reporting obligations pursuant to any settlement agreement entered into with any Governmental Authority; (iii) a defendant in any qui tam/False Claims Act litigation; or (iv) the recipient of or served with any search warrant, subpoena, civil investigation demand, contact letter or any other material inquiry related to compliance with Healthcare Laws from any Governmental Authority.

(f) The Company maintains and adheres to corporate compliance program for oversight of the Business designed to ensure compliance with Healthcare Laws and applicable Office of Inspector General compliance program guidance.

2.14 Intellectual Property.

(a) The Company has no (i) Company Registered Intellectual Property, or (ii) Company Licensed Intellectual Property. The Company is the sole owner of all right, title and interest in and to Company Intellectual Property and has the sole right to use Company Intellectual Property without the payment of any royalties or other amounts to any other Person, or have the right to use the Intellectual Property, other than Company Intellectual Property pursuant to a valid and subsisting license with a third party free and clear of all Liens.

(b) The Company has taken all commercially reasonable steps to obtain, maintain and protect the Company Intellectual Property, including, where appropriate, registering Company Intellectual Property. (i) To the Knowledge of the Company, the Company has not nor has it been infringing upon, nor has the Company received any notice with respect to any alleged infringement or unlawful or improper use of, any Intellectual Property owned or alleged to be owned by others, (ii) no reasonable basis exists upon which a claim may be asserted against the Company or any employees, officers, directors or equityholders for infringement, misappropriation or unlawful or improper use of any Intellectual Property, (iii) the Company has made any written claim or sent any written notice that a Person is infringing on the Company Intellectual Property, (iv) to the Knowledge of the Company, no Person is infringing on the Company Intellectual Property and (v) no order, holding, decision or judgment has been rendered by any Governmental Authority or court, and no agreement, consent or stipulation exists which would limit the Company's use or enjoyment of, any right in the Company Intellectual Property necessary to conduct its business as currently conducted.

14

(c) The Company has obtained and possesses valid licenses to use all of the software programs present on the computers and other software-enabled electronic devices that it owns or leases or that it has otherwise provided to its employees for their use in connection with the Business.

(d) All former and current employees, consultants and contractors of the Company have executed written instruments with the Company that assign to the Company all rights, title and interest in and to any and all inventions, improvements, discoveries, writings and other works of authorship, and information relating to the business of the Company or any of the products or services being researched, developed, manufactured or sold by the Company or that may be used with any such products or services and Intellectual Property relating thereto. To the Knowledge of the Company, no party to any such agreement is in material breach thereof. No employee or independent contractor has, or has asserted, any right, title or interest in or to any of the Company Intellectual Property.

(e) To the Knowledge of the Company, the IT Systems (i) are sufficient and in good working condition to effectively perform for the operation of the business, (ii) have not had any material malfunction or failure that has not been remedied in all material respects, and (iii) do not contain any virus, worm, Trojan horse or similar disabling code or program. To the Knowledge of the Companies, there have been no unauthorized intrusions or breaches of security with respect to the IT Systems. The Company has implemented and maintain commercially reasonable measures designed to protect the integrity and security of the IT Systems under their control, as well as commercially reasonable data backup and continuity, system redundancy, and disaster avoidance and recovery procedures.

(f) Following the Closing, Buyer will be permitted to cause to be exercised all of the rights of the Company under the Company Intellectual Property to the same extent the the Company would have been able had the Transactions not occurred. Following the Closing, all Company Intellectual Property will be fully transferable, alienable or licensable by Buyer without restriction and without payment of any kind to any Person.

2.15 <u>Privacy and Security Compliance</u>. The Company is in compliance in all material respects with (i) the applicable requirements of the regulations governing the privacy of individually identifiable health information and the regulations governing the security of such information maintained in electronic form promulgated pursuant to the Health Insurance Portability and Accountability Act of 1996 ("<u>HIPAA</u>") the implementing regulations adopted pursuant thereto at 45 C.F.R. Parts 160, 162 and 164 and any such other any other applicable federal privacy law or rule, including the American Recovery and Reinvestment Act of 2009, Public Law 111-5 ("<u>HITECH</u>") and any applicable state privacy laws, and (ii) all other applicable Privacy Laws governing the privacy, security, breach, integrity, accuracy, creation, transmission, receipt, maintenance, use, disclosure, or other protection of individually identifiable information created, received, maintained, transmitted, or destroyed by the Company, to the extent not preempted by HIPAA. The Company is in compliance in all material respects with all regulations and agreements regarding business associates, as that term is defined by HIPAA. Within the past six (6) years, the Company has not received any written complaint, nor, to the Knowledge of the Company, no any of its Representatives. To the Knowledge of the Company, within the past six (6) years, the Company has not fice of a patient's protected health information or individually identifiable information. To the Knowledge of the Company has not term is defined by HIPAA, involving any patient protected health information. To the Knowledge of the Company has not fice of the Company, no information security or privacy incident has occurred in the past six (6) years that would require notification to any Governmental Authority.

2.16 <u>Insurance</u>. The Company has made available to the Buyer all of the insurance policies and binders maintained by or on behalf of the Company with respect to its directors and officers, properties, assets, equipment, operations, employees, consultants and business, together with a claims history for the past three (3) years (collectively, "<u>Company Insurance Agreements</u>"). All Company Insurance Agreements are valid, binding and in full force and effect in accordance with their terms and all premiums with respect thereto covering all periods up to and including the Closing Date have been paid or will be paid when due, and the Company has not been: (a) in default with respect to its Liabilities under any such insurance policies or (b) denied insurance coverage. No material default exists with respect to the obligations of the Company under any such Company Insurance Agreements. The Company has not received any written notice of cancellation or termination with respect to any such policy which has not been replaced on substantially similar terms.

2.17 <u>Taxes</u>.

(a) The Company has duly and timely filed all income and other Tax Returns required to be filed by it under Applicable Laws with the appropriate Tax Authority in all jurisdictions in which such Tax Returns are required to be filed (after giving effect to any valid extensions of time in which to make such filings). All such Tax Returns are true, accurate and complete in all respects and have been prepared in compliance with all Applicable Laws. All Taxes (whether or not shown on any Tax Return) imposed or otherwise due and owing by or with respect to the business and activities of the Company have been paid in full.

(b) The Equityholder has made available to Buyer copies of all income Tax Returns of the Company for all taxable years ending on or after December 31, 2019.

(c) The unpaid Taxes of the Company (i) did not, as of the date of the Interim Balance Sheet, exceed the reserve for Tax liabilities (other than any reserve for deferred Taxes established to reflect timing differences between book and Tax income) of the Company set forth on the face of the Interim Balance Sheet; and (ii) do not exceed that reserve as adjusted for the passage of time through the Closing Date in accordance with the past custom and practice of the Company in filing its Tax Returns. Since the date of the Interim Balance Sheet, the Company has not incurred any liability for Taxes arising from

extraordinary gains or losses, as that term is used in GAAP, outside the ordinary course of business consistent with past custom and practice.

16

(d) No outstanding deficiency or adjustment in respect of Taxes has been proposed or asserted in writing or assessed by any Tax Authority against or with respect to the Company. There are no outstanding refund claims with respect to any Tax or Tax Return of the Company, or the business or activities of the Company.

(e) The Company thereof has withheld and paid all Taxes required to have been withheld and paid in connection with any amounts paid or owing to any employee, independent contractor, creditor, stockholder, or other third party. Neither the Company nor any of the Company's employees, is subject to liability as a "responsible person" under Code section 6672. The Company does not have any liability for Taxes, whether absolute or contingent with respect to any misclassification of any person as an independent contractor rather than as an employee.

(f) There are no liens for Taxes (other than current Taxes not yet due and payable or not yet delinquent) on any of the assets of the Company.

(g) The Company has not consented to extend, nor is a beneficiary of an agreement to extend, the time in which any Tax may be assessed or collected by any Tax Authority other than consents with respect to periods for which the applicable statute of limitations has since expired.

(h) The Company has not requested or been granted, nor is the beneficiary of, an extension of the time for filing any Tax Return that has not yet been filed (other than any automatic extensions of time that are allowable under Applicable Law).

(i) No Action related to Taxes or Tax audit has been initiated in writing or is currently pending against the Company (or, to the Knowledge of the Company, threatened) in any jurisdiction.

(j) The Company has timely withheld and timely paid all Taxes that are required to have been withheld and paid by it in connection with amounts paid or owing to any employee, independent contractor, creditor or other Person.

(k) There are no outstanding rulings of, or requests for rulings by, any Tax Authority addressed to the Company that are, or if issued would be, binding on the Company after the applicable Closing Date.

(1) There are no tax sharing, allocation, indemnification or similar agreements (other than Contracts that do not primarily relate to Taxes) in effect as between the Company or any predecessor or Affiliate thereof and any other party under which Buyer or the Company could be liable for any Taxes or other claims of any party after the Closing Date.

(m) Neither the Company nor Buyer (by reason of owning any interest in the Company) will be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of any (i) installment sale or open transaction disposition made prior to the Closing Date; (ii) prepaid amount received prior to the Closing Date, or (iii) closing agreement under Code Section 7121 (or other comparable agreement) entered into prior to the Closing Date.

(n) The Company is not and has never been a party to a "reportable" or "listed" transaction within the meaning of Sections 6662, 6662A, 6011, 6012, 6111, or 6707A of the Code or Treasury Regulations Section 1.6011-4 promulgated thereunder or pursuant to notices or other guidance published by the Internal Revenue Service (irrespective of the effective dates).

(o) The Company is not required to pay, gross up, or otherwise indemnify any employee or contractor for any Taxes, including potential Taxes imposed under Section 409A or Section 4999 of the Code. The Company is not a party to any contract, agreement, plan, or other arrangement covering any employee or former employee or contractor or former contractor that, individually or collectively, could give rise to a (or already has resulted in a) payment or provision of any other benefit (including accelerated vesting) that could not be deductible by reason of Section 280G of the Code or could be subject to an excise Tax under Section 4999 of the Code (in each case, determined without regard to the exceptions provided for in Section 280G(b)(5) of the Code). The Company's "nonqualified deferred compensation plans" within the meaning of Section 409A of the Code have been maintained in operational and documentary compliance with Section 409A of the Code and the Treasury Regulations promulgated thereunder and no such "nonqualified deferred compensation plan" has or will result in any participant incurring income acceleration or Taxes under Section 409A of the Code, as of the Closing Date, no outstanding equity interest of the Company is subject to a "substantial risk of forfeiture" for purposes of Section 83 of the Code.

(p) The Company has been, in the past five (5) years, a party to a transaction reported or intended to qualify as a reorganization under Section 368 of the Code. The Company has not constituted a "distributing corporation" or a "controlled corporation" (within the meaning of Section 355(a)(1)(A) of the Code) in a distribution of shares that was reported or otherwise constitutes a distribution of shares under Section 355 of the Code in the two (2) years prior to the date of this Agreement or that could otherwise constitute part of a "plan" or "series of related transactions" (within the meaning of Section 355(e) of the Code) that includes the transactions contemplated by this Agreement.

2.18 Employment Matters.

(a) Section 2.18(a) of the Disclosure Schedule contains a complete and accurate list of all current employees of the Company and sets forth for each such individual the following information as of the date of the Closing: (i) date of hire; (ii) title/position; (iii) full-time or part-time status; (iv) hourly rate or salary amount; (v) incentive-based compensation; (vi) fringe benefits; (vii) classification as exempt or non-exempt under the Fair Labor Standards Act ("FLSA") and other Applicable Laws; (vii) employment status (including whether any such employee is on approved leave of absence (whether family leave, workers' maternity or parental leave, workers' compensation, short-term and long-term disability, medical leave or otherwise); and (viv) employer of record. Each employee identified as "exempt" from overtime and/or minimum wage requirements of the FLSA or other Applicable Laws is properly and correctly so classified. Section 2.18(a) of the Disclosure Schedule also contains a separate accurate and complete list of the individual independent contractors currently engaged to provide services to the Company and the individuals listed in Section 2.18(a) of the Disclosure Schedule as independent contractors or employees have been properly so classified, as the case may be.

(b) Except as set forth in <u>Section 2.18(a) of the Disclosure Schedule</u>, the engagement of each employee and independent contractor of the Company is terminable at will without any penalty, liability or severance obligation.

(c) The Company has paid or will timely pay all compensation, including wages, accrued and unused paid vacation and other paid leave, commissions, bonuses, fees, or other compensation payable to their respective employees or independent contractors for services performed on or before the Closing Date. Other than as set forth in <u>Section 2.18(a) of the Disclosure Schedule</u>, there are no Contracts between the Company and any employees or independent contractors regarding any compensation, wages, commissions, bonuses, or fees.

(d) The Company is not party to any collective bargaining agreement, nor has it made any proposals regarding the terms of any collective bargaining agreement. The Company is not subject to any: (i) pending or, to the Knowledge of the Company, threatened unfair labor practice complaint before a Governmental Authority; (ii) pending or, to the Knowledge of the Company, threatened labor strike, slowdown, work stoppage, lockout, or other organized labor disturbance, or (iii) union organization efforts or attempts by any union to represent employees as a collective bargaining agent.

(c) There are no claims, disputes, grievances, or controversies pending or, to the Knowledge of the Company, threatened involving any of their respective prospective or actual employees or group of employees, interns, volunteers or independent contractors. To the Knowledge of the Company, there are no written threats, charges, investigations, administrative proceedings or formal complaints of discrimination (including discrimination based upon sex, age, marital status, race, national origin, sexual orientation, disability or veteran status) pending before the Equal Employment Opportunity Commission, the National Labor Relations Board, the U.S. Department of Labor, the U.S. Occupational Health and Safety Administration, the Workers Compensation Appeals Board, or any other Governmental Authority against the Company pertaining to any of their respective prospective or actual employees or group of employees, interns, volunteers or independent contractors.

(f) In the last six (6) years: (i) to the Knowledge of the Company, no allegations of sexual harassment or sexual misconduct have been made involving their respective current or former directors, officers, managers or independent contractors; and (ii) the Company has not entered into any settlement agreements related to allegations of sexual harassment or sexual misconduct by any current or former director, officer, manager or independent contractors.

(g) The Company is in compliance in all material respects with Applicable Laws regarding employment and employment practices, including all Applicable Laws relating to wages, hours, paid sick leave, overtime, collective bargaining, employment discrimination, civil rights, safety and health, workers' compensation, pay equity, immigration and employment eligibility verification laws (including the Immigration and Nationality Act and the Immigration Reform and Control Act of 1986, and all related regulations promulgated thereunder), classifications of employees and independent contractors for wage and hour and tax purposes, and the collection and payment of withholding and/or social security Taxes.

19

2.19 Employee Benefit Matters.

(a) Section 2.19(a) of the Disclosure Schedule sets forth an accurate and complete list of all benefit plans, policies, or arrangements under which any current or former employee, director, or consultant of the Company, or any beneficiary of such individuals, has any present or future right to benefits, or to which the Company has or may have any Liability or obligation to contribute (including on behalf of an ERISA Affiliate), including: (i) any "employee benefit plan" (within the meaning of Section 3(3) of ERISA, whether or not subject to ERISA); (ii) any profit-sharing, pension, Code Section 125 "cafeteria" or flexible spending account, deferred compensation, bonus, stock or equity-based incentive, educational assistance, severance, retention, change in control, vacation or paid time off, or health or welfare plan, agreement or arrangement; or (iii) any other material plan, agreement or arrangement providing for compensation, "fringe benefits" or prequisites related to or provided in the course of employenent with the Company, in each case whether or not in writing (each such plan, program, or arrangement an "Employee Benefit Plan")

(b) With respect to each Employee Benefit Plan, the Company has made available to Buyer a copy of the following documents as applicable to each such Employee Benefit Plan: (i) the plan document or, if an oral arrangement, a written summary of the material terms of the Employee Benefit Plan; (ii) the three most recent annual reports (Form 5500); (iii) each trust agreement, the most recent summary plan description and any summaries of material modification; (iv) all nondiscrimination testing reports for the three most recently completed plan years; (v) all related insurance contracts and administrative service agreements, (vi) all other material documents pursuant to which such Employee Benefit Plan is maintained, funded, and administered, (vii) the most recent Internal Revenue Service determination or opinion letter, and (viii) all material correspondence with any Governmental Authority.

(c) Neither the Company nor any ERISA Affiliate maintains, sponsors, contributes to, has any obligation to contribute to, or has any current or potential obligation or liability under or with respect to any (i) "defined benefit plan" (as defined in Section 3(35) of ERISA) or any other plan subject to the funding requirements of Section 412 or 430 of the Code or Section 302 or Title IV of ERISA, (ii) "multiemployer plan" (as defined in Section 3(37) of ERISA), (iii) plan, program, agreement or arrangement that provides for post-retirement or post-termination medical, life insurance or other similar benefits (other than health continuation coverage required by COBRA for which the covered individual pays the full cost of coverage), (iv) multiple employer plan (as described in section 413(c) of Code or Section 210 of ERISA), or (v) "multiple employer welfare arrangement" (as defined in Section 3(40) of ERISA). Neither the Company nor any ERISA Affiliate has any current or potential liability or obligation under Title IV of ERISA.

2	n	
L	υ	

(d) Each Employee Benefit Plan has been operated and administered in all material respects in accordance with its terms, and is and has been in compliance in all material respects with ERISA, the Code and all other Applicable Laws. The obligations of all Employee Benefit Plans that provide health, welfare or similar insurance are fully insured by bona fide third-party insurers. No Employee Benefit Plan is maintained through a human resources and benefits outsourcing entity, professional employer organization, or other similar vendor or provider, except to provide third-party administrative or recordkeeping services to an Employee Benefit Plan sponsored by the Company. With respect to each Employee Benefit Plan, all payments, premiums, contributions, distributions, reimbursements and accruals for all periods ending prior to or as of the Closing Date shall have been made or accrued on the Interim Balance Sheet in accordance with GAAP and there is no unfunded liability which is not reflected on the Interim Balance Sheet.

(e) Each Employee Benefit Plan which is intended to be qualified under Section 401(a) of the Code is so qualified and is the subject of a favorable determination letter or, in the case of a prototype or volume submitter plan, an opinion letter from the Internal Revenue Service that the form of the plan document meets the applicable requirements of Section 401(a) of the Code, and nothing has occurred since the date of such determination or opinion letter that could adversely affect the qualified status of such Employee Benefit Plan.

(f) With respect to each Employee Benefit Plan, (i) no "prohibited transaction," within the meaning of Section 4975 of the Code or Sections 406 and 407 of ERISA, and not otherwise exempt under Section 408 of ERISA, has occurred; (ii) there are no actions, suits or claims pending, or, to the Knowledge of the Company, threatened or anticipated (other than routine claims for benefits) against any such Employee Benefit Plan or fiduciary thereto or against the assets of any such Employee Benefit Plan; (iii) no breach of fiduciary duty (as determined under ERISA) or any other failure to act or comply in connection with the administration or investment of the assets of such Employee Benefit Plan has occurred; and (iv) there are no audits, inquiries or proceedings pending or, to the Knowledge of the Company, threatened by any Governmental

Authority with respect to any Employee Benefit Plan.

(g) The execution and delivery of this Agreement and the consummation of the Transactions will not, either alone or in connection with any other event or circumstance, (i) result in any termination or severance payment becoming due to any employee of the Company; (ii) accelerate the time of payment or vesting, or otherwise increase the amount of compensation or benefits due any officer, director, employee, or consultant of the Company; or (iii) result in any amount of compensation failing to be deductible by reasons of Section 280G of the Code.

(h) Each Employee Benefit Plan that is a "nonqualified deferred compensation plan" (as defined under Section 409A(d)(1) of the Code) is and has been in documentary and operational compliance with Section 409A of the Code. The Company does not have any "gross-up" or indemnity obligations for Taxes imposed under Section 4999 or 409A of the Code.

(i) The Company has, for purposes of each Employee Benefit Plan, correctly classified those individuals performing services for the Company as common law employees or independent contractors. The Company does not have any leased employees within the meaning of Section 414(n) of the Code.

21

(j) Each Employee Benefit Plan can be amended or terminated by the Company (or the applicable ERISA Affiliate) at any time (whether before or after the Closing Date) without any liability, obligation, or damages (other than routine administrative charges).

(k) None of the Employee Benefit Plans are maintained, contributed to, or required to be contributed to outside of the United States or for the benefit of any employee or other service provider that provides services or resides outside of the United States.

2.20 <u>Affiliate Transactions</u>. Except as set forth on <u>Section 2.20 of the Disclosure Schedule</u>, no Related Party, is, or has been within the past six (6) years, directly or indirectly, a party to any Contract or transaction with the Company or which is pertaining to the business of the Company or has any interest in any property, real or personal or mixed, tangible or intangible, used in or pertaining to the business of the Company. <u>Section 2.20 of the Disclosure Schedule</u> hereto describes: (a) all affiliated services provided to or on behalf of the Company by the Equityholder or their respective Affiliates and to or on behalf of the Equityholder and its Affiliates by the Company and all affiliate transactions or Contracts among the Company, on the one hand, and the Equityholder or any of its Affiliates, on the other hand (including, in each case, the costs charged to or by the Company), and (b) a description of all relationships and affiliations among all parties to any arrangements or transactions disclosed in accordance with clause (a) of this <u>Section 2.20</u>.

2.21 <u>Illegal Payments</u>. Neither the Company nor the Equityholder nor any of their respective directors, officers, employees or agents have offered, authorized, made or received on behalf of the Company any illegal payment or contribution of any kind, including any payment in violation of any Applicable Laws, directly or indirectly, including, without limitation, any kickbacks, bribes, payments, gifts or gratuities, to any Person, or U.S. or foreign national, state or local government officials, employees or agents or candidates therefor or other Persons. The Company has maintained systems of internal controls (including, but not limited to, accounting systems, purchasing systems and billing systems) and written policies to ensure compliance with Applicable Laws regarding illegal payments and to ensure that all books and records accurately and fairly reflect, in reasonable detail, all transactions and dispositions of funds and assets. Neither the Company nor the Equityholder nor any of their respective directors, officers, employees or agents, are the subject of any allegation, voluntary disclosure, investigation, prosecution or other enforcement action related to illegal payment laws.

2.22 <u>Brokers</u>. No broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with the Transactions or any other transactions, whether past, present or future, based upon arrangements made by or on behalf of the Equityholder, any Beneficial Owner or the Company.

ARTICLE 3 Representations and warranties of buyer

Except as disclosed in (x) the Company's Annual Report filed on Form 10-K or any other documents filed or furnished by Apollo Medical Holdings, Inc. with the SEC (including any exhibits and any other information incorporated by reference therein), Buyer hereby represents and warrants to the Equityholder that the statements contained in this <u>Article 3</u> are true and correct on the date hereof and shall be true and correct on the Closing Date as if made thereon.

22

3.1 <u>Organization and Authority of Buyer</u>. Buyer is a corporation duly organized, validly existing and in good standing under the Applicable Laws of its jurisdiction of incorporation. Buyer has the requisite organizational power and authority to own, lease, operate and use its assets and properties and to carry on its business as now conducted and currently proposed to be conducted. Buyer is not in default under or in violation of any provision of its Governing Documents. Buyer has all requisite power and authority to execute and deliver this Agreement and the applicable Transaction Documents and to consummate the transactions contemplated hereunder and thereunder and thereunder and thereunder. This Agreement and each Transaction Document to which Buyer is or will be a party has been duly executed and delivered by the Buyer, and this Agreement and the Transaction Documents constitute legal, valid and binding obligations of the Buyer, enforceable against Buyer in accordance with their respective terms. No other proceedings on the part of the Buyer are necessary to approve and authorize the execution, delivery and performance of this Agreement and the Transaction Documents to which it is a party and the consummation of the transactions contemplated hereby.

3.2 <u>Non-Contravention; Consents</u>. The execution and delivery by the Buyer of this Agreement and each Transaction Document to which Buyer is a party, and, assuming the accuracy of the Equityholders, Beneficial Owners and Company's representations and warranties set forth in <u>Article 2</u>, the performance by Buyer of any actions contemplated hereunder or thereunder, does not and will not, directly or indirectly (with or without notice or lapse of time): (a) conflict with or violate any provision of the Governing Documents of Buyer; (b) assuming the Required Consents are made and obtained, require notice, consent or approval under, materially conflict with, materially violate, result in a material breach of, result in the acceleration of material obligations, loss of a material benefit or increase in material Liabilities or fees under, create in any Person the right to terminate, cancel or materially modify, or cause a material default under or give rise to any material rights or penalties under (i) any provision of Applicable Law relating to Buyer, (ii) any provision of any Governmental Order to which Buyer or any of its properties are subject, or (iii) any provision of any material Contracts to which Buyer be required under the Securities Act, the Exchange Act, the DGCL, the HSR Act or other applicable Antitrust Laws, applicable state securities and "blue sky" laws or the rules and regulations of Nasdaq, require a registration, filing, application, notice, consent, approval, order, qualification or waiver with, to or from any Governmental Authority, except as, individually or in the aggregate, would not reasonably be expected to constitute or result in a Material Adverse Effect.

3.3 Legal Proceedings. There are no Actions pending or, to Buyer's knowledge, threatened against or by Buyer or any Affiliate of Buyer that, individually or in the aggregate, would have a Material Adverse Effect on Buyer's ability to execute, deliver or perform this Agreement or any Transaction Document to which it is a party, or to timely consummate the Transactions.

3.4 <u>Brokers</u>. Except for such amounts as the Buyer shall pay, no broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with the Transactions or any other Transaction Document based upon arrangements made by or on behalf of the Buyer.

ARTICLE 4 <u>Covenants</u>

Confidentiality. None of the Equityholder, any Beneficial Owner or any Affiliate of the Equityholder or Beneficial Owner (each, a "Restricted Person") shall, 4.1 and shall cause each of such Restricted Person's Affiliates and its and their Representatives not to, disclose, and treat and hold as strictly confidential, all Confidential Information and refrain from using any Confidential Information. On the request of Buyer at any time after the date hereof, each Restricted Person shall deliver promptly to Buyer or destroy all tangible embodiments of the Confidential Information which are in such Restricted Person's, its Affiliates' or their respective Representatives' possession or under any of their control and provide confirmation thereof in writing to Buyer. In the event that any Restricted Person or any of such Restricted Person's Affiliates or their respective Representatives is requested or required to disclose any Confidential Information, such Restricted Person shall notify Buyer promptly so that Buyer may seek an appropriate protective order or waive compliance with the provisions of this Section 4.1. If, in the absence of a protective order or the receipt of a waiver hereunder, such Restricted Person or any of such Restricted Person's Affiliates or their respective Representatives is compelled to disclose any Confidential Information to any governmental entity, such Restricted Person, Affiliate or Representatives may disclose such Confidential Information to such governmental entity; provided, that such Restricted Person, Affiliate or Representatives shall use best efforts to obtain an order or other assurance that confidential treatment shall be accorded to such Confidential Information required to be disclosed. Notwithstanding the foregoing, for purposes of this Agreement, Confidential Information shall not include information which is or becomes generally available to the public other than as a result of a disclosure by any Restricted Person or any of such Restricted Person's Affiliates or their respective Representatives in violation of this Agreement or other confidentiality obligation to which any of them is bound. Nothing in this Section 4.1 is intended to conflict with 18 U.S.C. § 1833(b) or create liability for disclosures of trade secrets that are expressly allowed by 18 U.S.C. § 1833(b). Accordingly, each Restricted Person has the right to disclose in confidence trade secrets to federal, state, and local governments officials, or to an attorney, for the sole purpose of reporting or investigating a suspected violation of law, and to disclose trade secrets in a document filed in a lawsuit or other Action, but only if the filing is made under seal and protected from public disclosure.

4.2 <u>Publicity</u>. Notwithstanding anything to the contrary contained in this Agreement, except as may be required to comply with the requirements of any Applicable Law or by the applicable rules of any securities exchange upon the reasonable advice of counsel (it being understood that each such Party shall use reasonable commercial efforts to provide the other Parties to this Agreement a draft of such press release, public statement or public communication in advance of such disclosure), from and after the date hereof, no Party shall make, and each Party shall cause their respective Affiliates not to make, any press release or similar public announcement or public communication relating to this Agreement or the Transactions unless specifically approved in advance by the other Parties, which approval shall not be unreasonably withheld, conditioned or delayed.

24

4.3 <u>Further Assurances</u>. From time to time, as and when requested by any Party, each Party shall execute and deliver, or cause to be executed and delivered, all such documents and instruments and shall take, or cause to be taken, all such further or other actions as such other Party may reasonably deem necessary or desirable to consummate the Transactions, including executing and delivering to the other Parties hereto such assignments, deeds, bills of sale, consents and other instruments as the other Parties hereto or their counsel may reasonably request as necessary or desirable for such purpose.

4.4 <u>Conduct of Business Prior to the Closing</u>

(a) Except for matters permitted or required by the terms of this Agreement or Transaction Documents, or except as required by Applicable Law, from the date of this Agreement through the Closing, the Company shall conduct the Business in the Ordinary Course of Business and preserve intact the current business organization, use its Commercially Reasonable Efforts to keep available the services of current officers, employees and agents of the Company and maintain in good state relations and good will with customers, suppliers, licensors, licensees, distributors, landlords, creditors, employees, agents and others having business relationships with the Company. The Company shall refrain from taking any action that would result in any of the conditions set forth in <u>Section 6.1</u> not being satisfied. In addition (and without limiting the generality of the foregoing), except as otherwise contemplated by the terms of this Agreement or the Transaction Documents, or except as required by Applicable Law, from the date of this Agreement through the Closing, the Company shall not do any of the following without the prior written consent of Buyer:

(i) amend the Governing Documents;

(ii) issue any capital stock or other equity interests or any option, warrant or right relating thereto or any securities convertible into or exchangeable for any capital stock or other equity interests;

(iii) adopt, modify, terminate or amend any Employee Benefit Plan (or any plan that would be an Employee Benefit Plan if adopted) or enter into, adopt, extend (beyond the Closing Date), renew or amend any collective bargaining agreement or other Contract with any labor organization, union or association;

(iv) pay any bonus or make any profit sharing or similar payment to, or materially increase the amount of the wages, salary, commissions, fees, fringe benefits or other compensation or remuneration payable to, any of its directors, managers, members, officers, employees or independent contractors, other than pursuant to written Contracts in effect prior to the date of this Agreement or in the Ordinary Course of Business or in connection with the performance of this Agreement;

(v) hire any employee or retain any independent contractor, other than in the Ordinary Course of Business and with annual compensation no greater than \$50,000, or in connection with the performance of this Agreement;

(vi) (i) borrow any amount, receive any committed loan facilities or, incur or guarantee any Indebtedness (including contingently as a guarantor or otherwise) or other Liabilities, except for Transaction Expenses and current liabilities incurred in the Ordinary Course of Business or (ii) grant, incur or suffer to exist any Lien, or received any notice of any Lien on any Equity Interests issued by it;

(vii) permit or allow the Shares or any of the assets or properties of the Company to become subjected to any Lien (other than Permitted Liens);

(viii) pay, loan or advance any amount to, or sell, transfer or lease any of its assets to, or enter into any agreement or arrangement with any of its respective Affiliates or Related Parties;

(ix) make any material change in any method of accounting or accounting practice or policy;

(x) acquire by merging or consolidating with, or by purchasing a substantial portion of the assets or capital stock or other equity interest of, or by any other manner, any business or any corporation, partnership, association or other business organization or division thereof or otherwise acquire any assets (other than inventory) other than in the Ordinary Course of Business;

(xi) make or incur capital expenditures (including entering into any capital lease) that are not currently budgeted and that, in the aggregate, are in excess of Ten Thousand Dollars (\$10,000);

(xii) other than in the Ordinary Course of Business, sell, lease, license or otherwise dispose of any of their assets that are material, individually or in the aggregate, to the Business;

(xiii) terminate or allow to be terminated any insurance policy in effect as of the date hereof, or fail to maintain, with financially responsible insurance companies, insurance on tangible assets of the Company and on the Business in such amounts and against such risks and losses as are consistent with past practice;

(xiv) form any Subsidiary or acquire the Equity Interests of any Person;

(xv) commence or settle any material Action, other than the settlement of an Action for amounts less than Ten Thousand Dollars (\$10,000) which only involve the payment of monetary damages and does not impose any equitable remedies upon the Company and there is no admission of fault or any violation of Applicable Law;

(xvi) enter into, amend or make any modification to any Material Contract or Permit, in each case, outside the Ordinary Course of Business;

(xvii) make any material change to any of the cash management practices of the Business, including but not limited to, discontinue the payment of its accounts payable that are payable in the Ordinary Course of Business or deviate from or alter any of its practices, policies or procedures in paying accounts payable or collecting accounts receivable;

26

(xviii) commence, institute, cancel, compromise, assign, waive, release, settle or agree to settle any right or claim (or series of related rights and claims) in excess of Ten Thousand Dollars (\$10,000) or outside the Ordinary Course of Business;

(xix) make, change or revoke any material Tax election, adopt or change any method of Tax accounting, settle or compromise any material Action with respect to Taxes, amend any material Tax Return, consent to any extension or waiver of the statute of limitations for the assessment or collection of any Tax, or take any reporting position with respect to Taxes that is inconsistent with past practice or that is otherwise not in the Ordinary Course of Business, in each case if such action would have the effect of materially increasing the Tax Liability or materially reducing any Tax asset of Buyers or the Company in respect of any Post-Closing Tax Period; or

(xx) terminate any employee, or enter into any employment Contract, written or oral, or modification of the terms of any such existing Contract, or enter into any collective bargaining agreement or relationship, or grant any increase in the base compensation of any of its current or former directors, officers, managers, members, partners, employees or consultants, or payment of or agreement to become obligated to pay any bonus, severance or change of control payments or other consideration of any nature whatsoever (other than salary, commissions or consulting fees) to any of its current or former directors, officers, managers, members, partners, employees or consultants;

interest in the Company;

(xxi) declare, set aside, make or pay any non-cash dividend or other non-cash distribution on or with respect to any Equity Interest or ownership

(xxii) accelerate the collection of or discount any accounts receivable, delay the payment of accounts payable or defer expenses, reduce inventories or otherwise increase cash on hand, except in the ordinary course of business consistent with past practices;

(xxiii) agree, commit or offer (in writing or otherwise) to take any of the actions described in clauses(i) through (xxiii) of this Section 4.4.

4.5 Access to Information. The Company shall afford to the Buyer and its Representatives, reasonable access, upon notice during normal business hours during the period prior to the Closing and during the Exclusivity Period, to all the personnel, independent accountants, properties, books, Contracts, commitments, Tax Returns and records of the Company, and, during such period shall make available to the Buyer any information concerning the Company as Buyer may reasonably request. In particular, but without limitation, from and after the date of this Agreement, the Buyer and its Representatives shall have the right and privilege, subject to compliance with Applicable Law, upon reasonable written request, of entering upon all properties leased or occupied by the Company and of reviewing the books and records of the Company regarding such properties from time to time as needed to make any inspections, evaluations, surveys or tests which Buyers may deem necessary or appropriate. Any visit or investigation by Buyer shall be conducted in such a manner subject to compliance with Applicable Law and as not to interfere unreasonably with the conduct of the Business.

4.6 Approvals and Consents.

(a) On the terms and subject to the conditions of this Agreement, each Party shall use its reasonable best efforts to cause the Closing to occur, including taking all reasonable actions necessary to comply promptly with all Applicable Laws that may be imposed on it or any of its Affiliates with respect to the Closing. Prior to and after the Closing, the Equityholder and Beneficial Owners shall provide information within his possession that is reasonably requested by Buyer regarding the Company and the Business in order to assist Buyer in its filings to regulatory approvals.

(b) In furtherance and not in limitation of the foregoing clause (a), the Company and Buyer shall promptly execute and file, or join in the execution and filing of, any application, notification, or other document that may be necessary in order to obtain the authorization, approval, or consent of any Governmental Authority, whether federal, state, local, or foreign, that may be reasonably required, or that Buyer may reasonably request, in connection with the consummation of the Transactions.

4.7 <u>No Negotiation</u>. Until such time as this Agreement shall be terminated pursuant to <u>Article 7</u>, neither the Company nor any of his Affiliates or Representatives shall directly or indirectly solicit, initiate, encourage, entertain or respond to any inquiries or proposals from, discuss or negotiate with, provide any information to or consider the merits of any inquiries or proposals from any Person (other than Buyer) or otherwise participate in a transaction or potential transaction concerning any sale of any, all or substantially all of the assets of the Company, any direct or indirect sale of equity interests or other securities the Company, or any merger, consolidation or similar transaction involving the Equityholder, any Beneficial Owner or the Company. The Equityholder shall notify Buyer of any such inquiry or proposal, and the material terms thereof within forty-eight (48) hours of receipt or awareness of the same by such Party. As of the date of this Agreement, neither the Company nor any of its Affiliates or Representatives is
involved in or is aware of any activities, discussions or negotiations being conducted prior to the date of this Agreement with any Persons other than Buyer with respect to any of the foregoing.

4.8 Efforts: Regulatory Filings and Consents. Subject to the terms hereof, the Buyer and the Company shall each, as applicable (a) use its reasonable best efforts to take, or cause to be taken, all actions, and do, or cause to be done, and to assist and cooperate with the other Party in doing, all things necessary, proper or advisable to consummate and make effective the Transactions as promptly as practicable; (b) use its reasonable best efforts to make, as promptly as practicable, all necessary filings, and thereafter make any other required submissions, with respect to this Agreement and the Transactions required under any other applicable law; (c) use its reasonable best efforts to obtain, as promptly as practicable, from any Governmental Authority or any other third party any consents, licenses, permits, waivers, approvals, authorizations or orders required to be obtained or made by it or any of its subsidiaries in connection with the authorization, execution and delivery of this Agreement and the consummation of the Transactions; and (d) execute or deliver any additional instruments or documents necessary to consummate the transactions contemplated by, and to fully carry out the purposes of, this Agreement.

4.9 <u>Non-Competition; Non-Solicitation; Non-Disparagement</u>.

(a) In further consideration for the Buyer's payment of the amounts to be paid to the Equityholder and Beneficial Owners under this Agreement and the Transaction Documents, and in order to protect the value of the Company and the business acquired by the Buyer under this Agreement, each Beneficial Owner and Equityholder hereby agrees, for itself and on behalf of its owners and trustees, as applicable, that during the period commencing on the Closing Date and ending on the fifth (5th) anniversary of the Closing Date (the "<u>Restricted Period</u>"), each Beneficial Owner and Equityholder shall not, and shall cause its respective Affiliates not to, directly or indirectly, acquire, engage or participate in (including by lending any Beneficial Owner's name, providing financing, or owning or assisting another in), or hold any economic, financial or other interest (including equity or debt securities) in, act as a stockholder, partner, consultant, agent, employee, co-venturer, member or representative of, render any services to, or otherwise operate or hold an interest (whether an equity interest or otherwise) or enter the employ of, or render any services to, in any Person engaged or seeking to engage in (1) any business the same or similar to the Protected Business within the State of California (the "<u>Restricted Territory</u>"), or (2) the provision of business, management, marketing or other support services to any medical or other healthcare practice anywhere within the Restricted Territory; <u>provided</u>, that nothing contained herein shall be construed or prohibit the any Beneficial Owner or the Equityholder from purchasing up to an aggregate of two percent (2%) of any class of the outstanding voting securities of any other Entity whose securities are listed on a national securities exchange, so long as each Beneficial Owner, the Equityholder and their respective Affiliates do not participate in any way in the management or control of such Person.

(b) In further consideration for the Buyer's payment of the amounts to be paid to the Equityholder and Beneficial Owners under this Agreement and the Transaction Documents, and in order to protect the value of the Company and the Protected Business, the Equityholder and Beneficial Owners hereby agree that during the Restricted Period, such Person shall not, directly or indirectly, hire, solicit, interfere with, induce or attempt to hire, solicit, interfere with or induce any (i) employee or contractor of the Protected Business, (ii) business relationship of the Protected Business to divert, reduce, adversely modify or terminate its business relationship with Buyer, its Subsidiaries or Affiliates (collectively, the "Buyer Parties") or the Company or (iiv) planned or contemplated potential acquisition or investment candidates of any Buyer Parties or any Company.

(c) The Equityholder and each Beneficial Owner hereby agrees that during the Restricted Period, such Person shall not make any negative comments or otherwise disparage the Company, Buyer Party or any of their respective Affiliates. Nothing herein or otherwise shall preclude Equityholder or any Beneficial Owner from making truthful statements that are necessary to comply with Applicable Law or Governmental Authority, or to defend or enforce such Person's or the Company's rights under this Agreement, any Transaction Document or any other agreement between or among the Parties.

(d) The Equityholder and each Beneficial Owner understands that the restrictions set forth in this<u>Section 4.9</u> are intended to protect the interests of the Company and the Buyer Parties in the Company's proprietary information, goodwill and established employee, customer, supplier, consultant and vendor relationships, and agree that such restrictions are reasonable and appropriate for this purpose. The Equityholder and each Beneficial Owner further acknowledges and agrees that the time, scope, geographic area and other provisions of this <u>Section 4.9</u> have been specifically negotiated by sophisticated parties and absent the Equityholder's and each Beneficial Owner's agreement to and compliance with the restrictions set forth in this <u>Section 4.9</u> the Buyer would not have entered into the transactions contemplated by this Agreement or the Transaction Documents. The Restricted Period shall be extended with respect to the Equityholder and each Beneficial Owner by each day that such Person or its Affiliates is in breach of this <u>Section 4.9</u>. The Equityholder's and each Beneficial Owner's obligations under this <u>Section 4.9</u> are independent of any other obligation of the Company, Buyer Party or their respective Affiliates has to the Equityholder, including any such obligation under this <u>Agreement</u>.

29

(e) The Parties expressly agree that the subject matter, length of time, geographical scope, and range of activities, as applicable, contained in this <u>Section 4.9</u> are reasonable in light of the circumstances as they exist on the date upon which this Agreement has been executed. However, should a determination nonetheless be made by a court of competent jurisdiction at a later date that the character, duration or geographical scope of the restrictive covenants set forth in this <u>Section 4.9</u> is unreasonable in light of the circumstances as they then exist, then it is the intention and the agreement of the Parties hereto that this Agreement shall be construed by the court in such a manner as to impose only those restrictions on the conduct of the Equityholder and each Beneficial Owner that may be enforceable under Applicable Law, to the fullest extent of such enforceability to assure the Buyer Parties of the intended benefit of this Agreement. If, in any judicial proceeding, a court shall refuse to enforce all of the separate covenants deemed included herein because, taken together, they are more extensive than necessary to assure the Buyer Parties of the intended benefit of this Agreement, it is expressly understood and agreed among the Parties hereto that those of such covenants that, if eliminated, would permit the remaining separate covenants to be enforced in such proceeding shall, for the purpose of such proceeding, be deemed eliminated from the provisions hereof.

4.10 <u>Termination of 401(k) Plan</u>. The Company shall take all actions necessary and appropriate (including fully vesting participants in their account balances) to terminate the Company 401(k) Plan, effective no later than the day immediately preceding the Closing Date. The form of the resolutions terminating the Company 401(k) Plan shall be reasonably satisfactory to the Buyer, which shall be afforded a reasonable opportunity to review such resolutions. The provisions of this <u>Section 4.10</u> are solely for the benefit of the parties to this Agreement, and no current or former employee of the Company shall be regarded for any purpose as a third-party beneficiary of this Agreement. In no event shall the terms of this <u>Section 4.10</u> be deemed to (a) establish, amend or modify any Employee Benefit Plan, (b) alter or limit the ability of the Company or Buyer to amend, modify or terminate any Employee Benefit Plan or any other benefit or employment plan, program or Contract, or (c) confer upon any current or former employee, officer, director or consultant any right to employment or continued employment or continued service with the Company, Buyer, or any of their respective Affiliates.

4.12 <u>PP&E</u>. On or immediately following the date hereof, the Corporation will purchase all of the PP&E of AHMS.

ARTICLE 5 TAX MATTERS

5.1 <u>Tax Elections</u>. Without the prior written consent of Buyer, not to be unreasonably withheld, conditioned or delayed, the Equityholder shall not, to the extent it may affect, or relate to, the Company, make, change or rescind any material Tax election, or amend any Tax Return. This covenant is in addition to that of the Company under <u>Section 4.4(a)</u>.

5.2 <u>Pre-Closing Tax Period Returns</u>. The Equityholder shall prepare, or cause to be prepared, at the Equityholder's sole expense, all Tax Returns required to be filed by the Company after the Closing Date with respect to a Pre-Closing Tax Period, other than Tax Returns for Straddle Periods, and Buyer shall cause the Company to execute and file timely such Tax Returns as so prepared.

5.3 <u>Straddle Period Tax Returns</u> Buyer shall prepare, or cause to be prepared, all Tax Returns required to be filed with respect to a Straddle Period, and Buyer and the Equityholder shall pay, when due, all Taxes reflected as due and payable on such Tax Returns to the extent allocated to such Party under <u>Section 5.5</u>.

5.4 <u>Manner of Preparation</u>. Any such Tax Return referred to in Section 5.2 or Section 5.3 shall be prepared in a manner consistent with past practice (unless otherwise required by Applicable Law) and without a change of any election or any accounting method and shall be submitted by the Party preparing the Tax Return to the other Party (together with schedules, statements and, to the extent requested by Buyer or the Equityholder, as applicable, supporting documentation) at least forty-five (45) days prior to the due date (including extensions) of such Tax Return. If the Party reviewing the Tax Return objects to any item on any such Tax Return, it shall, within fifteen (15) days after delivery of such Tax Return, notify the Party that prepared the Tax Return, in writing, that it so objects, specifying with particularity any such item and stating the specific factual or legal basis for any such objection. If a notice of objection shall be duly delivered, Buyer and the Equityholder shall negotiate in good faith and use their reasonable best efforts to resolve such items. If Buyer and the Equityholder are unable to reach such agreement within ten (10) days after receipt by Buyer of such notice, the disputed items shall be resolved by a Neutral Accounting Firm selected by Buyer (the "Accounting Referee") and any determination by the Accounting Referee shall resolve any disputed items within twenty (20) days of having the item referred to it pursuant to such procedures as it may require. If the Accounting Referee's resolution. The costs, fees and expenses of the Accounting Referee shall be filed as prepared and the Equityholder, on the other hand, as determined by the Accounting Referee that is unsuccessfully disputed by Buyer, on the one hand, or the Equityholder, on the other hand, as determined by the Accounting Referee that is unsuccessfully disputed by Buyer, on the one hand, or the Equityholder, on the other hand, as determined by the Accounting Referee that is unsuccessfully disputed by Buyer, on the one hand

31

5.5 <u>Apportionment</u>. In the case of Taxes that are payable with respect to a Straddle Period, the portion of any such Taxes that are treated as Taxes of the Company for any Pre-Closing Tax Period shall be:

(a) in the case of Taxes based upon, or related to, income or receipts, deemed equal to the amount which would be payable if the taxable year ended with the Closing Date; and

(b) in the case of other Taxes, deemed to be the amount of such Taxes for the entire period multiplied by a fraction the numerator of which is the number of days in the period ending on the Closing Date and the denominator of which is the number of days in the entire period.

5.6 Tax Indemnification

(a) The Beneficial Owners shall indemnify the Buyer Indemnitees and hold each of them harmless from and against any Damages attributable to (i) all Taxes (or the non-payment thereof) of the Company for all taxable periods ending on or before the Closing Date and the portion through the end of the Closing Date for any Pre-Closing Tax Period; (ii) all Taxes of any member of an affiliated, consolidated, combined or unitary group of which the Company (or any predecessor thereof) is or was a member on or prior to the Closing Date, including pursuant to Treasury Regulation section 1.1502-6 or any analogous or similar state, local or non-U.S. law or regulation; (iii) any and all Taxes of any Person (other than the Company) imposed on the Company as a transferee or successor, by contract or pursuant to any law, which Taxes relate to an event or transaction occurring on or before the Closing Date; (iv) all Taxes imposed on the Company as a result of an election made by Company under Code section 108(i); (v) all Taxes imposed on the Company being deemed to be a "responsible person" under Code section 6672 (or any corresponding provision of state or local law); (vi) any and all employment Taxes imposed with respect to any compensatory payments associated with the transactions contemplated by this Agreement; and (vii) third party costs and expenses associated with preparing any Tax Return for Pre-Closing Tax Periods.

(b) The Beneficial Owners shall reimburse Buyer for any Taxes of the Company that are the responsibility of Beneficial Owners pursuant to this Section 5.6 within fifteen (15) Business Days after payment of such Taxes by the Buyer or the Company.

(c) The indemnification under this Section 5.6 shall not be subject to any basket, cap, other limitations set forth in Article 8 or elsewhere in this

Agreement.

5.7 <u>Refunds.</u> Any credits or refunds of Taxes of the Company with respect to Pre-Closing Tax Periods shall belong to the Equityholder and shall be paid to the Equityholder for distribution to the Equityholder promptly upon actual receipt of such refund in cash or application of such credit actually to reduce other Taxes due and owing by Buyer. The amount of any refund or credit which is required to be paid to the Equityholder shall be reduced by the amount of any Taxes on such refund or credit and reasonable out-of-pocket expenses that Buyer or any Affiliates thereof incur (or will incur) with respect to such refund or credit.

5.8 <u>Contests</u>. Buyer shall promptly notify Equityholder following receipt of any notice of audit or other proceeding relating to any Tax Return that includes a Pre-Closing Tax Period (the "<u>Prior Period Returns</u>"). The Equityholder shall control any and all audits or other proceedings and litigation (a <u>'Tax Contest</u>") relating to any Prior Period Return, including the filing of an amended Tax Return, other than Prior Period Returns relating to any Straddle Period. With respect to any Tax Contest that it controls, the Equityholder (i) shall give prompt notice to Buyer of any Tax adjustment proposed in writing pursuant to such Tax Contest, (ii) shall afford Buyer the opportunity to participate in the conduct of such Tax Contest, including without limitation the right to participate in conferences with Tax Authorities and submit pertinent material in support of Buyer's position, and (iii) shall not settle such Tax Contests relating to a Tax Return for a Straddle Period, including the filing of an amended Tax Return for such Straddle Period; provided that neither Equityholder nor Buyer shall settle or compromise any Tax Contest relating to a Tax Return for a Straddle Period without the consent of the other Party, which consent shall not be unreasonably withheld, conditioned or delayed. 5.9 <u>Cooperation and Exchange of Information</u> Equityholder and Buyer shall provide each other with such cooperation and information as either of them reasonably may request of the other in filing any Tax Return pursuant to this <u>Article 5</u> or in connection with any audit or other proceeding in respect of Taxes of the Company. Such cooperation and information shall include providing copies of relevant Tax Returns or portions thereof, together with accompanying schedules, related work papers and documents relating to rulings or other determinations by Tax Authorities, making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder, and the provision of any powers of attorney or similar authorizations, or other designations, as may be necessary to effectuate the intent of this Article 5. The Equityholder and Buyer shall retain, or cause to be retained, all Tax Returns, schedules and work papers, records and other documents in their possession relating to Tax matters of the Company for any taxable period beginning before the Closing Date until the expiration of the statute of the taxable periods to which such Tax Returns and other documents relate, without regard to extensions except to the extent notified by the other Party in writing of such extensions for the respective Tax periods. Prior to transferring, destroying or discarding any Tax Returns, schedules and work papers, records and other documents in its possession relating to Tax matters of the Company for any taxable period beginning before the Closing Date, the Equityholder or Buyer (as the case may be) shall provide the other Party with reasonable written notice and offer the other Party the opportunity to take custody of such materials.

5.10 <u>Post-Closing Actions</u>. None of Buyer, the Company, or any of their Affiliates shall (a) make any election under Section 338 or Section 336 of the Code (or any similar election pursuant to any provision of federal, state or local Applicable Law) with respect to the Company in connection with the Transactions, except as otherwise expressly provided herein, (b) amend any Tax Return or change or revoke any election, relating to a Tax period (or portion thereof) ending on or before the Closing Date, (c) initiate any voluntary disclosure or similar proceeding with any Tax Authority relating to a Tax period (or portion thereof) ending on or before the Closing Date, or (d) make any other Tax election or take any other action that has the effect of increasing the amount of Taxes for which Equityholder may be liable under the terms of this Agreement or under applicable law, in each case without the prior written consent of the Equityholder.

33

5.11 <u>Treatment of Indemnification Payments</u>. All indemnification payments made under this Agreement shall be treated for Tax purposes by the Parties as an adjustment to the Purchase Price, unless otherwise required by Applicable Law.

5.12 <u>Consistent Treatment</u>. Buyer and Equityholder shall, unless otherwise required by a "determination" within the meaning of Section 1313 of the Code (i) treat and report the transactions described above in all respects consistently with the provisions of this Agreement for purposes of any Tax; and (ii) not take any actions or positions inconsistent with the obligations of the parties set forth herein.

5.13 <u>Transfer Taxes</u>. Notwithstanding anything to the contrary herein, any transfer, sales, stamp, use documentary, property, or similar Taxes applicable to, imposed upon or arising out of the Transactions (such Taxes, "<u>Transfer Taxes</u>") will be borne entirely by Buyer. Buyer and Equityholder shall cooperate in preparing and filing all filings, Tax Returns, reports and forms as necessary or appropriate to comply with the provisions of all Applicable Law in connection with the payment of such Transfer Taxes, and shall cooperate in good faith to minimize the amount of any such Transfer Taxes payable in connection therewith, provided that Buyers shall be responsible for preparing all such Tax Returns and related forms or filings, shall provide each such Tax Return or form or filing to the Equityholder for review and approval at least five (5) Business Days prior to the due date thereof, and Equityholder shall not have any obligation to execute any such form or filing unless Buyer revises such form or filing for any reasonable comments provided by the Equityholder in respect thereof.

5.14 <u>Overlap</u>. To the extent that any obligation or responsibility pursuant to <u>Article 8</u> may overlap with an obligation or responsibility pursuant to this Article 5, the provisions of this Article 5 shall govern.

ARTICLE 6

Conditions to Closing

The conditions to the obligations of the Equityholder and the Buyer with respect to the Closing shall be set forth in Sections 6.1 and 6.2, and 6.3.

6.1 <u>Conditions to the Obligations of All Parties with Respect to the Closing</u>. The obligations of each Party to effect the closing of the Transactions contemplated to take place at the Closing shall be subject to the fulfillment, at or prior to the Closing, of each of the following conditions: (a) the closing conditions to the CFC Purchase Agreement shall have been satisfied or waived, and both the Initial Closing and Second Closing shall have occurred; (b) no Applicable Law or Judgment enacted, entered, promulgated, enforced or issued by any Governmental Authority or other legal restraint or prohibition preventing the consummation of the Transactions contemplated to take place at the Closing shall be in effect; and (c) no Action shall be pending or threatened which (i) prevents or challenges consummation of the Transactions contemplated to take place at the Closing; or (ii) causes such Transactions to be rescinded following consummation.

34

6.2 <u>Closing Conditions to Obligations of Buyer</u>. The obligations of Buyer to effect the closing of the Transactions contemplated to take place at the Closing shall be subject to the fulfillment or Buyer's waiver, at or prior to the Closing, of each of the following conditions:

(a) The representations and warranties of the Beneficial Owners, Equityholder and the Company set forth in<u>Article 2</u>, other than the Fundamental Representations, shall be true and correct in all material respects as of the Closing Date as though made on and as of the Closing Date, except to the extent that such agreements or covenants are qualified by the term "material," or contain terms such as "Material Adverse Effect," in which case the Company and the Equityholder shall have performed and complied with all of such covenants (as so written, including the term "material," or "Material Adverse Effect,") in all respects through the Closing, except that representations and warranties that are made as of a specific date need be true and correct only as of such date. The Fundamental Representations shall be true and correct in all respects as of the Closing Date as though made on and as of the Closing Date, except that the Fundamental Representations that are made as of a specific date need be true and correct only as of such date.

(b) The Company, the Beneficial Owners, and the Equityholder shall have duly performed and complied in all material respects with all agreements and covenants required by this Agreement and each of the other Transaction Documents to be performed or complied with by them prior to or on the Closing Date, except to the extent that such agreements or covenants are qualified by the term "material," or contain terms such as "Material Adverse Effect," in which case the Company, Beneficial Owners and the Equityholder shall have performed and complied with all of such covenants (as so written, including the term "material," or "Material Adverse Effect,") in all respects through the Closing.

- (c) From the date of this Agreement, there shall not have occurred any Material Adverse Effect.
- (d) At or prior to the Closing, the Equityholder shall cause to be delivered to the Buyer:
 - (i) the Employment Agreement, duly executed by Ron Brandt;

- (ii) the Management Services Agreements, each duly executed by the Company;
- (iii) the Note, duly executed by the Equityholder and each Beneficial Owner;
- (iv) the Stockholders Agreement, duly executed by the Equityholder and each Beneficial Owner;

35

(v) evidence that the offer letters described in <u>Section 4.11</u> have been delivered, together with all other obligations contemplated thereunder, in form and substance reasonably satisfactory to the Buyer;

(vi) certificates, if any, representing all of the Purchased Shares, duly endorsed in blank or accompanied by transfer powers;

(vii) a W-9 executed by the Equityholder; and

(viii) resignations of the directors, managers and officers of the Company;

(ix) a certificate, dated as of the Closing Date and signed by a duly authorized officer of the Company, that each of the conditions set forth in Section 6.2(a), Section 6.2(b) and Section 6.2(c) have been satisfied;

(x) a certificate of the Secretary (or equivalent officer) of the Company certifying in their corporate capacity (i) that attached thereto are true and complete copies of all resolutions adopted by the board of directors of the Company authorizing the execution, delivery and performance of this Agreement and the Transaction Documents and the consummation of the Transactions, (ii) that all such resolutions are in full force and effect and are all the resolutions adopted in connection with the Transactions, (iii) the names and signatures of the officers of the Company authorized to sign this Agreement and the Transaction Documents, (iv) that there have been no amendments to the incorporation or formation documents since the date such documents were delivered pursuant to Section 6.2(d)(xi), and (v) that attached thereto are true and complete copies of all Governing Documents of the Company;

State of California;

(xi)

(xii) evidence, reasonably satisfactory to the Buyer, of termination of the Company 401(k) Plan in accordance with<u>Section 4.10;</u>

the incorporation document and all amendments thereto of the Company, duly certified as of a recent date by the Secretary of State of the

(xiii) a good standing certificate (or its equivalent) as of a recent date from the secretary of state of the State of California; and

(xiv) such other documents or instruments as Buyer reasonably requests and are reasonably necessary to consummate the Transactions.

(e) The (i) third party consents, notices and filings and (ii) notices, filings, consents and authorizations identified by Buyer shall have been duly delivered and obtained (collectively, the "Required Consents").

(f) The Equityholder shall have delivered to Buyer evidence of termination of the Contracts between Equityholder, on the one hand, and any of its Affiliates, Related Parties, or any Affiliate of any Related Party, on the other hand, as identified by Buyer.

(g) All actions to be taken by the Equityholder, Beneficial Owners and Company in connection with consummation of the transactions contemplated hereby and all certificates, opinions, instruments, and other documents required to effect the transactions contemplated hereby will be reasonably satisfactory in form and substance to the Buyer.

36

(h) The Continuing Employees (as defined in the CFC Purchase Agreement) shall have received offer letters from the Company, in form and substance reasonably satisfactory to Buyer.

6.3 <u>Closing Conditions to Obligations of the Equityholder</u>. The obligations of the Equityholder and Beneficial Owners to consummate the Transactions shall be subject to the fulfillment or the Equityholder's waiver, at or prior to the Closing, of each of the following conditions:

(a) The representations and warranties set forth in <u>Article 3</u> shall be true and correct in all respects as of the Closing Date as though made on and as of the Closing Date (disregarding all qualifications and exceptions contained therein relating to materiality, Material Adverse Effect or similar qualifiers), except: (i) that representations and warranties that are made as of a specific date need be true and correct only as of such date and (ii) for breaches and inaccuracies the effect of which would not, individually or in the aggregate, have a Material Adverse Effect.

(b) The Buyer shall have duly performed and complied in all material respects with all agreements and covenants required by this Agreement and each of the other Transaction Documents to be performed or complied with by them prior to or on the Closing Date; provided, that, the condition contained in this section shall only be deemed to be unsatisfied if such failure to have performed and complied in all material respects with all obligations and covenants required by this Agreement is not cured within ten (10) Business Days, to the extent curable.

(c) At or prior to the Closing, the Buyer shall cause to be delivered to the Equityholder:

- (i) the Management Services Agreements, duly executed by the Buyer and Metropolitan IPA, as applicable;
- (ii) the Note, duly executed by Apollo Care;
- (iii) the Stockholders Agreement, duly executed by the Buyer.
- (d) At or prior to Closing, the Buyer shall make the payments described in <u>Section 1.1(b)</u> above.

7.1 <u>Termination of the Agreement</u>. Notwithstanding anything to the contrary in this Agreement, this Agreement may be terminated and the Transactions abandoned at any time prior to the Closing:

(a) by the mutual written consent of the Equityholder, Beneficial Owners and Buyer;

37

(b) by Buyer, in the event of any material breach by any Beneficial Owner's, the Equityholder's or Company's covenants, obligations, agreements, representations or warranties contained herein and the failure of the Equityholder, Beneficial Owners or Company to cure, if curable, such breach within fifteen (15) days after receipt of written notice from Buyer requesting such breach to be cured; provided, however, that the right to terminate this Agreement under this Section 7.1(b) will not be available to Buyer if Buyer is in material breach of any of its covenants, obligations or agreements under this Agreement;

(c) by the Equityholder in the event of any material breach by Buyer of any of Buyer's covenants, obligations, agreements, representations or warranties contained herein and the failure of Buyer to cure, if curable, such breach within fifteen (15) days after receipt of written notice from the Equityholder requesting such breach to be cured; <u>provided</u>, <u>however</u>, that the right to terminate this Agreement under this <u>Section 7.1(c)</u> will not be available to the Equityholder if Equityholder or the Company is in material breach of any of its respective covenants, obligations or agreements under this Agreement;

(d) by the Equityholder or Buyer if the Closing does not occur on or prior to the date that is July 24, 2024 (such date, the <u>Outside Date</u>"); <u>provided</u>, <u>however</u>, that the right to terminate this Agreement under this <u>Section 7.1(d)</u> will not be available to any Party whose material breach of any of its covenants, obligations or agreements under this Agreement has been the cause of or resulted in the failure of the Closing to occur on or prior to such date; or

(e) by Buyer or the Equityholder if there will be in effect a final, non-appealable Judgment of a court of competent jurisdiction in effect precluding consummation of the Transactions; provided, however, that the right to terminate this Agreement under this <u>Section 7.1(e)</u> will not be available to any Party whose failure to fulfill any of its covenants, obligations or agreements under this Agreement has been the cause of or resulted in such Judgment.

7.2 Effect of Termination. If this Agreement is terminated for any reason, the provisions of this Section 7.2 and Article 10 shall remain in full force and effect. If this Agreement is terminated as provided in Section 7.1(a), this Agreement shall forthwith become void (except as stated in the immediately preceding sentence) and there shall be no Liability or obligation hereunder on the part of any Party hereto or their respective Representatives. If this Agreement is terminated as provided in subsections (a), (b), (c) or (e) of Section 7.1, such termination shall be without prejudice to any rights that the terminating Party may have against any breaching Party or any other Person under the terms of this Agreement or otherwise.

ARTICLE 8 Indemnification

8.1 Survival.

(a) Each representation and warranty contained herein shall survive the Closing Date until, and will expire and be of no force and effect on the two (2)year anniversary hereof, except that the representations and warranties in Sections 2.1 (Organization and Good Standing; Subsidiaries); 2.2 (Authority, Power and Enforceability); 2.3 (Capitalization); 2.4 (Non-Contravention; Consents); 2.10 (Legal Proceedings; Judgments); 2.11 (Compliance with Applicable Laws); 2.13 (Healthcare Regulatory Matters); 2.17 (Taxes); 2.17 (Employee Benefit Matters); 2.20 (Brokers) (collectively, the "Fundamental Representations") shall survive the Closing until the date that is sixty (60) days after the expiration of the applicable statute of limitations with respect to the subject matter of any such representation or warranty. All covenants and agreements contained in this Agreement and the Transaction Documents, and the certifications underlying all certificates delivered hereunder shall survive the Closing for the period contemplated by its terms.

38

(b) No Indemnified Person shall be entitled to make any claim in respect of any representation, warranty, covenant or agreement after the expiration of its applicable survival date as set forth in <u>Section 8.1(a)</u> (each, a "<u>Survival Date</u>"), except that any bona fide claim initiated by an Indemnified Person prior to the expiration of the applicable Survival Date in accordance with the provisions hereof shall survive until it is settled or resolved pursuant to this Agreement to the extent that an Indemnified Person provides written notice of such breach or inaccuracy (which notice shall describe the applicable breach or inaccuracy in reasonable detail) to the party to provide indemnity prior to the applicable Survival Date.

(c) The representations and warranties of each of the Parties set forth in this Agreement are material and shall be deemed to have been relied upon by the Party or Parties to whom they are made and shall survive as set forth above, regardless of any investigation on the part of such Party or its Representatives (each Party reserving all rights hereunder in connection with any breach or alleged breach).

8.2 Indemnification by the Equityholder and Beneficial Owners

(a) Subject to the other terms and conditions of this <u>Article 8</u>, the Equityholder and each Beneficial Owner shall, jointly and severally, indemnify and defend the Buyer and after the Closing Date, the Company, and their respective Affiliates and their respective successors, assigns, heirs and Representatives (collectively, the "<u>Buyer Indemnitees</u>") against, and shall hold each of them harmless from and against, and shall pay and reimburse each of them for, any and all Damages incurred or sustained by, or imposed upon, the Buyer Indemnitees based upon, arising out of, with respect to or by reason of:

(i) any breach or inaccuracy of any of the representations or warranties concerning the Equityholder, each Beneficial Owner or the Company in this Agreement or in any of the other Transaction Documents (including any certificated delivered by or on behalf of such Person in connection herewith and therewith);

(ii) any failure to perform or breach of any agreement or covenant or obligation concerning the Equityholder, each Beneficial Owner or the Company contained in this Agreement or in any of the other Transaction Documents (including any certificated delivered by or on behalf of such Person in connection herewith and therewith);

(iii) any Indebtedness relating to Equityholder, any Beneficial Owner or the Company not discharged on or prior to the Closing;

(iv) any Transaction Expenses of the Equityholder, any Beneficial Owner or the Company not discharged on or prior to the Closing;

(v) to the extent cash amount in the Company's bank account is less than Schedule 8.2

(vi) any Action by a Governmental Authority outstanding or arising from facts or circumstances that occurred prior to the Closing Date (whether known or unknown to the Company, any Beneficial Owner or the Equityholder at the Closing);

(vii) any fraud committed by or on behalf of the Equityholder, any Beneficial Owner the Company or their Affiliates;

(viii) any allegation, claim, charge or threatened claim or charge by any current or former employee or independent contractor of the Company involving or relating to facts or circumstances that occurred prior to the Closing Date.

(b) All of the representations and warranties set forth in this Agreement or any certificate or schedule that are so qualified as to "material," "materiality," "material respects," "Material Adverse Effect" or words of similar import or effect shall be deemed to have been made without such qualification for purposes of determining: (i) whether a breach of such representation or warranty has occurred and (ii) the amount of Damages resulting from, arising out of or relating to any such breach of such representation or warranty.

8.3 <u>Indemnification by the Buyer</u>. Subject to the other terms and conditions of this <u>Article 8</u>, Buyer shall indemnify and defend the Equityholder, Beneficial Owners and such Person's successors, assigns, heirs and Representatives (collectively, the "<u>Equityholder Indemnitees</u>") against, and shall hold each of them harmless from and against, and shall pay and reimburse each of them for, any and all Damages incurred or sustained by, or imposed upon, the Equityholder Indemnitees based upon, arising out of, with respect to or by reason of:

(a) any breach or inaccuracy of any of the representations or warranties made by Buyer in this Agreement or in any of the other Transaction Documents (including any certificated delivered by or on behalf of such Person in connection herewith and therewith); or

(b) any breach of any covenant or obligation of Buyer contained in this Agreement or in any of the other Transaction Documents (including any certificated delivered by or on behalf of such Person in connection herewith and therewith).

8.4 <u>Certain Limitations</u>. The indemnification provided for in<u>Section 8.2</u> and <u>Section 8.3</u> shall be subject to the following additional limitations:

(a) Notwithstanding anything to the contrary herein, any claims for indemnification by a Buyer Indemnitee with respect to Fundamental Representations will be recovered by any or a combination of the following: (i) by recovering such Damages directly from the Equityholder or any Beneficial Owner, or (2) by offset against any amounts payable by Buyer or any of its Affiliates to the Equityholder or any Beneficial Owner under this Agreement and any Transaction Document. Without limiting the generality of the foregoing, Buyer shall have the right to set-off any amount which it or any Buyer Indemnified Party is entitled to receive pursuant to this Article 8 against amounts otherwise payable by the Buyer or any Affiliate to the Company, Equityholder or any Beneficial Owner pursuant to this Agreement, the CFC Purchase Agreement, any document delivered in connection herewith or therewith or any other agreement, arrangement or understanding. Neither the exercise or failure to exercise such right of set-off or to give notice of a claim will constitute an election of remedies or limit any Buyer Indemnified Party in any manner in the enforcement of any other remedies that may be available to it.

40

(b) For all purposes of this <u>Article 8</u>, "Damages" shall be net of any amounts paid or payable to an Indemnified Person under any insurance policy or Contract in connection with the facts giving rise to the right of indemnification hereunder; <u>provided</u>, <u>however</u>, that the amount deemed to be paid under such insurance policies shall be net of the deductible for such policies and net of any costs incurred to recover such amounts. In calculating any Damage, there shall be deducted any Tax benefit, credit or refund to which the applicable Indemnified Person actually realizes or receives as a result of such Damage.

(c) If the amount to be netted hereunder from any indemnification payment required hereunder is determined after payment by an Indemnifying Person to an Indemnified Person of any amount otherwise required to be paid as indemnification pursuant hereto, the Indemnified Person shall repay, promptly after such determination, any amount that the Indemnifying Person would not have had to pay pursuant hereto had such determination been made at the time of such payment.

(d) Each Indemnified Person shall use its reasonable commercial efforts to mitigate any indemnifiable Damage.

(e) Upon making any payment to an Indemnified Person for any indemnification claim pursuant to this <u>Article 8</u>, the Indemnifying Person shall be subrogated, to the extent of such payment, to any rights which the Indemnified Person may have against any third parties with respect to the subject matter underlying such indemnification claim and the Indemnified Person shall assign any such rights to the Indemnifying Person.

8.5 <u>Indemnification Procedures</u>. The Party making a claim under this <u>Article 8</u> is referred to as the "<u>Indemnified Person</u>", and the Party against whom such claims are asserted under this <u>Article 8</u> is referred to as the "<u>Indemnifying Person</u>".

(a) <u>Third Party Claims</u>. If, on or after the Closing Date, any Indemnified Person receives notice of the assertion or commencement of any Action made or brought by any Person who is not a Party or an Affiliate of a Party or a Representative of the foregoing (a "<u>Third Party Claim</u>") against such Indemnified Person with respect to which the Indemnifying Person is obligated to provide indemnification under this Agreement, the Indemnified Person shall give the Indemnifying Person reasonably prompt written notice thereof, but in any event not later than thirty (30) days after receipt of such notice of such Third Party Claim. The failure to give such prompt written notice shall not, however, relieve the Indemnifying Person of its indemnification obligations, except and only to the extent that the Indemnifying Person forfeits rights or defenses by reason of such failure. Subject in all respects to the R&W Insurance Policy, if an Indemnified Person gives notice to the Indemnifying Person pursuant of the assertion of a Third Party Claim, then the Indemnifying Person shall be entitled to participate in the defense of such Third Party Claim and, to the extent that it wishes to assume the defense of such Third Party Claim, with counsel of its choosing. After notice from the Indemnifying Person of its election to assume the defense of such Third Party Claim, the Indemnifying Person shall not, so long as it diligently conducts such defense, be liable to the Indemnified Person under this <u>Article 8</u> for any fees of other counsel or any other expenses with respect to the defense of such Third Party Claim. The assumption of the defense of a chird Party Claim shall not be an admission as to the liability of the Indemnifying Person in respect of such Third Party Claim. The assumption of the defense of a Chird Party Claim shall not be an admission as to the liability of the Indemnifying Person in respect of such Third Party Claim. The Equityholder, Beneficial Owners and Buyer shall cooperate with each other in all reasonable res

case may be, shall keep the other Person fully informed of the status of such Third Party Claim and any related Actions at all stages thereof where such Person is not represented by its own counsel. The Parties agree to provide reasonable access to the other Parties to such documents and information as may be reasonably requested in connection with the defense, negotiation or settlement of any such Third Party Claim; <u>provided</u>, <u>however</u>, that the Parties shall cooperate in such a manner as to preserve in full (to the extent possible) the confidential information and the attorney-client and work-product privileges of the other Parties. In connection therewith, each Party agrees that: (i) it will use Commercially Reasonable Efforts, in respect of any Third Party Claim in which it has assumed or participated in the defense, to avoid production of confidential information (consistent with Applicable Law and rules of procedure); and (ii) all communications between any Party and counsel responsible for or participating in the defense of any Third Party Claim shall, to the extent possible, be made so as to preserve any applicable attorney-client or work-product privilege.

(c) <u>Settlement of Third Party Claims</u>. If a firm offer is made to settle a Third Party Claim without leading to liability or the creation of a financial or other obligation on the part of the Indemnified Person and provides, in customary form, for the unconditional release of each Indemnified Person from all liabilities and obligations in connection with such Third Party Claim and the Indemnifying Person desires to accept and agree to such offer, the Indemnifying Person shall give written notice to that effect to the Indemnified Person. If the Indemnified Person has assumed the defense pursuant to <u>Section 8.5(a)</u>, it shall not agree to any settlement without the written consent of the Indemnifying Person (which consent shall not be unreasonably withheld or delayed).

(d) Direct Claims. Any Action by an Indemnified Person on account of any Damages which does not result from a Third Party Claim (a 'Direct Claim') shall be asserted by the Indemnified Person giving the Indemnifying Person reasonably prompt written notice thereof, but in any event not later than sixty (60) days after the Indemnified Person becomes aware of such Direct Claim. The failure to give such prompt written notice shall not, however, relieve the Indemnifying Person of its indemnification obligations, except and only to the extent that the Indemnifying Person forfeits rights or defenses by reason of such failure. Such notice by the Indemnified Person shall describe the Direct Claim in reasonable detail, shall include copies of all material written evidence thereof and shall indicate the estimated amount, if reasonably practicable, of the loss that has been or may be sustained by the Indemnified Person. The Indemnifying Person shall have sixty (60) days after its receipt of such notice to respond in writing to such Direct Claim, and whether and to what extent any amount is payable in respect of the Direct Claim and the Indemnified Person shall allow the Indemnifying Person's investigation by giving such information and assistance (including access to the Companies' premises and personnel and the right to examine and copy any accounts, documents or records) as the Indemnifying Person or any of its professional advisors may reasonably request. If the Indemnifying Person shall be free to pursue such remedies as may be available to the Indemnifying Person shall be deemed to have rejected such claim, in which case the Indemnified Person shall be free to pursue such remedies as may be available to the Indemnifying Person on the terms and subject to the provisions of this Agreement. If there is a dispute regarding a Direct Claim the dispute shall be handled in accordance with terms and conditions set forth in this Agreement.

42

8.6 <u>No Contribution</u>. Anything to the contrary herein notwithstanding, none of the Beneficial Owners or the Equityholder shall have any right to seek any indemnification or contribution from or remedy against the Company whether arising prior to or after the Closing Date in respect of any breach of any representation or warranty by the Company or the failure of the Company to comply with any covenant or agreement to be performed by the Company on or prior to the Closing Date and each of the Beneficial Owners and the Equityholder hereby waives any such claim they may have against the Company with respect thereto whether at law, in equity or otherwise.

ARTICLE 9 Definitions

9.1 Certain Definitions. In addition to the other definitions contained in this Agreement, the following terms will, when used in this Agreement, have the following respective meanings:

"Action" means any claim, litigation, action, cause of action, counter-claim, cross-claim, demand, suit, arbitration, inquiry, notice of violation, proceeding, summons, subpoena, audit, examination or investigation of any nature, civil, criminal, administrative, regulatory or otherwise, whether at Law or in equity, conducted or heard by or before, or otherwise involving, any court or other Governmental Authority, any arbitrator or arbitration panel or mediator, any governmental, judicial, administrative or adversarial proceeding (public or private), any action, complaint, claim, lawsuit, legal proceeding, whistleblower complaint, litigation, arbitration or mediation, any hearing, or any other dispute, including any adversarial proceeding arising out of this Agreement.

"<u>Affiliate</u>" means, with respect to any Person, any Person who directly or indirectly owns or controls, is owned or controlled by, or is under direct or indirect common ownership or control with such other Person. Without limiting the generality of the foregoing, a Person shall be deemed to "own" another person if it owns, directly or indirectly, more than fifty percent (50%) of the capital stock or other equity interest of such other Person and a Person shall be deemed to "control" another Person if such Person possesses, directly or indirectly, the power to direct or cause the direction of the management and policies of such specified Person, and/or if such other Person is at such time a, direct or indirect, beneficial holder of at least ten percent (10%) of any class of the equity interests of such specified Person.

43

"<u>Applicable Law</u>" means any federal, state, local, municipal, foreign or other law, statute, legislation, constitution, principle of common law, resolution, ordinance, code, edict, decree, proclamation, treaty, convention, rule, regulation, proposed regulation, listing standard, ruling, directive, pronouncement, requirement, specification, determination, decision, opinion or interpretation that is, or has been issued, enacted, adopted, passed, approved, promulgated, made, implemented or otherwise put into effect by or under the authority of any Governmental Authority or the governing body of any national securities exchange.

"Business" means (i) the Management Services, and (ii) any other business activity conducted by the Company during the twelve (12) month period immediately preceding the Closing or currently contemplated to be conducted.

"Business Day" means each day other than a Saturday, Sunday or other day on which commercial banks in California are authorized or required by Applicable Law to close.

"COBRA" means Part 6 of Subtitle B of Title I of ERISA, Section 4980B of the Code, and any similar Applicable Law.

"Code" means the Internal Revenue Code of 1986, as amended.

"Company 401(k) Plan" means the 401(k) plan sponsored or maintained by the Company.

"Company Intellectual Property" means all Intellectual Property owned or purported to be owned by the Company.

"<u>Company Licensed Intellectual Property</u>" means all Intellectual Property pursuant to any licenses, sublicenses or other agreements under which the Company: (i) is granted rights by others in Intellectual Property (other than commercial off the shelf software that is made available for a total cost of less than \$10,000) or (ii) has granted rights to others in Company Intellectual Property. "<u>Company Registered Intellectual Property</u>" means all Company Intellectual Property registered, filed, or issued under the authority of any domain name registrar, any state, federal, local or international government or other Governmental Authority, including all patents, registered copyrights, registered trademarks, and domain names, and all applications for any of the foregoing.

"<u>Confidential Information</u>" means, all information of a nonpublic, confidential, proprietary or commercially sensitive nature (whether or not specifically labeled or identified as "confidential"), in any form or medium, that relates to the Protected Business, including: (a) internal business information, processes and procedures (including historical and projected financial information and budgets and information relating to strategic and staffing plans and practices, business, training, marketing, promotional and sales plans and practices, cost, rate and pricing structures and accounting and business methods); (b) requirements of and specific contractual arrangements with customers, suppliers, contractors, vendors, employees or other business relations and their confidential information; (c) trade secrets, know-how, source code and methods of operation, techniques, formulae and systems relating to the products or services of the Purchased Business and data, data bases, analyses, records, reports, manuals, handbooks documentation and models and relating thereto; (d) inventions, innovations, improvements, discoveries, business units of the Protected Business; (f) acquisition plans, targets and strategies of the Protected Business; and (g) financial and other business or strategic information of the Protected Business. Notwithstanding the foregoing, Confidential Information shall not include information which is or becomes generally available to the public other than as a result of a disclosure by the Equityholder or the Beneficial Owners' Affiliates or their respective Representatives in violation of this Agreement or other confidentiality obligation to which any of the mis bound.

44

"<u>Contracts</u>" means any contract, agreement, understanding, arrangement, warranty, deed, power of attorney, purchase order, work order, commitment, assignment, certificate, covenant, assurance, undertaking, option, lease, license, insurance policy, sale and purchase order, commitment and other instruments of any kind, whether written or oral, to which a Person is a party, is otherwise bound or has any current or future obligation or Liability.

"Damages" shall include any loss, damage, Liability, claim, demand, settlement, Judgment, award, fine, penalty, Tax, fee (including any legal fee, expert fee, accounting fee or advisory fee), charge, cost or expense (including any cost of investigation, penalty, fee and disbursement of counsel).

"Disclosure Schedule" means the disclosure schedule prepared by the Equityholder, Beneficial Owners and Company and delivered to the Buyer incident to the execution and delivery of this Agreement.

"Employment Agreement" means the Employment Agreement, in substantially the form attached hereto as Exhibit A, by and between the Company and Ron Brandt, to be dated as of the Closing Date.

"Entity" means any firm, corporation, partnership, joint venture, limited liability company, Governmental Authority, unincorporated organization, trust, association or other entity.

"Equity Interests" means, with respect to any Person, (a) any capital stock, partnership or membership interest, joint venture interest, unit of participation or other similar interest (however designated) in such Person and (b) any option, warrant, call, right (including purchase rights, conversion rights, exchange rights, preemptive rights, co-sale rights, rights of first refusal and similar rights) or other Contract which would entitle any other Person to acquire any such interest in such Person.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" means any Person that at any relevant time is or was considered a single employer with the Company under Section 414 of the Code.

"GAAP" means generally accepted accounting principles in the United States, as in effect as of the date of this Agreement, set out in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or such other principles as may be approved by a significant segment of the accounting profession in the United States, that are applicable to the circumstances as of the date of determination, consistently applied in accordance with past practice.

"Governing Documents" means, with respect to any particular entity: (a) if a corporation, the articles or certificate of incorporation and the bylaws; (b) if a general partnership, the partnership agreement and any statement of partnership; (c) if a limited partnership, the limited partnership agreement and the articles or certificate of limited partnership; (d) if a limited liability company, the articles of organization or certificate of formation and operating agreement, regulations, limited liability company agreement, or company agreement; I if another type of Person, any other charter or similar document adopted or filed in connection with the creation, formation or organization of the Person; (e) all equityholders' agreements, voting agreements, joint venture agreements, registration rights agreements or other agreements or documents relating to the organization, management or operation of any Person or relating to the rights, duties and obligations of the equityholders of any Person; and (f) any amendment or supplement to any of the foregoing.

"Governmental Authority" means any federal, state, local or foreign government or political subdivision thereof, or any agency or instrumentality of such government or political subdivision, or any self-regulated organization or other non-governmental regulatory authority or quasi-governmental authority (to the extent that the rules, regulations or orders of such organization or authority have the force of Applicable Law), or any arbitrator, court or tribunal of competent jurisdiction.

"Governmental Order" means any order, writ, judgment, injunction, decree, stipulation, determination or award entered by or with any Governmental Authority.

"Healthcare Laws" means all relevant includes but is not limited to (a) any and all applicable federal, state and federal civil or criminal healthcare Laws applicable to the Businesslocal fraud and abuse laws, including Medicaid, Medicare, all applicable anti-kickback, false claims, self-referral, and insurance fraud laws, such as (i) the federal Anti-kickback Anti-Kickback Statute (42 U.S.C. § 1320a-7b(b)), (ii) the Stark Law (42 U.S.C. § 1395nn and §1395(q)), (iii) the civil False Claims Act (31 U.S.C. § 3729 et seq.), the administrative False Claims Law(iv) Sections 1320a-7, and 1320a-7a and 1320a-7b of Title 42 of the United States Code, and (v) any criminal laws relating to health care fraud and abuse, including but not limited to 18 U.S.C. Sections 286, 287, 1347 and 1349 and the health care fraud criminal provisions under HIPAA the regulations promulgated pursuant to such statutes; (b) any applicable laws, rules and regulations governing the health insurance program for the aged and disabled established by Title XVIII of the Social Security Act (42 U.S.C. § 1320a-7b(a)), the Civil Money Penalties Law§ 1395 et seq.) and any applicable statutes succeeding thereto (c) any applicable laws, rules and regulations governing the health care fraud and as (42 U.S.C. § 1320a-7), orş 1396 et seq.) and any applicable statutes succeeding thereto; (d) to the extent applicable, the Medicare Prescription Drug, Improvement, and Modernization Act of 2003 (Pub. L. No. 108-173) and the regulations promulgated pursuant thereto; (e) HIPAA and applicable laws and regulations regives and any applicable statutes and regulatione proves and the regulation at of 2003 (Pub. L. No. 108-173) and the regulations promulgated pursuant thereto; (e) HIPAA an

licensure, credentialing, insurance entity licensure, scope of practice, fee-splitting and laws relating to the corporate practice of a licensed profession and the maintenance of a provider network, including any willing or authorized provider laws; (h) applicable laws and regulations regarding the licensure, qualification, or authority to conduct third party administrator services or similar services or to take on reimbursement or expense risk, under agreements with health plans, Medicare Advantage or managed Medicaid health plans, other insurers, or self-insured plans; and (i) any comparable state or federal laws in respect of or similar to any of the foregoing, each of (a) through (i) as may be amended from time to time and the regulations promulgated pursuant to such Applicable Laws, and comparable state laws.

"HHS" means the U.S. Department of Health and Human Services.

"Indebtedness" means, without duplication: (a) with respect to any Person, all obligations (including all obligations in respect of principal, accrued interest, penalties, fees, costs and premiums, whether or not contingent and including any prepayment penalties, premiums and any other fees, expense, breakages costs, indemnities, "make-whole amounts", expense reimbursement obligations and other monetary obligations in connection therewith) of such Person for any (i) indebtedness for borrowed money (including overdraft facilities), whether accrued, absolute, contingent, unliquidated or otherwise, known or unknown, whether due or to become due; (ii) deferred price of property, goods or services (other than trade payables incurred in the Ordinary Course of Business that are less than ninety (90) days past due); (iii) any overpayments of capitation revenue, regardless of age; (iv) reimbursement and other obligations for surety bonds and letters of credit and similar transactions to the extent drawn; (v) obligations evidenced by notes, bonds, debentures or similar Contracts; (vi) breakage fees with respect to Contracts relating to interest rate protection, swap agreements and collar agreements; (vii) obligations arising from checks issued but not presented for payment within a specified timeframe from the issue date; and (viii) guaranties of any of the foregoing; and (b) with respect to the Company, all obligations (including all obligations in respect of principal, accrued interest, penalties, fees, costs and premiums, whether or not contingent and including any prepayment penalties, premiums and any other fees, expense, breakages costs, indemnities, "make-whole amounts", expense reimbursement obligations and other monetary obligations in connection therewith) of the Company for any (i) payable or other indebtedness owing to the Equityholders, any of their respective Affiliates or any Related Party of the Equityholders or any of their respective Affiliates, whether accrued, absolute, contingent, unliquidated or otherwise, known or unknown, whether due or to become due; (ii) any unfunded or underfunded liabilities with respect to any defined benefit plan and any employer contribution obligations of the Company under a Company 401(k) plan owed or payable by the Company as of the Closing, in each case including the employer's share of all employment, payroll and similar Taxes thereon; (iii) any unpaid commission, severance, deferred compensation, bonus, profit sharing or similar types of payments, whether or not accrued, attributable to any period prior to the Closing, in each case including the employer's share of all employment, payroll and similar Taxes thereon; (iv) contingent purchase price payment, earnout or similar payment; (v) any outstanding balances due to vendors and/or suppliers of the Company or Equityholder; and (v) any contingent purchase price payment, earnout or similar payment.

47

"Intellectual Property" means all: (a) patent registrations, patent applications, patent disclosures, improvements and inventions, (b) trademarks, service marks, trade dress, trade names, logos, dba's, fictitious names, and corporate names and registrations and applications (including intents to use) for registration thereof, together with all of the goodwill associated therewith, (c) copyrights (registered or unregistered) and copyrightable works and registrations and applications for registration thereof, (d) mask works and registrations and applications for registration thereof, (e) computer software, data, data bases and documentation thereof, (f) trade secrets, trade compositions, technologies, inventions (whether patentable or unpatentable and whether or not reduced to practice), know-how, manufacturing and production processes and techniques, research and development information, drawings, specifications, designs, plans, manuals, proposals, technical data, copyrightable works, financial, marketing and business plans and customer and supplier lists and other proprietary information, (g) domain names, (h) Software, (i) other intellectual property rights, (j) license agreements or other rights pertaining to the foregoing, (k) rights of enforcement of any of the foregoing, and (l) copies and tangible embodiments therefore in whatever form or medium.

"IT Systems" means computers, software, hardware, servers, websites, workstations, routers, hubs, switches, data communications lines, firmware, networks, and all other information technology equipment owned or used by any of the Company.

"Judgment" means any: (a) order, judgment, injunction, edict, decree, ruling, pronouncement, determination, decision, opinion, verdict, sentence, subpoena, writ or award issued, made, entered, rendered or otherwise put into effect by or under the authority of any court, administrative agency or other Governmental Authority or any arbitrator or arbitration panel; or (b) Contract with any Governmental Authority entered into in connection with any Action.

"Knowledge of the Company" or any other similar knowledge qualification, means the actual knowledge after reasonable inquiry of any of the following persons: Ronald Brandt and Allison Brandt.

"Liability" or "Liabilities" means, with respect to any Person, any liability or obligation of such Person of any kind, character or description, whether known or unknown, absolute or contingent, accrued or unaccrued, disputed or undisputed, liquidated or unliquidated, secured or unsecured, joint or several, vested or unvested, executory, determined, determinable or otherwise and whether or not the same is required to be accrued on the financial statements of such Person.

"Lien" means any charge, mortgage, pledge, security interest, lien, or Lien, other than those that customarily arise under securities Applicable Laws in private transactions.

48

"<u>Management Services Agreements</u>" means (i) the Management Services Agreement, in substantially the form attached hereto as <u>Exhibit B-1</u>, by and between the Company and the Buyer, to be dated as of the Closing Date, and (ii) the Management Services Agreement, in substantially the form attached hereto as <u>Exhibit B-2</u>, by and between the Company and Metropolitan IPA, a California professional corporation ("<u>Metropolitan IPA</u>"), to be dated as of the Closing Date.

"<u>Material Adverse Effect</u>" means any change, event or effect that, when taken individually or together with all other related changes, events or effects, has had or would reasonably be expected to have a material adverse effect on the Business, properties, assets, results of operations or condition, financially or otherwise, taken as a whole, of the Company, or hinder, delay or prevent the consummation of the Transactions; <u>provided</u>, <u>however</u>, that a "Material Adverse Effect" shall not include any change or effect: (a) in general economic or business conditions; (b) financial, banking or securities markets of the United States in general (including any disruption thereof and any decline in the price of any security or any market index); (c) in national or international political or social conditions, including the engagement by the United States in hostilities, whether or not pursuant to the declaration of a national emergency or war, or the occurrence of any military or terrorist attack upon the United States or any of its territories, possessions or diplomatic or consular offices or upon any military installation, equipment or personnel of the United States; (d) affecting generally the industries or markets in which the Company operates; (e) related to the announcement taken at the written request, or with the express written permission, of the Buyer or its Representatives or consummation of any of the Transactions; (f) resulting from any act taken by the Equityholder, Beneficial Owner or the Company pursuant to or in accordance with this Agreement taken at the written request, or with the express written permission, of the Buyer or its Representatives; (i) resulting from changes in reimbursement policies or fee schedules by Governmental Authority; (j) adverse effects related to the COVID-

19 pandemic; or (k) items which are cured prior to the earlier of the applicable Closing Date or termination of this Agreement in accordance with <u>Article 7</u>, so long as, in the case of each of the foregoing clauses (a) through (j), any such effect, change, event or condition does not affect the Company in a materially disproportionate manner relative to other participants in similar industries in which the Company operates.

"Neutral Accounting Firm" means an independent accounting firm of nationally recognized standing that is not at the time it is to be engaged hereunder rendering services to any Party, or any Affiliate of any Party, and has not done so within the two (2) year period prior thereto.

"Note" means the Promissory Note, in substantially the form attached hereto as Exhibit C, by and between the Company and Apollo Care Connect, Inc., a Delaware corporation("Apollo Care"), to be dated as of the Closing Date.

"Ordinary Course of Business" means an action taken by a Person if that action (a) is consistent in nature, scope and magnitude with the past practices of such Person and is taken in the ordinary course of the normal, day-to-day operations of such Person; (b) does not require special or separate authorization by the board of directors or shareholders of such Person (or by any Person or group of Persons exercising similar authority) and does not require any other separate or special authorization of any nature; and (c) is similar in nature, scope and magnitude to actions customarily taken, without any separate or special authorization, in the ordinary course of the normal, day-to-day operations of other Persons that are in the same line of business as such Person.

49

"Permits" means material approvals, permits, consents, certificates and licenses, if any, of all Governmental Authorities required for the operation of the Business.

"Permitted Liens" means (a) landlords', lessors', mechanics', materialmen's, warehousemen's, carriers', workers', manufacturer's or repairmen's Liens or other similar Liens arising or incurred in the Ordinary Course of Business; (b) Liens for Taxes, assessments and other governmental charges not yet due and payable or due but not delinquent or being contested in good faith by appropriate legal proceedings; (c) Liens set forth on <u>Section 2.8 of the Disclosure Schedule</u>; (d) with respect to real property, (i) easements, quasi-easements, licenses, covenants, rights-of-way, rights of re-entry or other similar restrictions, including any other agreements, conditions or restrictions that would be shown by a current title report or other similar report or listing, which do not materially impair the occupancy or use of the real property for the purposes for which it is currently used in connection with the Companies' businesses, (ii) any conditions that may be shown by a current survey or physical inspection, which do not materially impair the occupancy or use of the real property for the purposes for which it is currently used in connection with the Companies' businesses, (iii) and or other similar requirements or restrictions which are not violated by the current use and operation of the applicable real property (except for any violations that would not materially affect the use and occupancy of any such real property as currently used and occupied) and (iv) any and all service contracts and agreements affecting any real property, in each case, which do not impair in any material respect the current use or occupancy of the real property (except for any violations that would not materially impair restrictions of record identified in any title reports obtained by Buyer (or any of its Representatives) or made available in the Electronic Data Room; (f) pledges to secure the payment of workers' compensation, unemployment insurance, social security benefits or obligations arising under similar Applicable Laws, or to secur

"Person" means any individual or Entity.

"Post-Closing Tax Period" means with respect to the Company any taxable period beginning after the Closing Date and, with respect to any taxable period beginning on or before and ending after the Closing Date, the portion of such taxable period beginning after the Closing Date.

"Pre-Closing Tax Period" means with respect to the Company any taxable period ending on or before the Closing Date and, with respect to any taxable period beginning on or before and ending after the Closing Date, the portion of such taxable period ending on and including the Closing Date.

"Privacy Laws" means all Applicable Laws and industry self-regulatory programs concerning the access, collection, use, processing, analysis, retention, storage, protection, transfer, disclosure and/or disposal of individually identifiable information including, but not limited to state consumer protection Laws, state breach notification Laws, and state social security number protection Laws. Such Privacy Laws include but are not limited to, the California Consumer Privacy Act of 2018, the California Confidentiality of Medical Information Act (CMIA), the Federal Trade Commission Act, the federal Privacy Act of 1974, the Telephone Consumer Protection Act, the CAN-SPAM Act, the Fair Credit Reporting Act and its state law equivalents, the California Online Privacy Protection Act, the California Consumer Privacy Act, and all other U.S. state laws concerning privacy, data protection, and/or data security, each as amended from time to time, as well as the Digital Advertising Alliance's Self-Regulatory Principles for Online Behavioral Advertising, the Payment Card Industry Data Security Standard and all other requirements of the payment card brands and any state or federal Applicable Law or regulation or industry standard binding upon the Company the purpose of which is to protect the privacy and/or security of individually identifiable information.

50

"Protected Business" means the Business and any other business activities conducted by the Company, the Equityholder or any Beneficial Owner, AHMS, CFC IPA, Buyer or any Buyer Parties during the twelve (12) month period immediately preceding the Closing or currently contemplated to be conducted.

"Related Party" means, with respect to any Person, any shareholder, member, manager, partner, employee, officer, director of, or consultant to, such Person or any individual related by blood, marriage or adoption to any such Person or any Entity in which any such Person owns any beneficial interest.

"Representative" means, with respect to any Person, any and all directors, managers, members, partners, employees, attorneys, accountants, financial advisors, officers and independent contractors, and other agents of such Person.

"Software" means any and all (a) computer programs and other software, including software implementations of algorithms, models, and methodologies, whether in source code, object code or other form, including libraries, subroutines and other components thereof; (b) computerized databases and other computerized compilations and collections of data or information, including all data and information included in such databases, compilations or collections (whether machine readable or otherwise) and rights therein; (c) screens, user interfaces, command structures, report formats, templates, menus, buttons and icons; and (d) all documentation, including development, diagnostic, support, user and training documentation related to any of the foregoing.

"Stockholders Agreement" means the Stockholders Agreement of the Company, in substantially the form attached hereto as Exhibit D, by and between the Company, the Equityholder and each Beneficial Owner, to be dated as of the Closing Date.

"Straddle Period" means with respect to the Company any taxable period that begins on or before and ends after the Closing Date.

"Subsidiary" means with respect to any Person, any Entity of which fifty percent (50%) or more of the outstanding voting securities or other voting Equity Interests are owned, directly or indirectly, by the referenced Person.

"Tax" or "Taxes" (and with correlative meaning, "Taxable") means any United States federal, state or local, or non-United States, income, gross receipts, franchise,

estimated, alternative minimum, add-on minimum, sales, use, transfer, registration, value added, excise, natural resources, severance, stamp, withholding, occupation, premium, windfall profit, environmental, customs, duties, real property, personal property, capital stock, net worth, intangibles, social security, unemployment, disability, payroll, license, employee or other tax or similar levy, of any kind whatsoever, including any interest, penalties, fines or additions to tax in respect of the foregoing.

51

"Tax Authority" means any governmental entity or any subdivision, agency, commission or authority thereof having jurisdiction over the assessment, determination, collection or imposition of any Tax.

"<u>Tax Return</u>" means any return, declaration, report, claim for refund, information return or other document (including any related or supporting estimates, elections, schedules, statements or information), including any amendment thereof, filed or required to be filed in connection with the determination, assessment or collection of any Tax or the administration of any laws, regulations or administrative requirements relating to any Tax.

"Transaction Documents" means, with respect to a Party, all agreements, certificates and other instruments to be delivered by such Party pursuant to this Agreement and any other documents, certificates, or instruments executed or delivered in connection with the transactions contemplated hereby, including without limitation, (a) the CFC Purchase Agreement, (b) the Employment Agreement, (c) the Management Services Agreement, (d) the Stockholders Agreement, and in each case, all agreements, certificates and other instruments to be delivered thereunder and any other documents, certificates, or instruments executed or delivered in connection with the transactions contemplated thereby.

"<u>Transaction Expenses</u>" means, without duplication, with respect to any Person, all fees, costs and expenses (including all legal fees and expenses, all fees and expenses payable to any investment banker, broker or finder, and all fees and expenses of any audit firm or accountants) that have been incurred in connection with the Transactions on behalf of or for the benefit of such Person, including any change of control, severance, sale, retention, success or similar payments, bonuses, deferred compensation or promises to payments made by any of the Company to any Person which are triggered, whether in part or whole, by the Transactions.

"Transactions" means the purchase and sale of the Purchased Shares and the other transactions contemplated hereby.

"Treasury Regulations" means any proposed, temporary or other applicable Treasury Regulations promulgated under the Code.

52

9.2 Interpretation. As used in this Agreement, except as otherwise indicated in this Agreement or as the context may otherwise require: (a) the words "include," "includes," and "including" are deemed to be followed by "without limitation" whether or not they are in fact followed by such words or similar import; (b) the word "or" is not exclusive; (c) references to an "Article," "Section, "preamble," recital," or any other subdivision, or to an "Appendix," "Exhibit," "Schedule," or "Disclosure Schedule" are to an article, section, preamble, recital, or subdivision of this Agreement, or to an appendix, exhibit, schedule, or disclosure schedule to this Agreement, "hereby," "hereof," "herein," and comparable words refer to all of this Agreement, including the Appendices, Exhibits, Schedule, or Disclosure Schedule to this Agreement; (e) any pronoun in masculine, feminine, or neuter form shall include any other gender; (f) any word in the singular form includes the plural and vice versa; (g) references to any Applicable Law are to it as amended, modified, supplemented, and restated now or from time to time after the date of this Agreement, and to any corresponding provisions of successor Applicable Laws, and, unless the context requires otherwise, any reference to any statute shall be deemed also to refer to all rules and regulations promulgated thereunder; (h) references to any Person include such Person's respective successors and permitted assigns (and in the case of a natural person, such Person's heirs, estate, and personal representatives); (i) references to a "day" or number of "days" (without the explicit qualification of "Business") refer to a calendar day or number of calendar days; (j) references to a "Company" or the "Companies" shall include, unless the context requires otherwise, reference to such Company or the Companies and its/their, direct and indirect, Subsidiaries; and (k) the term "made available" means posted to the "iHealth, Inc." virtual data room hosted at https://collaborate.shepp

ARTICLE 10 Miscellaneous

10.1 <u>Notices</u>. All notices, requests, consents, claims, demands, waivers and other communications hereunder shall be in writing and shall be deemed to have been given (a) when delivered by hand (with written confirmation of receipt); (b) when received by the addressee if sent by a nationally recognized overnight courier (receipt requested); (c) on the date sent by facsimile or e-mail of a PDF document (with confirmation of transmission) if sent during normal business hours of the recipient, and on the next Business Day if sent after normal business hours of the recipient or (d) on the third (3rd) day after the date mailed, by certified or registered mail, return receipt requested, postage prepaid. Such communications must be sent to the respective Parties at the following addresses (or at such other address for a Party as shall be specified in a notice given in accordance with this <u>Section 10.1</u>):

If to any Beneficial Owner, the	1601 N Sepulveda, Blvd. #110
Equityholder or the Company	Manhattan Beach, CA 90266
(prior to the Closing):	Attention: Ron Brandt
	Email: brandt@ihealthmd.com

53

with a copy (which shall not constitute notice) to:

If to Buyer:

Craig Steinberg, Esq Email: craig@csteinberglaw.com

Network Medical Management, Inc. 1668 S. Garfield Avenue, 2nd Floor Alhambra, CA 91801 Attention: Chandan Basho Email: cbasho@apollomed.net with a copy (which shall not constitute notice) to:

Russ August & Kabat, LLP 12424 Wilshire Blvd., 12th Floor Los Angeles, CA 90025 Attention: Lauren Geisser and Christine Shin Email: <u>lgeisser@raklaw.com</u> and <u>cshin@raklaw.com</u>

10.2 <u>Assignment</u>. This Agreement shall be binding upon and inure to the benefit of the Parties hereto and their respective heirs, successors and permitted assigns. Neither this Agreement nor the obligations of any Party hereunder shall be assignable or transferable by such Party without the prior written consent of the other Parties hereto. Subject to the preceding sentence, this Agreement shall be binding upon, inure to the benefit of and be enforceable by the parties to this Agreement and their respective successors and assigns. Notwithstanding anything to the contrary, Buyer may transfer and assign its rights and obligations under this Agreement and any Transaction Document without the prior written consent of any other Party by way of collateral assignment to any bank, financing institution or other lender providing financing to Buyer now or in the future or to an Affiliate of Buyer or to any successor to Buyer by merger, consolidation or reorganization.

10.3 <u>No Third Party Beneficiaries</u>. This Agreement is for the sole benefit of the Parties hereto and their successors and permitted assigns and nothing herein expressed or implied shall give or be construed to give to any Person any legal or equitable rights hereunder, other than (i) the Parties hereto and such successors and assigns, (ii) the Indemnified Parties and (iii) the Released Parties. Without limiting the generality of the foregoing, (a) no employee of the Companies shall have any rights, as an employee, under this Agreement or under any of the Transaction Documents to which he or she is not personally a party and (b) other than as an express party to this Agreement or the other Transaction Documents, if applicable, no creditor of any of the Parties shall have any rights under this Agreement or any of the Transaction Documents.

10.4 <u>Counterparts</u>. This Agreement may be executed and delivered in one or more counterparts, each of which when executed and delivered shall be an original, and all of which when executed shall constitute one and the same instrument. The exchange of copies of this Agreement and of signature pages by facsimile or other electronic transmission shall constitute effective execution and delivery of this Agreement as to the Parties and may be used in lieu of the original Agreement for all purposes. Signatures of the Parties transmitted by facsimile or other electronic means shall be deemed to be their original signatures for all purposes.

54

10.5 <u>Amendment and Modification; Waiver</u>. This Agreement may only be amended, modified or supplemented by an agreement in writing signed by each Party hereto. No waiver by any Party hereto of any of the provisions hereof shall be effective unless explicitly set forth in writing and signed by the Party so waiving. No waiver by any Party shall operate or be construed as a waiver in respect of any failure, breach or default not expressly identified by such written waiver, whether of a similar or different character, and whether occurring before or after that waiver. No failure to exercise, or delay in exercising, any right, remedy, power or privilege arising from this Agreement shall operate or be construed as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege.

10.6 <u>Expenses</u>. Except as otherwise expressly provided by this Agreement, whether or not either Closing is effected, all Transaction Expenses shall be borne by the Party incurring the same.

10.7 <u>Governing Law; Submission to Jurisdiction</u>.

(a) The validity and construction of this Agreement shall be governed by the internal laws of the State of Delaware, regardless of the laws that might otherwise govern under applicable principles of conflict of laws thereof.

(b) Except as otherwise provided herein, the successful or prevailing party shall be entitled to recover reasonable attorneys' fees and other costs incurred in connection with that Action.

(c) The Parties shall attempt in good faith to resolve any dispute, controversy or claim between the Parties that relates to the interpretation, carrying out of obligations, breach, termination or enforcement of this Agreement or in any way arising out of or connected with this Agreement ("Dispute").

(d) In the event of a Dispute among the Parties, prior to referring to arbitration as described in<u>Section 10.7(e)</u> the Parties must engage in a mediation under the then current mediation rules of JAMS, Inc. ("JAMS"). A single disinterested third party mediator familiar with Delaware law shall be selected by JAMS in accordance with its then current arbitration rules. No Party shall commence arbitration as described in <u>Section 10.7(e)</u> without first attempting in good faith to engage in a mediation as set forth in this <u>Section 10.7(d)</u>. The parties to the Dispute shall share the expenses of the mediator and the other costs of mediation on a pro rata basis.

(e) If any Dispute cannot be resolved by the Parties pursuant to this <u>Section 10.7</u> or otherwise, such Dispute shall be exclusively referred to and finally settled by arbitration in accordance with the Rules of Arbitration of JAMS. The arbitration proceedings shall take place in Los Angeles, California and shall be conducted in the English language. The number of arbitrators shall be three, and the presiding arbitrator shall be chosen by the two arbitrators nominated by the Parties with input from the parties. If the presiding arbitrator is not chosen by the two arbitrators within 30 days of the date of confirmation of the later of the two party nominated arbitrators, he or she shall be appointed by JAMS. The award of the arbitral tribunal shall be final and binding upon the Parties and shall be the sole and exclusive remedy between the parties regarding any claims, counterclaims, issues, or requests for declaratory, accounting, or other relief presented to the arbitral tribunal. The costs of arbitration shall be apportioned by the arbitral tribunal in the award.

(f) EACH OF THE PARTIES TO THIS AGREEMENT HEREBY IRREVOCABLY WAIVES ALL RIGHT TO A TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

10.8 <u>Specific Performance</u>. The Parties agree that irreparable damage may occur and that the Parties would not have any adequate remedy at law in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that prior to the valid termination of this Agreement pursuant to <u>Section 7.1</u>, the Parties shall be entitled to seek an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement in any federal or state courts located in the State of California without proof of actual damages, this being in addition to any other remedy to which they are entitled at law or in equity.

10.9 <u>Remedies Cumulative</u>. Except as otherwise set forth herein, the various rights and remedies herein provided will be cumulative and not exclusive of any other rights or remedies herein provided or any rights or remedies provided by Applicable Law.

10.10 Construction.

(a) Inasmuch as this Agreement is the result of negotiations between sophisticated Parties of equal bargaining power represented by counsel, no inference in favor of or against any Party will be drawn from the fact that any portion of this Agreement has been drafted by or on behalf of such Party.

(b) Any capitalized terms used in any Exhibit or Schedule but not otherwise defined therein, shall have the meaning as defined in this Agreement.

(c) The heading references herein are for convenience purposes only, do not constitute a part of this Agreement and will not be deemed to limit or affect any of the provisions hereof.

10.11 <u>Severability</u>. The provisions of this Agreement will be deemed severable and the invalidity or unenforceability of any provision will not affect the validity or enforceability of the other provisions hereof. If any provision of this Agreement, or the application thereof to any Person or any circumstance, is invalid or unenforceable, (a) a suitable and equitable provision will be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision, and (b) the remainder of this Agreement and the application of such provision to other Persons or circumstances will not be affected by such invalidity or unenforceability, nor will such invalidity or unenforceability affect the validity or enforceability of such provision, or the application thereof, in any other jurisdiction.

10.12 <u>Entire Agreement</u>. This Agreement, the Disclosure Schedules and the Transaction Documents contain the entire understanding of the Parties with respect to the subject matter hereof and thereof and thereof and supersede all prior oral or written agreements, negotiations, understandings, statements or proposals with respect to the subject matter hereof and thereof.

56

10.13 <u>Time is of the Essence</u>. Time is of the essence of this Agreement.

10.14 <u>No Rescission</u>. No party shall be entitled to rescind the transactions contemplated hereby by virtue of any failure of any party's representations and warranties herein to have been true or any failure by any party to perform its obligations hereunder. Notwithstanding anything to the contrary in this Agreement, the Closing will not occur unless and until such time Exhibits A through D have been fully executed.

10.15 <u>Release</u>.

For and in consideration of the amount to be paid to the Equityholder and Beneficial Owners under this Agreement and the Transaction Documents, (a) and the additional covenants and promises set forth in this Agreement, each Beneficial Owner and the Equityholder, on behalf of itself, himself or herself and its, his and her assigns, heirs, beneficiaries, creditors, representatives, agents and Affiliates (the "Releasing Parties"), hereby fully, finally and irrevocably releases, acquits and forever discharges the Buyer, the Company, their respective Subsidiaries and their respective officers, directors, managers, managing member, partners, general partners, limited partners, managing directors, members, trustees, shareholders, Representatives, employees, principals, agents, Affiliates, parents, subsidiaries, joint ventures, predecessors, successors, assigns, beneficiaries, heirs, executors, personal or legal representatives, insurers and attorneys of any of them (collectively, the "Released Parties") from any and all commitments, actions, debts, claims, counterclaims, suits, causes of action, damages, demands, liabilities, obligations, costs, expenses and compensation of every kind and nature whatsoever, past, present or future, at law or in equity, whether known or unknown, contingent or otherwise, which such Releasing Parties, or any of them, had, has or may have had at any time in the past until and including the date of this Agreement against the Released Parties, or any of them, including but not limited to any claims which relate to or arise out of such Releasing Party's prior relationship with the Company, AHMS, CFC IPA, any Buyer Party (including their respective present and former subsidiaries, parent entities or any predecessors-in-interest) (collectively, for the purposes of this Section 10.15, "Causes of Action"); provided, that this release shall not affect or impair: (i) any Causes of Action, rights of the Releasing Parties or any obligations of the Released Parties to the Releasing Parties arising under this Agreement or in any Transaction Document, (ii) any Causes of Action or rights to reimbursement, indemnification or contribution of any Releasing Party in his, her or its capacity as an officer, director, manager, stockholder, unitholder or employee of a Released Party (whenever arising) under the governing documents of the applicable Releasing Party, the Tail Policies or Applicable Law or (iii) if the Releasing Party is an employee of the Company as of the date hereof, any rights of the Releasing Parties to payments in respect of such Releasing Parties' employment with the Company .

(b) The Releasing Parties hereby represent to the Released Parties that the Releasing Parties: (i) have not assigned any Causes of Action or possible Causes of Action against any Released Party, (ii) fully intend to release all Causes of Action against the Released Parties including without limitation, unknown and contingent Causes of Action (other than those specifically reserved above) and (iii) have consulted with counsel with respect to the execution and delivery of this general release and has been fully apprised of the consequences hereof. The Releasing Parties hereby covenant and agree not to make any claim or cross-claim or to take any proceedings against any other Person who or which might claim contribution from or indemnification by any Affiliate of the Company under the provisions of any statute or otherwise except claims relating to rights arising from this Agreement or in any Transaction Document.

57

(c) Each Releasing Party acknowledges that such Releasing Party has been advised to consult with legal counsel and is familiar with the provisions of California Civil Code Section 1542, a statute that otherwise prohibits the release of unknown claims, which provides as follows:

A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS WHICH THE CREDITOR OR RELEASING PARTY DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE, WHICH IF KNOWN BY HIM OR HER WOULD HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR OR RELEASED PARTY.

(d) Each Releasing Party being aware of said code section agrees to expressly waive any rights such Releasing Party may have thereunder, as well as under any other statute or common law principles of similar effect.

(e) Each of the Releasing Parties hereby acknowledges that he, she or it has been informed that the Company and/or its Affiliates and successors may from time to time enter into agreements for additional types of financing, including without limitation, recapitalizations, mergers and initial public offerings of equity interests of the Company and/or its Affiliates and successors, and also may pursue acquisitions or enter into agreements for the sale of the Company and/or its Affiliates or successors or all or a portion of the Company or its Affiliates or successor's assets, which may result in or reflect an increase in equity value or enterprise value and that any and all claims based on such increases in equity value or enterprise value (without limitation) are encompassed within the scope of the release described in this <u>Section 10.15</u>.

(f) The Releasing Parties hereby irrevocably covenant to refrain from, directly or indirectly (i) asserting any Causes of Action, or commencing,

instituting or causing to be commenced, or continuing with any claim, action or proceeding for a Cause of Action, and this Agreement may be raised by any Released Party or Buyer as an estoppel to any such claims, actions or proceedings and (ii) making any claim or commencing any action or proceeding against any Person (or assisting or encouraging any other Person in connection therewith) in which any claim, action or proceeding would arise against any Released Party for contribution or indemnity or other relief from, over and against any Released Party or which otherwise results in a Released Party suffering or incurring any Damages, whether under common law, equity, statute, Contract or otherwise, with respect a Cause of Action. Without in any way limiting any of the rights and remedies otherwise available to any Released Party, the Releasing Parties shall indemnify and hold harmless Buyer and each Released Party from and against all Causes of Action whether or not involving third-party claims, action or proceedings, arising directly or indirectly from or in connection with the assertion by or on behalf of each Releasing Party or any of its Affiliates of any claim, action or proceeding or other matter which is, or is purported to be, a Cause of Action. It is the intention of the Equityholder, each Beneficial Owner and the other Releasing Parties hart be ab to each Cause of Action hereinabove specified. In furtherance of this intention, the Equityholder, each Beneficial Owner and the other Releasing Parties hereby expressly waive any and all rights and benefits conferred upon it by the provisions of Applicable Law with respect to any Cause of Action 10.15 shall be given full force and effect according to each and all of its express terms and provisions.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY LEFT BLANK]

58

IN WITNESS WHEREOF, each Party has duly executed and delivered this Agreement as of the date first above written.

BUYER:

NETWORK MEDICAL MANAGEMENT, INC.

By: /s/ Chandan Basho Name: Chandan Basho Title: Chief Financial Officer

[Signature Page to Stock Purchase Agreement]

IN WITNESS WHEREOF, each Party has duly executed and delivered this Agreement as of the date first above written.

COMPANY:

I HEALTH, INC.

By: /s/ Ronald Brandt Name: Ronald Brandt Title: Chief Executive Officer

BENEFICIAL OWNERS:

/s/ Ronald Brandt RONALD BRANDT

/s/ Allison Brandt ALLISON BRANDT

EQUITYHOLDER:

RONALD LEE BRANDT AND ALLISON LEIGH BRANDT FAMILY TRUST DATED DECEMBER 16, 2003

/s/ Ronald Brandt Ronald Brandt, Co-Trustee

/s/ Allison Brandt

Allison Brandt, Co-Trustee

[Signature Page to Stock Purchase Agreement]



Apollo Medical Holdings, Inc. Announces Definitive Agreement to Acquire Assets of Community Family Care Medical Group IPA, Inc. and Health Plan

ALHAMBRA, Calif., Nov. 7, 2023 /PRNewswire/ -- Apollo Medical Holdings, Inc. ("ApolloMed," and together with its subsidiaries and affiliated entities, the "Company") (NASDAQ: AMEH), a leading physician-centric, technology-powered healthcare company focused on enabling providers in the successful delivery of value-based care, today announced that it and its affiliated professional entity have entered into a definitive agreement to acquire assets relating to Community Family Care Medical Group IPA, Inc. ("CFC"), including the CFC independent physician association ("IPA"), the CFC Health Plan ("HP") and CFC's management services organization ("MSO") entities.



CFC's scaled provider group, with more than 350 primary care physicians and 500 specialists on platform, manages the healthcare of over 200,000 members in the Los Angeles, California area. The group serves patients across Medicare, Medicaid, and Commercial payers. CFC, which became an ApolloMed Care Enablement client in 2020, has a Restricted Knox Keene ("RKK") license for Medicaid members, and will seamlessly integrate as an ApolloMed Care Partner as it continues to utilize our Care Enablement platform.

Brandon Sim, Co-CEO of ApolloMed, stated, "We're incredibly excited to announce our acquisition of CFC, a scaled, unique network of mission-driven providers with a long track record of strong performance, especially in the Medicaid market in Los Angeles. We believe that pairing CFC's impressive track record and clinical capabilities with ApolloMed's scaled value-based care platform will result in even better outcomes for those we serve. We also anticipate that CFC's demonstrated ability to take full risk successfully in Medicaid with its RKK license will allow us to increase our ability to manage a person's total health across our entire Medicaid business. Importantly, this acquisition of an existing client shows the value and synergy of leveraging both our Care Enablement and Care Partners business segments, transitioning from a vendor-client relationship into one in which we are responsible for the total cost of care of their scaled membership base in a de-risked and accretive manner. We believe that our partnership with CFC and its providers demonstrates our deep commitment to delivering accessible, high-quality, and high-value healthcare to local communities across Los Angeles."

Dr. Accie Mitchell, CEO of CFC, commented, "We are pleased to strengthen our existing relationship with ApolloMed through this transaction, which is a natural progression towards the continued growth and success of CFC. Leveraging ApolloMed's Care Enablement solution has empowered our physicians to deliver better care and enabled the company to deliver a multi-year, growing track record of profitability while managing Medicaid populations under value-based care contracts. We are incredibly excited for the next chapter of CFC's growth and success in serving those who most need it across Los Angeles and beyond."



For 2023, CFC as a whole is expected to generate approximately \$190 million in revenue. CFC's adjusted EBITDA for the full year is expected to be approximately \$25 million, reflecting a margin of approximately 13%.

Additional details regarding ApolloMed's acquisition of CFC can be found on the Company's Investor Relations website.

Transaction Summary

Under the terms of the agreement, subject to satisfaction of customary conditions, the purchase price of \$202 million is made of \$152 million of cash on hand, subject to customary adjustments, \$20 million of equity in ApolloMed stock, and up to \$30 million in performance-based milestone payments.

ApolloMed's acquisition of the CFC IPA is expected to close in Q1 2024. The acquisition of the CFC HP and the MSO entities are expected to close by the end of Q1 2024, subject to regulatory approval.

Advisors

Sheppard, Mullin, Richter & Hampton LLP is serving as legal counsel for CFC. Russ August & Kabat LLP and Nossaman LLP are serving as legal counsel for ApolloMed.

About Apollo Medical Holdings, Inc.

ApolloMed is a leading physician-centric, technology-powered, risk-bearing healthcare management company. Leveraging its proprietary end-to-end technology solutions, ApolloMed operates an integrated healthcare delivery platform that enables providers to successfully participate in value-based care arrangements, thus empowering them to deliver high quality care to patients in a cost-effective manner.

Headquartered in Alhambra, California, ApolloMed's subsidiaries and affiliates include management services organizations (MSOs), affiliated independent practice associations (IPAs), and entities participating in the Centers for Medicare & Medicaid Services Innovation Center (CMMI) innovation models. For more information, please visit <u>www.apollomed.net</u>.

Forward Looking Statements

This press release contains forward-looking statements within the meaning of the Private Securities Litigation Reform Act of 1995, Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended, such as statements about the Company's operational focus and strategic growth plans. Forward-looking statements reflect current views with respect to future events and financial performance and therefore cannot be guaranteed. Such statements are based on the current expectations and certain assumptions of the Company's management, and some or all of such expectations and assumptions may not materialize or may vary significantly from actual results. Actual results may also vary materially from forward-looking statements due to risks, uncertainties and other factors, known and unknown, including the risk factors described from time to time in the Company's reports to the Securities and Exchange Commission ("SEC"), including without limitation the risk factors discussed in the Company's Annual Report on Form 10-K for the year ended December 31, 2022, filed with the SEC and any subsequent quarterly reports on Form 10-Q.

FOR MORE INFORMATION, PLEASE CONTACT:

Investor Relations Asher Dewhurst investors@apollomed.net